

Environmental - Remediation - Engineering - Laboratories - Drilling

# PRELIMINARY SITE INVESTIGATION

108 Station Street, Wentworthville NSW

Prepared for Beaini Projects Pty Ltd

> Ref: ES6206v2 28<sup>th</sup> April 2015

HEAD OFFICE: PO Box 398 Drummoyne NSW 1470 Aargus Pty Ltd ACN 050 212 710 • Aargus Holdings Pty Ltd ACN 063 579 313 Aargus Australia Pty Ltd ACN 086 993 937 • Aargus Recruitment Pty Ltd ACN 098 905 894 Telephone: 1300 137 038 • Facsimile: 1300 136 038 • Email: admin@aargus.net • Website: www.aargus.net

Other office locations in NSW - QLD - VIC - SA and 4 overseas countries

#### CONTROLLED DOCUMENT DISTRIBUTION AND REVISION REGISTER

#### **DISTRIBUTION LIST**

Copy No.	Custodian	Location
1	Nick Kariotoglou	Aargus Pty Ltd (Library)
2	Beaini Projects Pty Ltd	PO Box 5006 Kingsdene NSW

**Note:** This distribution list identifies the current custodians of controlled copies of the subject document. It is expected that these custodians would be responsible for:

- the storage of the document
- ensuring prompt incorporation of amendments
- making the document available to pertinent personnel within the organization
- encouraging observance of the document by such personnel
- making the document available for audit

#### **DOCUMENT HISTORY**

Document No.	Revision No.	Issue Date	Description
ES6206	0	01.04.2015	Initial Issue
ES6206v2	1	28.04.2015	<b>Revised Issue</b>

Approved for release by:

Mark Ketty

Mark Kelly Environmental Manager

Date:

28.04.2015



## TABLE OF CONTENTS

T	ABLI	E OF CONTENTS	3
L	IST C	DF TABLES	4
L	IST C	DF FIGURES	4
L	IST C	DF APPENDICES	5
A	BBRI	EVIATIONS	6
E	XECU	UTIVE SUMMARY	7
1	I	NTRODUCTION	9
-	1.1	BACKGROUND	
	1.2	OBJECTIVE	
	1.3	SCOPE OF WORKS	
2	S	SITE IDENTIFICATION AND DESCRIPTION	. 11
	2.1	SITE IDENTIFICATION	
	2.2	SITE INSPECTION	
	2.3	TOPOGRAPHY AND SURFACE WATER DRAINAGE	. 13
	2.4	SURROUNDING LAND USES	. 13
3	S	SITE HISTORY	. 14
	3.1	LAND TITLES	. 14
	3.2	AERIAL PHOTOGRAPHS	. 15
	3.3	EPA RECORDS	
		3.1 CLM Act 1997	
	3.4	3.2 POEO Register WORKCOVER NSW RECORDS	
	3.5	SECTION 149 CERTIFICATES	
	3.6	COUNCIL SEARCH RECORDS	. 17
	3.7	INDUSTRIAL PROCESSES AND PRODUCTS MANUFACTURED	. 17
	3.8	FORMER CHEMICAL STORAGE AND TRANSFER AREAS	
	3.9	PRODUCT SPILL & LOSS HISTORY	
		DISCHARGES TO LAND, WATER AND AIR	
		COMPLAINTS HISTORY	
		HISTORICAL USE OF ADJACENT LAND DISCUSSION AND SUMMARY OF SITE HISTORY	
4		ENVIRONMENTAL SETTING	
4			
	4.1 4.2	SENSITIVE ENVIRONMENTAL RECEPTORS	
	4.2 4.3	SOIL	
	4.4	ACID SULFATE SOILS	
	4.5	HYDROGEOLOGY	
	4.6	LOCAL METEOROLOGY	
5	A	AREAS OF POTENTIAL ENVIRONMENTAL CONCERN	. 23
6	I	PRELIMINARY CONCEPTUAL SITE MODEL	. 24



6.	1 CONC	EPTUAL SITE MODEL	24
	6.1.1	Data Gaps	26
7	CONCL	USION AND RECOMMENDATIONS	27
LIM	ITATION	S	28
REF	ERENCE	S	30

## LIST OF TABLES

Table 1: Site Identification	11
Table 2: Surrounding Land Uses	13
Table 3: Land Title Information	14
Table 4: Summary of Historical Aerial Photos	15
Table 5: Summary of Registered Groundwater Bore Records	21
Table 6: Summary of Potential Areas and Contaminants of Concern	23
Table 7: Conceptual Site Model	25

## LIST OF FIGURES

- Figure 1: Site Locality
- Figure 2: Site Lot and DP Numbers
- Figure 3: Site Features



## LIST OF APPENDICES

- APPENDIX A: SITE PLANS
- APPENDIX B: SITE SURVEY PLANS
- APPENDIX C: SITE PHOTOGRAPHS
- APPENDIX D: LAND TITLE INFORMATION
- APPENDIX E: CURRENT AND HISTORICAL AERIAL PHOTOGRAPHS
- APPENDIX F: NSW EPA RECORDS
- APPENDIX G: GROUNDWATER BORE SEARCH
- APPENDIX H: LOCAL METEOROLOGY
- APPENDIX I: IMPORTANT INFORMATION ABOUT YOUR REPORT



## ABBREVIATIONS

AIP	Australian Institute of Petroleum Ltd
ANZECC	Australian and New Zealand Environment and Conservation Council
AST	Aboveground Storage Tank
BGL	Below Ground Level
BTEX	Benzene, Toluene, Ethyl benzene and Xylene
COC	Chain of Custody
DA	Development Approval
DP	Deposited Plan
DQOs	Data Quality Objectives
EPA	Environment Protection Authority
ESA	Environmental Site Assessment
HIL	Health-Based Soil Investigation Level
LGA	Local Government Area
NEHF	National Environmental Health Forum
NEPC	National Environmental Protection Council
NHMRC	National Health and Medical Research Council
OCP	Organochlorine Pesticides
OPP	Organophosphate Pesticides
PAH	Polycyclic Aromatic Hydrocarbon
PCB	Polychlorinated Biphenyl
PID	Photo Ionisation Detector
PQL	Practical Quantitation Limit
QA/QC	Quality Assurance, Quality Control
RAC	Remediation Acceptance Criteria
RAP	Remediation Action Plan
RPD	Relative Percentage Difference
SAC	Site Assessment Criteria
SVC	Site Validation Criteria
TCLP	Toxicity Characteristics Leaching Procedure
TPH	Total Petroleum Hydrocarbons
UCL	Upper Confidence Limit
UST	Underground Storage Tank
VHC	Volatile Halogenated Compounds
VOC	Volatile Organic Compounds



## **EXECUTIVE SUMMARY**

Aargus Pty Ltd (Aargus) was appointed by Beaini Projects Pty Ltd (the 'client') to undertake a Preliminary Site Investigation (PSI) within the property located at 108 Station Street, Wentworthville NSW (the 'site'). This report will outline suitability of the site while maintaining its existing use, and to seek to amend the existing FSR and Height of the Building.

A site investigation was requested by Holroyd City Council to determine the potential for onsite contamination and is a required document for the Planning Proposal

At the time of the site inspection (Friday 27<sup>th</sup> March 2015), the site was rectangular in shape and is currently used for commercial purposes, the majority of the property is covered by asphalt/bitumen or concrete sealed surfaces, including an asphalt driveway within the northern area of the property, hydrocarbon staining was noted on all sealed surfaces within the carpark area of the property, cracks and potholes were noted on the sealed surfaces within the car park area of the property, a high voltage electrical transformer box was noted within the northern portion of the car park, and a grease pit and chemical storage area was noted at the rear of the building within the car park area of the property.

The land title information provided suggested that the site was used by private individuals since 1918. In 1963, the site was transferred to Ralian Pty Ltd and leased to T.R.N. Holdings Pty Ltd. The other lessors are private names. In 1989, it was transferred to Isikit Pty Ltd and then was under different business entities from 1990 to 2002. The site was transferred to Michael Elias and Samira Elias in 2002 and it remains the same name until now.

The land use of the site appeared to have been commercial from at least 1991 to present with structures remaining generally unchanged. The general land use of the immediate site vicinity seems to have been consistently commercial properties to the north and south of the site with low to medium density residential to the east and low density residential to the west, with new commercial properties constructed after 1991 to the north and south.



The findings of the assessment indicated the following areas of potential environmental concern:

- Potential importation of uncontrolled fill that may contain various contaminants;
- Current or past use of pesticides;
- Car park areas where leaks and spills from cars may have occurred;
- Chemical Storage Areas;
- High Voltage Electrical Transformer box where leaks may have occurred; and
- Asbestos based building materials.

The contaminants that may be present in some of these areas were considered to be of low significance in terms of risk to the human and environmental receptors identified.

Based on the information collected during this investigation, the site will be suitable subject to the completion of a Detailed Site Investigation (and after remediation and validation, if required). It is also recommended that a hazardous material assessment on the building structures within the site be undertaken followed by an asbestos clearance certificate.



## 1 INTRODUCTION

#### 1.1 Background

Aargus Pty Ltd (Aargus) was appointed by Beaini Projects Pty Ltd (the 'client') to undertake a Preliminary Site Investigation (PSI) within the property located at 108 Station Street, Wentworthville NSW (the 'site'). This report will outline suitability of the site while maintaining its existing use, and to seek to amend the existing FSR and Height of the Building. The location of the property is presented in Figure 1 of Appendix A.

A site investigation was requested by Holroyd City Council to determine the potential for onsite contamination and is a required document for the Planning Proposal

#### 1.2 Objective

The primary objectives of this PSI are as follows:

- Identify potential areas where contamination may have occurred from current and historical activities;
- Identify potential contaminants associated with potentially contaminating activities;
- Assess the potential for soils and groundwater to have been impacted by current and historical activities; and
- Assess the suitability of the site to amend the exiting FSR and Height of the Building, while maintaining its existing land use.



#### 1.3 Scope of Works

The scope of works for this PSI includes:

- Review of the physical site setting and site conditions based on a site inspection, including research of the location of sewers, drains, holding tanks and pits, spills, patches of discoloured vegetation, etc. (where applicable);
- Research and review of the information available, including previous environmental investigations, current and historical titles information, review of aerial photographs, groundwater bore searches, EPA notices, council records, anecdotal evidence, site survey and site records on waste management practices;
- Development of a preliminary Conceptual Site Model (CSM) to demonstrate the interactions between potential sources of contamination, exposure pathways and human/ecological receptors identified; and
- Recommendations for additional investigations should any data gaps be identified or possible strategies for the management of the site, where relevant.

This report was prepared with reference to the NSW Environment Protection Authority (EPA) "Guidelines for Consultants Reporting on Contaminated Sites" (2011).



## 2 SITE IDENTIFICATION AND DESCRIPTION

#### 2.1 Site Identification

Site identification information and land use is summarised in the table below.

Lot and DP Number (Address)	Lot B in DP410947
Coordinates (SE corner)*	Latitude: 33°48′36″ Longitude:150°58′17″
Approx. Site Area	1,916m <sup>2</sup>
Local Government Area	Holroyd
Parish	Prospect
County	Cumberland
Current Land Zoning**	B2 - Local Centre
Proposed Land Use	Commercial
Current Site Owner	Michael Elias and Samina Elias
Site End Users	Residents (adults & children), workers

#### Table 1: Site Identification

Notes: \* refer to <u>http://maps.six.nsw.gov.au/</u>

\*\* refer to Holroyd City Council Zoning Map http://passthrough.fw-http://www.holroyd.nsw.gov.au/your-development/policies-plans/holroyd-local-environment-plan-2013/

The site boundary and Lot and DP numbers are presented in Figure 2 of Appendix A. A survey plan provided by the client is included in Appendix B.



#### 2.2 Site Inspection

A site visit was carried out on Friday 27<sup>th</sup> March 2015 by an Aargus field scientist to inspect the site for any potential sources of contamination and document any observations made regarding the current site conditions. At the time of the site inspection, the following observations were made:

- The site was rectangular in shape and is currently used for commercial purposes.
- The majority of the property is covered by asphalt/bitumen or concrete sealed surfaces, including an asphalt driveway within the northern area of the property.
- The main access to the site is along Station Street on the western boundary.
- No signs of stress were observed on the vegetation surrounding the property.
- The site boundaries were defined by the buildings western (front) façade on the western boundary, asphalt driveway on the northern boundary, a colour-bond fence along the eastern boundary and the adjoining commercial building's concrete façade on the southern boundary.
- No surface standing water was noticed at the site.
- Hydrocarbon staining was noted on all sealed surfaces within the carpark area of the property.
- Cracks and potholes were note on the sealed surfaces within the car park area of the property.
- A high voltage electrical transformer box was noted within the northern portion of the car park.
- A grease pit and chemical storage area was noted at the rear of the building within the car park area of the property.

The site features are presented in Figure 3 of Appendix A. Site photographs are included in Appendix C.



### 2.3 Topography and Surface Water Drainage

The following observations were made during the site inspection carried out on the Friday 27<sup>th</sup> March 2015:

- The site topography is sloping towards the south along Station Street, at approximately a 3% slope and towards the east from Station Street to the eastern boundary of the property at approximately a 10% slope.
- Stormwater runoff from the site is expected to flow in a southerly direction along Station Street and easterly across the property.

#### 2.4 Surrounding Land Uses

The surrounding land uses identified are described in the table below:

#### **Table 2: Surrounding Land Uses**

Orientation	Description
North	Commercial properties
East	Medium Density Residential properties
South	Commercial properties
West	Station Street then Low density residential

In summary, the surrounding land uses comprised of commercial and low to medium density residential properties.



## 3 SITE HISTORY

#### 3.1 Land Titles

A review of historical documents held at the NSW Department of Lands offices was undertaken to identify the current and previous land owners, and potential land uses. The results of the title search are summarised in the following table.

Year	Lot 2773 in DP629153 (77 Monash Rd, Doonside)	
2002-Current	Michael Elias	
	Samira Elias	
2012-2014	(Lease) Amit Kumar	
2006 2000	Reema Nagpal	
2006-2008	(Lease) David Bayssari	
2005-2010	Maria Bayssari (Lease) Kathirvel Vel Satha Sevam	
1995-2002	Antonia Da Silva Duarte	
1998 2002	Maria Duarte	
1998-2002	(Lease) KC.Macrae Holdings Pty Ltd	
1989-1995	Isikit Pty Ltd	
1993-1998	(Lease) KC.Macrae Holdings Pty Ltd	
1991-1993	(Lease) Shamar May	
1990-1991	(Lease) J.K.N Holdings Pty Ltd	
	Prior Vol. 15519 Fol.74	
1988-1989	Ralian Pty Ltd	
	Prior Vol. 8442 Fol.95	
1963-1988	Ralian Pty Ltd	
1984-1988	(Lease) Jennifer Anne Wall	
	Kathleen Margret Ferrari	
1982-1984	(Lease) Piero Batolacci	
1982-1988	Inelda Batolgcci	
	(Lease) T.R.N. Holdings Pty Ltd	
1982-1988	(Lease) Frank Pellegrino	
1982-1988	(Lease) Author Norman Roleate	
	Prior Vol.6092 Fol.121	
1954-1988	Albert Reginald Payne	
	Prior Vol.2894 Fol.250	
1941-1954	Albert Reginald Payne	
1939-1941	Residential Estates Limited	
1929-1939	William Ashton Gorden	
1918-1929	Thomas Booke	

#### **Table 3: Land Title Information**



In summary, the land title information provided suggested that the site was used by private individuals since 1918. In 1963, the site was transferred to Ralian Pty Ltd and leased to T.R.N. Holdings Pty Ltd. The other lessors are private names. In 1989, it was transferred to Isikit Pty Ltd and then was under different business entities from 1990 to 2002. The site was transferred to Michael Elias and Samira Elias in 2002 and it remains the same name until now.

A copy of the historical land titles information obtained by Aargus can be found in Appendix D.

#### 3.2 Aerial Photographs

Selected aerial photographs obtained from the NSW Department of Lands were reviewed to describe the site features and surrounding areas at various timelines. A summary of the review is presented in the table below. Copies of current and historical aerial photographs are presented in Appendix E.

Year	Site	Surrounding areas		
1928	The site appears to be vacant with no building structures within the site.	N: Low Density Residential S: Low Density Residential E: Low Density Residential W: Road then Low Density Residential		
1951	The site appears to be vacant with no building structures within the site.	N: Commercial property S: Low Density Residential E: Low Density Residential W: Road then Low Density Residential		
1970	Some small structures appear to be present within the site.	N: Commercial properties S: Commercial properties E: Low Density Residential W: Road then Low Density Residential		
1991	The site appears to show a building within the western portion of the site. The eastern portion of the site appears to have been converted into car parking.	N: Commercial properties S: A new large commercial property E: Appears to be vacant W: Road then Low Density Residential		
2015	The site layout appeared to be similar to that observed in the 1991 photo.	N: New commercial property has been constructed S: New commercial property has been constructed E: New medium density residential property has been constructed W: Road then Low Density Residential		

|--|



In summary, land use of the site appeared to have been commercial from at least 1991 to present with structures remaining generally unchanged.

The general land use of the immediate site vicinity seems to have been consistently commercial properties to the north and south of the site with low to medium density residential to the east and low density residential to the west, with new commercial properties constructed after 1991 to the north and south.

#### 3.3 EPA Records

#### 3.3.1 CLM Act 1997

The NSW EPA publishes records of contaminated sites under Section 58 of the Contaminated Land Management (CLM) Act 1997. The notices relate to investigation and/or remediation of site contamination considered to pose a significant risk of harm under the definition in the CLM Act. However, it should be noted that the EPA record of Notices for Contaminated Land does not provide a record of all contaminated land in NSW.

A search of the EPA database revealed that the subject site is not listed. However, two (2) other properties were listed within the Holroyd City Council Local Government Area. There are 7 former and 2 current notices related to these surrounding properties. However all of the listed properties were situated 1.0km or greater away from the site and were not considered posing an adverse impact on the site.

Copies of the EPA records are included in Appendix F.

#### 3.3.2 POEO Register

A search of the POEO Register revealed that the site was not listed. However, six (6) other properties were listed within the Holroyd City Council Local Government Area issued with Penalty Notices. A copy of the POEO register search is included in Appendix F.



#### 3.4 WorkCover NSW Records

The WorkCover NSW database was requested to be accessed in order to disclose evidence of past storage of Dangerous Goods within the site. We are currently awaiting information from WorkCover NSW and will update/amend this report once we have gathered the relevant information.

#### 3.5 Section 149 Certificates

The client has advised us that at the time of this report there was no Planning Certificate – Section 149 (2) of the Environmental Planning & Assessment Act 1979 for the site.

#### 3.6 Council Search Records

The Holroyd City Council database was requested to be accessed in order to disclose file records relating to the site. We are currently awaiting information from Council and will update/amend this report once we have gathered the relevant information.

#### 3.7 Industrial Processes and Products Manufactured

A review of the industrial processes and/or products manufactured at the site was conducted, with no information at present pertaining to the site.

#### 3.8 Former Chemical Storage and Transfer Areas

A review of the former chemical storage and transfer areas and/or products manufactured at the site was conducted, with no information at present pertaining to the site. The site inspection revealed the following:

• A high voltage electrical transformer box was noted within the northern portion of the car park.



• A grease pit and chemical storage area was noted at the rear of the building within the car park area of the property.

#### 3.9 Product Spill & Loss History

It was indicated by the client, that to their knowledge no serious land or water contamination had occurred.

The majority of the site is currently occupied by a building and/ or sealed surfaces. At the time of the inspections, the sealed surfaces were in generally good condition with some staining observed and crack and potholes detected within the car park area of the property.

#### 3.10 Discharges to Land, Water and Air

It was indicated by the client, that to their knowledge no discharges to land, water or air had occurred at the site.

#### 3.11 Complaints History

There was no information obtained or provided indicating any relevant complaint history pertaining to the site.

#### 3.12 Historical Use of Adjacent Land

It was indicated by the client that to their knowledge, the adjacent lands to the site have been used primarily commercial developments, with the exception of the eastern boundary which consist of a medium density residential property.



#### 3.13 Discussion and Summary of Site History

Based on available information, the site historical usage is summarised as follows:

- The land title information provided suggested that the site was used by private individuals since 1918. In 1963, the site was transferred to Ralian Pty Ltd and leased to T.R.N. Holdings Pty Ltd. The other lessors are private names. In 1989, it was transferred to Isikit Pty Ltd and then was under different business entities from 1990 to 2002. The site was transferred to Michael Elias and Samira Elias in 2002 and it remains the same name until now.
- The land use of the site appeared to have been commercial from at least 1991 to present with structures remaining generally unchanged.
- The general land use of the immediate site vicinity seems to have been consistently commercial properties to the north and south of the site with low to medium density residential to the east and low density residential to the west, with new commercial properties constructed after 1991 to the north and south.



## **4 ENVIRONMENTAL SETTING**

#### 4.1 Sensitive Environmental Receptors

The nearest down-gradient watercourse is Finlaysons Creek located approximately 180m east of the site. The nearest recreational area is Friend Park, located approximately 20m northwest of the site.

#### 4.2 Geology

The Geological Map of Sydney (Geological Series Sheet 9130, Scale 1:100,000, 1983) published by the Department of Mineral Resources indicates the residual soils within the site to be underlain by Ashfield Shale of the Wianamatta Group comprising shale, carbonaceous claystone, claystone, laminite, fine to medium grained lithic sandstone and rare coal.

#### 4.3 Soil

The Soil Landscape Map of Sydney (soil Landscape Series Sheet 9130, Scale 1:100,000, 2002), prepared by the Soil Conservation Service of NSW, indicates that the site is located within the Blacktown landscape area and typically consists of highly plastic and relatively impermeable residual soil.

#### 4.4 Acid Sulfate Soils

To determine whether there is a potential for acid sulphate soils to be present within a site, reference was made to the NSW Department of Land & Water Conservation (DLWC) *Acid Sulphate Soil Risk Map* (Edition Two, December 1997, Scale 1:25,000) for Map 90 "Prospect-Parramatta". The map shows that at the site location and in the immediate vicinity of the site location that there were no known occurrences of acid sulphate soil materials.



#### 4.5 Hydrogeology

Based on available information, our desktop study indicates that groundwater from site is likely to be flowing in a general easterly direction towards Finlaysons Creek (approximately 180m east of the site).

A search of the Department of Natural Resources (DNR) borehole database information revealed eight (8) groundwater bores within a 1km radius of the site. A summary of the relevant information provided by the registered groundwater bore record search is provided in the following table. A copy of the groundwater bore search records are in Appendix G.

GW Bore ID	Approximate Location	Intended Purpose	Depth (m bgl)	Standing Water Level (m bgl)	Water Bearing Zones	Salinity (µS/cm)
GW113429	600m S	Monitoring	-	-	-	-
GW113430	600m S	Monitoring	-	-	-	-
GW113431	600m S	Monitoring	-	-	-	-
GW113432	600m S	Monitoring	-	-	-	-
GW113433	600m S	Monitoring	-	-	-	-
GW113434	600m S	Monitoring	-	-	-	-
GW113435	600m S	Monitoring	-	-	-	-
GW113436	600m S	Monitoring	-	-	-	-

Table 5: Summary of Registered Groundwater Bore Records

- No Information available.

The registered groundwater bores within a 1km radius of the site were located down-gradient. Note: At the time of this report no groundwater bore records information available, however they are unlikely to be used for human consumption since the site is not located within the SEPP boundary for the Sydney Drinking Water Catchment.

Based on geological information, a confined aquifer may be present within water bearing zones in the underlying shallow sandstone stratum. No information was provided regarding salinity records, however, groundwater is likely to be fresh or brackish based on other site investigations conducted by Aargus in the area.



#### 4.6 Local Meteorology

The monthly rainfall of the local area can be represented by the data collected by Bureau of Meteorology (BOM) from the rainfall gauge located in Bathurst Street Greystanes, which is located approximately 3km southwest of the site. Records collected from 2001 till December 2014 indicated that the average annual rainfall amount is 748.1mm.

Reference can be made to Appendix H – Local Meteorology.



## 5 AREAS OF POTENTIAL ENVIRONMENTAL CONCERN

Based on the site inspection, site history, previous reports and review of available information from the desktop study, the potential Areas of Environmental Concern (AEC) and their associated Contaminants of Concern (CoC) for the site were identified. These are summarised in the following table.

Potential AEC	Potentially contaminating activity	Potential CoCs	Likelihood of Site Impact	Justification
Entire site	Importation of fill material from unknown origin	Metals, TPH, BTEX, PAH, OCP, PCB, Phenols, Cyanides, Asbestos	Low	Based on the site observations and site topography, the presence of imported fill material is likely to be minimal.
	Potential for pesticides to have been sprayed or injected on or underneath building pad and sealed surfaces	OCP	Low	The site is not known to have been used for agricultural purposes from the 1950s when OCPs were first introduced into Australia. If use of OCPs has occurred, the impact is likely to have been localised and limited to the topsoil layer.
Car parking areas	Leaks from vehicles	Metals, TPH, BTEX, PAH	Low	Staining areas were noted on the asphalt car park surfaces. Also some cracking and potholes of the car park surfaces was visible.
Chemical Storage Area	Leaks containers	Metals, TPH, BTEX, PAH, Phenols	Low	Staining areas were noted on the sealed surfaces.
High Voltage Electrical Transformer Box	Leaks from Transformer Box	Metals, PCB, Phenols, Cyanides	Low	No staining was noted around the transformer box on the asphalt car park surfaces
Building Structures	Potential Asbestos/Fibro Features	Asbestos	Low	If present, these will be removed by licensed contractors.

#### Table 6: Summary of Potential Areas and Contaminants of Concern



## 6 PRELIMINARY CONCEPTUAL SITE MODEL

#### 6.1 Conceptual Site Model

The Preliminary Conceptual Site Model (CSM) presented in the table below provides a representation of the potential risks associated with the linkages between the following elements:

- Potential contamination sources and their associated contaminants of concern identified in Section 5. Only potential areas of concern with a significance rating of low to high were included;
- Potential human receptors that may be impacted by site contamination are current and future end-users, construction workers and the general public within the immediate vicinity;
- Potential environmental receptors identified in Section 4;
- Potential exposure pathways; and
- Whether each source-pathway-receptor pollution linkage are complete, limited or not present, based on current and future site conditions.



Potential Sources	Potential Receptor	Potential Exposure Pathways	Complete Linkages	Risk	Justification
Contaminated soil from placement of uncontrolled fill	Site users or the general public	Dermal contact, inhalation or ingestion of exposed impacted soils	Limited (Current)	Low	Direct contact with impacted soils is limited as the entire site has sealed surfaces.
across the site. Use of OCPs.			No (Future)	Negligible	If present, contaminated soils are likely to be remediated and removed off-site for disposal.
Hydrocarbon spills and leakages from vehicles	The aquatic ecosystems at Finlaysons Creek	Migration of impacted groundwater and surface water run-off	Yes (Current)	Low	No obvious sources of inorganic contamination were observed on site that could migrate off site with surface water run-off.
Chemical spills from storage area and Electrical Transformer. Asbestos.			No (Future)	Negligible	If present, contaminated soils and groundwater is likely to be remediated and any remaining residual contamination would likely be at acceptable concentrations upon reaching the Finlaysons Creek
	Underlying Bedrock Aquifer	Leaching and migration of contaminants through groundwater infiltration	Limited (Current)	Low	Due to the existing concrete sealed surfaces, expected shallow bedrock, leachability of metals and other inorganics vertical migration of contaminants is likely to be limited.
			No (Future)	Low	If present, contaminated soils and groundwater is likely to be remediated and any remaining residual contamination would likely be at acceptable concentrations upon reaching the underlying aquifer

## **Table 7: Conceptual Site Model**



#### 6.1.1 Data Gaps

Based on the CSM, the following data gaps were identified with respect to the pollution linkages identified and should be determined if there is any proposed future excavation works within the property:

- The presence, origin and thickness of fill material (whether sourced from site or imported);
- Confirmation if contamination has occurred from current and historical site activities through collection and laboratory analysis of soil samples;
- WorkCover search results in regards to Licences to Keep Dangerous Goods pertaining to the site were not undertaken and need to be obtained and read to further define the CSM; and
- The Council records were not obtained and need to be read to further define CSM.



## 7 CONCLUSION AND RECOMMENDATIONS

The findings of the assessment indicated the following areas of potential environmental concern:

- Potential importation of uncontrolled fill that may contain various contaminants;
- Current or past use of pesticides;
- Car park areas where leaks and spills from cars may have occurred;
- Chemical Storage Areas;
- High Voltage Electrical Transformer box where leaks may have occurred; and
- Asbestos based building materials.

The contaminants that may be present in some of these areas were considered to be of low significance in terms of risk to the human and environmental receptors identified.

Based on the information collected during this investigation, the site will be suitable subject to the completion of a Detailed Site Investigation (and after remediation and validation, if required). It is also recommended that a hazardous material assessment on the building structures within the site be undertaken followed by an asbestos clearance certificate.

Thank you for the opportunity to undertake this work. We would be pleased to provide further information on any aspects of this report.

For and on behalf of

Aargus Pty Ltd Written By:

**Con Kariotoglou** Project Manager

**Reviewed By:** 

Park Kethe

Mark Kelly Environmental Manager



## LIMITATIONS

The Aargus assessment is based on the result of limited site investigations and sample testing. Neither Aargus, nor any other reputable consultant, can provide unqualified warranties nor does Aargus assume any liability for site conditions not observed or accessible during the time of the investigations.

Despite all reasonable care and diligence, the materials encountered and concentrations of contaminants measured may not be representative of conditions between the locations sampled and investigated. There is always some disparity in subsurface conditions across a site that cannot be fully defined by investigation. Hence it is unlikely that measurements and values obtained from sampling and testing during environmental works carried out at a site will characterise the extremes of conditions that exist within the site. In addition, site characteristics may change at any time in response to variations in natural conditions, chemical reactions, truck movement or contractor movement of soils and other events, e.g. groundwater movement and or spillages of contaminating substances. These changes may occur subsequent to Aargus investigations and assessment.

This report and associated documentation and the information herein have been prepared solely for the use of the client and interested parties at the time or writing the report and is valid (for the purposes of management or transport of material) for a period of one month only from the date of issue. Any other reliance assumed by third parties on this report shall be at such parties' own risk. Any ensuing liability resulting from use of the report by third parties cannot be transferred to Aargus.

Whilst this report provides a review of site conditions encountered at sampling locations within the investigation, it should be noted that if materials are proposed to moved from site - Part 5.6, Section 143 of the Protection of the Environment Operations (POEO) Act 1997 states that is an offence for waste to be transported to a place that cannot lawfully be used as a facility to accept that waste. It is the duty of the owner and transporter of the waste to ensure that all material removed from a site must be accompanied by an appropriate waste classification report and materials are disposed of appropriately. An environmental or



validation report does not constitute a waste classification report and results are treated differently. Aargus accepts no liability for the unlawful disposal of waste materials from any site. Aargus does not accept any responsibility for the material tracking, loading, management, transport or disposal of waste from the site. If material is to be removed from a site, before disposal of any material to a licensed landfill is undertaken, the site owner must ensure an appropriate waste classification exists for all materials on the site planning to be removed, the waste producer will need to obtain prior consent from the licensed landfill/recycler. The receiving site should check to ensure that the material received matches the description provided in the report.

Opinions are judgements, which are based on our understanding and interpretation of current regulatory standards, and should not be construed as legal opinions.

Appendix I – Important information about your environmental site report should also be read in conjunction with this report.



## REFERENCES

This report was prepared with reference to the following guiding documents:

- ANZECC/NHMRC (1992) "Australian and New Zealand Guidelines for the Assessment and Management of Contaminated Sites". Australian and New Zealand Environment and Conservation Council and the National Health and Medical Research Council, Canberra;
- ANZECC National Water Quality Management Strategy "Australian Water Quality Guidelines for Fresh and Marine Waters", 1992.
- CRC Care Technical Report No. 13 Soil Vapour Assessment (August 2009);
- Department of Urban Affairs and Planning EPA (1998) "Managing Land Contamination – Planning Guidelines – SEPP 55 – Remediation of Land";
- National Environment Protection (Assessment of Site Contamination) Amendment Measure 2013 (No.1);
- NSW DEC, "Guidelines for the Assessment and Management of Groundwater Contamination" (March 2007);
- NSW DEC "Guidelines for the NSW Site Auditor Scheme" (2006, 2<sup>nd</sup> edition). NSW Environment Protection Authority, Sydney;
- NSW DECC (2009) "Waste Classification Guidelines, Part 1: Classifying Waste";
- NSW DECCW, "Vapour Intrusion: Technical Practice Note", (September 2010);
- NSW EPA "Guidelines for Assessing Service Station Sites" (1994). NSW Environment Protection Authority, Sydney;
- NSW EPA "Guidelines for Consultants Reporting on Contaminated Sites" (2011). NSW Environment Protection Authority, Sydney;
- NSW EPA "Guidelines on the Duty to Report Contamination under the Contaminated Land Management Act 1997" (2009). NSW Environment Protection Authority, Sydney;
- NSW EPA "Sampling Design Guidelines" (1995). NSW Environment Protection Authority, Sydney.



# **APPENDIX** A







#### Source: http://maps.six.nsw.gov.au/

PROJECT DETAILS			DRAWING	DETAILS		
Project Title	Preliminary Site Investigation		Figure No.	1	Rev No.	0
Project No.	ES6206		Scale	As above	Size	A4
Client	Beaini Projects Pty Ltd		Drawn by	СК	Date	27.03.2015
Site Address	108 Station Street, Wentworthville NSW	Aargus	Approved by	СК	Date	27.03.2015

## SITE PLAN – LOT & DEPOSITED PLAN



Environment - Remediation - Geotechnical Engineering

## SITE FEATURES





SITE FEATURES				
1	Commercial property			
2	Asphalt/Bitumen carpark			
3	Surrounding commercial properties			
4	Surrounding medium-high density residential properties			
5	Surrounding low density residential properties			

PROJECT DETAILS			DRAWING D	DETAILS		
Project Title	Preliminary Site Investigation		Figure No.	3	Rev No.	0
Project No.	ES6206		Scale	As above	Size	A4
Client	Beaini Projects Pty Ltd		Drawn by	СК	Date	27.03.2015
Site Address	108 Station Street, Wentworthville NSW	Aargus	Approved by	СК	Date	27.03.2015

Environment – Remediation – Geotechnical Engineering

# **APPENDIX B**







2.5 5 7.5 10 12.5	DRAWN:	PCJ	NI NI	IEW SOUTH SURVEYS Pty Ltd	CLIEN
SCALE 1:1259 A1	SURVEYED: CHECKED:	AS DH M.I.S. NSW		and & Engineering Surveyors O. BOX 119 Granville NSW 2142 obile: 0405 416 914 Fax: (02)9885 0630 mail:info@newsouthsurveys.com.au	PROJ
# **APPENDIX C**

## SITE PHOTOGRAPHS



#### SITE PHOTOGRAPHS

Client:	Beaini Projects Pty Ltd	
Project:	Preliminary Site Investigation	
Site Location:	108 Station Street, Wentworthville NSW	
Job No.:	ES6206	
Photos Taken By:	СК	



#### Photograph Nº 1



View of 108 Station Street, Wentworthville frontage Looking southeast. Inspected on 27.03.2015

#### Photograph Nº 3



View of asphalt sealed driveway to rear carpark Looking east. Inspected on 27.03.2015

#### Photograph Nº 5



View of carpark showing typical stains on surfaces Inspected on 27.03.2015

#### Photograph Nº 2



View of site showing sealed pathway within western (Station Street) boundary. Looking southeast. Inspected on 27.03.2015

#### Photograph Nº 4



View of rear carpark. Looking south. Inspected on 27.03.2015

#### Photograph N° 6



View of carpark showing cracks and holes within surfaces. Inspected on 24.03.2015

#### SITE PHOTOGRAPHS

Client:	Beaini Projects Pty Ltd
Project:	Preliminary Site Investigation
Site Location:	108 Station Street, Wentworthville NSW
Job No.:	ES6206
Photos Taken By:	СК



#### Photograph Nº 7



View of carpark showing high voltage electrical transformer box. Looking southeast. Inspected on 27.03.2015

#### Photograph Nº 9



View of rear of property showing chemical storage shed Looking west. Inspected on 27.03.2015

Photograph Nº 11



View of adjacent medium-high density residential property on eastern (downgradient) boundary. Looking east. Inspected on 27.03.2015

#### Photograph Nº 8



View of rear of property showing grease pit and chemical /oil storage area. Looking north. Inspected on 27.03.2015

#### Photograph Nº 10



View of rear carpark showing unsealed area within south east corner of carpark Looking east. Inspected on 27.03.2015

#### Photograph Nº 12



View of adjacent commercial property on northern (upgradient) boundary. Looking north. Inspected on 27.03.2015

# **APPENDIX D**

## LAND TITLE INFORMATION



	Land & Pro Information	n T	HISTORICAL TITLE SEARCH Certificate issued under Section 96G of the Real Property Act 1900	
			fied to: 27/3/2015 9:54AM	
			<b>Reference:</b> B/410947	Page 1
			: SEE PRIOR TITLE(S) : VOL 15519 FOL 74	
	Recorded	Number	Type of Instrument	C.T. Issue
	29/7/1989		TITLE AUTOMATION PROJECT	LOT RECORDED FOLIO NOT CREATED
	4/10/1989		CONVERTED TO COMPUTER FOLIO	FOLIO CREATED CT NOT ISSUED
$\checkmark$	6/9/1990	Z193326	LEASE	EDITION 1
	12/3/1991	Z525410	LEASE	EDITION 2
	21/5/1991	Z525411	LEASE	
$\checkmark$	21/5/1991	Z606154	LEASE	EDITION 3
	15/6/1992	E524114	CAVEAT	
Ø	12/2/1993	I113618	WITHDRAWAL OF CAVEAT	
Ø	24/3/1993	I191282	LEASE	
	24/3/1993	I191283	LEASE	
	24/3/1993	I19 <b>1</b> 284	LEASE	
$\checkmark$	24/3/1993	I191285	TRANSFER BY MORTGAGEE UNDER	
	24/3/1993	I191286	POWER OF SALE REQUEST	EDITION 4
	25/3/1993	I191287	MORTGAGE	EDITION 5
	5/8/1994	U506603	TRANSFER OF LEASE	
	5/8/1994	U506604	VARIATION OF LEASE	
	5/8/1994	U506605	TRANSFER OF LEASE	

#### END OF PAGE 1 - CONTINUED OVER

doccop1

PRINTED ON 27/3/2015

The Registrar General certifies that at the date and time specified above the information set out in this search constitutes the historical record of all dealings recorded in or action taken in respect of the mentioned title which is required to be kept by the Registrar General under section 32(7) of the Real Property Act 1900.

STRAR GEN

78



## **HISTORICAL TITLE SEARCH**

Certificate issued under Section 96G of the Real Property Act 1900

A division of the Department of Finance & Services

**No.** 78

Search certified to: 27/3/2015 9:54AM

Computer Folio Reference: B/410947

	Recorded	Number	Type of Instrument	C.T. Issue
	3/11/1994	U760035	DETERMINATION OF LEASE	
V	3/11/1994	U760036	LEASE	EDITION 6
	6/2/1995	U968462	TRANSFER GRANTING EASEMENT	EDITION 7
	24/3/1995	0111404	DISCHARGE OF MORTGAGE	
$\checkmark$	24/3/1995	0111405	TRANSFER	
	24/3/1995	0111406	MORTGAGE	EDITION 8
	11/9/1995	0459584	TRANSFER RELEASING EASEMENT	EDITION 9
	2/2/1996	0885641	LEASE	EDITION 10
	16/4/1997	2881706	LEASE	EDITION 11
	28/4/1998	3944925	DISCHARGE OF MORTGAGE	
	.28/4/1998	3944926	CHANGE OF NAME	
	28/4/1998	3944927	MORTGAGE	EDITION 12
$\checkmark$	9/11/1998	5382069	LEASE	EDITION 13
	19/11/1998	5407660	TRANSFER OF LEASE	
	19/11/1998	5407661	MORTGAGE OF LEASE	
	13/12/2001	8200306	DEPARTMENTAL DEALING	
	26/9/2002	8987302	DISCHARGE OF MORTGAGE	
$\checkmark$	26/9/2002	8987303	TRANSFER	
	26/9/2002	8987304	MORTGAGE	EDITION 14
	9/12/2003	AA234680	LEASE	EDITION 15

END OF PAGE 2 - CONTINUED OVER

doccop1

PRINTED ON 27/3/2015 78

The Registrar General certifies that at the date and time specified above the information set out in this search constitutes the historical record of all dealings recorded in or action taken in respect of the mentioned title which is required to be kept by the Registrar General under section 32(7) of the Real Property Act 1900.





## **HISTORICAL TITLE SEARCH**

Certificate issued under Section 96G of the Real Property Act 1900

A division of the Department of Finance & Services

**No**. 78

Search certified to: 27/3/2015 9:54AM Computer Folio Reference: B/410947

	Recorded	Number	Type of Instrument	C.T. Issue
	12/2/2004	AA411196	LEASE	EDITION 16
	30/3/2005	AB376909	VARIATION OF MORTGAGE	EDITION 17
	14/9/2005	AB768965	MORTGAGE OF LEASE	
J	/ 28/9/2005	AB803964	LEASE	EDITION 18
	7/12/2006	AC795298	VARIATION OF MORTGAGE	EDITION 19
	8/8/2007	AD320375	DISCHARGE OF MORTGAGE	
	8/8/2007	AD320376	MORTGAGE OF LEASE	
	17/1/2008	AD684219	DISCHARGE OF MORTGAGE	
	17/1/2008	AD684220	MORTGAGE	
$\checkmark$	17/1/2008	AD684221	LEASE	EDITION 20
	16/2/2009	AE132213	WITHDRAWN - DISCHARGE OF MORTGAGE	
	16/2/2009	AE132215	WITHDRAWN ~ MORTGAGE OF LEASE	
	21/1/2010	AF265509	LEASE	EDITION 21
	23/12/2010	AF963964	MORTGAGE OF LEASE	
	7/2/2011	AG44293	LEASE	EDITION 22
	23/9/2011	AG516342	DISCHARGE OF MORTGAGE	
	23/9/2011	AG516343	MORTGAGE	EDITION 23
C	/14/2/2012	AG789227	LEASE	EDITION 24

END OF PAGE 3 - CONTINUED OVER

doccop1

PRINTED ON 27/3/2015 78

The Registrar General certifies that at the date and time specified above the information set out in this search constitutes the historical record of all dealings recorded in or action taken in respect of the mentioned title which is required to be kept by the Registrar General under section 32(7) of the Real Property Act 1900.

SUMAR GEAL

Page 3

Destaura Const



A division of the Department of Finance & Services

## **HISTORICAL TITLE SEARCH**

Certificate issued under Section 96G of the Real Property Act 1900

**No**. 78

#### Search certified to: 27/3/2015 9:54AM

Computer Folio Reference: B/410947

Page 4

Recorded	Number	Type of Instrument	C.T. Issue
14/10/2013	AI84710	VARIATION OF LEASE	

\*\*\* END OF SEARCH \*\*\*

doccop1

The Registrar General certifies that at the date and time specified above the information set out in this search constitutes the historical record of all dealings recorded in or action taken in respect of the mentioned title which is required to be kept by the Registrar General under section 32(7) of the Real Property Act 1900.

STEAR GLARE

78

				or conveyancing pu	poses Only.
•	97-011	<i>[</i> 1 ]	ANSFER Property Act, 1900 Office		0 111405 B
	\$5.00		£0/292526	5420582 913T 0 <del>4</del> 500	
(A)	LAND TRANSFERRED Show no more than 20 References to Title. If appropriate, specify the share transferred.	FOLIO 3	IDENTIFIER B,	/410947	
(B)	LODGED BY	245	12( PH	WEALTH BANK OF	X 434
(C)	TRANSFEROR	ISIKIT PTY	LIMITED AC	J 002 455 218	
(D) (E) (F)	acknowledges receipt of the consideration of and as regards the land specified above transubject to the following ENCUMBRANCES TRANSFEREE		sferee an estate in f		3.7.191,284
(G)	ANT TENANCY:		and MARIA DU	ARTE	
(H)	We certify this dealing correct for the purposed Signed in my presence by the Transferor with THE COMMON SEAL OF JSIKI LIMITED was hereunto all the authori signature of Witness of of directors in the pres	ho is personally k T PTY. ixed by the board	snown to me.	BATED A.C. K.	3.003 - 7.6 
	Signed in my presence by the Transferee w	ho is personally k		ector -	
	Signature of Witness				
	Name of Witness (BLOCK LETTER	s)		-1-	2-
	Address of Witness		Solic	itor for Agains of The	ansferee



LAND

## TITLE SEARCH

Computer Folio Certificate issued under Section 96D of the Real Property Act 1900

### Search certified to:

No. 77

27/3/2015 9:54 AM

COMPUTER FOLIO REFERENCE

B/410947

EDITION No. & DATE OF CURRENT CERTIFICATE OF TITLE

Page 1

LOT B IN DEPOSITED PLAN 410947 AT WENTWORTHVILLE LOCAL GOVERNMENT AREA HOLROYD PARISH OF PROSPECT COUNTY OF CUMBERLAND TITLE DIAGRAM DP410947

FIRST SCHEDULE

\_\_\_\_\_\_

MICHAEL ELIAS

SAMIRA ELIAS

AS JOINT TENANTS

#### SECOND SCHEDULE (9 NOTIFICATIONS)

\_\_\_\_\_

#### 1 RESERVATIONS AND CONDITIONS IN THE CROWN GRANT(S)

- 2 M791690 RIGHT OF WAY APPURTENANT TO THE LAND ABOVE DESCRIBED AFFECTING THE LAND SHOWN SO BURDENED IN PLAN WITH M791690
- 3 M791691 RIGHT OF WAY AFFECTING THE LAND SHOWN SO BURDENED IN PLAN WITH M791691
- 4 U968462 EASEMENT TO DRAIN WATER 1.2 WIDE AFFECTING THE PART OF THE LAND ABOVE DESCRIBED SHOWN SO BURDENED IN PLAN WITH U968462
- 5 SP50270 EASEMENT TO DRAIN WATER 2 AND 3 METRE(S) WIDE APPURTENANT TO THE LAND ABOVE DESCRIBED
- 6 AF265509 LEASE TO MICHAEL JAMES DIXON BEING SHOPS 3 & 4 , 116 STATION STREET, WENTWORTHVILLE. EXPIRES: 30/6/2013.
  - AF963964 MORTGAGE OF LEASE AF265509 TO COMMONWEALTH BANK OF AUSTRALIA

AI84710 VARIATION OF LEASE AF265509 EXPIRY DATE NOW

END OF PAGE 1 - CONTINUED OVER

#### doccop1

\*

PRINTED ON 27/3/2015

The Registrar General certifies that at the date and time specified above the person(s) described in the First Schedule was the registered proprietor of an estate in fee simple (or other such estate or interest set out in the Schedule) in the land described, subject to any exceptions, encumbrances, interests, and entries which appear in the Second Schedule.

\* ANY ENTRIES PRECEDED BY AN ASTERISK DO NOT APPEAR ON THE CURRENT EDITION OF THE CERTIFICATE OF TITLE WARNING: THE INFORMATION APPEARING UNDER NOTATIONS HAS NOT BEEN FORMALLY RECORDED IN THE REGISTER.



Registrar General

77

(T 8987303)



division of the Department of Finance & Services

## **TITLE SEARCH**

Computer Folio Certificate issued under Section 96D of the Real Property Act 1900

**No.** 77

\_\_\_\_\_

Search certified to:

27/3/2015 9:54 AM

COMPUTER FOLIO REFERENCE

B/410947

EDITION No. & DATE OF CURRENT CERTIFICATE OF TITLE

Page 2

SECOND SCHEDULE (9 NOTIFICATIONS) (CONTINUED)

31/12/2014.

- 7 AG44293 LEASE TO JANSI RANI JOHN OF SHOPS 10 TO 12, 108-120 STATION STRETT, WENTWORTHVILLE. EXPIRES: 31/8/2012. OPTION OF RENEWAL: 3 YEARS.
- 8 AG516343 MORTGAGE TO BANK OF WESTERN AUSTRALIA LIMITED
- 9 AG789227 LEASE TO AMIT KUMAR & REEMA NAGPAL OF SHOP 5, 108-120 STATION STREET, WENTWORTHVILLE. EXPIRES: 21/9/2014. OPTION OF RENEWAL: 3 YEARS WITH A FURTHER OPTION OF 3 YEARS.

NOTATIONS

\_\_\_\_\_

UNREGISTERED DEALINGS: NIL

\*\*\* END OF SEARCH \*\*\*

doccop1

PRINTED ON 27/3/2015

The Registrar General certifies that at the date and time specified above the person(s) described in the First Schedule was the registered proprietor of an estate in fee simple (or other such estate or interest set out in the Schedule) in the land described, subject to any exceptions, encumbrances, interests, and entries which appear in the Second Schedule.

\* ANY ENTRIES PRECEDED BY AN ASTERISK DO NOT APPEAR ON THE CURRENT EDITION OF THE CERTIFICATE OF TITLE WARNING: THE INFORMATION APPEARING UNDER NOTATIONS HAS NOT BEEN FORMALLY RECORDED IN THE REGISTER. 77

79 Req:C145579 /Doc:CT 15519-074 CT /Rev:23-Dec-2010 /Sts:OK.OK /Prt:27-Mar-2015 09:56 /Pgs:ALL /Seq:1 of 2 Ref:lp sixdre /Src:W ATE OF TITLE **OPERTY ACT, 1900** NEW SOUTH WALES 15519Vol. First Title Old System EDITION 1988 Prior Title Vol. 8442 Fol. 95 ISSUED **~** I certify that the person named in the First Schedule is the registered proprietor of an estate in fee simple (or such other  $\sim$ estate or interest as is set out below) in the land described subject to the recordings appearing in the Second Schedule Fol and to the provisions of the Real Property Act, 1900. ŝ 5 CIRY SEE AUTO FOLIO ALTERING OR ADDING TO THIS CERTIFICATE OR ANY NOTIFICATION HEREON LAND REFERRED TO Lot B in DP410947 at Wentworthville in the Municipality of Holroyd Parish of Prospect County (Page I) Vol. of Cumberland. Title Diagram: DP410947. FIRST SCHEDULE H161492 RALIAN PTY. LIMITED. OF THE REG £ B SECOND SCHEDULE 1. Reservations and conditions in the Crown grant. EDZ2. M618564 Easement for drainage appurtenant to the land above described affecting the land shown AUTHEN so burdened in plan with M618564. RWZ3. M791690 fasement for right of way appurtenant to the land above described affecting the land shown so burdened in plan with M791690 OUGH AND 4. R286491 Mortgage to National Australia Bank Land the land shown so burdened in plan with M791691. 5. M791691 Easement for right of way affecting the land shown so burdened in plan with M791691. PERSONS ARE CAUTIONED AGAINST RW 6. V445930 Lease to Jennifer Anne Wall and Kathleen Margaret Formosa, as Joint Tenants of Shop 5. Ralian Centre, 108-120 Station Street, Wentworthville together with and reserving rights. THR Expires 31-12-1986. Option of renewal. u(L) 7. V445930PLease V445931 Transfer. Lessees now Isabel Mary Smith and Merridee Lyn Renshall, as RULED Joint Tenants. 8. X267542\_Mortgage to Custom Credit Corporation limited. X586112 9. X267543 Lease to J.K.N. Holdings Pty. Limited of Shop No.4, Ralian Centre, 108-120 Station ŝ 1 Street, Wentworthville. Expires 25-11-1989. Option of renewal 5 years. 10. X267544 Lease to Frank Pellegrino and Joan Pellegrino, as Joint Tenants of Shop No.3, Ralian NTRI ΠŢ Centre, 108-120 Station Street, Wentworthville. Expires 25-11-1988. Option of renewalλÛ, 5 years. LON 11. X267546 Lease to Piero Bartolacci and Inelda Bartolacci, as Joint Tenants of Shops Nos. 1 and 2, shown in plan with X267546. Expires 25-11-1989.

LT 2/64

(Page 2 of 2 pages) 15519 Fol. 74 Vol 0. 55 FIRST SCHEDULE (continued) **REGISTERED PROPRIETOR Registrar** General CANCELLED SEE AUTO FOLIO SECOND SCHEDULE (continued) PARTICULARS CANCELLATION **Registrar General** X385210 Transfer - Easement to drain water appurtenant to the land within described affecting the land shown as "site of proposed drainage easement" in plan with M618564. Registered 18-3-1988 1988316-6-1988 X586116 Mortgage to Farrow Mortgage Services Pty. Limited Registered (19-6-Farrow Mortgage Dervices Lange Ist Y18420 agistered 16-3-1989. x26/546 Lease Y418231 Transfer to Devadi Pty. Limited. Registered 26-7-1989.



Jna	ensee: Midware S mpion Legal	PRIVACY NOTE: this information is legally requ	Act 1900	8987303V	
	STAMP DUTY	Office of State Revenue use only		SECTION 18(2)	01122807-001 \$\$*******2_00
(A)	TORRENS TITLE	If appropriate, specify the part transferred Follo identifier B/410947		· · · · · · · · · · · · · · · · · · ·	
( <b>B</b> )	LODGED BY	Delivery Box (407) Reference (optional): DHF	ohone (A) AC		CODE T TW (Sheriff)
(C)	TRANSFEROR	ANTONIO DA SILVA DUARTE and MAR	IA DUARTE		
(D)	CONSIDERATION	The transferor acknowledges receipt of the cor	sideration of \$ 2	,280,000.00	and as regards
(E)	ESTATE	the land specified above transfers to the transfe	eree an estate in t	ieo simple.	
(F)	Share Transferred				
(G)		Encumbrances (if applicable):			
(H)	TRANSFEREE	MICHAEL ELIAS and SAMIRA ELIAS			
(I)		TENANCY: Joint Tenants			
ወ	DATE	······································			•
	personally acqua	person(s) signing opposite, with whom I am inted or as to whose identity I am otherwise this instrument in my presence.	Certified corr 1900 by the tr	ect for the purposes of the Rea ansferor.	Property Act
		vesna Nitrolouska	Signature of t	ia Decar	Ce)
		ss: 90-92 Auchley SI	HMA	in Derati	to
	a successfully and trachle	petersham new 2019	Har	au warne	~~~,
			Certified corr	ect for the purposes of the Rea erson whose signature appears	l Property Act
•		•		/	
			Signature:	#	
				ame: PAT SPITHILL apacity: Solicitor for the transfe	eree

1988	STAMP DUTY			onveyai		J	
	# 2-					191285 G	
	<i></i>	TRANSFER BY MO UNDER POWER	OF SALE	E TP	to t	R -	Jel 1
		REAL PROPERTY AC (See instructions for Completion	T, 1900 on back of form))		₽		
	Torrens Title Reference	II Part Only, Dete	ale Whole and Give De	ntails	Loc	ation	· ·
DESCRIPTION OF LAND Note (a)	FOLIO IDENTIFIER B/4109	947	WHOLE	-	MOSMAN	- - - - -	ر مستحد مرتبط و در ۱۵ مرتبع مرتبط و در ۱۵ مرتبع مرتبط و در ۱۵ مرتبط مرتبط و مرتبط و در ۱۵ مرتبط مرتبط و مرتبط و مرتبط و مرتبط و مرتبط و مرتبط و مرتبط و مرتبط و مرتبط و
TRANSFEROR Note (b)	FARROW MORFGAGE SERVICE Level 11, 459 Collins S as mortgagee exercising	treet, Melbourne	JIDATION) AC	N 006 1:	25 757		
	(the abovenamed THANSFEROR) being the reg						
MORTGAGOR S	dated 24 May 1988, from And a	њемовтолоов, <i>RALIAN 1</i> 8 February 1989 respec	<i>PTY LIMITEL</i> tively	0.		;	
001449114 Note (d)	hereby acknowledges receipt of the consideratio and, as mongages under the abovementioned m	n of \$ 790,000.00 origage, transfers in exercise of power of	sale an estate in fee si	imple in the lar	a above described to the upder	mentioned TRANSFE	100 L
TRANSFEREE G	ISIKIT PTY LIMITED ACN of 4/76 Campbell Parade		· .				Ha/2827
TENANCY Note (e)	-oc jolist Ionantsfremants in common						H.
PRIOR	aubjects for following PRIOR ENCUMBRANCE	s y Lease No. 2193326					
ENCLIMBRANCES Note (f)	. Lease Z525410		form N	IO. Z525	411 4 Taina N	T. MEDELEN	
note (i)			. 3 pease p		411 4. /Lechie r	No. Z606154	
	5. Lease d		b. hease f	¢'	7. Lease \$	ULCES OF	
	5. lease d		6. Lease ) The Common	Seal of Fam	7. Lease #	LAN SERVICES PT	
**************************************	5. lease d	ug for 1992	6. Lease ) The Common	Seal of Fam	7. Lease #	LAN SERVICES PT	N CHILDREN
ta Nu CC EXECUTION	5. lease d	warbon 1992 purposes of the Real Property Act, 1900.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Fact m) was here: mes Patrick corge Hodgs rices Pty Lid the Suprem	7. Lease 5 now Martgage Services Pty anto affixed in the Dewney for and on behaf oi) (Liquidator of Farrow (in Liquidation)) pursue	THE COMMC	
\$2.00	5. Lease d DATE g.th. No	purposes of the Real Property Act, 1900.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Fact m) was here: mes Patrick corge Hodgs rices Pty Lid the Suprem	7. Lease 5 now Martgage Services Pty anto affixed in the Dewney for and on behaf oi) (Liquidator of Farrow (in Liquidation)) pursue	THE COMMC	
H EXECUTION Note (g)	5. Lease of DATE <u>J. M. Mo</u> We hereby carily this dealing to be correct for the Stand in mycrosence by the transferor who is p	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Fact m) was here: mes Patrick corge Hodgs rices Pty Lid the Suprem	7. Lease 5 now Martgage Services Pty anto affixed in the Dewney for and on behaf oi) (Liquidator of Farrow (in Liquidation)) pursue	SANGES P) SANGES P) THE COMMC SEAL	
H EXECUTION Note (g)	5. Lease of DATE <u>J. M. Mo</u> We hereby carily this dealing to be correct for the Stand in mycrosence by the transferor who is p	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Fact m) was here: mes Patrick corge Hodgs rices Pty Lid the Suprem	7. Lease 5 now Martgage Services Pty anto affixed in the Dewney for and on behaf oi) (Liquidator of Farrow (in Liquidation)) pursue	THE COMMC	
H EXECUTION Note (g)	5. Lease of DATE <u>J. M. Mo</u> We hereby carily this dealing to be correct for the Stand in mycrosence by the transferor who is p	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Fact m) was here: mes Patrick corge Hodgs rices Pty Lid the Suprem	7. Lease 5 now Martgage Services Pty anto affixed in the Dewney for and on behaf oi) (Liquidator of Farrow (in Liquidation)) pursue	THE COMMC	
H EXECUTION Note (g)	5. Lease of DATE <u>J. M. Mo</u> We hereby carily this dealing to be correct for the Stand in mycrosence by the transferor who is p	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Fact m) was here: mes Patrick corge Hodgs rices Pty Lid the Suprem	7. Lease 5 now Martgage Services Pty anto affixed in the Dewney for and on behaf oi) (Liquidator of Farrow (in Liquidation)) pursue	LA SENTICES PO SEAL OF NAME	
H EXECUTION Note (g)	5. Lease of DATE <u>J. M. Mo</u> We hereby carily this dealing to be correct for the Stand in mycrosence by the transferor who is p	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Fact m) was here: mes Patrick corge Hodgs rices Pty Lid the Suprem	7. Lease 9 now Montgage Services Pay unto affixed in the Dewney for and on beha on (Liquidator of Farrow (In Liquidation)) pursual te Court of Victoria made inther presence of	LA SENTICES PO SEAL OF NAME	
H EXECUTION Note (g)	5. Lease of DATE <u>J. M. Mo</u> We hereby carily this dealing to be correct for the Stand in mycrosence by the transferor who is p	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Fact m) was here: mes Patrick corge Hodgs rices Pty Lid the Suprem	7. Lease 9 now Montgage Services Pay unto affixed in the Dewney for and on beha on (Liquidator of Farrow (In Liquidation)) pursual te Court of Victoria made inther presence of	LA SENTICES PO SEAL OF NAME	
H EXECUTION Note (g)	5. Lease of DATE <u>J. M. Mo</u> We hereby carily this dealing to be correct for the Stand in mycrosence by the transferor who is p	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Fact m) was here: mes Patrick corge Hodgs rices Pty Lid the Suprem	7. Lease 9 now Montgage Services Pay unto affixed in the Dewney for and on beha on (Liquidator of Farrow (In Liquidation)) pursual te Court of Victoria made inther presence of	LA SENTICES PO SEAL OF NAME	
H EXECUTION Note (g)	5. Lease of DATE <u>J. M. Mo</u> We hereby carily this dealing to be correct for the Stand in mycrosence by the transferor who is p	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Fact m) was here: mes Patrick corge Hodgs rices Pty Lid the Suprem	7. Lease 9 now Montgage Services Pay unto affixed in the Dewney for and on beha on (Liquidator of Farrow (In Liquidation)) pursual te Court of Victoria made inther presence of	LA SENTICES PO SEAL OF NAME	
H EXECUTION Note (g)	5. Lease of DATE <u>J. M. Mo</u> We hereby carily this dealing to be correct for the Stand in mycrosence by the transferor who is p	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G- Merigage Ser- toran Order of	Seal of Farm m) was here mes Patrick corge Hodgs rices Ply Ltd (the Supression 992 (fithe fithe 992 (fithe fithe source)	7. Lease 7. Lease row Montgage Services Pay unto affixed in the Dewney for and on beha (i) (Liquidation) pursual (in Liquidation)) pursual is Court of Victoria made inther presence of: Signature of Tran Maille Signature of Tran	Sforces	
H EXECUTION Note (g)	5. Lease d DATE <u>J.M. No</u> We hereby carily this dealing to be correct for the Signed in mycrosonce by the transferor who is p Signature of Witness CULLITE JELL Name of Witness (BLOCK LETTER 535 SULAKE, STALET MELSON Address and occupation of Witness Signature of Witness Signature of Witness Marne of Witness (BLOCK LETTER Name of Witness (BLOCK LETTER Name of Witness (BLOCK LETTER Name of Witness (BLOCK LETTER 13 - Address and occupation of Witness	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony Ga Marigage Services In March, P	Seal of Farm m) was herein mes Patrick corge Hodgs rices Pty Ltd (the Suprem 992 if the fi 992 if the fi 	7. Lease 7 row Mortgage Services Pty moto affixed in the Dewney for and on beha on (Liquidation) pursuit (In Liquidation) pursuit in the presence of the Signature of Tran Signature of Tran ANNE ELIZAE	Sforces	
Note (g) Note (g) Not	5. Lease of DATE <u>J.M. No</u> We hereby carily this dealing to be convect for the	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony Ga Merigage Ser- to an Order of 11th March, H	Seal of Farm m) was herein mes Patrick corge Hodgs rices Pty Ltd (the Suprem 992 if the fi 992 if the fi 	7. Lease 7. Lease row Montgage Services Pay unto affixed in the Dewney for and on beha (i) (Liquidation) pursual (in Liquidation)) pursual is Court of Victoria made inther presence of: Signature of Tran Maille Signature of Tran	Sforces	
Note (g) Note (	5. Lease d DATE <u>J.M. No</u> We hereby carily this dealing to be correct for the Signed in mycrosonce by the transferor who is p Signature of Witness CULLITE JELL Name of Witness (BLOCK LETTER 535 SULAKE, STALET MELSON Address and occupation of Witness Signature of Witness Signature of Witness Marne of Witness (BLOCK LETTER Name of Witness (BLOCK LETTER Name of Witness (BLOCK LETTER Name of Witness (BLOCK LETTER 13 - Address and occupation of Witness	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony Ga Merigage Ser- to an Order of 11th March, H	Seal of Farm m) was herein mes Patrick corge Hodgs rices Pty Ltd (the Suprem 992 if the fi 992 if the fi 	7. Lease 7 row Mortgage Services Pty moto affixed in the Dewney for and on beha on (Liquidation) pursuit (In Liquidation) pursuit in the presence of the Signature of Tran Signature of Tran ANNE ELIZAE	Sforces	
Note (g) Note (g) Not	5. Lease d DATE <u>J.M.</u> Mo We hereby carily this dealing to be correct for the Signature of Witness Signature of Witness Signature of Witness (BLOCK LETTER S35 SOURCE, STALET MELSOU Address and occupation of Witness Signature of Witness Marme of Witness (BLOCK LETTER 3 Address and occupation of Witness Name of Witness (BLOCK LETTER 3 Address and occupation of Witness LODGED BY Arch	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony Ga Merigage Ser- to an Order of 11th March, H	Seal of Farm m) was herein mes Patrick corge Hodgs rices Pty Ltd (the Suprem 992 if the fi 992 if the fi 	7. Lease 7 row Montgage Services Pay unto affixed in the Dewney for and on beha on (Liquidator) pursue (In Liquidation) pursue in the presence of Signature of Tran Bignature of Tran ANNE ELIZABI COATION OF DOCUMENTS Herewith.	Sforces	
Note (g) Note (	5. Lease d DATE <u>J.M.</u> No We hereby carily this dealing to be conect for the Signature of Witness Signature of Witness Signature of Witness (BLOCK LETTER 53.5 SOURCE, STATET MELSOU Address and occupation of Witness Signature of Witness Signature of Witness Marne of Witness (BLOCK LETTER 3. Address and occupation of Witness Name of Witness (BLOCK LETTER 3. Address and occupation of Witness I LODGED BY Address and occupation of Witness Marne of Witness (BLOCK LETTER 3. Address and occupation of Witness I LODGED BY Address and occupation of Witness	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony Ga Merigage Ser- to an Order of 11th March, H	Seal of Farm m) was herein mes Patrick corge Hodgs rices Pty Ltd (the Suprem 992 if the fi 992 if the fi 	7. Lease 7 row Mortgage Services Pay unto affixed In the Dewney for and on beha on (Liquidator of Farrow (In Liquidation)) pursuit is Court of Victoria made inther presence of: Signature of Tran ANNE ELIZAEI DECATION OF DOCUMENTS Herewith. In L.T.Q. with	Sforces	
Note (g) Note (	5. Lease d DATE <u>J.M.</u> Mo We hereby carily this dealing to be correct for the Signed in mysresence by the transferor who is p Signeture of Witness WHITELEL Name of Witness (BLOCK LETTER 535 BUAKE, STALET MELGOL Address and occupation of Witness Signature of Witness Signature of Witness Name of Witness (BLOCK LETTER 3. Address and occupation of Witness Name of Witness (BLOCK LETTER 3. Address and occupation of Witness ICOGED BY Address and occupation of Witness LODGED BY Address and occupation of Witness LODGED BY Address and occupation of Witness LODGED BY Address and occupation of Witness Address and occupation of Witness	<ul> <li>purposes of the Real Property Act, 1900.</li> <li>personally known to me.</li> <li>(5)</li> <li>(5)</li> <li>(5)</li> <li>(6)</li> <li>(7)</li> <li>(7)&lt;</li></ul>	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony Ga Merigage Ser- to an Order of 11th March, H	Seal of Farm m) was herein mes Patrick corge Hodgs rices Pty Ltd (the Suprem 992 if the fi 992 if the fi 	7. Lease 7 row Montgage Services Pay into affixed in the Dewney for and on beha (in Liquidator of Farrow (in Liquidation)) pursual inther presence of: Signature of Tran ANNE ELIZABI DCATION OF DOCUMENTS Herewith. In L.T.Q. with Produced by	Steres 28, 14	
Note (s) Note (	5. Lease d DATE <u>J.M.</u> Mo We hereby carily this dealing to be correct for the Signed in mysresence by the transferor who is p Signeture of Witness WHITELEL Name of Witness (BLOCK LETTER 535 BUAKE, STALET MELGOL Address and occupation of Witness Signature of Witness Signature of Witness Name of Witness (BLOCK LETTER 3. Address and occupation of Witness Name of Witness (BLOCK LETTER 3. Address and occupation of Witness ICOGED BY Address and occupation of Witness LODGED BY Address and occupation of Witness LODGED BY Address and occupation of Witness LODGED BY Address and occupation of Witness Address and occupation of Witness	e purposes of the Real Property Act, 1900. personally known to me.	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony G. Mcrigage Service rean Order of 11th March, H Soliciton CT OT Secondary	seal of Farm m) was here mes Patrick corge Hodgs rices Ply Lud (the Suprem 992 ) fithe fit 992 ) fithe fit search of the fit of the Lot HER	7.       Lease         row Montgage Services Pay unto affixed in the Dewney for and on beha on (Liquidator of Farrow (In Liquidation)) pursuit (In Liquidation))         Elements       Signature of Train ANNE ELIZABI DOCATION OF DOCUMENTS         Horewith.       In L.T.Q. with Produced by         Market State       Signature of Train ANNE ELIZABI DOCATION OF DOCUMENTS	In State	
Note (g) Note (	5. Lease d DATE <u>J.M.</u> Mo We hereby carily this dealing to be correct for the Signed in mean end with the transferror who is p Signeture of Witness WHITE JEL Name of Witness (BLOCK LETTER 535 GUARS, STARET, MELGOL Address and occupation of Witness Signature of Witness Signature of Witness Signature of Witness Name of Witness (BLOCK LETTER 3 Address and occupation of Witness LODGED BY Address and occupation of Witness Ref. Definery Box Number Checked Passed REG	<ul> <li>purposes of the Real Property Act, 1900.</li> <li>personally known to me.</li> <li>(5)</li> <li>(5)</li> <li>(5)</li> <li>(6)</li> <li>(7)</li> <li>(7)&lt;</li></ul>	6. Lease ) The Common (In Liquidatio presence of Ja of Anthony Ga Mengage Service them Order of 11th March, H Soliciton CT OT	seal of Farm m) was here mes Patrick eorge Hodgs rices Ply Lid (the Suprem 992) of the fr 992 of the fr 992 of the fr 100 million 100 mill	7. Lease 7 row Montgage Services Pay into affixed in the Dewney for and on beha (in Liquidator of Farrow (in Liquidation)) pursual inther presence of: Signature of Tran ANNE ELIZABI DCATION OF DOCUMENTS Herewith. In L.T.Q. with Produced by	In State	

. •	Form: 07L Release: 2.8 www.lpma.nsw.j	30v.au	LEASE New South Wales Real Property Act 1900	AG7892	27P		
	by this form to the Register is m	r the establis	of the Real Property Act 1900 (RP Act) authorises shment and maintenance of the Real Prope to any person for search upon payment of a fee	the Registrar General to collect the in rty Act. Register, Section 968, BP	formation required		
	STAMP DUTY	Office of S	tate Revenue use only				
(A)	TORRENS TITLE	Property le					
		B/41094	7 part being Shop 5, 108-120 St	ation Street, Wentworthvi	lle		
(B)	LODGED BY	Document	Name, Address or DX, Telephone, and Custo	mer Account Number if any	CODE		
		Collection Box	Paramonte Legal DX 8263 PARRAMATTA PH:(02) 9635 7388				
(C)	LESSOR	Michael	Reference: MH:2110138 ELIAS & Samira ELIAS	RELODGE			
				t 4 FEB 2012			
(D)			uses to the lessee the property referred to above. cs (if applicable):	TIME: 121	5		
(E)	LESSEE	Amit KUM	AR & Reema NAGPAL				
(F)		TENANCY;	Joint Tennet				
(G)	<ol> <li>TERM 3 Y</li> <li>COMMENCIN</li> <li>TERMINATING</li> </ol>	G DATE	22 SEPTEMBER 2011 2   SEPTEMBER 2014				
	set out in cla	ise 4	W for a period of 3 YEARS x 3 YEARS of ANNEXURE B AND ITEM 12				
	6. Together with	1 and reservic	HASE set out in clause N.A. of N ig the RIGHTS set out in clause N.A.	of N.A.	·		
	8. Incorporates	<ol> <li>Incorporates the provisions or additional material set out in ANNEXURE(S) A&amp;B hereto.</li> <li>Incorporates the provisions set out in N.A. with the Land and Property Management Authority as No. N.A.</li> </ol>					
	9. The RENT is						
	ALL HANDWRITING	MUST BE IN B	LOCK CAPITALS. Page 1 of 17	LAND AND PROPIRTY MANA	GEMENT AUTHORITY		

.

for p

DATE

(H) I certify that the person(s) signing opposite, with whom I am personally acquainted or as to whose identity I am otherwise satisfied, signed this instrument in my presence.

Signature of witness:

Name of witness: Address of witness:

Mireille Haddad Solicitor Suite 5, 2 Horwood Place Parramatta NSW 2150

I certify that the person(s) signing opposite, with whom I am personally acquainted or as to whose identity I am otherwise satisfied, signed this instrument in my presence.

Signatu witne

Name of witness: VATIND 5R KUMAR BACHLA Address of witness: 319 BLACK TOWN ROAD ROSPECT MSW 2148

Certified correct for the purposes of the Real Property Act 1900 by the lessor.

Signature of lessor:

Certified correct for the purposes of the Real Property Act 1900 by the lessee.

Signature of lessee:

STATUTORY DECLARATI 1

solemnly and sincerely declare that-

1. The time for the exercise of option to

in expired lease No. 2. The lessee under that lease has not exercised the option.

has ended; and

I make this solemn declaration conscientiously believing the same to be true and by virtue of the provisions of the Oaths Act 1900.

Made and subscribed at

on

**(I)** 

Signature of witness:

in the presence of-

Signature of lessor:

in the State of New South Wales

Full name of witness:

Address of witness:

Qualification of witness: [tick one]

Justice of the Peace

D Practising Solicitor

Other qualified witness *[specify]* 

\* As the Land and Property Management Authority may not be able to provide the services of a justice of the peace or other qualified witness, the statutory declaration should be signed and witnessed prior to lodgment.

ALL HANDWRITING MUST BE IN BLOCK CAPITALS.

#### ANNEXURE A

,

.

Annexure A 👘

÷

SEE A SOLICITOR ABOUT THIS LEASE

Landtord: MICHAEL ELIAS & SAMIRA ELIAS

#### Tenant: AMIT KUMAR & REEMA NAGPAL T/AS FEMINA BEAUTY SALON

This annexure consists of 2 pages.

NOTE: Any alterations and additions to Lease Covenants In Annexure B must be made by additional clauses in Annexure A. The printed clauses in Annexure B are to remain in their copyright form without alteration.

#### SCHEDULE OF ITEMS (continued)

ltem 10 (cls 2.3, 13.1)	A.	The guarantor:	Reema Nagpal	& Amil Kumar
(ci 13.7)	8,	Limit of guarantor's liability:	N/A	
ltem 11 (cl 3)	Additic	onal leased property:	N/A	
ltem 12 (cl 4)	Option	to renew		
(0 4)	А.	Further period of 3 years from $22$	/09/2014 to .21/0	9/2017.
	8.	Further period of 3 years from $22$	7/09/201 <b>7</b> to /99	109/2020
	C.	Maximum period of tenancy under	this lease and pe	rmitted renewals: years
	D.	First day option for renewal can be	exercised:	22 /03/2014
	E.	Last day option for renewal can be	exercised: 2	2 /06/2014
ltem 13	А.	Rent		
(ci 5)	For the	lease period:		
		From 22 November 2011 to the first rent review date:		\$40,040.00 (including GST) per annum by monthly instalments of \$3,336.67
		Afterwards:		At the new yearly rent beginning on each review date by monthly instalments of one twelfth of the new yearly rent.
		For the further period in item 12A: From the commencement date to the first rent review date:		
		(for example: Current market rent)		Indexed to CPI
		Afterwards:		At the new yearly rent beginning on each review date by monthly instalments of one twelfth of the new yearly rent.

Page 3 of 17

M.E & Z

•		For the further period i		Annexure
		From the commencent to the first rent review (for example: Current	date:	Indexed to CPI
		Afterwards:		At the new yearly rent beginning on each revi date by monthly instalments of one twelfth of I new yearly rent.
ltem 13 (ci 15)	В.	Goods and Services	Tax	
	Clause	e 15 provides for paymen	t by the tenant of Goods	and Services Tax unless otherwise here indicated:
ltem 14 (cl 5)	Outgo	pings		
(0.0)	Α.	Share of outgoings:	100%	
	В.	Outgoings -		
		(a) Water Usage		
		for the land or the build	ding of which the property	y is part, fairly apportioned to the period of this lease.
ltern 15 (cl 5.1.5)	Intere	st rate:	%	
ltem 16 (cl 5.4)	Rent i	review		
	Rent r	review date	Method of rent revie	w If Method 1 applies, increase by (the increase should show percenta or amount)
	10 Se	ptember 2012 ptember 2013 ptember 2014	Method 2 Method 2 Method 2	Not Applicable
	. Metho	d 1 is a fixed amount or p d 2 is Consumer Price Ind d 3 is current market rent	dex.	
	Metho	d 2 applies unless anothe	r method is stated.	
ltem 17 (cl 6.1)	Permi	tted use: Beauty Salon		
	Amou	nt of required public lia	bility insurance: \$10 M	illion Dollars
item 18 (cl 8.1.1)	•			
(cl 8.1.1) Item 19	Bank (	Guarantee		
(cl 8.1.1)		Guarantee 10.00, which is equivalent	to three (3) months rent,	inclusive of GST

0 6

.

•

Page 4 of 17

J.E

82 Req:C145582 /Doc:DL AG789227 /Rev:17-Feb-2012 /Sts:OK.OK /Prt:27-Mar-2015 10:20 /Pgs:ALL /Seq:5 of <u>17</u> Ref:lp;:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

Page 1 of 13 pages Annexure B

SEE A SOLICITOR ABOUT THIS LEASE

#### ANNEXURE B

#### Landlord: MICHAEL ELIAS & SAMIRA ELIAS

Tenant: AMIT KUMAR & REEMA NAGPAL T/AS FEMINA BEAUTY SALON

This annexure consists of 14 pages.

©2005 COPYRIGHT of the Law Society of New South Wales which has approved this annexure as printed to clause 15. WARNING: Unauthorised reproduction in whole or in part is an infringement of copyright.

NOTE: Any alterations and additions to Lease Covenants in Annexure B must be made by additional clauses in Annexure A. The printed clauses in Annexure B are to remain in their copyright form without alteration.

#### CONTENTS

CLAUSE	SUBJECT	PAGE	CLAUSE	SUBJECT	PAGE
1	Form of this Lease	2	9	Access	8
2	Partles	2	10	Transfer and Sub-Lease	8
3	The Property	2	11	Landlord's other Obligations	9
4	Lease Period	2	12	Forfeiture and End of Lease	10
5	Money	3	13	Guarantee	11
6	Use	6	14	Exclusions, Notices and Special Clauses	11
7	Condition and Repairs	6	15	Goods and Services Tax	12
8	Insurance and Damage	7		· · · · · · · · · · · · · · · · · · ·	

#### **RETAIL LEASE CERTIFICATE**

If section 16 of the Retail Leases Act 1994 applies to this lease, and the term plus any further terms are less than 5 years, the term will be extended unless a section 16 certificate is given. Sections 16(1) and (2) provide -

- 16(1) The term for which a retail shop lease is entered into, together with any further term or terms provided for by any agreement or option for the acquisition by the lessee of a further term as an extension or renewal of the lease, must not be less than 5 years. An agreement or option is not taken into account if it was entered into or conferred after the lease was entered into.
- 16(2) If a lease is entered into in contravention of this section, the validity of the lease is not thereby affected but the term of the lease is extended by such period as may be necessary to prevent the lease contravening this section.

I certify that I am a solicitor not acting for the landlord and that at the request of the tenant I explained to the tenant before the tenant entered into this lease -

I the effect of sections 16(1) and (2); and

I that the giving of this certificate would result in section 16 not applying to this lease.

Date		Signature				
			NAME (BLOCH			
	\$4\$\$\$\$		*******			******
Ø	ß	Page 5 of 1		M.F		

	, ,	
. •	·	Page 2 of 13 pages Annexure B
	CLAUSE 1	FORM OF THIS LEASE
•		What are the parts to this lease?
	1.1	There are three parts to this lease - a lease form, Annexure A and this annexure.
	1.2	This lease is a deed even if it is not registered.
	CLAUSE 2	PARTIES
		Who are the parties to this lease?
	2.1	The landtord. The landlord is also called the lessor (or in the case of a sublease, the sublessor) and is named on page 1 of this lease.
	2.2	The tenant. The tenant is also called the lessee (or in the case of a sublease, the sublessee) and is named on page 1 of this lease.
	2.3	The guarantor, if a guarantor is named in item 10 in the schedule.
	2.4	If a party consists of two or more persons, obligations of that party can be enforced against any one or more of them.
	CLAUSE 3	THE PROPERTY
		What property is leased?
	3.1 <sup>.</sup>	The property leased is described on page 1 of this lease.
	3.2	The landlord's fixtures are included in the property leased.
	3.3	If anything else is leased (such as furniture belonging to the landlord) and is described in item 11 in the schedule It is included in the property.
	3.4	If the property has facilities and services shared in common with other persons in the same building as the property, clause 11.3.2 applies to those common facilities. The tenant shares the common facilities with the landlord, and with other tenants of the landlord. The tandlord can set reasonable rules for sharing these common facilities.
	CLAUSE 4	LEASE PERIOD
		How long is this lease for?
	4.1	This lease is for the period stated in item 1 in the schedule, commences on the date stated in item 2 in the schedule and ends on the date stated in item 3 in the schedule.
	4.2	If a further period, commencing when this lease ends, is stated in item 12A in the schedule then the tenant has the option to renew this lease for that period.
	4.3	The tenant can renew this lease more than once if that is stated in item 12B in the schedule. However the period of tenancy under this lease and under any renewal(s) is, in total, not longer than the maximum period stated in item 12C in the schedule.

. (~

٦

٠

ME \$.2

.

Page 3 of 13 pages Annexure B

4.4 The tenant can exercise the option only if 4.4.1 the tenant serves on the landlord a notice of exercise of option not earlier than the first day stated in item 12D in the schedule and not later than the last day stated in item 12E in the schedule;

- 4.4.2 there is at the time of service no rent or outgoing that is overdue for payment; and
- 4.4.3 at the time of service all the other obligations of the tenant have been complied with or fully remedied in accordance with the terms of any notice to remedy given by the landlord.

If this lease is extended by legislation, items 12D and 12E in the schedule are adjusted accordingly.

- 4.5 After exercising the option the tenant must continue to pay all rents and outgoings on time and continue to comply with all of the tenant's obligations under this lease. If the tenant does not do so, the landlord may treat any breach as being a breach of the new lease as well as of this lease.
- 4.6 A new lease will be the same as this lease except for -
  - 4.6.1 the new rent;
  - 4.6.2 the commencement date and the termination date;
  - 4.6.3 the omission of clauses 4.2, 4.9, 4.4, 4.5 and 4.6 and items 12A and 12B in the schedule in the last lease allowed in item 12 in the schedule;
  - 4.6.4 Item 12B becoming item 12A;
  - 4.6.5 adjustment of item 12C in the schedule; and
  - 4.6.6 adjustment of items 12D and 12E in the schedule. The number of days between the dates stated in items 12D and 12E in the schedule of the new lease and the termination date of the new lease and the number of days between each date stated in items 12D and 12E in the schedule of this lease and the termination date of this lease are to correspond.

If the new rent is to be current market rent it will be decided in the same way that current market rent is to be decided under Method 3 stated in clause 5 assuming that this lease and the new lease were one continuous lease and the commencement date of the new lease was a rent review date.

#### CLAUSE 5 MONEY

- What money must the tenant pay?
- 5.1 The tenant must pay to the landlord or as the landlord directs -
  - 5.1.1 the rent stated in item 13A in the schedule;
  - 5.1.2 the share stated in item 14A in the schedule of those outgoings stated in item 14B in the schedule;
  - 5.1.3 the reasonable cost to the landlord of remedying a default by the tenant;
  - 5.1.4 the reasonable cost to the landlord of dealing with any application by the tenant for the landlord's consent under this lease (whether or not it is given);
  - . 5.1.5 interest on these moneys at the rate stated in item 15 in the schedule when payment is more than 14 days overdue, calculated from the due date to the date of payment;
  - 5.1.6 registration fee for registration of this lease at Land and Property Information NSW (payable on delivery to the landlord's solicitor of the executed lease);
  - 5.1.7 stamp duty on this lease (payable on delivery to the landlord's solicitor of the executed lease) if not previously paid by the tenant to the Office of State Revenue;
  - 5.1.8 if the tenant defaults, the landlord's reasonable legal costs relating to the default;
  - 5.1.9 the landlord's reasonable costs and expenses in connection with the preparation of this lease but only that part of those costs and expenses which are permitted to be recovered by a landlord under section 14 and section 45 of the Retail Leases Act, 1994; and
  - 5.1.10 Goods and Services Tax as provided for in clause 15.
  - The first month's instalment of rent is to be paid by the commencement date. Each later month's instalment of rent is to be paid in advance.

Page 7 of 17

ME \$.Z

5.2

Page 4 of 13 pages Annexure B 5.3 A payment under clause 5.1.2 must be paid on the next rent day after a request for payment is made by the landlord. · A request for payment can be made after the landlord has paid an outgoing; or 5.3.1 5,3,2 after the landlord has received an assessment or account for payment of an outgoing. If item 14B in the schedule refers to land tax -If the property is a strata lot, the relevant land tax is land tax on that lot; f if the property is not a strata lot but is part of a building, the relevant land tax is land tax on the land on A which the building is situated, plus any land of the landlord used or available for use by or for the benefit of tenants conducting business in the building or in connection with trading in the building; and in either case, the land tax must be calculated as if the land was the only land owned by the landlord 1 and there was no special trust or non-concessional company involved. When and how is the rent to be reviewed? 5.4 The rent is to be reviewed on the rent review dates stated in item 16 in the schedule. If this lease is extended by legislation, the rent review dates include each anniversary of the latest rent review date stated in item 16 in the schedule (or if none is stated each anniversary of the commencement date) which falls during the extension. 5.5 The tenant must continue to pay rent at the old rate until the new rate is known. After that, the tenant is to pay the new rent from the next rent day. By that rent day the tenant is also to pay any shortfall between the old and new rate for the period since the rent review date. Alternatively, the landlord is to refund to the tenant any overpayment of rent. 5.6 There are three different methods described here for fixing the new rent on a rent review date. The method agreed by the landlord and the tenant is stated at item 16 in the schedule. The tenant is entitled to a reduction if the method produces a rent lower than the rent current just before the review date. Method 1. By a fixed amount or percentage. 5.7 In this case the new rent beginning on each review date is stated in item 16 in the schedule. Method 2. By reference to Consumer Price Index. 5.8 In this case take the yearly rent as of the last review date or if none, the rent at the commencement date (\$X), Ħ

- divide that rent by the Consumer Price Index Number for Sydney (All Groups) for the quarter ended just before that date (CPI 1).
- multiply the result by the Consumer Price Index Number for Sydney (All Groups) for the quarter ended just before the review date (CPI 2).

The product is the new rent for the year beginning on the review date (\$Y), written as a formula -

Page 8 of 17

ME \$.2

•	
۱ ۹ - ۲ ۰	Page 5 of 13 pages Annexure B
5.9 -	The landlord must calculate the new rent after each review date and give the tenant written notice of the new rent.
5.10	If the Australian Bureau of Statistics makes a change in the reference base of the index and there is a published co-relation between the old and new base then the published co-relation is to be applied to convert the CPI 1 figure to the new reference base. If there is none then the fandlord and the tenant agree to accept the calculations of the landlord's solicitor who must be retained to determine a fair co-relation between the old and the new series of numbers.
5.11	If the index used to calculate the new rent is discontinued the landlord may substitute another index that, as , nearly as practicable, serves the same purpose and, if there is no such index, then the rent will be fixed by Method 3.
	Method 3. By reference to current market rent.
5.12	In this case the rent is to be the current market rent. This can be higher or lower than the rent payable at the rent review date and is the rent that would reasonably be expected to be paid for the property, determined on an effective rent basis, having regard to the following matters-
	<ul> <li>5.12.1 the provisions of this lease;</li> <li>5.12.2 the rent that would reasonably be expected to be paid for the property if it were unoccupied and offered for renting for the same or a substantially similar use to which the property may be put under this lease;</li> <li>5.12.3 the gross rent, less the landlord's outgoings payable by the tenant;</li> <li>5.12.4 where the property is a retail shop, rent concessions and other benefits that are frequently or generally offered to prospective tenants of unoccupied retail shop; and</li> <li>5.12.5 the value of goodwill created by the tenant's occupation and the value of tenant's fixtures and fittings are to be ignored.</li> </ul>
5.13	The landlord or the tenant can inform the other in writing at least 60 days before the rent review date of the rent that the landlord or tenant thinks will be the current market rent at the review date.
5.14	If the landlord and the tenant agree on a new rent then that rent will be the new rent beginning on the rent review . date and the landlord and the tenant must sign a statement saying so.
5.15	If the landlord and the tenant do not agree on the amount of the new rent 30 days before the rent review date, the current market rent will be decided by a valuer appointed under clause 5.16.
5.16	The landlord and the tenant can either agree upon a valuer or can ask the President of the Law Society of New South Wales to nominate a person who is a licensed valuer to decide the current market rent. Where the property is a retail shop, the valuer appointed must be a specialist retail valuer.
5.17	The valuer will act as an expert not an arbitrator. The landlord and the tenant can each make submissions in writing to the valuer within 14 days after they receive notice of the valuer's appointment but not later unless the valuer agrees.
5.18	The valuer's decision is final and binding. The valuer must state how the decision was reached.
5.19	If the valuer -
	<ul> <li>5.19.1 does not accept the nomination to act; or</li> <li>5.19.2 does not decide the current market rent within 1 month after accepting the nomination; or</li> <li>5.19.3 becomes incapacitated or dies; or</li> <li>5.19.4 resigns,</li> </ul>
	then another valuer is to be appointed in the same way.
	0

P 72-

Page 9 of 17

ME \$.7

•

	Page	e of 13 page
5.20	The landlord and tenant must each pay half the valuer's costs.	Annexur
5.21	If the landlord and tenant do not agree upon a valuer and neither asks for a valuer to be nominated	d before -
	5.21.1 the next rent review date passes; or 5.21.2 this lease ends without the tenant renewing it; or	
	5.21.3 this lease is transferred after the rent review date with the landlord's consent; or 5.21.4 the property is transferred after the rent review date	
	then the rent will not change on that rent review date.	
CLAUSE 6	USE	
	How must the property be used?	
6.1	The tenant must -	
	6.1.1 use the property for the purpose stated in item 17 in the schedule and not for any other pu open for business at times usual for a business of the kind conducted by the tenant;	upose;
	5.1.3 keep the property clean and dispose of waste properly; and	
	6.1.4 comply with all laws relating to strata schemes and all other laws regulating how the proobtain any consents or licences needed, comply with any conditions of consent, and keel licences or registrations needed for the use of the property or for the conduct of the tent there.	en current a
6.2	The landlord can consent to a change of use and cannot withhold consent unreasonably.	
6.3	The tenant must not -	
	6.3.1 do anything that might invalidate any insurance policy covering the property or that migh premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the landlord consents in which case the tenant must pay the increased premium unless the tenant must pay the tenant must	t increase t
	6.3.2 use the property as a residence or for any activity that is dangerous, offensive, noxi immoral or that is or may become a nuisance or annoyance to the landlord or to the own of any neighbouring property; or	ous illegat
	6.3.3 hold any auction, bankrupt or fire sale in the property; or	
	6.3.4 display signs or advertisements on the outside of the property, or that can be seen from unless the landlord consents (but the landlord cannot withhold consent unreasonably); or	n the outsid
	6.3.5 overload the floors or walls of the property.	
CLAUSE 7	CONDITION AND REPAIRS	
24	Who is to repair the property?	
7.1	The landlord must	
	7.1.1 maintain in a state of good condition and serviceable repair the roof, the ceiling, the exter external doors and associated door jambs, and the floors of the property and must fix struct maintain the number of the property and must fix struct	mal walls ar ctural defect
	<ul> <li>7.1.2 maintain the property in a structurally sound condition; and</li> <li>7.1.3 maintain essential services.</li> </ul>	

P 1

.

M.E J.Z

•	Page 7 of 13 page
7.2	Annexure . The tenant must otherwise maintain the property in its condition at the commencement date and promptly or repairs needed to keep it in that condition but the tenant does not have to -
	7.2.1 alter or improve the property; or 7.2.2 fix structural defects; or
	7.2.2 repair fair wear and tear.
7.3	The lenant must also -
	7.3.1 reimburse the landlord for the cost of fixing structural damage caused by the tenant, apart from fa wear and tear;
	7.3.2 maintain and decorate the shop front if the property has one; and
	7.3.3 decorate the inside of the property in the last 3 months of the lease period (however it ends) - 'decorate here means restoring the surfaces of the property in a style and to a standard of finish originally use e.g. by repainting.
7.4	If an authority requires work to be done on the property and it is structural work or work needed to make the
	property safe to use then the landlord must do the work unless it is required only because of the way the tenar uses the property. But if it is any other work or is required only because of the way the tenant uses the proper then the tenant must do the work.
7.5	If the tenant fails to do any work that the tenant must do the landlord can give the tenant a notice in writin stating what the tenant has failed to do. After the notice is given the tenant must -
	<ul><li>7.5.1 do the work immediately if there is an emergency; and</li><li>7.5.2 do the work promptly and diligently in any other case.</li></ul>
·	If the tenant does not do the work, the landlord can do it and the tenant must reimburse the landlord for the co of the work.
7.6	The tenant must not make any structural alterations to the property. Any other alterations require the landlord consent in writing (but the landlord cannot withhold consent unreasonably).
CLAUSE 8	INSURANCE AND DAMAGE
	What insurances must the tenant take out?
8.1	The tenant must keep current an insurance policy covering -
	8.1.1 liability to the public in an amount not less than the amount stated in item 18 in the schedule (for eac accident or event); and
	<ul> <li>8.1.2 damage or destruction from any cause to all plate glass in the windows and other portions of the property</li> </ul>
	- and must produce to the landlord, upon request, the policy and the receipt for the last premium.
	What happens if the property is damaged?
8.2	If the property or the building of which it is part is damaged (a term which includes destroyed) -
	8.2.1 the tenant is not liable to pay rent, or any amount payable to the landlord in respect of outgoings an other charges, that is attributable to any period during which the property cannot be used under the second during which the property cannot be used under the second during which the property cannot be used under the second during which the property cannot be used under the second during which the property cannot be used under the second during which the property cannot be used under the second during which the property cannot be used under the second during which the property cannot be used under the second during which the property cannot be used under the second during the s
	<ul> <li>8.2.2 if the property is still useable under this lease but its useability is diminished due to the damage, the tenant's liability for rent and any amount in respect of outgoings attributable to any period during which useability is diminished is reduced in proportion to the reduction in useability caused by the damage;</li> </ul>
A <sup>1</sup>	
1	Page 11 of 17 $M \in \mathcal{J} - \mathcal{L}$

•

.

	Page 8 of 13 pag
	Annexure 8.2.3 if the landlord notifies the tenant in writing that the landlord considers that the damage is such as make its repair impracticable or undesirable, the landlord or the tenant can terminate this lease giving not less than 7 days notice in writing of termination to the other and no compensation is payab
	in respect of that termination; 8.2.4 if the landlord fails to repair the damage within a reasonable time after the tenant requests the landlo to do so the tenant can terminate this lease by giving not less than 7 days notice in writing
	<ul> <li>termination to the landlord; and</li> <li>8.2.5 nothing in clause 8.2 affects any right of the landlord to recover damages from the tenant in respect any damage or destruction to which the clause applies.</li> </ul>
CLAUSE 9	ACCESS
	What are the landlord's rights of access to the property?
9.1	The tenant must give the landlord (or anyone authorised in writing by the landlord) access to the property at a reasonable time for the purpose of-
	9.1.1 inspecting the condition of the property, or how it is being used; or
	9.1.2 doing anything that the landlord can or must do under this lease or must do by law; or
	9.1.3 viewing the property as a valuer, prospective buyer or mortgagee; or
	9.1.4 fixing a notice in a reasonable position on the outside of the property saying that it is for sale; or
	<ul> <li>9.1.5 viewing the property as a prospective tenant not earlier than 6 months before the lease period ends; o</li> <li>9.1.6 fixing a notice not earlier than 6 months before the lease period ends in a reasonable position on the</li> </ul>
	outside of the property saying that it is to let; or 9.1.7 inspecting, cleaning or repairing another property or any services to another property.
9.2	The landlord must give the tenant al least 2 days written notice for access (except in an emergency). The day the giving of the notice and any Saturday, Sunday or public holiday on which the property is not open for business are not counted.
9.3	The landlord must promptly make good any damage caused to the property and to any of the tenani belongings which results from exercising these rights.
9.4	The tenant must give to the landlord a copy of any notice relating to the property or relating to any neighbourir property immediately after receiving the notice.
CLAUSE 10	TRANSFER AND SUB-LEASE
	Can this lease be transferred or the property shared or sub-let?
10.1	The tenant must not transfer this lease without consent.
10.2	The landlord can withhold consent only if -
	<ul> <li>10.2.1 the proposed transferee proposes to change the use to which the property is put; or</li> <li>10.2.2 where the property is a retail shop, the proposed transferee has financial resources or retailing skil inferior to those of the proposed transferor and otherwise the proposed transferee has financial resources or business experience inferior to those of the proposed transferor; or</li> <li>10.2.2 the teamt has teamt here to make the proposed transferor and otherwise the proposed transferor; or</li> </ul>
	10.2.3 the tenant has not complied with clause 10.3 and, where the property is a retail shop, clause 10.4.
10.3	• A request for the landlord's consent to a transfer of lease must be made in writing and the tenant must provid the landlord with such information as the landlord may reasonably require concerning the financial standing an business experience of the proposed transferee.

 $\left( \sum_{i=1}^{n} \right)$ N

Page 12 of 17

M.E & Z

Page 9 of 13 pages Annexure B

- 10.4 Where the property is a retail shop, before requesting the consent of the landlord to a proposed transfer of this lease, the tenant must furnish the proposed transferee with a copy of any disclosure statement given to the tenant in respect of this lease, together with details of any changes that have occurred in respect of the information contained in the disclosure statement (being changes of which the tenant was aware or could reasonably be expected to be aware). For the purpose of enabling the tenant to comply with this obligation, the tenant can request the landlord to provide the tenant with a copy of the disclosure statement concerned and, if the landlord is unable or unwilling to comply with such a request within 14 days after it is made, this clause 10.4 does not apply.
- 10.5 Where the tenant has complied with clause 10.3 and where required to do so clause 10.4 and the landlord has not within 42 days after the request was made given notice in writing to the tenant either consenting or withholding consent the landlord is taken to have consented.
- 10.6 The tenant has to pay in connection with any consent the landlord's reasonable legal costs, the reasonable costs of obtaining any mortgagee's consent, the stamp duty and the registration fee for the transfer.
- 10.7 Where the property is a retail shop, the tenant can sublet, grant a license or concession, share or part with the possession of the whole or any part of the property or mortgage or otherwise charge or encumber the tenant's estate or interest in this lease only with the written consent of the landlord which can be refused in the landlord's absolute discretion. Otherwise, the tenant cannot do any of these things.

#### CLAUSE 11 LANDLORD'S OTHER OBLIGATIONS

#### What are the landlord's other obligations?

- 11.1 So long as the tenant does all the things that must be done by the tenant under this lease the landlord must allow the tenant to possess and use the property in any way permitted under this lease without interference from the landlord, or any person claiming under the landlord or having superior title to the title of the landlord.
- 11.2 The landlord must pay all outgoings for the land or the building of which the property is part when they fall due.
- 11.3 If the property is part of a building owned or controlled by the landlord -
  - 11.3.1 the landlord must maintain in reasonable structural condition all parts of the building that the tenant can use under this lease; and
  - 11.3.2 if the property has facilities and service connections shared in common with other persons the landlord must -
    - 11.3.2.1 allow reasonable use of the facilities and service connections including
      - the right for the tenant and other persons to come and go to and from the property over the areas provided for access;
      - access by the tenant to service connections; and
      - the right for the tenant's customers to park vehicles in any area set aside for customer parking, subject to any reasonable rules made by the landlord.

ME & Z

- 11.3.2.2 maintain the facilities and service connections in reasonable condition.
- 11.4 The landlord must ensure that this lease is registered.
- 11.5 If a consent is needed for this lease, from someone such as a mortgagee or head landlord of the property, then the landlord must get the consent.

12

8<u>2 Reg:C145582</u> /Doc:DL AG789227 /Rev:17-Feb-2012 /Sts:OK.OK /Prt:27-Mar-2015 10:20 /Pgs:ALL /Seq:14 of 17 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

Page 10 of 13 pages Annexure B **CLAUSE 12** FORFEITURE AND END OF LEASE When does this lease end? 12.1 This lease ends -12.1.1 on the date stated in item 3 in the schedule; or 12.1.2 if the landlord lawfully enters and takes possession of any part of the property; or 12.1.3 if the landlord lawfully demands possession of the property. 12.2 The landlord can enter and take possession of the property or demand possession of the property if -12.2.1 the tenant has repudiated this lease; or rent or any other money due under this lease is 14 days overdue for payment; or 12.2.2 the tenant has failed to comply with a landlord's notice under section 129 of the Conveyancing Act 12.2.3 1919: or 12.2.4 the tenant has not complied with any term of this lease where a landlord's notice is not required under section 129 of the Conveyancing Act 1919 and the landlord has given at least 14 days written notice of the landlord's intention to end this lease. 12.3 When this lease ends, unless the tenant becomes a tenant of the property under a new lease the tenant must -12.3.1 return the property to the landlord in the state and condition that this lease requires the tenant to keep it in: and 12.3.2 have removed any goods and anything that the tenant fixed to the property and have made good any damage caused by the removal. Anything not removed becomes the property of the landlord who can keep it or remove and dispose of it and charge to the tenant the cost of removal making good and disposal. 12.4 If the landlord allows the tenant to continue to occupy the property after the end of the lease period (other than under a new lease) then -12.4.1 the tenant becomes a monthly tenant and must go on paying the same rent and other money in the same way that the tenant had to do under this lease just before the lease period ended (apportioned and payable monthly); 12.4.2 the monthly tenancy will be on the same terms as this lease, except for n clause 4: n clauses 5.4 to 5.21 inclusive; and clause 6.2 unless consent has previously been given; either the landlord or the tenant can end the monthly tenancy by giving, at any time, one month's 12.4.3 written notice to the other expiring on any date; and 12.4.4 anything that the tenant must do by the end of this lease must be done by the end of the monthly tenancy. 12.5 Essential terms of this lease include -12.5.1 the obligation to pay rent not later than 14 days after the due date for payment of each periodic instalment (and this obligation stays essential even if the landlord, from time to time, accepted late payment): 12.5.2 the obligations of the tenant in clause 5.1.2 (dealing with outgoings); the obligations of the tenant in clause 6.1 (dealing with use); 12.5.3 the obligations of the tenant in clause 7 (dealing with repairs); and 12.5.4 12.5.5 the obligations of the tenant in clause 10 (dealing with transfer and sub-lease).

Page 14 of 17

ME 22

Page 11 of 13 pages Annexure B

- 12.8 If there is a breach of an essential term the landlord can recover damages for losses over the entire period of this lease but must do every reasonable thing to mitigate those losses and try to lease the property to another tenant on reasonable terms.
- 12.7 The landlord can recover damages even if -
  - 12.7.1 the landlord accepts the tenant's repudiation of this lease; or
  - 12.7.2 the landlord ends this lease by entering and taking possession of any part of the property or by demanding possession of the property; or
  - 12.7.3 the tenant abandons possession of the property; or
  - 12.7.4 a surrender of this lease occurs.

#### CLAUSE 13 GUARANTEE

- What are the obligations of a guarantor?
- 13.1 This clause applies if a guarantor of the tenant is named in item 10A in the schedule and has signed or executed this lease or, if this lease is a renewal of an earlier lease, the earlier lease.
- 13.2 The guarantor guarantees to the landlord the performance by the tenant of all the tenant's obligations (including any obligation to pay rent, outgoings or damages) under this lease, under every extension of it or under any renewal of it or under any tenancy and including obligations that are later changed or created.
- 13.3 If the tenant does not pay any money due under this fease, under any extension of it or under any renewal of it or under any tenancy the guarantor must pay that money to the landlord on demand even if the landlord has not tried to recover payment from the tenant.
- 13.4 If the tenant does not perform any of the tenant's obligations under this lease, under any extension of it or under any renewal of it or under any tenancy the guarantor must compensate the landlord even if the landlord has not tried to recover compensation from the tenant.
- 13.5 If the tenant is insolvent and this lease or any extension or renewal of it is disclaimed the guarantor is liable to the landlord for any damage suffered by the landlord because of the disclaimer. The landlord can recover damages for losses over the entire period of this lease or any extension or renewal but must do every reasonable thing to mitigate those losses and try to lease the property to another tenant on reasonable terms.
- 13.6 Even if the landlord gives the tenant extra time to comply with an obligation under this lease, under any extension of it or under any renewal of it or under any tenancy, or does not insist on strict compliance with the terms of this lease or any extension of it or renewal of it or of any tenancy, the guarantor's obligations are not affected.
- 13.7 If an amount is stated in item 10B in the schedule the guarantor's liability under this clause is limited to that amount.
- 13.8 The terms of this guarantee apply even if this lease is not registered, even if any obligation of the tenant is only 'an equitable one, and even if this lease is extended by legislation.

#### CLAUSE 14 EXCLUSIONS, NOTICES AND SPECIAL CLAUSES

- 14.1 No covenant or power is implied in this lease by section 84 or 85 of the Conveyancing Act 1919.
- 14.2 A document under or relating to this lease is -
  - 14.2.1 served if it is served in any manner provided in section 170 of the Conveyancing Act 1919; and

Page 15 of 17

M.E. d.E

Page 12 of 13 pages Annexure B

14.2.2 served on the tenant if it is left at the property.

- 14.3 This lease is subject to any legislation that cannot be excluded.
- 14.4 In this lease, 'retail shop' means premises to which the Retail Leases Act 1994 applies,

#### CLAUSE 15 GOODS AND SERVICES TAX

Unless item 13B has been completed in a way that indicates that this clause is not to apply:

- 15.1 The rent and all other monies payable to the landlord are exclusive of Goods and Services Tax (GST). Whenever the tenant becomes liable to pay rent or other monies payable under this lease in respect of a taxable supply made by the landlord the tenant must also pay an additional 10% to cover GST. This percentage of 10% assumes that GST payable on the value of a taxable supply is 10%. If the GST rate is different then this percentage will instead be the GST rate.
- 15.2 Outgoings in item 14B are to be calculated after deducting any input tax credit to which the landlord is entitled.

M.E Ø.F

82 Req:C145582 /Doc:DL AG789227 /Rev:17-Feb-2012 /Sts:OK.OK /Prt:27-Mar-2015 10:20 /Pgs:ALL /Seq:17 of 17 Ref:lpi.vixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

> Page 13 of 13 pages Annexure B

> > 1

•

.

#### **ADDITIONAL SPECIAL CONDITIONS**

CLAUSE 16	BANK GUARANTEE
16.1	If a number of months appears in item 19 in the Schedule, clauses 16.2 to 16.5 apply;
16.2	On or before the commencement date of this lease, the lessee will deliver to the lessor or the lessor's agent, a guarantee by a bank in the State of New South Wales in the form of an unconditional and irrevocable undertaking to pay, drawn in favour of the lessor (unlimted as to time), in a form acceptable to the lessor and for an amount equivalent to the number of months referred to in item 19 in the schedule;
16.3	The lessor is entitled to claim under the guarantee an amount equal to any moneys due but unpaid by the lessee to the lessor under this lease;
16.4	The lessee agrees to vary the amount of the guarantee immediately upon each rent review so that the amount at all times represents the equivalent of the number of months referred to in the schedule;
16.5	The lessor will deliver the guarantee (or so much of it as if then held by the lessor) to the lessee on the last of:-
	16.5.1:- the terminating date of this lease;
	16.5.2:- the expiry date of any holding over under this lease; and
	16.5.3:- the date that the lessee has no further obligations under this lease or at law.
CLAUSE 17	RENT FREE PERIOD
17.1	Subject to the Lessee complying with all of the Lessee's obligations pursuant to this Lease, the Lessor agrees to grant to the Lessee a rent free period of 2 months, commencing on the Commencement date and as specified in Item 20 of Schedule A;
17.2	Notwithstanding this Clause 17, the Lessee must assume all responsibility for any outgoings payable pursuant to this Lease;
17.3	This Clause 17 will not apply to any option Lease.
CLAUSE 18	FIRE SAFETY REGULATIONS
18.1	The Lessee agrees to comply with the Standard Fire Safety Regulations and must submit a copy of the Annual Fire Safety Statement to the Lessor annually on the anniversary of this Lease, as well as a copy of the Service Certificate every 6 months from the Commencement date;
CLAUSE 19	POTENTIAL DEVELOPMENT SITE
19.1	The Lessor has disclosed to the Lessee, and the Lessee understands, that the premises the subject of this Lease is a potential development site;
19.2	The Lessor may, during any time in any option period, require the Lessee to vacate the premises by giving them written notice no less than 3 months prior to the date on which they are required to vacate;
19.3	Notwithstanding anything else herein contained, should the Lessor exercise his option in Clause 19.2 requiring the Lessee to vacate the premises, then the Lessor agrees that upon completion of the redeveloped premises, the Lessee will have the first right of option to renegotiate a new lease with the Lessor.

`.**`** 

•

۰, ۱

Page 17 of 17

ME \$ 2

Form: 07L Licence: 01-08-084	LEASE	
licensee: Midware S	Systems New South Wales	
Champion Legal	Real Property Act 1900	AD684221M
STAMP DUTY	PRIVACY NOTE: this information is legally required and will be	
STAMP DUTT	Office of State Revenue use only	NEW SOUTH WALES DUTY
		13-09-2005 0002950778-001
		DUPLICATE OUTIABLE ANOUNT \$ ***********************************
		DUTY \$ ####88#######2.00
A) TORRENSTITLE	Property leased: if appropriate, specify the part or premises	
:	Part Folio Mentifier B/410947 holog Promises (courses)	
	Part Folio Identifier B/410947 being Premises known as Wentworthville	Shop 5/108-120 Station Street
B) LODGED BY	Delivery Name, Address or DX and Telephone Box	CODE
	124E LLPN: 123820V L8, 170 Phillip St., Sydney NS	SIM DOOL
	PO Box A250 Sydney South N Ph: 02 9230 6900	ISW 123
LODGED	MAK:1230	
DIESSOR		
- 9 JAN 2008	MICHAEL ELIAS & SAMIRA ELIAS	
l l		
NE: 2.39-	The lessor leases to the lessee the property referred to above.	
	Encumbrances (if applicable): AD 684.220	
) Lessee	10004220	
	DAVID BAYSSARI & MARIA BAYSSARI	
	DAVID DATSSARI & MARIA DATSSARI	
)	TENANCY: Joint	
_		OFF L 4A 411196
i) 1. TERM:	TWO YEARS	OFF L 4A 411196
	DATE: 12 January 2006	11-B
3. TERMINATING	DATE: 11 January 2008	U
<ol> <li>TERMINATING</li> <li>With an OPTIC</li> </ol>	ON TO RENEW for a period of Two years	
<ol> <li>TERMINATING</li> <li>With an OPTIC set out in claus</li> </ol>	ON TO RENEW for a period of Two yearsse No. 18of Annuexure "A"	
<ol> <li>TERMINATING</li> <li>With an OPTIC set out in claus</li> <li>With an OPTIC</li> </ol>	ON TO RENEW for a period of Two years se No. 18 of Annuexure "A" ON TO PURCHASE set out in clause No. N.A. of N.A.	NA
<ol> <li>TERMINATING</li> <li>With an OPTIC set out in claus</li> <li>With an OPTIC</li> <li>Together with</li> </ol>	ON TO RENEW for a period of Two years se No. 18 of Annuexure "A" ON TO PURCHASE set out in clause No. N.A. of N.A.	N.A.
<ol> <li>TERMINATING</li> <li>With an OPTIC set out in claus</li> <li>With an OPTIC</li> <li>Together with</li> <li>Incorporates the</li> </ol>	ON TO RENEW for a period of Two years se No. 18 of Annuexure "A" ON TO PURCHASE set out in clause No. N.A. of N.A. and reserving the RIGHTS set out in clause No. N.A. of	.1
<ol> <li>TERMINATING</li> <li>With an OPTIC set out in claus</li> <li>With an OPTIC</li> <li>With an OPTIC</li> <li>Together with</li> <li>Incorporates th</li> <li>Incorporates th</li> </ol>	ON TO RENEW for a period of Two years se No. 18 of Annuexure "A" ON TO PURCHASE set out in clause No. N.A. of N.A. and reserving the RIGHTS set out in clause No. N.A. of he provisions set out in ANNEXURE "A" hereto. he provisions set out in MEMORANDUM filed at Land and Prope	.1

Total Pages (office use only)

### Number additional pages sequentially

84 Reg:C145584 /Doc:DL AD684221 /Rev:18-Jan-2008 /Sts:OK.OK /Prt:27-Mar-2015 10:21 /Pgs:ALL /Seq:2 of 32 Ref:1pi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

12, 1,06 DATE

(H) I certify that the person(s) signing opposite, with whom I am personally acquainted or as to whose identity I am otherwise satisfied, signed this instrument in my presence.

Signature of witness: Y Mille. Name of witness: Y - 104 y Nehme. Address of witness: X | Kings view way Gilen Wood 2765 Certified correct for the purposes of the Real Property Act 1900 by the lessor.

Signature of lessor: Y

1900 by the lessee.

Signature of lessee:

in expired lease No.

y dikligi

Note: where applicable, the lessor must complete the statutory declaration below

VIE

Certified correct for the purposes of the Real Property Act

I certify that the person(s) signing opposite, with whom I am personally acquainted or as to whose identity I am otherwise satisfied, signed this instrument in my presence.

Signature of witness:

Name of witness:

Address of witness:

ROSARIO LEOTTA Esti: Solicitor & Conveyancer Suite 3, First Floor, 245 The Boulevarde, FAIRFIELD HEIGHTS NSW 2168 Talephone: 9609 1444

#### (I) STATUTORY DECLARATION

Ĩ,

solemnly and sincerely declare that -

- 1. The time for the exercise of option to
- 2. The lessee under that lease has not exercised the option.

I make this solemn declaration conscientiously believing the same to be true and by virtue of the provisions of the Oaths Act 1900.

Made and subscribed at	in the State of New South Wales
on	in the presence of:
Signature of witness:	Signature of lessor:
Name of witness:	
Address of witness:	

Qualification of witness:

has ended;

#### ANNEXURE "A" AND SCHEDULE

#### LANDLORD: MICHAEL ELIAS & SAMIRA ELIAS

**TENANT:** DAVID BAYSSARI & MARIA BAYSSARI

PREMISES: SHOP 5/108-120 STATION STREET WENTWORTHVILLE

#### 1. COVENANTS NEGATIVED

The covenants, powers and provisions implied in Leases by Sections 84 and 85 of the Conveyancing Act, 1919 are hereby expressly negatived except to the extent to which they are or may be included in the covenants hereinafter contained.

#### 2. TERM

#### Term of Lease

- 2.1 Subject to the provisions of this Lease, the Lease commences on the Commencement Date as set out in Item 1 (a) of the Schedule and ends on the Expiry Date as set out in Item 1 (b) of the Schedule.
- 3. RENT

Rent

- 3.1 The Tenant covenants with the Landlord to pay free from all deductions the annual sum set out in Item 2 (a) of the Schedule for rent for the first year of the term by monthly payments set out in Item 2 (b) of the Schedule each in advance.
- 3.2 For the second year of the term the annual sum set out in Item 2(c) of the Schedule by monthly payments set out in Item 2(d) of the Schedule each in advance the first payment to be made on the commencement of the second year of the term and thereafter on the same day of each succeeding month.
- 3.3 The Tenant must pay the first instalment on or before the Commencement Date.
- 3.4 The Tenant shall not withhold or be entitled to withhold the whole or part of any such payment by way of deduction, set off or counter claim in respect of any claim for damages or for compensation which the Tenant shall make or has made against the Landlord.

Please initial foot of Y eveny, sage

A.Z

JN: M:\Docs\50049\193934.doc

#### Monthly tenancy

- 3.5 Should the Tenant continue to occupy the Premises beyond the Expiry Date with the Consent of the Landlord otherwise than pursuant to a further lease granted by the Landlord, the Tenant shall do so as a monthly tenant only, on the terms and conditions of this Lease with necessary changes applicable to a monthly tenancy except that:
- 3.5.1 the monthly rent will be equal to one month's proportion of the rent hereby reserved increased annually by the percentage set out in Item 4 of the Schedule.
- 3.5.2 the first payment of rent is to made on the next day after the Expiry Date and thereafter payable monthly in advance
- 3.5.3 the tenancy may be determined at the will of either the Landlord or Tenant by one month's Notice in writing expiring on any day of the month and the tenancy will end one month after service of that Notice.

#### 4 OUTGOINGS

4.1 - 4.3 Intentionally Deleted

#### Telephones, electricity, etc

4.4 The Tenant will pay to the proper authorities all charges for both rental and use of the telephone services for gas and electricity consumed on the Premises and for all water and sewerage usage charges, excess water, greasetrap charges and all meter rents and if the Tenant makes default in the payment thereof it shall be optional for the Landlord to pay the same and in addition to the Landlord's remedies the Landlord may recover any amounts to be paid as if the same were overdue rent hereunder.

#### **Registration of Strata Plan**

- 4.5 Should during the term of this Lease or any renewal thereof or during any holding over period a strata plan be registered in respect of the Land of which the Premises forms part so that the Premises becomes a lot in the strata plan then in addition to the rent payable hereunder the Tenant shall forthwith and in any event no longer than fourteen days after demand shall have been made therefore pay to the Landlord the following outgoings payable with respect to the Premises:
- 4.5.1 Council rates and charges.
- 4.5.2 Water sewerage and drainage rates and charges payable to Sydney Water.
- 4.5.3 Land tax PROVIDED that this shall be calculated on the basis that the Premises is the only land in respect of which the Landlord is required to lodge a Return for land tax purposes and the Land concerned was not subject to a special trust (within the meaning of the Land Tax Management

JN: M:\Docs\50049\193934.doc

MEDY

\$ 1.3
Act 1956) and the Landlord was not a company classified under Section 29 of that Act as a non-concessional company.

- 4.5.4 Insurance premiums payable in respect of each risk or insurable interest which the Landlord may in its discretion effect.
- 4.5.5 all levies or contributions payable to the Owners Corporation (created upon registration of the strata plan in respect of the Property of which the Demised Premises forms part) pursuant to Part 3 Division 2 of the Strata Schemes Management Act 1996 or which may otherwise be properly raised by the Owners Corporation against the Landlord.

### Abatement of rent

- 4.6 The Lessee covenants with the Lessor to pay the initial rent herein referred to and such additional rent as may be payable from time to time pursuant to this Lease at the times and in the manner herein provided without any deduction or abatement whatsoever except as may be hereinafter provided that:
- 4.6.1 If after the commencement of the said term the Demised Premises or any part thereof be destroyed or damaged so as to be unfit for occupation or use or is inaccessible due to that damage the rent hereby reserved and any amount payable to the Lessor in respect of outgoings or a fair proportion thereof according to the nature and extent of the damage sustained shall be suspended until the Demised Premises shall have again been rendered fit for occupation or use.
- 4.6.2 (a) If the Lessor notifies the Lessee in writing that the Lessor considers that the damage is such as to make its repair impracticable or undesirable, the Lessor or the Lessee may terminate the Lease by giving not less than seven days' Notice in writing to the other and no compensation is payable in respect of that termination.
  - (b) If the Lessor fails to repair the damage within a reasonable time after the Lessee requests the Lessor in writing to do so, the Lessee may terminate the Lease by giving not less than seven days' Notice in writing of termination to the Lessor.
  - (c) the aforesaid provisions do not affect any right of the Lessor to recover damages from the Lessee in respect of any damage or destruction to which those provisions apply.
- 4.6.3 Nothing in this section is to limit any liability of a party to the Lease to pay compensation to another party to the Lease in respect of damage to the Premises or the Building of which the Premises forms part.
- 4.6.4 Nothing in this section prevents the parties to a Lease from terminating the Lease by agreement if the Premises or the Building of which it forms part is damaged or destroyed.

ME 0.92

01.6

4.6.5 If the Demised Premises or any part thereof be adversely impacted due to redevelopment of the Building so as to be unfit or restricted for occupation or use or is inaccessible due to building works the rent hereby reserved and any amount payable to the Lessor in respect of outgoings or a fair proportion thereof according to the nature and extent of the redevelopment sustained shall be suspended until the Demised Premises shall have again been rendered fit for occupation or use.

## **BREACH OF ESSENTIAL TERMS**

- 5.1 In the event of a breach of an essential term of this Lease and if such breach shall continue for a period of seven (7) days after Notice in writing to the Tenant by the Landlord of such breach then thenceforth it shall be lawful for the Landlord or any person or persons duly authorised by the Landlord in that behalf to enter into or upon the Premises (forcibly if necessary) or any part thereof in the name of the whole and to repossess and enjoy the same as if these presents had not been made without prejudice to any right of action or remedy of the Landlord in respect of any antecedent breach by the Tenant of any essential term hereof.
- 5.2 Notwithstanding anything herein contained upon the Landlord becoming entitled to re-enter or determine this Lease the residue of the term hereof for the time being unexpired shall at the option of the Landlord immediately upon Notice of the exercise of such option being given by the Landlord to the Tenant become reduced to one week and thereafter the tenancy hereby created shall be and remain a tenancy from week to week at a weekly rate equal to one week's proportion of the rent hereby reserved (as adjusted hereunder from time to time) commencing from the date of service of such Notice and shall be determinable by either the Landlord or the Tenant giving to the other one week's Notice in writing expiring on any day of the week.
- 5.3 Upon the expiration or sooner determination of the term of this Lease the Tenant shall surrender to the Landlord all keys giving access to all parts of the Premises held by the Tenant the Tenant's servants or invitees irrespective of whether or not the same have been supplied by the Landlord, and remove at the Tenant's expense all lettering and distinctive marks or signs put up by the Tenant or by the Landlord for the Tenant on any part or parts of the Premises or the Building.
- 5.4 Each covenant by the Tenant which is specified in this paragraph is an essential term of this Lease:
- 5.4.1 The covenant to pay rent and outgoings throughout the Lease term and any holding over period not later than the due date for the payment of each instalment of rent (Clause 2.1);
- 5.4.2 Clause 6.1 & 6.3; (To Repair);
- 5.4.3 Clause 7.1; (Alterations to Premises);
- 5.4.4 Clause 10; (Use of Premises);

MÉ S.G

DD-3

# 5.4.5 Clause 13; (Assignment of Lease);

- 5.5 In respect of the Tenant's obligation to pay rent, the acceptance by the Landlord of arrears or of any late payment of rent shall not constitute a waiver of the essentiality of the Tenant's obligation to pay rent in respect of those arrears or of the late payments or in respect of the Tenant's continuing obligation to pay rent during the Lease term.
- 5.6 The Tenant covenants to compensate the Landlord in respect of any breach of an essential term of this Lease and the Landlord is entitled to recover damages from the Tenant in respect of such breaches. The Landlord's entitlement under this Clause is in addition to any other remedy or entitlement of the Landlord (including to terminate this Lease).
- 5.7 In the event that the Tenant's conduct (whether acts or omissions) constitutes a repudiation of the Lease (or of the Tenant's obligations under the Lease) or constitutes a breach of any Lease covenants, the Tenant covenants to compensate the Landlord for the loss or damage suffered by reason of the repudiation or breach.
- 5.8 The Landlord shall be entitled to recover damages against the Tenant in respect of repudiation or breach of covenant for the damage suffered by the Landlord during the entire term of this Lease.
- 5.9 The Landlord's entitlement to recover damages shall not be affected or limited by any of the following:
- 5.9.1 If the Tenant shall abandon or vacate the leased Premises;
- 5.9.2 If the Landlord shall elect to re-enter or to terminate the Lease;
- 5.9.3 If the Landlord shall accept the Tenant's repudiation;
- 5.9.4 If the parties conduct shall constitute a surrender by operation of law.
- 5.10 The Landlord shall be entitled to institute legal proceedings claiming damages against the Tenant in respect of the entire Lease term, including the periods before and after the Tenant has vacated the lease Premises and before and after abandonment, termination, repudiation, acceptance of repudiation or surrender by operation of law referred to in clause 5.9, whether the proceedings are instituted either before or after such conduct.
- 5.11 In the event of the Tenant vacating the leased Premises, whether with or without the Landlord's Consent, the Landlord shall be obliged to take reasonable steps to mitigate his damages and to endeavour to lease the Premises at a reasonable rent and on reasonable terms. The Landlord's entitlement to damages shall be assessed on the basis that the Landlord should have observed the obligation to mitigate damages contained in this paragraph. The Landlord's conduct taken in pursuance of the duty to mitigate damages shall not by itself constitute acceptance of the Tenant's breach or repudiation or a surrender by operation of law.

) D.\$

ME D.2

## **TENANT'S OBLIGATIONS**

6. **Positive use obligations** 

The Tenant must:

# 6.1 Maintenance and Repair

- 6.1.1 Repair and to keep the Premises, the exterior facade and all additions thereto and the Landlord's fixtures thereon and all drains soil and other pipes sewers sanitary and water apparatus glass pavings wall fences and railing and appurtenances in good and substantial repair and maintained paved and cleansed in every respect (excepting fair wear and tear damage by fire explosion carthquake aircraft flood tempest Act of God riot and civil commotion save where any insurance moneys are irrecoverable through the neglect and default or misconduct of the Tenant) AND ALSO will and so often as it shall be necessary to renew any fixtures belonging to the Premises or substitute other fixtures of a similar description and value to the satisfaction of the Landlord except where such substitution is due to damage by fire explosion earthquake aircraft flood tempest Act of God riot or civil commotion save where any insurance moneys are irrecoverable through the neglect and default or misconduct of the Tenant.
- 6.1.2 Paint the whole of the Premises, being the interior and exterior (if applicable), as reasonably required and prior to the expiration of the Lease.
- 6.1.3 Where applicable, clean and pump out every six months the inlet and drainage lines leading the grease arrester and the outlet pipes from the arrester to the Sydney Water Sewerage pipelines.

# 6.2 Replace fixtures

Replace from time to time during the said term such of the Landlord's fixtures and fittings as may have become worn out or unfit for use (excepting fair wear and tear) or lost by substituting others of a like quality and in new condition.

# 6.3 Enter renew and repair

6.3.1 The Landlord may by itself or through an independent property consultant enter upon the Premises every six months upon giving 48 hours notice (except in case of emergency as to the existence of which the unfettered opinion of the Landlord shall be conclusive, when no Notice shall be required) and view the state of repair thereof and the Landlord may require the Tenant by Notice in writing served upon the Tenant to rectify any defect (excluding latent defects and defects not arising from the act or omission of the Tenant its servants or agents) or carry out any repairs within 30 days in accordance with any covenant expressed or implied in the Lease.

Q D.B

мЕ &.У

- 6.3.2 In default of the Tenant repairing any defects according to the Notice the Landlord may from time to time enter the Premises and execute and carry out the required repairs as if it were the Tenant and for that purpose the Landlord its architects contractors and workmen and agents may enter upon the whole or any part of the Premises and there remain for the purpose of doing executing and effecting and carrying out any such repair AND the expenses and costs of carrying out such work shall forthwith be payable by the Tenant to the Landlord.
- 6.3.3 The Tenant will permit the Landlord upon 48 hours Notice (except in case of emergency as to the existence of which the unfettered opinion of the Landlord shall be conclusive when no Notice shall be required) to enter execute and carry out repairs renovations maintenance or alterations to the Premises or to any part thereof or to the Common Areas or any part thereof in compliance with the Landlord's obligations under the provisions of this Lease or otherwise deemed necessary or desirable by the Landlord and for the purpose the Landlord its architects, contractors and workmen and agents may enter upon the whole or any part of the Premises and there remain for the purpose of doing executing effecting and carrying out such repairs renovations maintenance or alteration PROVIDED ALWAYS THAT in the exercise of such power no undue inconvenience is caused to the Tenant.
- 6.3.4 If at any time any authority having jurisdiction or authority over or in respect of the Premises or the user thereof requests or requires any alterations, or additions, conversions, improvements or other works to be made to the Premises which the Landlord elects to do and for which the Tenant is not liable under its covenants herein contained the Landlord its architects contractors workmen servants and others and the servants and workmen of any of its contractors may enter into and on the Premises and there remain at all times for the purpose of making any such alterations additions conversions improvements or other works or any of them as aforesaid PROVIDED ALWAYS THAT in the exercise of any such power no undue inconvenience is caused to the Tenant,

## 6.4 Plant and equipment

Keep any air conditioning and ventilating equipment and all other plant and equipment installed in the Premises during the continuance of this Lease in proper working condition.

### 6.5 Pests

Take all proper precautions to keep the Premises free of white ants borers vermin and other like pests in particular to ensure that the Premises shall not be or create a hazard to health or otherwise infringe the provisions of Public Health Act or other legislation the provisions of which may be applicable thereto from time to time.

S lib

ME \$2

# 6.6 **Compliance with the Statutes**

Comply with all Statues, Ordinances, Proclamations, Orders, Regulations or Notices relating to the Premises and do and execute or cause to be done or executed all such works acts deeds matters and things as under or by virtue of any of the said Statutes, Ordinances, Proclamations, Orders, Regulations and Notices for the time being in force or shall be properly directed or necessary to be done or executed upon or in respect of the Premises or any part thereof whether by the owner Landlord Tenant or occupier in consequence of the occupation and use of the Premises by the Tenant.

# 6.7 Inspection by prospective Tenants

At all reasonable times permit the Landlord to exhibit to prospective tenants or purchasers the Premises and will at all times within the three months immediately preceding the termination of the this Lease allow the Landlord to affix and exhibit where the Landlord shall think fit at any time the usual "For Salc" and the usual "To Be Let" Notice and in each case with the name and address of the Landlord and/or its agent thereon and the Tenant will not remove any such Notice without the written Consent of the Landlord.

# 7. Negative Use Obligations

The Tenant must not:

# 7.1 Alterations

Without the previous written Consent of the Landlord (which shall not be unreasonably withheld):

- 7.1.1 erect permit or suffer to be erected or carried out any structural alteration or works on the Premises PROVIDED that if any structural alterations or works shall be carried out by the Tenant with the Consent of the Landlord as aforesaid the Tenant shall at the expiration of the term hereof (if so required by the Landlord) restore the Premises to the state and condition in which they were prior to the carrying out of such works having regard to the condition of the Building through fair wear and tear during the term hereof.
- 7.1.2 change the exterior colour of the Premises or of the Building in which the same are situate or any part thereof from the colour nor the size location or composition of any sign symbol or advertisement that may have been approved by the Landlord.
- 7.1.3 install any exterior lighting or plumbing facilities shades canopies awnings or window boxes or amplifiers or similar devices or use any advertising medium which may be heard or experienced outside the Premises such as loud speakers, tapes, CDs, television, video or radio broadcasts.
- 7.1.4 unless herein provided, mark paint drill write upon or in any way deface any wall ceiling floor wood stone or ironwork of the Premises.

PR D.D

ME

7.1.5 in any manner to interfere with or attempt to control any part of the fire alarm or sprinkler systems or any other machinery of the Landlord of any kind now installed or hereafter to be installed in the Premises PROVIDED HOWEVER and it is hereby specifically agreed by and between the parties hereto that the Landlord shall not be responsible for and shall incur no liability for failure of any of the aforesaid equipment or machinery or for the ineffectual operation thereof or for any damage or loss occasioned or arising in connection with the same to the Tenant or to any person claiming by through or under the Tenant and the Tenant hereby indemnifies and agrees to keep indemnified the Landlord against all or any such claim.

## 7.2 Additions

Should the Landlord at the request of the Tenant construct or erect or permit the Tenant to construct or erect any walls, partitions or ceilings within the Premises whether at the cost of the Tenant or not the Tenant shall not be entitled to remove such walls partitions or ceiling on vacating the Premises (but the same shall remain on the Premises for the benefit of but at no cost to the Landlord).

# 7.3 Advertisement

Not affix or exhibit or suffer to be affixed or exhibited to or on any part of the exterior of the Premises any advertisements signs symbols posters placards neon lights or other light or other object or thing other than such as may relate solely to the Premises or some business lawfully carried on there and so that the same shall be displayed only in such position on and height about the Premises and be of such materials colour dimensions and general character as shall be first approved in writing by the Landlord nor hang place deposit or expose or permit or suffer any hanging placing depositing or exposing outside any part of the Buildings comprised in the Premises any goods articles or thing for sale.

## 7.4 Water closets

Not use the water closets or other water apparatus for any other purpose than those for which they are constructed and no sweepings, rubbish, rags, trade effluent or deleterious matter of any description shall be thrown into water closets sewers drains or watercourses serving the Premises, and any damage thereby resulting to such water closets or sewers drains or watercourse and the cost of repairing the same shall be borne by the Tenant.

## 7.5 Nuisance

Generally not do or permit or suffer to be done or permitted upon or in connection with the Premises anything which shall or tend to be a nuisance annoyance or cause of damage to the Landlord or to any adjoining or neighbouring property or to the owner or occupier thereof and to pay the Landlord all costs charges and expenses which may be incurred by the Landlord in abating any such nuisance and for executing all works as may be necessary to comply with any Notice served by a local or public authority or neighbouring premises for abating such

R AB

AF d.E

nuisance created as a consequence of the occupation and use of the Premises by the Tenant.

#### 7.6 Heavy objects

Not do or bring or permit or suffer to be done or to be brought in or upon the Premises anything which may be put on the Premises or any Buildings of which the same form part any weight or impose any strain in excess of that which such Premises are calculated to bear with due margin for safety and in particular not to suspend any weights from the roof trusses.

#### 7.7 **Obstruction of windows**

Not obstruct or permit or suffer to be obstructed any of the windows, doorways, lights or ventilators belonging to the Premises.

#### 7.8 Rubbish

- 7.8.1 Not permit any rubbish or garbage to accumulate on the Premises or any part thereof unless confined in suitable containers so located as not to be visible to members of the public visiting the Premises.
- 7.8.2Dispose of all rubbish, garbage and trade waste by way of minimum weekly domestic, trade or commercial disposal services in accordance with all laws, ordinances and local council policies and as approved or reasonably instructed by the Landlord.

#### 7.9 **Development Application**

Not do or omit or permit or suffer to be done or omitted anything on or in connection with the Premises the doing or omission of which shall be a contravention of any Act of Parliament or Ordinance or the conditions of any development application approval or of any Notices Orders Licences Consents Permissions and Conditions (if any) served made granted or imposed thereunder and to indemnify the Landlord against all actions proceedings damages penalties costs charges claims and demands in respect of such acts and omissions or any of them except as may be caused by or result from the acts or omissions of the Landlord, its agents or employees, and against the cost of any application for planning permission and the works and things done in pursuance thereof.

#### 7.10 Auction sale

Not conduct or permit an auction sale in the Premises.

#### 7.11 **Driveways and car parking**

Not park or allow motor vehicles to be parked in the driveway nor to obstruct the driveway and not to park nor allow any employees or invitees of the Tenant to park motor vehicles on the Premises other than in allocated motor vehicles parking spaces.

A DB

ME d.X

# 8 Indemnities

- 8.1 The Tenant agrees to:
- 8.1.1 Occupy, use and keep the Premises at the risk of the Tenant and hereby releases to the full extent permitted by law the Landlord and its contractors and employees in the absence of any negligence on their part from all claims and demands of every kind and from all liabilities which may arise in respect of any accident or damage to property or injury to any person in or near the Premises or the Building and the Tenant expressly agrees that in the absence of any such negligence as aforesaid the Landlord shall have no responsibility or liability for any loss or damage to fixtures or personal property of the Tenant.
- 8.1.2 Hereby indemnify the Landlord and its contractors and employees in the absence of any negligence on their part from and against all actions claims demands losses damages costs and expenses incurred by the Landlord or for which the Landlord may become liable in respect of damage to property or injury to any person which may be suffered or sustained in upon or near any part of the Premises or the Building whether in the occupation of the Landlord or of the Tenant or of any other person.
- 8.1.3 Without limiting the generality of clauses 8.1.1 and 8.1.2 the Tenant will and does hereby indemnify the Landlord from and against all claims, actions, demands, losses, damages, costs and expenses incurred by the Landlord or with respect to which the Landlord may become liable in respect of or arising from:
- 8.1.3.1 the negligent or careless use, misuse waste or abuse by the Tenant or any contractor, sub-contractor, licensee, invitee, client customer or visitor of the Tenant or any other person claiming through or under the Tenant of the water, gas, electricity, lighting and other installations in and facilities of the Premises or the Building or arising from any faulty fittings or fixtures of the Tenant;
- 8.1.3.2 overflow or leakage of water (including rain water) in or from the Premises or caused or contributed to by any act or omission on the part of the Tenant or any other person referred to in sub-clause 8.1.3.1 of this clause;
- 8.1.3.3 loss, damage or injury from any cause whatsoever to the Premises or the Buildings or to any property or person within or without the Premises or the Building occasioned or attributed to by any act, omission, neglect, breach or default by the Tenant or any other person as aforesaid.
- 9. Costs

The Tenant shall pay all stamp duty and all the Landlord's legal costs which are recoverable under the Retail Leases Act 1994 (as amended), charges and expenses of and incidental to the preparation, completion stamping, registration of, and the obtaining of any Mortgagee's approval to, this Lease; and of any assignment or

& Mb

ME \$.Z

sub-letting and of any surrender or other termination of this Lease otherwise than by effluxion of time; and in case of default by the Tenant in performing or observing any covenants herein contained or implied the Tenant shall pay to the Landlord all legal costs and other costs, charges and expenses for which the Landlord shall become liable in consequence of or in connection with such default within seven days of the Landlord requesting him so to do.

## 10. Use of Premises

During the term hereby the Tenant will not at any time use the Premises other than as set out in Item 6 of the Schedule.

## DAMAGE TO PREMISES

### Damages

- 11.1 Should any damage occur to the Premises the Tenant shall forthwith give Notice thereof in writing to the Landlord and in particular upon the happening thereof to give to the Landlord Notice of any accident to or defects in the water pipes, gas pipes, lights or fittings used in connection with the water, gas or electrical services and the Tenant shall forthwith make good and repair any such damage or defect to the extent that the Tenant may be required to do so pursuant to the provisions of this Lease.
- 11.2 If the Premises or any part thereof be destroyed or damaged by any of the insured risks and the insurance money under any of the Insurance Policies against the same effected by the Landlord being wholly or partly irrecoverable by reasons solely or in part by any act or default of the Tenant then and in every such case the Tenant will forthwith (in addition to the said rent and any other moneys payable hereunder) pay to the Landlord the whole or (as the case may require) a fair proportion of the cost (including professional and other fees) of completely rebuilding and reinstating the same.
- 11.3 If after the commencement of the said term the Premises or any part thereof be destroyed or damaged so as to be unfit for occupation or use or is inaccessible due to that damage the rent hereby reserved and any amount payable to the Landlord in respect of outgoings or a fair proportion thereof according to the nature and extent of the damage sustained shall be suspended until the Premises shall have again been rendered fit for occupation or use.
- 11.3.1 If the Landlord notifies the Tenant in writing that the Landlord considers that the damage is such as to make its repair impracticable or undesirable, the Landlord or the Tenant may terminate the Lease by giving not less than seven days' Notice in writing to the other and no compensation is payable in respect of that termination.
- 11.3.2 If the Landlord fails to repair the damage within a reasonable time after the Tenant requests the Landlord in writing to do so, the Tenant may terminate the Lease by giving not less than seven days' Notice in writing of termination to the Landlord.

Q.J.B

MF \$.9

- 11.3.3 the aforesaid provisions do not affect any right of the Landlord to recover damages from the Tenant in respect of any damage or destruction to which those provisions apply.
- 11.4 Nothing in this section is to limit any liability of a party to the Lease to pay compensation to another party to the Lease in respect of damage to the Premises or the Building of which the Premises forms part.
- 11.5 Nothing in this section prevents the parties to a Lease from terminating the Lease by agreement if the Premises or the Building of which it forms part is damaged or destroyed.

## 12. Insurance

- 12.1 The Landlord shall keep insured or cause to be kept insured at all times the Premises.
- 12.2 The Tenant shall effect and keep current during the term hereof and any holding-over hereunder a Public Risk Policy for an amount in respect of any single accident or event of not less than that set out in Item 7 of the Schedule. Such Policy shall indemnify the Landlord against any actions, suits, claims and demands of any kind arising from the use by the Tenant, its employees and visitors of the Buildings.
- 12.3 The Tenant shall not at any time during the said term, do, permit or suffer to be done any act, matter or thing whereby any insurance in respect of or in relation to the Premises or the Building or the Land may be vitiated or rendered void or voidable or (except with the approval of the Landlord in writing) whereby the rate of premium on any insurance shall be liable to be increased.
- 12.3.1 In the event the Tenant's use of the premises causes the rate of premium to increase, the Tenant will pay 100% of the increase of the premiums.
- 12.4 The Tenant shall not store chemicals, inflammable liquids or solids acetylene gas or alcohol, volatile or explosive oils, compounds or substances upon or about the Premises or elsewhere in the Building and will not use any of such substances of fluids in or about the Premises or elsewhere in the Building for any purpose and will not in any way create any actual or potential fire hazard in the Premises or elsewhere in the Building. The Tenant will permit the Landlord at all times to enter upon the Premises and to abate any actual or potential fire hazard in or about the Premises and if such hazard in the reasonable opinion of the Landlord, is caused by or results from the use or occupancy of the Premises by the Tenant, the Tenant shall pay to the Landlord on demand the costs and expenses incurred by the Landlord in carrying out such work.
- 12.5 The Tenant will insure in the joint names of the Landlord and the Tenant in the full replacement value thereof all plate glass windows, doors, display showcases and shelves against the risk of breakage, hait, windstorm, flood, riot

) D·I

ME p. 2

and civil commotion, explosion, earthquake, cost of temporary shattering, accidental damage to window frames, showcase frames and fittings.

- 12.6 The Tenant shall from time to time as and when required by Notice in writing from the Landlord forthwith pay all extra premiums payable in respect of insurance on the Building and its contents if any be required on account of extra risk caused by the use to which the Premises are put by the Tenant or by the bringing or keeping on the Premises of any materials or substances.
- 12.7 All Policies of Insurance liable or required to be effected by the Tenant hereunder, whether in respect of the property or risk of the Landlord or the Tenant, shall be taken out with an insurance office or company approved of by the Landlord. The Landlord hereby declares that such approval will not be withheld where the Tenant intends to effect such insurance with a reputable and responsible insurance company.
- 12.8 The Tenant will in respect of any Policy of Insurance to be effected by the Tenant hereunder, if so required, produce to the Landlord the Policy of Insurance and the receipt for the premium last paid in respect thereof.

## ASSIGNMENT, SUB-LETTING, ETC

- 13.1 The Tenant covenants not to assign this Lease except in accordance with the provisions contained in this Clause.
- 13.2 If the Tenant desires to assign this Lease:
- 13.2.1 the Tenant shall, before requesting the Landlord's consent, furnish the proposed assignee with a copy of any Disclosure Statement given to the Tenant in respect of the Lease, together with details of any changes that have occurred in respect of the information contained in that Disclosure Statement since it was given to the Tenant (being changes of which the Tenant is aware or could reasonable be expected to be aware). If the Tenant cannot provide the proposed assignee with a copy of the Disclosure Statement, the Tenant may request the Landlord to provide a copy of that Disclosure Statement and if the Landlord is unable or unwilling to provide that copy within fourteen days after the request, the Tenant need not comply with this paragraph;
- 13.2.2 the Tenant shall request the Landlord's Consent to the assignment of this Lease in writing and shall furnish with that request:
- 13.2.2.1 information regarding the financial resources and financial standing and the business experience and retailing skills of the proposed assignee;
- 13.2.2.2 particulars of the use of the Leased Premises intended by the proposed assignce;

13.2.2.3 confirmation that the Tenant has complied with Clause 13.2.1

\$ 1.8

ME \$2

13.3	The Landlord:
13.3.1	agrees to deal expeditiously with the Tenant's request for Consent to assign this Lease;
13.3.2	is entitled to require the Tenant to furnish to the Landlord:
13.3.2.1	details of the Tenant's financial resources and retailing skills at the time of the request for Consent;
13.3.2.2	such further information as the Landlord may reasonably require concerning the financial standing and business experience of the proposed assignee;
13.3.3	is entitled to withhold Consent to the assignment of this Lease in any of the following circumstances:
13.3.3.1	if the proposed assignee proposes to change the use to which the Premises are put (unless the Landlord consents to the change of use in accordance with the provisions of this Lease);
13.3.3.2	if the proposed assignee has financial resources and retailing skills that are inferior to those of the Tenant;
13.3.3.3	if the Tenant has failed to comply with the provisions contained in this Clause for requesting and obtaining Consent to the assignment.
13.4	For the purposes of the Tenant making a request and the Landlord considering that request under this Clause:
13.4.1	if the assignce is a corporation, the retailing skills of the assignce include the retailing skills of those of its directors who have actively participated in the assignce's business and intend to actively participate in the business in the Leased Premises;
13.4.2	if the Landlord Consents to a change of use by the assignce, the assignce's retailing skills for that use should be compared with the Tenant's retailing skills for the Tenant's use of the Leased Premises;
13.5.	The Landlord may require the assignee to execute an Instrument in which the assignee will assume liability under the Lease for the residue of the lease term:
13.5.1	may require, if the assignee is a corporation whose shares are not listed on an Australian Stock Exchange, the Landlord may require, as a condition of the Landlord's Consent to the assignment, that at least two of its directors or shareholders (chosen by the Landlord) execute personal Guarantees in respect of the payment of rent and the observance and performance of the Lease covenants by the assignee during the lease term. It is agreed that in

JN: M:\Docs\50049\193934.doc

P.B

. -

ME 2.2

any change in the principal shareholding altering the effective control of the Tenant (if a company) shall be deemed to be an assignment of this Lease.

- 13.5.2 will require the Tenant will pay the Landlord's reasonable legal and other expenses of investigating the Tenant's Application for Consent and of preparing, executing, stamping and registering all the required Instruments or Dealings and the costs of obtaining the Mortgagee's Consent.
- 13.6 If the Tenant has complied with the requirements of Clause 13.5, the Landlord shall indicate, within forty two days after the Tenant has made its request under Clause 13.2 and furnished any further details required under Clause 13.3, whether the Landlord consents or withholds consent. If the Landlord does not provide that indication within forty two days, it shall be taken to have consented to the assignment, subject to the conditions in Clause 13.5.
- 13.7 The Tenant will not during the continuance of this Lease transfer demise sub-let or part with possession of the Premises (or any part thereof) or by any act or deed procure the Premises (or any part thereof) to be transferred sub-let unto or put into possession of any person or persons without the previous Consent in writing of the Landlord. Such consent shall not be unreasonably withheld where:
- 13.7.1 the Tenant not being in default under the covenants and agreements on the Tenant's part herein contained proposes sub-let to a sub-Tenant who proves to the satisfaction of the Landlord that such proposed sub-Tenant is a respectable responsible, solvent, fit and proper person capable of adequately carrying on the business in the Premises whilst performing and observing the terms and conditions of this Lease and the Tenant must provide the Landlord with such information as the Landlord may reasonably require concerning the financial standing and business experience of the proposed assignee, and
- 13.7.2 the Tenant pays to the Landlord all proper costs, charges and expenses incurred by the Landlord of and incidental to any enquiries which may be made by or on behalf of the Landlord as to respectability, responsibility, solvency, fitness and suitability of any proposed sub-Tenant, and
- 13.7.3 the Tenant pays the Landlord's solicitors costs in respect of the sub-lease; and
- 13.7.4 the sub-Tenant being a company the directors and also the principal shareholders of such company unconditionally guarantee to the Landlord the performance by the Company of the terms and conditions contained in this Lease or in the proposed sub-lease (as the case may be). It is agreed that in any change in the principal shareholding altering the effective control of the Tenant (if a company) shall be deemed to be an assignment of this Lease.

Ø J.b

MF \$ 2

# LANDLORD'S OBLIGATIONS

14.1 The Landlord HEREBY COVENANTS with the Tenant that the Tenant paying the rent hereby reserved and observing and performing the covenants conditions and agreements herein contained on the Tenant's part to be observed and performed shall and may quietly enjoy the Premises during the said term without any interruption by the Landlord.

# 14.2 Tenant to be given Notice of alterations and refurbishment

The Landlord must not commence to carry out any alteration or refurbishment of the Building of which the Premises forms part which is likely to adversely affect the business of the Tenant unless:

- 14.2.1 the Landlord has notified the Tenant in writing of the proposed alteration or refurbishment at least three months before it is commenced; or
- 14.2.2 the alteration or refurbishment is necessitated by an emergency and the Landlord has given the Tenant the maximum period of notice that is reasonably practicable in the circumstances.

# 14.3 Disturbance

- 14.3.1 Subject to 14.3.3, the Tenant is to be compensated for disturbance if the Landlord:
- 14.3.1.1 inhibits access of the Tenant to the Premises in any substantial manner; or
- 14.3.1.2takes any action that would inhibit or alter, to a substantial extent, the<br/>flow of customers to the Premises; or
- 14.3.1.3 unreasonably takes any action that causes significant disruption of, or has a significant adverse effect on, trading of the Tenant in the Premises; or
- 14.3.1.4 fails to take all reasonable steps to prevent or put a stop to anything that causes significant disruption of, or which has a significant adverse effect on, trading of the Tenant in the Premises and that is attributable to causes within the Landlord's control; or
- 14.3.1.5 fails to rectify any breakdown of plant or equipment under the Landlord's care or maintenance, and the Landlord does not rectify the matter as soon as reasonably practicable after being requested in writing by the Tenant to do so, the Landlord is liable to pay the Tenant reasonable compensation for any loss or damage (other than nominal damage) suffered by the Tenant as a consequence.
- 14.3.2 In determining whether a Landlord has acted unreasonably for the purposes of sub-clause 14.3.1.3, due consideration is to be given to whether the Landlord

D.B

ME D.Z

has acted in accordance with recognised shopping centre management practices.

- 14.3.3 In respect of all occurrences the likelihood of which have been specifically drawn to the Tenant's attention in the Disclosure Statement given to the Tenant before entering into this Lease, the Tenant is precluded from making any claim for compensation for any loss or damage suffered by the Tenant as provided in Section 34 of the Retail Leases Act 1994 and the Landlord is exempted from any liability for compensation. In particular the Tenant acknowledges that the Landlord may during the term, the option period or any holding over period carry out a redevelopment of the site. This redevelopment may cause inconvenience, loss of Tenant's business and/or interruption to the Tenant's business as disclosed. The Tenant acknowledges that he is prevented from making a claim for compensation during the term, the renewed term or any holding over period under Section 34 of The Retail Leases Act as the plan to carry out extensive redevelopment during the term, the renewed term or any holding over period has been disclosed.
- 14.4.4 The provisions implied by this section do not apply to any action taken by the Landlord:
- 14.4.4.1 as a reasonable response to an emergency situation; or
- 14.4.4.2 in compliance with any duty imposed by or under an Act or resulting from a requirement imposed by a public or local authority acting under the authority of an Act.

### **Bond or Bank Guarantee**

- 15.1 On signing of the Lease the tenant must deposit with the Landlord the amount set out in Item 8 of the Schedule as security for the due and punctual observance and performance of all the covenants obligations and provisions of the Tenant's part contained herein.
- 15.2 If at any time the Tenant fails to so duly and punctually observe and perform the Tenant's covenants in the Lease, then the Landlord may in the Landlord's sole discretion at any time appropriate and apply so much of or the whole of the deposit as may be necessary in the opinion of the Landlord to compensate the Landlord for loss or damage sustained or suffered by the Landlord by reason of such breach by the Tenant.
- 15.3 Any such appropriation by the Landlord shall not prejudice any other right of the Landlord arising from such breach.
- 15.4 Should the deposit or any portion thereof be appropriated by the Landlord as aforesaid then the Tenant shall upon demand by the Landlord and within seven days from the date thereof pay to the Landlord the amount of the sum so appropriated.

ME DZ=

).ß

- 15.5 Should the Tenant comply with all the said covenants obligations and provisions and duly and punctually pay all of the rental hereby reserved and all other sum payable by the Tenant to the Landlord hereunder the said deposit less any sums appropriated by the Landlord in accordance with this clause and not paid by or received from the Tenant shall be refunded to the Tenant on the expiration of the term of this Lease or of any extension or renewal thereof.
- 15.6 The Landlord shall be at liberty to pay the said deposit less any sums appropriated by the Landlord in accordance with this clause and not paid or received from the Tenant to any assignee or transferee of the Landlord's interest in the Premises in the event that such interest of the Landlord is assigned or transferred and thereupon the Landlord shall be discharged from all liability to the Tenant or any other person with respect to such deposit.
- 15.7 The deposit paid by the Tenant to the Landlord as security for the performance of the Tenant's obligations under the Lease must be held by the Landlord on behalf of the Tenant in an account bearing interest. The Landlord must account to the Tenant for interest earned on the deposit but the Landlord is entitled to retain any such interest and deal with it as money paid by the Tenant to the Landlord to form part of the deposit concerned.

## Termination

- 16.1 At the expiration or sooner determination of the said term the Tenant must:
- 16.1.1 quietly yield up unto the Landlord the Premises together with all additions and improvements made thereto and all fixtures in or upon the Premises or which during the said term may have been affixed or fastened to or upon the same and in such state and condition as shall in all respects be consistent with full and due performance by the Tenant of the covenants on the part of the Tenant herein contained.
- 16.1.2 If so required by the Landlord, remove any wall paper, carpet, floor tiles or any other covering which may have been affixed to the Premises whether by the Tenant or not and whether with the Consent of the Landlord or not and to resurface the walls ceiling and floors, in a manner approved by the Landlord consistent with the nature and character of the Building.
- 16.1.3 Repair any damage caused to the Premises by the removal of any of the Tenants fixtures and in particular make good any damage to the walls and ceiling caused by the removal of the air conditioning unit in the restaurant area and the cool room.
- 16.1.4 Remove at the Tenant's own expense all signs, advertisements or Notices inscribed painted or affixed by the Tenant (or by the Landlord at the Tenant's request) on or in any part of the Building and shall make good to the satisfaction of the Landlord any damage or disfigurement existing after such removal. In the event that the Tenant shall fail to make good any such damage or disfigurement this may be made good by the Landlord and the Tenant shall

e D.B

ME J.Z

pay to the Landlord the cost thereof within seven days of the Landlord notifying the Tenant in writing of the amount thereof.

16.1.5 In the event that any part of the Premises which has been previously painted has not been painted during the period of two years immediately preceding the expiration or sooner determination of the said term then the Tenant shall to the satisfaction of the Landlord paint every such previously painted surface with good quality paint to a standard and in colours approved by the Landlord consistent with the nature and character of the Building.

## NOTICES

- 17.1 Any Notice required to be given under this Lease shall be in writing.
- 17.2 Any Notice to the Tenant (if a Corporation) shall be sufficiently served if left addressed to the Tenant on the Premises or sent by registered post or left at its registered office of the Tenant and (if the Tenant shall be a person) shall be sufficiently served if left addressed to the Tenant on the Premises or sent by registered post or left addressed to the Tenant at the Tenant's last known address in Australia.
- 17.3 Any Notice to the Landlord (if a Corporation) shall be sufficiently served if left addressed to the Landlord at or sent by registered post to the registered office of the Landlord and if the Landlord shall be a person shall be sufficiently served if sent by registered post to the address of the Landlord stated herein or at such other address as the Landlord may have in writing stipulated to the Tenant.
- 17.4 Any Notice sent by registered post shall be deemed to have been duly served at the expiration of twenty-four hours after the time of posting and in proving service it shall be sufficient to prove that the envelope containing the Notice was duly addressed to the Tenant or the Landlord (as the case may be) in accordance with this Clause and left at or posted to the address to which it is so addressed.

### **OPTION TO RENEW**

### **Tenant's option**

- 18.1 The Landlord must grant a new Lease to the Tenant, at the cost of the Tenant on the Expiry Date to commence on the next day only if:
- 18.1.1 the Tenant gives the Landlord notice in writing no less than 6 months and no more than 9 months before the Expiry Date stating that it wants a new Lease for a further term set out in Item 9 of the Schedule
- 18.1.2 the Tenant in the meantime continues to duly and punctually pay the rent reserved by this Lease and duly observe and perform the covenants, obligations and agreements in accordance with the terms of the Lease
- 18.1.3 when the Tenant gives the Notice and on the Expiry date, the Tenant is not in breach of this Lease

£ 1).B

MF p.Z

## New Lease

18.2.	The new Lease is to be identical with this Lease except that:	
18.2.1.	this clause 18 and clauses 3.1 to 3.4 are to be deleted, and a new cla	

- use 3.1 to 3.4 are to be inserted in accordance with the terms set out in clause 18.4 hereunder.
- 18.2.2. the term, commencement date, expiry date and review dates are to be as set out in item 10 of the schedule
- the rent from the commencement date of the new Lease will be determined in 18.2.3 accordance with clause 18.3.

### **Market Review**

- The Landlord shall within thirty five days of the receipt of Notice to be given 18.3 by the Tenant in clause 18.1 inform the Tenant in writing of the Landlord's assessment of the rent for the first year of the new term.
- 18.3.1 The Tenant may accept the Landlord's assessment and in that event the Tenant fails or refuses to accept the Landlord's assessment within twenty eight days after service of the assessment or if the Landlord and Tenant are unable to agree on some other amount as the current market rent for the first year of the new term the current market rent shall be determined in accordance with this clause 18.3.
- 18.3.2 The Parties must apply within 14 days to the President or the Principal Officer for the time being of the Australian Property Institute Inc (New South Wales Division) ("the Nominator") to nominate a person ("the Nominee") who is a licensed valuer and
- 18.3.2.1 has been a full Member of not less than ten years standing of the Australian Property Institute Inc (New South Wales Division);
- 18.3.2.2 must be the holder of a licence to practice as a Valuer of the kind of premises by this Lease;
- 18.3.2.3 must have at least ten years experience in valuing such kind of premises and be active in the market at the time of the appointment.
- 18.3.3 The Nominee shall act as an expert and not as an arbitrator. Each party must submit to the Nominee written Valuations and Submissions within 14 days after the Nominee has accepted the nomination to act, but may not make oral submissions or adduce any evidence. At the time of making any written Submissions or forwarding to the Nominee any written Valuation, that party shall forward to the other party a copy of all written materials submitted to the Nominee. Within 14 days after the Nominee has accepted the nomination to act each party may forward to the Nominee written comments on the other party's written Valuations and Submissions. The Nominee shall take into

Z D.B

ME DZ

consideration any written Submissions received within those periods but is not to be fettered by them and shall determine the current market rent in accordance with his own Judgment in the opinion he forms. The Nominee shall conclude his Determination in writing and shall inform the parties of it forty five days after having accepted the nomination to act in writing giving detailed reasons for his Determination and must specify the matters to which he had regard for the purposes of making his Determination). The Nominee's Determination is final and is binding on the parties.

- 18.3.4 If the Nominee:
- 18.3.4.1 fails to accept the nomination to act;
- 18.3.4.2 fails to determine the current market rent within sixty days after accepting the nomination to act;
- 18.3.4.3 becomes incapacitated or dies;
- 18.3.4.4 resigns as the Nominee;

then either party may request the nominator to appoint another Nominee in accordance with sub-clause 18.3.2.

18.3.5 The parties shall bear equally the total cost of the rent determination for the rent for the first year of the renewed term. Each party shall bear its own costs of any experts for making valuations for the purpose of written Submissions,

If the Tenant fails to contribute its share of the rent determination within 7 days of the request to do so either by the Landlord or the Nominee, such failure will be a fundamental breach of the Lease thereby terminating the Tenant's rights to take up the option for a new Lease.

- 18.3.6 If the current market rent for the first year of the renewed term shall not be determined by the date of commencement of the renewed term:
- 18.3.6.1 the Tenant shall pay interim rent calculated by:
- 18.3.6.2 increasing the current instalments of rent due by 80% of the Landlord's assessment of the increase in rent, if such increase is less than \$10,000.00, or
- 18.3.6.3 increasing the current instalments of rent due by 60% of the Landlord's assessment of the increase in rent, if such increase is more than \$10,000.00

until the new rent has been determined;

18.3.7 Immediately upon the rent has been determined the Tenant shall pay the additional amount (if any) due for rent from the date of commencement of the renewed term to the date when the next monthly instalment of rent is

2 1).8

ME SZ

payable on the rent day next succeeding the expiration of thirty days after the date when the Tenant has been informed of the rent determination.

18.3.8 If the new rent is less than the interim rent, the Landlord will reimburse the Tenant the over payments within 14 days.

### 18.4 "3. RENT

#### Rent

- 3.1 The Tenant covenants with the Landlord to pay free from all deductions the annual sum set out in Item 2 (a) of the Schedule for rent for the first year of the term by monthly payments set out in Item 2 (b) of the Schedule each in dvance.
- **3.2** The Tenant must pay the first instalment on or before the Commencement Date.
- 3.3 The Tenant shall not withhold or be entitled to withhold the whole or part of any such payment by way of deduction, set off or counter claim in respect of any claim for damages or for compensation which the Tenant shall make or has made against the Landlord.

#### 3.4 Review of Rent

As and from each anniversary of the commencement date hereof (such anniversary occurring during the term or during any holding over period) the yearly rental for the ensuing year shall be the rent paid for the preceding year increased by five (5%) per cent."

#### **INTEREST ON OVERDUE MONEY**

19. Without prejudice to the rights, powers and remedies of the Landlord otherwise under this Lease, the Tenant will pay to the Landlord interest as set out in Item 11 of the Schedule on any moneys due to the Landlord from the Tenant but unpaid for fourteen days after becoming due on any account whatsoever pursuant to this Lease. Such interest shall be computed from the due date for the payment of the money in respect of which the interest is chargeable until payment of such moneys in full and to be recoverable in a like manner as rent in arrears.

#### HEADINGS

20. The headings to each Clause or paragraph of this Lease are inserted only for the purpose of ease of reference and are not to be used for the interpretation of this Lease.

## LEASE DOCUMENTATION

21.1 If the Lease is not to be registered, the Landlord must provide the Tenant with an executed copy of the stamped Lease within one month after the Lease is returned to the Landlord or the Landlord's lawyer or agent following payment of stamp duty on the Lease.

£ 13

ME D.Z

- 21.2 If the Lease is to be registered, the Landlord must lodge the Lease for registration within one month after the Lease is returned to the Landlord or the Landlord's lawyer or agent following payment of stamp duty on the Lease and the Landlord must provide the Tenant with an executed copy of the stamped and registered Lease within one month after the Lease is returned to the Landlord or the Landlord's lawyer or agent following registration of the Lease.
- 21.3 The periods specified in clauses 21.1. and 21.2 are to be extended for delays attributable to the need to obtain any Consent from a head Landlord or mortgagee.
- 21.4 In clauses 21.1 and 21.2 "registered" means registered under the Real Property Act 1900.
- 21.5 This clause does not affect the operation of the Stamp Duties Act 1920 or the Real Property Act 1900.

### GOODS AND SERVICES TAX

- 22.1 Unless the Rent and/or Other Moneys are expressly stated in this Lease to be inclusive of GST, the Rent and/or Other Moneys, price, fees, rates, charges, tax and expenses (:the Amounts") payable by the Lessee to the Lessor under this Lease do not include any goods and services tax ("GST").
- 22.2 Where the Rent and/or Other Moneys are expressly stated in this Lease to be inclusive of GST, the GST is 1/11<sup>th</sup> of the GST inclusive Rent. If the rate of GST is varied at any time during the term, the GST inclusive Rent shall be adjusted in accordance with any changes in the rate of GST.
- 22.3 If any GST is incurred or payable by, or collectable from the Lessor in respect of any supply under this Lease the Lessee must pay the GST to the lessor in addition to the Amounts provided that the payment or reimbursement will not in the aggregate exceed 10% of the Amounts or such other percentage as may be the statutory GST rate from time to time.
- 22.4 The GST must be paid by the Lessee to the Lessor at the earliest of:
  - (i) 7 days prior to the date upon which the Lessor is required to remit the GST to the Australian Taxation Office; or
  - (ii) at the same time as the Amount is required to be paid by the Lessee to the Lessor.
- 22.5 The Lessor shall promptly provide the Lessee with a tax invoice in the form specified by the relevant GST regulations from time to time,
- 22.6 Any GST charged on Outgoings shall be reduced by the amount of input tax credits received by the Lessor in respect of GST on the cost of supply of those Outgoings to the Lessor.

E D.B

ME de

## DISCLOSURE STATEMENT

23. The Tenant acknowledges the receipt of the Disclosure Statement required to be given to the Tenant under the Retail Leases Act 1994.

## **GUARANTEE**

24.1 Intentionally Deleted

## **DEFINITION AND INTERPRETATION**

25. In this Lease unless the context otherwise requires:

"the Building" means the building or buildings and other structures erected or to be erected on the Land within which the Premises are situated together with any modifications, additions, extensions or alterations thereto with the fixtures, fittings, furnishings, plant, machinery and equipment of the Landlord therein from time to time.

"the Landlord" shall mean and include where the Landlord is a company its successors and assigns and where the Landlord is a person the Landlord's executors, administrators and assigns.

"the Tenant" shall mean and include where the Tenant is a company its permitted successors and assigns and where the Tenant is a person the Tenant's executors administrators and permitted assigns.

In the event of there being two or more Landlords or two or more Tenants under this Lease (including whilst the Lease or the reversion is held by legal personal representatives, successors or assigns), each of them is jointly and severally liable to perform covenants and obligations under this Lease.

"the Premises" or "the Premises" means the premises hereby leased including the fixtures, fittings and appurtenances belonging thereto.

"the Guarantor" or "the Guarantors" means the person or persons (if any) named herein as guarantor and includes the guarantor's executors administrators successors and assigns and every two or more of them jointly and severally:

Words importing:

- (i) The singular number include the plural numbers and vice versa.
- (ii) Any gender include all other genders.
- (iii) Persons include corporations.
- (iv) References to Statutes include all Statutes amending consolidating or replacing the Statute referred to.

MEDE

P.B

The expression "current market rental" means:

- 1. The best annual rent that can be reasonably obtained for the leased Premises;
- 2. On the basis that the Premises are available for leasing for the use to which the Premises may be put in accordance with the Lease with vacant possession by a willing Landlord to a willing Tenant for the residue of the current lease term (and Option for Renewal) at the review date:
- 3. On the terms and conditions contained in this Lease and such other matters as are relevant to the assessment of current market rent;
- 4. On the basis that the Tenant's Lease covenants and obligations shall have been fully performed at the review date:
- 5. Without taking into account any goodwill created by the Tenant's occupation of the Premises; improvements or fixture erected or installed at the Tenant's expense which the Tenant is permitted or required to remove at the termination of the Lease, except for permanent structural improvements to the leased Premises installed at the Tenant's expense which the Tenant is not permitted to remove at the termination of the Lease, which shall be taken into account;
- 6. Having regard to the rental values of comparable premises.
- 7. Rent concessions and other benefits that are frequently or generally offered to prospective Tenants of unoccupied premises are declared to be relevant matters to be taken into account in the assessment of current market rent.

The reference to "Consumer Price Index" "(CPI)" means the Consumer Price Index All Groups Sydney or the index officially substituted for it.

CPI adjustment of rent in accordance with the CPI:

- (i) take the yearly rent as of the last review date or if none, the rent at the commencement date (\$X),
- divide that rent by the Consumer Price Index Number for Sydney (All Groups) (ii) for the quarter ended just before that date (CPI 1),
- (iii) multiply the result by the Consumer Price Index Number for Sydney (All Groups) for the quarter ended just before the review date (CP1 2).

The product is the new rent for the year beginning on the review date (\$Y), written as a formula --

> \$X x CP2 = \$Y CPI 1

The Landlord must calculate the new rent after each review date and give the tenant written notice of the new rent.

? DB

ME \$2

If the Australian Bureau of Statistics makes a change in the reference base of the index and there is a published co-relation between the old and new base then the published co-relation is to be applied to convert the CPI I figure to the new reference base. If there is none then the landlord and the tenant agree to accept the calculations of the landlord's solicitor who must be retained to determine a fair co-relation between the old and the new series of numbers.

## **DEMOLITION**

- 26.1 In the event that the Landlord intends to demolish the Building of which the Premises forms part the Lessor may terminate the Lease by written Notice to the Tenant, but:-
  - (a) The Lease cannot be terminated on that ground unless and until the Landlord has provided the Tenant with details of the proposed demolition sufficient to indicate a genuine proposal to demolish the Building within a reasonably practicable time after the Lease is to be terminated.
  - (b) The Lease cannot be terminated by the Landlord on that ground without at least six months written Notice of termination.
  - (c) If Notice of termination on that ground is given to the Tenant, the Tenant may terminate the Lease by giving the Landlord not less than seven days written Notice of termination at any time within six months before the termination date notified by the Landlord.
- 26.2 If the Lease is for a term of twelve months or less, the period of six months in sub-section (1)(b) and (c) is shortened in each case to three months,
- 26.3 If a Lease is terminated on such a ground and demolition of the Building is not carried out within a reasonably practicable time after the termination date notified by the Landlord, the Landlord is liable to pay the Tenant reasonable compensation for damage suffered by the Tenant as a consequence of the early termination of the Lease unless the Landlord establishes that at the time Notice of termination was given by the Landlord there was a genuine proposal to demolish the Premises within that time.
- 26.4 "demolition" of the Building of which the Premises forms part includes any substantial repair, renovation or reconstruction of the Building that cannot be carried out practicably without vacant possession of the Premises.

## **DISPUTE RESOLUTION**

27. If a dispute arises out of or relates to this agreement or the breach, termination, validity or subject matter thereof, or as to any claim in tort, in equity or pursuant to any domestic or international statute or law, the parties to the agreement and to the dispute expressly agree to first endeavour in good faith to settle the dispute by mediation administered in accordance with the Retail Tenancy Unit before having recourse to arbitration or litigation..

Z D.B

ME \$.2

27.1	A party claiming that a dispute has arisen, must give written notice to the other parties to the dispute specifying the nature of the dispute.
27.2	On receipt of the notice specified 18.1 the parties to the dispute must within seven (7) days of receipt of said notice seek to resolve the dispute.
27.3	The mediation shall be conducted and administered in accordance with Retail Tenancy Mediation Rules which set out the procedures to be adopted, the process of selection of the mediator and the costs involved and which terms are hereby deemed incorporated.
28.5	This clause shall not merge upon completion.

ME J.2

,

JN: M:\Docs\50049\193934.doc

, CIB

. '

84 Req:C145584 /Doc:DL AD684221 /Rev:18-Jan-2008 /Sts:OK.OK /Prt:27-Mar-2015 10:21 /Pgs:ALL /Seq:31 of 32 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

29

#### SCHEDULE

#### ITEM:

, **'** 

- 1. Term (Clause 2.1) TWO YEARS
- 1(a) Commencement Date 12 January 2006
- 1 (b) Expiry Date 11 January 2008
- 2(a) \$33,852.00 per annum plus GST of \$3,385.20 = .\$37,237.20 per annum including GST
- 2(b) \$2821.00 per month plus GST of \$282.10 = \$3,103.10 per month including GST
- 2(c) \$33,852.00 per annum plus GST of \$3,385.20 = .\$37,237.20 per annum including GST
- 2(d) \$2821.00 per month plus GST of \$282.10 = \$3,103.10 per month including GST
- 3. Rent Review See 2(c) and 2(d) above
- 4. Percentage on Holding Over Rent (Clause 3.5) 5% annually
- 5. Outgoings (Clause 4.1.1) Not Applicable
- 6. Use of Premises (Clause 10) Pizza shop including eat-in, takeaway or delivery of pizza, retail sale of drinks, pasta, milk, salads, garlic bread, desserts and ancillary products.
- 7. Public liability insurance (Clause 12.2) \$10,000,000.00.
- 8. Bond (Clause 15.1) \$6,206.20 including GST
- 9. Option (Clause 18.1.1) Two years
- 10. Option Review Dates (Clause 18.2.2)
   Lease Commencement Date 12 January 2008
   Lease Expiry Date 11 January 2010
   Rent Review Dates –12 January 2009 to be increased by factor of 5% of the previous years rent.
- 11.Interest on Overdue Money (Clause 19)Twelve (12%) percent per annum
- 12. Guarantor(s) (Clause 24.1) Not Applicable

DIB

ME \$ 2

IN WITNESS WHEREOF of the parties hereto have hereunto set their hands and affixed their seals the day and year first hereinbefore written.

SIGNED by the said Landlord MICHAEL ELIAS & SAMIRA ELIAS in the presence of:

¥.....

χ

χ

SIGNED by the said Tenant DAVID BAYSSARI in the presence of:

, **†** 

. . . . . . . . . . .

.................

SIGNED by the said Tenant MARIA BAYSSARI in the presence of:

لكو 0

. . . . . . . . . . . . . .

2

85 Rég:Cl45585 /Doc:DL 5382069 /Rev: Rev.lpi:sixdre /Src:W /WARNING: A4 C	11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:1 of 19 opy Supplied by LPI NSW for Conveyancing Purposes Only.
B ISSU	LEASE Real Property Act 1900 LAW SOCIETY LEASE
© 1994 COPYRIGHT of the Law Society of New South Wales which has approved this page and the next page as printed.	Office of State 50°5551\$ 10/577267102 TI 2026 866010 ALTICI JUSTIS "M"S"N
PROPERTY LEASED	
Show up to 20 references to title. If part of the land in a title is being leased, specify the part.	FOLIO IDENTIFIER B/410947 being PART SHOP 3&4 116 STATION STREET, WENTWORTHVILLE
LODGED BY OFFICE USE	LTO Box Name, address or DX, and telephone osciph. G. Capogreco & Ascociatios olicitors & Altorneys 8 New Centerbury Road, Petersham 2049 O: Hox. 729 Petersham 2049 REFERENCES (marx156Ha23ders)
LESSOR (landlord)	ANTONIO DUARTE AND MARIA DUARTE
The lessor leases to the lessee the pr	operty described above subject to the following ENCUMBRANCES
1 2	
LESSEE (tenant) ACN	ACRAE HOLDINGS PTY LTD DO1 772 514
TENANCY:	(e.g. joint tenants)
	SCHEDULE OF ITEMS
1. LEASE PERIOD: FIVE (5)	) YEARS
2. COMMENCEMENT DATE: 1 J	JULY 1998
With an OPTIONSTO BENEW for	JUNE 2003 لما a period of FIVE (5) YEARS set out in clause 4
Month 5. With an OPTION TO PURCHAS	
6. Together with and reserving the	he <b>RIGHTS</b> set out in Not applicable
7. This lease includes ANNEXURE	ES A and B.
8. This lease includes the provisi	ons in MEMORANDUM No. filed in the LTO. Not applicable
SEE A SOLICITOR ABOUT THIS	S LEASE CHECKED BY (office use only)
AD. HD.	100-374 A

/Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:2 of 19 85 Rig:C145585 /Doc:DL 5382069 Rechlpi:Bixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only. . . We certify this dealing correct for the purposes of the Real Property Act 1900. DATE: 2 Signed in my presence by the lessor who is personally known to me Signature of witness Maria Dearte Antonio Dearte ANNA SousA NAME OF WITNESS (BLOCK LETTERS) 18 New Ca Address of witness Signature of lessor ••• ת. mman Seal Signed in my presence by the lessee who is personally known to me The Common Seal of K.C. Macrae Holdings Pty Ltd was affixed in accordance with Signature of witness its articles of association in the presence NAMB OF WITNESS (BLOCK LETTERS) DIRECTOR/SECRETARY DIRECTOR ......... Address of witness Signature of lessee Signed in my presence by the guarantor who is personally known to me (BLOCK LETTERS) Macha of witness Signature of guarantor I solemnly and sincerely declare in respect of every option to renew or purchase in Lease No. that the time for exercise of the option has ended and that the lessee under that lease has not exercised the option. I make this solemn declaration conscientiously believing it to be true and by virtue of the Oaths Act, 1900. in the presence of Signature of witness

NAME OF WITNESS (BLOCK LETTERS)

Address of witness

Signature of lessor

ANNEXURE	A to the Law Society Lease		Page 1 of L pages Annexure A
	NTONIO DUARTE & MARIA DUARTE	_	
	C.C. MACRAE HOLDINGS PTY LTD		
This annexur		AUN UUI //2 314	
····	CONDICION ON LUBIO.		
NOTE: Any a Annexure A	lterations and additions to Lease Covenants The printed clauses in Annexure B are to re	in Annexure B must be made by addit	tional clauses in
	. The printed clauses in Annexure B are to re	main in their copyright form without a	alteration.
Que titte	SCHEDULE OF ITEMS (continue		
Item 9		TTA-CAE IRISTOPHER MCRAEF & PATRIC	TANY MODAR
(cl 2.3) (cl 13.7)	B. Limit of guarantor's liability:	KISIUPHER MURADO O TAINIO.	IA MAY MUKAE
Item 10	Additional leased property:		
(cl 3)	Additional leased property.		
Item 11	/ Option to renew		
Part M /	A. Further period of 5	TOTAL FROM 1 THE V OLOD	
J. 10 (11) mg-	to 30 JUNE 200	years from 1 JULY 2003	
R.M. SPS King	B. Further period of 5 to 30 JUNE 2003	years from 1 JULY 200 <b>3</b>	
Pm.MKn	C. Maximum period of tenancy under this lease and permitted r	renewals: 1 <b>5</b> years	
MOA	D. First day option for renewal can	be exercised: 30 MARCH 2003	
R.M. 11/Ka	E. Last day option for renewal can	30 MAY 2003	
Item 12	Rent	30 MAXCH 2008 30 MAY 2008	
(cl 5)	For the lease period:		
	From the commencement date to the first rent review date:	\$ 80,, 402.40 a year by monthly instalments of \$	\$ 6,700.20
	Afterwards:	At the new yearly rent begin review date by monthly inst one twelfth of the new year	inning on each stalments of
	For the further period in item 11A: From the commencement date to the first rent review date: (for example: Current market ren		
	Afterwards:	At the new yearly rent begin	inning on each
1/1	est /	review date by monthly inst one twelfth of the new year	stalments of
	Tanfe O	ALSO D	rly rent.
1.0	llacker	1.1. Serve	
	•	••••••••••••••••••••••••••••••••••••••	

85 Reg:C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:4 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

Page 2 of $\mathcal{V}$ pag Annexure A	jes , , , , , , , , , , , , , , , , , , ,
	For the further period in item 11B: From the commencement date to the first rent review date: (for example: Current market rent)
	Afterwards: At the new yearly rent beginning on each review date by monthly instalments of one twelfth of the new yearly rent.
Item 13 (cl 5)	Outgoings
((1.5)	<ul> <li>A. Share of outgoings:</li> <li>B. Outgoings -</li> </ul>
	<ul> <li>(a) local council rates and charges; N/A</li> <li>(b) water sewerage and drainage charges; N/A</li> <li>(c) land tax; N/A</li> <li>(d) insurance N/A</li> </ul>
	for the land or the building of which the property is part, fairly apportioned to the period of this lease.
Item 14 (cl 5.1.5)	Interest rate: 12%
Item 15 (cl 5.4)	Rent review
(,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Rent review date 30/6/99 Method of rent review 1 If Method 1 applies, increase by (the increase should show a percentage or amount).
	Method 1 is a fixed amount or percentage. 5% Method 2 is Consumer Price Index. Method 3 is current market rent.
	Method 2 applies unless another method is stated.
Item 16 (cl 6.1)	Permitted use: PHARMACY, TOY SHOP, STORAGE AND OFFICE
Item 17 (cl 8.1.1)	Amount of required public liability insurance: \$10MILLION
jh n V	sola Macha

AD. HD

\*\*\*\*\*\*\*

85 Eeq:C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:5 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.



ANNEXURE B to the Law Society Lease

Landlord: ANTONIO DUARTE & MARIA DUARTE Tenant: K.C. MACRAE HOLDINGS PTY LTD ACN 001 772 514

This annexure consists of 12 pages.

©1994 COPYRIGHT of the Law Society of New South Wales which has approved this annexure as printed to clause 14. WARNING: Unauthorised reproduction in whole or in part is an infringement of copyright.

**NOTE:** Any alterations and additions to Lease Covenants in Annexure B must be made by additional clauses in Annexure A. The printed clauses in Annexure B are to remain in their copyright form without alteration.

#### CONTENTS

. .

CLAUSE	SUBJECT	PAGE CLAUSI	SUBJECT PAGE
1	Form of this Lease		Insurance and Damage
	Parties		Access
	The Property		Transfer and Sub-Lease
4	Lease Period		Landlord's other Obligations
5	Money		Forfeiture and End of Lease
6	Use		Guarantee
7	Condition and Repairs		Exclusions, Notices and Special Clauses 11

### **RETAIL LEASE CERTIFICATE**

If section 16 of the Retail Leases Act 1994 applies to this lease, and the term plus any further terms are less than 5 years, the term will be extended unless a section 16 certificate is given. Sections 16(1) and (2) provide —

16(1) The term for which a retail shop lease is entered into, together with any further term or terms provided for by any agreement or option for the acquisition by the lessee of a further term as an extension or renewal of the lease, must not be less than 5 years. An agreement or option is not taken into account if it was entered into or conferred after the lease was entered into.
16(2)

16(2) If a lease is entered into in contravention of this section, the validity of the lease is not thereby affected but the term of the lease is extended by such period as may be necessary to prevent the lease contravening this section.

I certify that I am a solicitor not acting for the landlord and that at the request of the tenant I explained to the tenant before the tenant entered into this lease —

- the effect of sections 16(1) and (2); and
- that the giving of this certificate would result in section 16 not applying to this lease.

Date Malandar A.D. M.D.

Signature	
NAME (BLOCK LETTERS)	
Al P. Kash	
11-12-2007	

Page 1 of 12 Annexure B 85 Epq:C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:6 of 19 "Ref:1pi:sixdre /Src:W /WARNING: A4-Copy Supplied by LPI NSW for Conveyancing Purposes Only.

•

٤,

#### Page 2 of 12 Annexure B

#### CLAUSE 1 FORM OF THIS LEASE

#### What are the parts to this lease?

- 1.1 There are three parts to this lease a lease form, Annexure A and this annexure.
- 1.2 This lease is a deed even if it is not registered.

#### CLAUSE 2 PARTIES

#### Who are the parties to this lease?

- 2.1 The landlord. The landlord is also called the lessor (or in the case of a sublease, the sublessor) and is named on page 1 of this lease.
- 2.2 The tenant. The tenant is also called the lessee (or in the case of a sub-lease, the sub-lessee) and is named on page 1 of this lease.
- 2.3 The guarantor, if a guarantor is named in item 9 in the schedule.
- 2.4 If a party consists of two or more persons, obligations of that party can be enforced against any one or more of them.

#### CLAUSE 3 THE PROPERTY

#### What property is leased?

- 3.1 The property leased is described on page 1 of this lease.
- 3.2 The landlord's fixtures are included in the property leased.
- 3.3 If anything else is leased (such as furniture belonging to the landlord) and is described in item 10 in the schedule it is included in the property.
- 3.4 If the property has facilities and services shared in common with other persons in the same building as the property, clause 11.3.2 applies to those common facilities. The tenant shares the common facilities with the landlord, and with other tenants of the landlord. The landlord can set reasonable rules for sharing these common facilities.

#### CLAUSE 4 LEASE PERIOD

#### How long is this lease for?

- 4.1 This lease is for the period stated in item 1 in the schedule, commences on the date stated in item 2 in the schedule and ends on the date stated in item 3 in the schedule.
- 4.2 If a further period, commencing when this lease ends, is stated in item 11A in the schedule then the tenant has the option to renew this lease for that period.
- 4.3 The tenant can renew this lease more than once if that is stated in item 11B in the schedule. However the period of tenancy under this lease and under any renewal(s) is, in total, not longer than the maximum period stated in item 11C in the schedule.
- 4.4 The tenant can exercise the option only if
  - 4.4.1 the tenant serves on the landlord a notice of exercise of option not earlier than the first day stated in item 11D in the schedule and not later than the last day stated in item 11E in the schedule;

85 R<sup>l</sup>eg:Cl45585 /Doc:DL 5382069 /Rev:ll-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:7 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.



Page 3 of 12 Annexure B

4.4.2 there is at the time of service no rent or outgoing that is overdue for payment; and
4.4.3 at the time of service all the other obligations of the tenant have been complied with or fully remedied in accordance with the terms of any notice to remedy given by the landlord.

If this lease is extended by legislation, items 11D and 11E in the schedule are adjusted accordingly.

4.5 After exercising the option the tenant must continue to pay all rents and outgoings on time and continue to comply with all of the tenant's obligations under this lease. If the tenant does not do so, the landlord may treat any breach as being a breach of the new lease as well as of this lease.

4.6

, t

, i

A new lease will be the same as this lease except for —

- 4.6.1 the new rent;
- 4.6.2 the commencement date and the termination date;
- 4.6.3 the omission of clauses 4.2, 4.3, 4.4, 4.5 and 4.6 and items 11A and 11B in the schedule in the last lease allowed in item 11 in the schedule;
- 4.6.4 item 11B becoming item 11A;
- 4.6.5 adjustment of item 11C in the schedule; and
- 4.6.6 adjustment of items 11D and 11E in the schedule. The number of days between the dates stated in items 11D and 11E in the schedule of the new lease and the termination date of the new lease and the number of days between each date stated in items 11D and 11E in the schedule of this lease and the termination date of this lease are to correspond.

If the new rent is to be current market rent it will be decided in the same way that current market rent is to be decided under Method 3 stated in clause 5 assuming that this lease and the new lease were one continuous lease and the commencement date of the new lease was a rent review date.

### CLAUSE 5 MONEY

### What money must the tenant pay?

- 5.1 The tenant must pay to the landlord or as the landlord directs
  - 5.1.1 the rent stated in item 12 in the schedule;
  - 5.1.2 the share stated in item 13A in the schedule of those outgoings stated in item 13B in the schedule;
  - 5.1.3 the reasonable cost to the landlord of remedying a default by the tenant;
  - 5.1.4 the reasonable cost to the landlord of dealing with any application by the tenant for the landlord's consent under this lease (whether or not it is given);
  - 5.1.5 interest on these moneys at the rate stated in item 14 in the schedule when payment is more than 14 days overdue, calculated from the due date to the date of payment;
  - 5.1.6 registration fee for registration of this lease at the Land Titles Office (payable on delivery to the landlord's solicitor of the executed lease);
  - 5.1.7 stamp duty on this lease (payable on delivery to the landlord's solicitor of the executed lease) if not previously paid by the tenant to the Office of State Revenue;
  - 5.1.8 if the tenant defaults, the landlord's reasonable legal costs relating to the default; and
  - 5.1.9 the landlord's reasonable costs and expenses in connection with the preparation of this lease.

5.2 The first month's instalment of rent is to be paid by the commencement date. Each later month's instalment of rent is to be paid in advance.

۴,

Page 4 of 12 Annexure B

5.3 A payment under clause 5.1.2 must be paid on the next rent day after a request for payment is made by the landlord.

A request for payment can be made --

- 5.3.1 after the landlord has paid an outgoing; or
- 5.3.2 after the landlord has received an assessment or account for payment of an outgoing.

If item 13B in the schedule refers to land tax ---

- if the property is a strata lot, the relevant land tax is land tax on that lot;
- if the property is not a strata lot but is part of a building, the relevant land tax is land tax on the land on which the building is situated, plus any land of the landlord used or available for use by or for the benefit of tenants conducting business in the building or in connection with trading in the building; and
- in either case, the land tax must be calculated as if the land was the only land owned by the landlord and there was no special trust or non-concessional company involved.

#### When and how is the rent to be reviewed?

5.4 The rent is to be reviewed on the rent review dates stated in item 15 in the schedule.

If this lease is extended by legislation, the rent review dates include each anniversary of the latest rent review date stated in item 15 in the schedule (or if none is stated each anniversary of the commencement date) which falls during the extension.

- 5.5 The tenant must continue to pay rent at the old rate until the new rate is known. After that, the tenant is to pay the new rent from the next rent day. By that rent day the tenant is also to pay any shortfall between the old and new rate for the period since the rent review date. Alternatively, the landlord is to refund to the tenant any overpayment of rent.
- 5.6 There are three different methods described here for fixing the new rent on a rent review date. The method agreed by the landlord and the tenant is stated at item 15 in the schedule. The tenant is entitled to a reduction if the method produces a rent lower than the rent current just before the review date.

#### Method 1. By a fixed amount or percentage.

5.7 In this case the new rent beginning on each review date is stated in item 15 in the schedule.

#### Method 2. By reference to Consumer Price Index.

- 5.8 In this case
  - take the yearly rent as of the last review date or if none, the rent at the commencement date (\$X),
  - divide that rent by the Consumer Price Index Number for Sydney (All Groups) for the quarter ended just before that date (CPI 1),
  - multiply the result by the Consumer Price Index Number for Sydney (All Groups) for the quarter ended just before the review date (CPI 2).

The product is the new rent for the year beginning on the review date (\$), written as a formula —

$$\frac{$X}{CPI 1} \times CPI 2 = $Y.$$

5.9

The landlord must calculate the new rent after each review date and give the tenant written notice of the new rent.
85 Ned:Cl45585 /Doc:DL 5382069 /Rev:ll-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:9 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.



Page 5 of 12 Annexure B

- 5.10 If the Australian Bureau of Statistics makes a change in the reference base of the index and there is a published co-relation between the old and new base then the published co-relation is to be applied to convert the CPI 1 figure to the new reference base. If there is none then the landlord and the tenant agree to accept the calculations of the landlord's solicitor who must be retained to determine a fair co-relation between the old and the new series of numbers.
- 5.11 If the index used to calculate the new rent is discontinued the landlord may substitute another index that, as nearly as practicable, serves the same purpose and, if there is no such index, then the rent will be fixed by Method 3.

# Method 3. By reference to current market rent.

- 5.12 In this case the rent is to be the current market rent. This can be higher or lower than the rent payable at the rent review date.
- 5.13 The landlord or the tenant can inform the other in writing at least 60 days before the rent review date of the rent that the landlord or tenant thinks will be the current market rent at the review date.
- 5.14 If the landlord and the tenant agree on a new rent then that rent will be the new rent beginning on the rent review date and the landlord and the tenant must sign a statement saying so.
- 5.15 If the landlord and the tenant do not agree on the amount of the new rent 30 days before the rent review date
  - 5.15.1 the current market rent will be decided by a valuer appointed under clause 5.16;
  - 5.15.2 the current market rent is the rent that, having regard to the terms and conditions of this lease and such other matters as are relevant to the assessment of current market rent, would be reasonably expected to be paid for the property if it were unoccupied and offered for renting for the use to which the property can be put in accordance with this lease;
  - 5.15.3 rent concessions and other benefits that are frequently or generally offered to prospective tenants of unoccupied retail shops, where the property is a retail shop, and otherwise of unoccupied comparable premises, are relevant matters; and
  - 5.15.4 the value of goodwill created by the tenant's occupation and the value of tenant's fixtures and fittings are to be ignored.
- 5.16 The landlord and the tenant can either agree upon a valuer or can ask the President of the Law Society of New South Wales to nominate a person who is a licensed valuer to decide the current market rent.
- 5.17 The valuer will act as an expert not an arbitrator. The landlord and the tenant can each make submissions in writing to the valuer within 14 days after they receive notice of the valuer's appointment but not later unless the valuer agrees.
- 5.18 The valuer's decision is final and binding. The valuer must state how the decision was reached.
- 5.19 If the valuer —

Ç,

- 5.19.1 does not accept the nomination to act; or
- 5.19.2 does not decide the current market rent within 60 days after accepting the
- nomination; or
- 5.19.3 becomes incapacitated or dies; or
- 5.19.4 resigns,

then another valuer is to be appointed in the same way.

5.20 The landlord and tenant must each pay half the valuer's costs.

85 Řed:Cl45585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:10 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

Page 6 of 1	12		•	,				·. ·
· · ·	в						· ·	
<b>F</b> 04		 			-	 		

- 5.21 If the landlord and the tenant do not agree upon a valuer and neither asks for a valuer to be nominated before
  - 5.21.1 the next rent review date passes; or
  - 5.21.2 this lease ends without the tenant renewing it; or
  - 5.21.3 this lease is transferred after the rent review date with the landlord's consent; or
  - 5.21.4 the property is transferred after the rent review date

then the rent will not change on that rent review date.

### CLAUSE 6 USE

### How must the property be used?

- 6.1 The tenant must
  - 6.1.1 use the property for the purpose stated in item 16 in the schedule and not for any other purpose;
  - 6.1.2 open for business at times usual for a business of the kind conducted by the tenant;
  - 6.1.3 keep the property clean and dispose of waste properly; and
  - 6.1.4 comply with all laws regulating how the property is used, obtain any consents or licences needed, comply with any conditions of consent, and keep current any licences or registrations needed for the use of the property or for the conduct of the tenant's business there.
- 6.2 The landlord can consent to a change of use and cannot withhold consent unreasonably.
- 6.3 The tenant must not
  - 6.3.1 do anything that might invalidate any insurance policy covering the property or that might increase the premium; or
  - 6.3.2 use the property as a residence or for any activity that is dangerous, offensive, noxious, illegal or immoral or that is or may become a nuisance or annoyance to the landlord or to the owner or occupier of any neighbouring property; or
  - 6.3.3 hold any auction, bankrupt or fire sale in the property; or
  - 6.3.4 display signs or advertisements on the outside of the property, or that can be seen from the outside, unless the landlord consents (but the landlord cannot withhold consent unreasonably); or
  - 6.3.5 overload the floors or walls of the property.

# CLAUSE 7 CONDITION AND REPAIRS

# Who is to repair the property?

- 7.1 The landlord must
  - 7.1.1 maintain in a state of good condition and serviceable repair the roof, the ceiling, the external walls, and the floors of the property and must fix structural defects;
  - 7.1.2 maintain the property in a structurally sound condition; and
  - 7.1.3 maintain essential services.
- 7.2 The tenant must otherwise maintain the property in its condition at the commencement date and promptly do repairs needed to keep it in that condition but the tenant does not have to —
  - 7.2.1 alter or improve the property; or
  - 7.2.2 fix structural defects; or
  - 7.2.3 repair fair wear and tear.

85 led: C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:11 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.



Page 7 of 12 Annexure B

The tenant must also ----

7.3

- 7.3.1 reimburse the landlord for the cost of fixing structural damage caused by the tenant, apart from fair wear and tear;
- 7.3.2 maintain and decorate the shop front if the property has one; and
- 7.3.3 decorate the inside of the property in the last 3 months of the lease period (however it ends) — 'decorate' here means restoring the surfaces of the property in a style and to a standard of finish originally used e.g. by repainting.
- If an authority requires work to be done on the property and it is structural work or work 7.4 needed to make the property safe to use then the landlord must do the work unless it is required only because of the way the tenant uses the property. But if it is any other work or is required only because of the way the tenant uses the property then the tenant must do the work.
- If the tenant fails to do any work that the tenant must do the landlord can give the tenant a 7.5 notice in writing stating what the tenant has failed to do. After the notice is given the tenant must ----
  - 7.5.1do the work immediately if there is an emergency; and
  - do the work promptly and diligently in any other case. 7.5.2

If the tenant does not do the work, the landlord can do it and the tenant must reimburse the landlord for the cost of the work.

The tenant must not make any structural alterations to the property. Any other alterations 7.6 require the landlord's consent in writing (but the landlord cannot withhold consent unreasonably).

# CLAUSE 8 INSURANCE AND DAMAGE

## What insurances must the tenant take out?

- 8.1 The tenant must keep current an insurance policy covering —
  - 8.1.1 liability to the public in an amount not less than the amount stated in item 17 in the schedule (for each accident or event); and
  - 8.1.2 damage or destruction from any cause to all plate glass in the windows and other portions of the property

and must produce to the landlord, upon request, the policy and the receipt for the last premium.

# What happens if the property is damaged?

- 8.2
- - 8.2,1 the tenant is not liable to pay rent, or any amount payable to the landlord in respect of outgoings or other charges, that is attributable to any period during which the property cannot be used under this lease or is inaccessible due to that damage;
  - if the property is still usable under this lease but its usability is diminished due to 8.2.2 the damage, the tenant's liability for rent and any amount in respect of outgoings attributable to any period during which useability is diminished is reduced in proportion to the reduction in usability caused by the damage;
  - 8.2.3 if the landlord notifies the tenant in writing that the landlord considers that the damage is such as to make its repair impracticable or undesirable, the landlord or the tenant can terminate this lease by giving not less than 7 days notice in writing of termination to the other and no compensation is payable in respect of that termination;

85.Reg:C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:12 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

Page 8 of 12 Annexure B		
8.2.4	if the landlord fails to repair the damage within a reasonable time after the tenant requests the landlord to do so the tenant can terminate this lease by giving not less	
8.2.5	than 7 days notice in writing of termination to the landlord; and nothing in clause 8.2 affects any right of the landlord to recover damages from the tenant in respect of any damage or destruction to which the clause applies.	

# CLAUSE 9 ACCESS

# What are the landlord's rights of access to the property?

9.1

The tenant must give the landlord (or anyone authorised in writing by the landlord) access to the property at any reasonable time for the purpose of -

- 9.1.1 inspecting the condition of the property, or how it is being used; or
- 9.1.2 doing anything that the landlord can or must do under this lease or must do by law; or
- 9.1.3 viewing the property as a valuer, prospective buyer or mortgagee; or
- 9.1.4 fixing a notice in a reasonable position on the outside of the property saying that it is for sale; or
- 9.1.5 viewing the property as a prospective tenant not earlier than 6 months before the lease period ends; or
- 9.1.6 fixing a notice not earlier than 6 months before the lease period ends in a reasonable position on the outside of the property saying that it is to let; or
- 9.1.7 inspecting, cleaning or repairing another property or any services to another property.
- 9.2 The landlord must give the tenant at least 2 days written notice for access (except in an emergency). The day of the giving of the notice and any Saturday, Sunday or public holiday on which the property is not open for business are not counted.
- 9.3 The landlord must promptly make good any damage caused to the property and to any of the tenant's belongings which results from exercising these rights.
- 9.4 The tenant must give to the landlord a copy of any notice relating to the property or relating to any neighbouring property immediately after receiving the notice.

# CLAUSE 10 TRANSFER AND SUB-LEASE

## Can this lease be transferred or the property shared or sub-let?

- 10.1 The tenant must not transfer this lease without consent.
- 10.2 The landlord can withhold consent only if
  - 10.2.1 the proposed transferce proposes to change the use to which the property is put; or
     10.2.2 where the property is a retail shop, the proposed transferee has financial resources or retailing skills inferior to those of the proposed transferor and otherwise the proposed transferee has financial resources or business experience inferior to those of the proposed transferor; or
  - 10.2.3 the tenant has not complied with clause 10.3 and, where the property is a retail shop, clause 10.4.
- 10.3 A request for the landlord's consent to a transfer of lease must be made in writing and the tenant must provide the landlord with such information as the landlord may reasonably require concerning the financial standing and business experience of the proposed transferee.

85 Req:C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:13 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.



Page 9 of 12 Annexure B

Where the property is a retail shop, before requesting the consent of the landlord to a proposed 10.4 transfer of this lease, the tenant must furnish the proposed transferee with a copy of any disclosure statement given to the tenant in respect of this lease, together with details of any changes that have occurred in respect of the information contained in the disclosure statement (being changes of which the tenant was aware or could reasonably be expected to be aware). For the purpose of enabling the tenant to comply with this obligation, the tenant can request the landlord to provide the tenant with a copy of the disclosure statement concerned and, if the landlord is unable or unwilling to comply with such a request within 14 days after it is made, this clause 10.4 does not apply.

Where the tenant has complied with clause 10.3 and where required to do so clause 10.4 and 10.5 the landlord has not within 42 days after the request was made given notice in writing to the tenant either consenting or withholding consent the landlord is taken to have consented.

- The tenant has to pay in connection with any consent the landlord's reasonable legal costs, the 10.6 reasonable costs of obtaining any mortgagee's consent, the stamp duty and the registration fee for the transfer.
- 10.7 Where the property is a retail shop, the tenant can sublet, grant a license or concession, share or part with the possession of the whole or any part of the property or mortgage or otherwise charge or encumber the tenant's estate or interest in this lease only with the written consent of the landlord which can be refused in the landlord's absolute discretion. Otherwise, the tenant . cannot do any of these things.

# CLAUSE 11 LANDLORD'S OTHER OBLIGATIONS

# What are the landlord's other obligations?

- 11.1 So long as the tenant does all the things that must be done by the tenant under this lease the landlord must allow the tenant to possess and use the property in any way permitted under this lease without interference from the landlord, or any person claiming under the landlord or having superior title to the title of the landlord.
- 11.2 The landlord must pay all outgoings for the land or the building of which the property is part when they fall due.
- If the property is part of a building owned or controlled by the landlord ----11.3
  - the landlord must maintain in reasonable structural condition all parts of the 11.3.1 building that the tenant can use under this lease; and
  - if the property has facilities and service connections shared in common with other 11.3.2 persons the landlord must -11.3.2.1
    - - the right for the tenant and other persons to come and go to and
      - from the property over the areas provided for access;
      - access by the tenant to service connections; and
      - the right for the tenant's customers to park vehicles in any area set aside for customer parking, subject to any reasonable rules made by the landlord.
    - maintain the facilities and service connections in reasonable condition. 11.3.2.2
- 11.4 The landlord must ensure that this lease is registered.
- 11.5 If a consent is needed for this lease, from someone such as a mortgagee or head landlord of the property, then the landlord must get the consent.

85 Req:C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:14 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

 $\mathbf{v}_{i}^{*}$ 

• ;

Page 10 of 12 Annexure B

# CLAUSE 12 FORFEITURE AND END OF LEASE

# When does this lease end?

- 12.1 This lease ends
  - 12.1.1 on the date stated in item 3 in the schedule; or
  - 12.1.2 if the landlord lawfully enters and takes possession of any part of the property; or
  - 12.1.3 if the landlord lawfully demands possession of the property.
- 12.2 The landlord can enter and take possession of the property or demand possession of the property if
  - 12.2.1 the tenant has repudiated this lease; or
  - 12.2.2 rent or any other money due under this lease is 14 days overdue for payment; or
  - 12.2.3 the tenant has failed to comply with a landlord's notice under section 129 of Conveyancing Act 1919; or
  - 12.2.4 the tenant has not complied with any term of this lease where a landlord's notice is not required under section 129 of the Conveyancing Act 1919 and the landlord has given at least 14 days written notice of the landlord's intention to end this lease.
- 12.3 When this lease ends, unless the tenant becomes a tenant of the property under a new lease the tenant must
  - 12.3.1 return the property to the landlord in the state and condition that this lease requires the tenant to keep it in; and
  - 12.3.2 have removed any goods and anything that the tenant fixed to the property and have made good any damage caused by the removal.

Anything not removed becomes the property of the landlord who can keep it or remove and dispose of it and charge to the tenant the cost of removal making good and disposal.

- 12.4 If the landlord allows the tenant to continue to occupy the property after the end of the lease period (other than under a new lease) then
  - 12.4.1 the tenant becomes a monthly tenant and must go on paying the same rent and other money in the same way that the tenant had to do under this lease just before the lease period ended (apportioned and payable monthly);
  - 12.4.2 the monthly tenancy will be on the same terms as this lease, except for ■ clause 4;
    - clauses 5.4 to 5.21 inclusive;
    - clause 6.2 unless consent has previously been given; and
    - clause 8 except for the provisions relating to rental suspension;
  - 12.4.3 either the landlord or the tenant can end the monthly tenancy by giving, at any time, one month's written notice to the other expiring on any date; and
  - 12.4.4 anything that the tenant must do by the end of this lease must be done by the end of the monthly tenancy.
- 12.5 Essential terms of this lease include
  - 12.5.1 the obligation to pay rent not later than 14 days after the due date for payment of each periodic instalment (and this obligation stays essential even if the landlord, from time to time, accepted late payment);
  - 12.5.2 the obligations of the tenant in clause 5.1.2 (dealing with outgoings);
  - 12.5.3 the obligations of the tenant in clause 6.1 (dealing with use);
  - 12.5.4 the obligations of the tenant in clause 7 (dealing with repairs); and
  - 12.5.5 the obligations of the tenant in clause 10 (dealing with transfer and sub-lease).
- 12.6 If there is a breach of an essential term the landlord can recover damages for losses over the entire period of this lease but must do every reasonable thing to mitigate those losses and try to lease the property to another tenant on reasonable terms.

85 Req:C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:15 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

¥. و • B

Page 11 of 12 Annexure B

12.7

The landlord can recover damages even if —

- 12.7.1 the landlord accepts the tenant's repudiation of this lease; or12.7.2 the landlord ends this lease by entering and taking possession of any part of the
  - property or by demanding possession of the property; or
- 12.7.3 the tenant abandons possession of the property; or
- 12.7.4 a surrender of this lease occurs.

# CLAUSE 13 GUARANTEE

## What are the obligations of a guarantor?

- 13.1 This clause applies if a guarantor of the tenant is named in item 9A in the schedule and has signed or executed this lease or, if this lease is a renewal of an earlier lease, the earlier lease.
- 13.2 The guarantor guarantees to the landlord the performance by the tenant of all the tenant's obligations (including any obligation to pay rent, outgoings or damages) under this lease, under every extension of it or under any renewal of it or under any tenancy and including obligations that are later changed or created.
- 13.3 If the tenant does not pay any money due under this lease, under any extension of it or under any renewal of it or under any tenancy the guarantor must pay that money to the landlord on demand even if the landlord has not tried to recover payment from the tenant.
- 13.4 If the tenant does not perform any of the tenant's obligations under this lease, under any extension of it or under any renewal of it or under any tenancy the guarantor must compensate the landlord even if the landlord has not tried to recover compensation from the tenant.
- 13.5 If the tenant is insolvent and this lease or any extension or renewal of it is disclaimed the guarantor is liable to the landlord for any damage suffered by the landlord because of the disclaimer. The landlord can recover damages for losses over the entire period of this lease or any extension or renewal but must do every reasonable thing to mitigate those losses and try to lease the property to another tenant on reasonable terms.
- 13.6 Even if the landlord gives the tenant extra time to comply with an obligation under this lease, under any extension of it or under any renewal of it or under any tenancy, or does not insist on strict compliance with the terms of this lease or any extension of it or renewal of it or of any tenancy, the guarantor's obligations are not affected.
- 13.7 If an amount is stated in item 9B in the schedule the guarantor's liability under this clause is limited to that amount.
- 13.8 The terms of this guarantee apply even if this lease is not registered, even if any obligation of the tenant is only an equitable one, and even if this lease is extended by legislation.

# CLAUSE 14 EXCLUSIONS, NOTICES AND SPECIAL CLAUSES

- 14.1 No covenant or power is implied in this lease by section 84 or 85 of the Conveyancing Act 1919.
- 14.2 A document under or relating to this lease is
  - 14.2.1 served if it is served in any manner provided in section 170 of the Conveyancing Act 1919; and
  - 14.2.2 served on the tenant if it is left at the property.
- 14.3 This lease is subject to any legislation that cannot be excluded.
- 14.4 In this lease, 'retail shop' means premises to which the Retail Leases Act 1994 applies.

85 Keq:C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:16 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

Page 12 of 12 Annexure B



# **IMPORTANT NOTES**

The following notes are for guidance and do not form part of this lease.

If you are a landlord, a solicitor will prepare this lease for you.

If you are a tenant, a solicitor can advise you about it.

- 1. This document creates legal rights and legal obligations.
- 2. Failure to register a lease can have serious consequences.
- 3. If an option for renewal is not exercised at the right time it will be lost.
- 4. The tenant can exercise an option for renewal even if there has been a breach of this lease in a case where section 133E of the Conveyancing Act 1919 applies. The landlord must give a prescribed notice within 14 days after the option is exercised if the landlord wants to rely on the breach to prevent the exercise of the option.
- 5. The Law Society of New South Wales is not to be responsible for any loss resulting from the use of this lease as printed whether authorised or not.

······

I certify that this and the preceding eleven pages are in exactly the same wording as Annexure B of the copyright Law Society Lease.

Solicitor for the Landlord

85. Req:C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:17 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

••	ð	i,	,#		••		
		•		•			



Page 1 of  ${\mathcal V}$  pages Disclosure Statement

# DISCLOSURE STATEMENT

(Schedule 2, Retail Leases Act)

# **ADVICE TO LESSEES**

1. Before signing agreements to lease or leases, lessees should ensure they fully understand the documents.

2. If there is any doubt, lessees should seek independent legal advice.

# TENANCY DETAILS

Address of shop/shop No. 4 &	3	Lettable area (sq. m.)				
Permitted use of shop PHARMACY, TOY SHOP, STORAGE AND OFFICE						
Lease period	Options		Options periods			
1 /7 /1998to30/6/2003	No 5 Yes		/ /			
Finishes, fixtures, fittings, equipme	ent and services to	be provided by th	e lessor.			
Lessee has to pay for the finishes, fixtures, fittings, equipment and services to be provided by the lessor: yes/no						
If yes, to what extent:						
Finishes, fixtures, fittings, equipment and services to be provided by the lessee						
Hours of access to shop outside trading hours. At all hours						
Date on which shop will be available for occupation by the lessee.						
Lessor's requirements as to quality and standard of fittings in shop.						
Rent: \$80,402.40 per annum						
Method for Calculating Rent:						
1. Commencement Date						
2. Formula 1						
3. Rent Reviews						
Frequency EARLY Nature						

# DETAILS AS TO INTEREST OF LESSOR

Is the Lessor:

I Owner of the shop

or Lessee of the shop

Give details of any rights and obligations of lessor under that lease that may affect the shop.

# DETAILS AS TO AGREEMENTS OR REPRESENTATIONS

Give details of any other agreements between lessor and lessee, or representations made by lessor or lessee including those relating to exclusivity or limitations on competing uses. e . . .

\* \* \* \* \*

.

# Page 2 of $\mathcal{V}$ pages Disclosure Statement

# **OUTGOINGS TO BE PAID BY THE LESSEE**

DETAILS OF OUTGOINGS	ESTIMATE \$ P.A.	DETAILS OF OUTGOINGS	ESTIMATE \$ P.A.
Air conditioning/ Ventilation	N/A	Audit fees	N/A
Building intelligence and Emergency systems	N/A	Car parking	N/A
Child minding	N/A	Cleaning	N/A
Electricity	N/A	Energy management systems	N/A
Fire protection	N/A	Gardening	N/A
Gas & oil	N/A	Insurance	
Land tax		Lifts and Escalators	
Local Government rates and charges		Management costs	
Pest control	N/A	Public address / Music	N/A
Repairs and Maintenance		Sinking fund for repairs and maintenance	
Security		Sewage disposal and Sullage	
Signs		Telephones (public)	N/A
Uniforms	N/A	Water, Sewerage and Drainage rates and charges	
Others (specify)			
TOTAL (both columns)			
Formula for apportionment of outgoings if the lessee is not liable for the total amount			
Additional outgoings to be borne by lessee			

NOTE: The Law Society Lease was prepared to suit a retail shop, NOT located in a retail shopping centre. Accordingly, the information contained in the final two pages of Schedule 2 under the heading "Retail shopping Centre Details" has not been reproduced. Users of this form should reinstate that information if applicable.

.

85 Redq:C145585 /Doc:DL 5382069 /Rev:11-Nov-1998 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:19 of 19 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

# CONSENT TO LEASE R.P.A.

ż

WESTPAC BANKING CORPORATION ARBN 007 457 141 (hereinafter called "the Bank") being the Mortgagee under Mortgage No 3944927 of the premises demised by the within Lease hereby consents to such Lease as from the registration thereof and not otherwise and subject to the following conditions and provisions namely:

- 1. THAT this consent shall be without prejudice to the rights powers and remedies of the Bank and its assigns under the said Mortgage which shall remain in full force and effect as if this consent had not been given except that so long as the covenants and provisions of the said Lease are duly observed and performed the Bank will in the event of the exercise of the power of sale or other power or remedy of the Bank or its assigns on default under the said Mortgage exercise the same subject to the then subsisting rights of the Lessee his executors administrators and assigns under the said Lease.
- THAT so long as the Bank or its assigns is or are registered as Mortgagee of the said premises the Lessee shall obtain the consent or approval of the Bank or its assigns in addition to the consent or approval of the Lessor in all cases where under the said Lease the consent or approval of the Lessor is required.
- 3. THAT upon the Bank or its assigns giving notice to the Lessee of demanding to enter into receipt of the rents and profits of the said premises the covenants on the part of the Lessee expressed or implied in the said Lease shall be deemed to have been entered into by the Lessee with the Bank and its assigns and all the rights powers and remedies of the Lessor under the said Lease shall vest in and be exercisable by the Bank and its assigns until such notice be withdrawn or the said Mortgage be discharged.
- 4. The Bank shall in no way be bound to perform and shall not incur any liability in respect of the covenants and agreements expressed or implied in the said Lease and on the part of the Lessor to be performed and observed.
- 5. The word "Lessee" where used herein shall mean and include the Lessee his executors administrators or permitted assigns.

DATED this 30 October 1998.

4

Signed for and on behalf of WESTPAC BANKING CORPORATION

rederick Mar Holmes

(Full name of Attorney) its Attorney who is personally known to me

..... 

(Witness Signature)



WESTPAC BANKING CORPORATION by its Attorney who states that at the time of executing this instrument no notice of revocation has been received of the Power of Attorney registered in the Land Titles Office Book 4059 No.831 under the authority of which this Instrument has been executed.

.....

Supervisor, NSW Service Centre

55538708-2-2-LD734-001

LD734.009 29/05/1998



Reų;C145586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:2 of 54 86 DATE OF EXECUTION 21 7/93 (H) We certify this dealing correct for the purposes of the Real Property Act, 1900 Signed in my presence by the lessor who is personally known to me  $L^{O}$ 121111 THE COMMON SEAL OF ISIKIT PTY 17 7 Signature of Witness LIMITED was affixed in accordance with ...... Name of Winess (BLOCK LETTERS) Director Director/Secretary its articles of association in the presence of: ...... ..... Signature of Lessor Address of Witness MACE Signed in my presence by the lessee who is personally known to me THE COMMON SEAL OF K.CMACRAE HOLDINGS Signature of Witness PTY LTD was affixed in accordance with Machas Name of Witness (BLOCK LETTERS) Director its articles of association in the presence of: Director/Secretary Signature of Lessee Address of Wimess I solemnly and sincerely declare that the time for the exercise of the Option to Renew/Purchase in expired lease No. (11) has ended and the lessee under that lease has not exercised the option. I make this solemn declaration conscientiously believing the same to be true and by virtue of the Oaths Act, 1900. in the presence of \*\*\*\*\* Signature of Witness Name of Witness (BLOCK LETTERS) . . . . . . . . . . . Signature of Lessor Address of Wilness

86 Rey:C145586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:3 of 54

## ANNEXTIRE "A"

DEED made		199
BETWEEN	The Party identified in Item 1 of the Reference Schedule	("Landlord")
AND	The Party identified in Item 2 of the Reference Schedule	("Tenant")
AND	The Party (if any) identified in Item of the Reference Schedule	u 15 ("Guarantor")

WITNESSES AND IT IS AGREED by the parties as follows:

### **1. Definitions and Interpretation**

#### 1.1 Definitions

In this Lease unless the context otherwise requires:

- "Air Conditioning Equipment" means the plant, chilled water piping, electrical (1) installations, ductwork and diffusers used to heat, cool, circulate and extract air throughout the Building;
- (2) "Australian Institute" means the Australian Institute of Valuers and Land Economists (Inc.) New South Wales Division or its successor or other organisation replacing it;
- (3) "Building" means all improvements of which the Premises form part erected on the Land including any modifications, extensions or alterations to those improvements and, where appropriate, includes the Land and the Landlord's Fixtures:
- (4) "Claim" includes any claim or legal action;
- (5) "Commencement Date" means the date of commencement of this Lease stated in Item 4 of the Reference Schedule;
- "Common Areas" means all parts of the Building and the Land designated by (6) the Landlord for common use by the Building's tenants or occupiers and their respective employees, invitees and licensees;
- (7) "Cost" means any cost, charge, expense, outgoing, payment or other expenditure of any nature and where appropriate, includes the reasonable fees and disbursements payable to contractors, consultants and lawyers;
- (8) "Default Rate" means, unless another rate of interest is stated in Item 13 of the Reference Schedule, 2% per annum above the highest overdraft rate charged by the Landlord's bank for commercial loans in excess of \$100,000.00 at the date for payment of any money and, in the absence of manifest error, a written statement by the Landlord's bank manager will be conclusive evidence of the rate;

AEW11067270001D

Vilada V P. Machae

050793aew

- 2 -

- (9) "Guarantor" means the person (if any) named in Item 15 of the Reference Schedule:
- (10) "Land" means the whole of the Land disclosed in the Torreus Title reference to the Lease and extends to and includes any other land which is acquired and used by the Landlord in conjunction with the Building;
- (11) "Landlord" means the person named in Item 1 of the Reference Schedule and includes:
  - (a) in the case of a company, its successors and assigns; and
  - (b) in the case of a person, his executors, administrators and assigns;
- (12) "Landlord's Fixtures" includes:
  - (a) the Landlord's plant, equipment, fittings, fixtures, furniture, furnishings, window coverings, blinds, floor coverings and light fittings in any part of the Premises;
  - (b) stopcocks, fire hoses, hydrants, fire prevention aids and other fire fighting equipment located in the Premises or servicing the Premises from Common Areas; and
  - (c) drains, basins, sinks, toilets and urinals in the Building;
- (13) "Law" includes the provisions of any statute, rule, regulation, proclamation, ordinance or by-law, present or future, whether state, federal or otherwise;
- (14) "Lease" means this Annexure "A" and the Form RPA to which this Annexure "A" is annexed;
- (15) "Managing Agent" means any agent appointed by the Landlord to manage the Building:
- (16) "Premises" means the part of the Building stated in Item 3 of the Reference Schedule and includes the Landlord's Fixtures;
- (17) "Reference Schedule" means the Reference Schedule to this Lease;
- (18) "Rent" means the amount stated in Item 7 of the Reference Schedule as varied from time to time under the provisions of this lease;
- (19) "Requirement" means any requirement, notice, order or direction received from or given by any statutory, public or other competent authority, present or future:
- (20) "Rules" means the rules and regulations (if any) made by the Landlord in respect of the Building;
- (21) "Services" means all gas, electricity, telephone, water, sewerage, fire prevention, ventilation, air conditioning, hydraulic, elevator and security services and all other services or systems provided in the Building or the Common Areas or to the Land or available for the use of the Building's tenants or occupiers;

N PMachae

Juladas

- (22) "Tenant" means the person named in Item 2 of the Reference Schedule and includes:
  - (a) in the case of a company, its successors and permitted assigns; and
  - (b) in the case of a person, his executors, administrators and permitted assigns;
- (23) "Tenant's Employees" means each of the Tenant's employees, contractors, agents, customers, clients, visitors, subtenants, licensees or others (with or without invitation) who may be upon the Premises, the Building or the Land;
- (24) "Tenant's Fittings" includes all fixtures, fittings, furniture, plant, equipment, partitions or other articles and chattels which are in the Premises and are not owned by the Landlord;
- (25) "Term" means the term of this Lease stated in Item 6 of the Reference Schedule;
- (26) "Termination Date" means the date of termination of this Lease stated in Item 5 of the Reference Schedule; and
- (27) "Valuer" means a person who is:
  - (a) a fully licensed member, of not less than 5 years' standing, of the Australian Institute and active in the relevant market at the time of his appointment; and
  - (b) has at least 3 years' experience in valuing the kind of premises leased by this Lease.

# 1.2 Construction

Reference to:

- (1) a person includes:
  - (a) a corporation and government body; and
  - (b) the legal representatives, successors and assigns of that person;
- (2) month or monthly means calendar month or calendar monthly; and
- (3) a right includes a remedy, authority or power.

# 1.3 Plurals and Genders

A word importing:

- (1) the singular includes the plural and vice versa; and
- (2) any gender includes all other genders.

Kullanden P. Machae

- 4 -

## 1.4 Parties Bound Jointly and Individually

If two or more persons are named as Tenant or Guarantor, any covenant or agreement made by or on behalf of the Tenant or the Guarantor binds them jointly and each of them individually.

### 1.5 Statutes and Regulations

References to statutes, regulations, ordinances or by-laws include all statutes, regulations, ordinances or by-laws amending, consolidating or replacing them.

## 1.6 Covenants

Every obligation undertaken by a party to this Lease will be deemed to be and be construed as a covenant by that person.

### 1.7 Severability

- (1) As far as possible all provisions of this Lease will be construed so as not to be invalid, illegal or unenforceable in any respect.
- (2) If any provision on its true interpretation is illegal, invalid or unenforceable, that provision will, as far as possible, be read down to the extent necessary to ensure that it is not illegal, invalid or unenforceable and so as to give it a valid operation of a partial character.
- (3) If any provision or part of it cannot be read down, that provision or part will be deemed to be void and severable and the remaining provisions of this Lease will not be affected or impaired.

### 1.8 Whole Agreement

- (1) The terms contained in this Lease comprise the whole agreement between the Landlord and the Tenant.
- (2) No further terms will be implied or arise between the Landlord and the Tenant under any collateral or other agreement made on or prior to execution of this Lease.

### 1.9 Headings

- (1) Headings and the table of contents to this Lease have been inserted for guidance only and will not affect the construction of this Lease.
- (2) References to clauses, schedules and annexures will be construed as references to clauses of and schedules and annexures to this Lease.

### 1.10 Organisations

- (1) If any organisation has ceased to exist, reference to that organisation will be deemed to be reference to an organisation having substantially the same objects as the original organisation.
- (2) Reference to the president of an organisation will, in the absence of a president, be read as reference to the senior officer for the time being of the organisation or any other person fulfilling the duties of president.

Kulandas J V P. Machae

### 1.11 Areas and Measurement

- (1) In this clause "method of measurement" means:
  - (a) the relevant method for the measurement of buildings utilised by the Building Owners and Managers Association of Australia Limited; or
  - (b) if there is no relevant method, the method or criteria which the Landlord selects as the most appropriate.
- (2) Unless the context otherwise requires, where any area of the Premises or the Building is to be calculated or measured for the purposes of this Lease, those calculations and measurements will be in accordance with the method of measurement.

### 1.12 Landlord's Consent

Unless otherwise stated, if the Landlord's consent or approval is required it:

- (1) may be granted or withheld at the Landlord's discretion; and
- (2) will not be effective unless in writing.

# 1.13 Written Notices

If a provision of this Lease requires a notice to be given by a party, it must be in writing unless the provision states that it may be given orally.

# 2. Exclusion of Statutory Provisions

2.1 Statutory Provisions Excluded

To the extent permitted by Law:

- (1) the covenants, powers and provisions (if any) implied in leases by virtue of any Law are expressly excluded; and
- (2) the application to this Lease of any Law having the effect of extending the Term, reducing or postponing the payment of Rent or otherwise affecting the operation of any of the covenants, terms and conditions of this Lease is excluded and negatived.

### 3. Term and Holding Over

### 3.1 Term

The Landlord leases to the Tenant and the Tenant takes a lease of the Premises for the Term commencing on the Commencement Date and terminating on the Termination Date subject to the provisions of this Lease.

Kertaker

, V R.Machae

- 6 -

# 3.2 Holding Over

If the Tenant continues to occupy the Premises beyond the Termination Date with the Landlord's consent except under a lease arising from the valid exercise of an option to renew then the following provisions will apply:

## (1) Monthly Tenancy

The Tenant will occupy the Premises as a monthly tenant at a total rental payable monthly in advance being an amount equal to one-twelfth of the aggregate of the Rent and any other money payable by the Tenant to the Landlord under this Lease as at the Termination Date, the first of the monthly payments to be made on the day following the Termination Date.

### (2) **Provisions of Lease to Apply**

The tenancy will otherwise continue on the terms and conditions of this Lease as far as applicable to a monthly tenancy.

# (3) Rent Review

The Rent or any part of it for the monthly tenancy may be reviewed whenever the Landlord determines it appropriate and any provisions in this Lease relating to calculation of, limitation on or right of review of the Rent or any part of it will not apply.

## (4) Termination

- (a) The monthly tenancy will be terminable at any time by either the Landlord or the Tenant giving to the other 1 month's notice expiring on any date.
- (b) If the Tenant defaults in the performance of its obligations under the monthly tenancy, the Landlord may terminate the monthly tenancy by giving the Tenant 24 hours' notice.

### 4. Rent and Rent Reviews

### 4.1 Rent

- (1) The Tenant must pay the Rent to the Landlord without demand, deduction or right of set-off by equal monthly instalments in advance on the first day of each month.
- (2) If necessary, the first and last instalments will be apportioned on a daily basis.
- (3) The first instalment will be paid on the Commencement Date.
- (4) All instalments must be paid as the Landlord directs.

### 4.2 Definitions

In this clause:

Konfarka A P. Machae

- (1) "index number" means the Consumer Price Index (All Groups) for Sydney published from time to time by the Australian Statistician; if the Australian Statistician updates the reference base of the index number, the index number will be appropriately adjusted so as to preserve the intended continuity of calculation by using the appropriate arithmetical factor determined by the Australian Statistician;
- (2) "index review date" means each of the dates (if any) stated in Item 8(a) of the Reference Schedule:
- (3) "market review date" means each of the dates (if any) stated in Item 8(b) of the Reference Schedule;
- (4) "quarter" means a 3 month period; and
- (5) "review date" means an index review date or a market review date as the case requires.

# 4.3 Annual Rent Review; Consumer Price Index

If Item 8(a) of the Reference Schedule has been completed by inserting index review dates, the following provisions will apply to the review of the Rent.

(1) Review

The Rent will be reviewed on each index review date to an amount being the greater of the follows:

- (a) the Rent at the index review date increased by 5%; or
- (b) an amount represented by A in the following formula:

$$A = \underline{B} \times D$$

$$C$$

- Where B = the index number released for the quarter ending immediately prior to the relevant index review date;
  - C = the index number released for the quarter ending immediately prior to the later of the Commencement Date or the date when the Rent was last reviewed; and
  - D = the Rent payable immediately prior to the index review date.
- (2) Suspension of Consumer Price Index Etc.
  - (a) If the Consumer Price Index (All Groups) for Sydney is suspended or discontinued, the words "index number" will mean the price index substituted by the Australian Statistician.
  - (b) If no price index is substituted, the words "index number" will mean an index which the parties agree most closely reflects changes in the cost of living.
  - (c) If the parties cannot agree on a substitute index, the president of the Australian Institute, at the request of either party, may appoint an

& R. Machae Kerlanda

- 8 -

expert to determine a substitute index which most closely reflects changes in the cost of living and the words "index number" will mean that index.

# (3) Alternative Method of Review

If the Consumer Price Index (All Groups) for Sydney is suspended or discontinued and the expert appointed under clause 4.3(2) is unable to determine a substitute index within 40 days of being appointed, the method of adjustment of Rent in relation to an index number will cease and the Rent will be increased by 8% per annum on each index review date.

### 4.4 Market Rent Review

If Item 8(b) of the Reference Schedule has been completed by inserting market review dates, the Rent will be reviewed on each market review date to an amount determined in accordance with the following procedure.

(1) Landlord's Notice

The Landlord may vary the Rent to an amount which it considers to be the market rent for the Premises by notice to the Tenant ("Landlord's notice") at any time within 3 months before and 6 months after a market review date.

- (2) Tenant's Notice
  - (a) Within 21 days after service of the Landlord's notice, the Tenant may serve on the Landlord a notice ("Tenant's notice") disputing the amount stated in the Landlord's notice, stating the amount which the Tenant considers to be the market rent for the Premises, and requiring the market rent to be determined under clause 4.4.
  - (b) If the Tenant fails to give the Tenant's notice within 21 days then the Rent payable until the next market review date will be the amount set out in the Landlord's notice.

### (3) Negotiations

Within 14 days of service of the Tenant's notice or such further time as they may agree upon in writing, the Landlord and the Tenant or their representatives will meet and attempt to resolve their dispute as to the market rent payable for the Premises at the market review date.

# (4) Valuer to Settle if Dispute

- (a) If the Landlord and the Tenant have not agreed on the market rent for the Premises within 21 days after service of the Tenant's notice, the dispute must be referred for determination by a Valuer to be nominated by the president of the Australian Institute at the request of either the Landlord or the Tenant.
- (b) The appointed Valuer will make a determination of the market rent as at the particular market review date acting as an expert and not as an arbitrator.

I N Prelachae Kellandas

- (c) The Valuer must make a written determination containing reasons as soon as reasonably possible after the date of his appointment.
- (d) The Valuer's determination will be final and binding on the Landlord and the Tenant.
- (e) The Valuer's Cost will be paid by the Landlord and the Tenant equally. Either party may pay the Valuer's Costs and recover one half of the amount paid from the other party.

# (5) Valuer to Give Notice of Acceptance of Appointment

A Valuer is not to act as an expert under this clause until he has given written notice of acceptance of his appointment to the Landlord and the Tenant.

- (6) Landlord's and Tenant's Submissions
  - (a) A Valuer who accepts appointment under this clause may confer with the Landlord or the Tenant and may require either party to supply information which the Valuer considers relevant to the determination.
  - (b) Any request for information must be complied with promptly in writing by the party to whom it is directed, who will make a copy of that information available to the other party.
  - (c) Either party may supply the Valuer with other information which it considers relevant and, if it does so, must make a copy of that information available to the other party.
  - (d) Information may be provided on a confidential basis and, if so, the party receiving it and the Valuer will treat the information as confidential and will not use that information other than for the purposes of this clause.
- (7) Written Submissions
  - (a) The Landlord and the Tenant may make written submissions to the Valuer in relation to the market rent within 30 days of receipt of written notice of the Valuer's acceptance of his appointment. A party making a written submission must at the same time make a copy of it available to the other party.
  - (b) The Valuer must not determine the market rent until the expiration of that period of 30 days.

# (8) Valuer's Criteria

- (a) In determining the market rent as at a market review date the Valuer may take into account any matters he considers relevant including taking into account or disregarding any written submissions received from the Landlord or the Tenant but in making his determination under this Lease the Valuer must:
  - (i) disregard:

P. Mackae Kinfontos

86 Req:C145586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:12 of 54

- (A) any goodwill attributable to the Premises by reason of the trade, business or activity carried on by the Tenant and the value of the Tenant's Fittings;
- (B) any state of disrepair of the Premises if that condition results from any work carried out or not carried out on the Premises by the Tenant or from the Tenant's breach of any provision of this Lease;
- (C) any money received under any sublease, subtenancy agreement or occupational arrangement in respect of the Premises which has not been approved by the Landlord;
- (D) any inducement provided or to be provided to the Tenant in connection with the granting of this Lease;
- (E) any inducement then being provided or to be provided to any other tenant in relation to the taking of a lease of any other premises; and
- (F) anything (including part of a submission received from the Landlord or the Tenant) which is not consistent with the matters to be disregarded or taken into account under clause 4.4(8); and
- (ii) take into account the following matters or, where the context requires, make the following assumptions:
  - (A) the provisions of this Lease;
  - (B) the Term and any option for renewal (disregarding the elapsed part of the Term);
  - (C) the rent and operating expenses paid or payable in respect of other premises of a quality, nature, size and location similar to the Premises;
  - (D) the use permitted by the relevant Laws and the provisions of this Lease;
  - (E) assume that the Tenant has observed and performed all of the provisions of this Lease; and
  - (F) assume that it is the Landlord's and the Tenant's express requirement that no reduction or adjustment will be made to the market rent on account of any inducement provided or to be provided to the Tenant to secure it as a tenant of the Premises or to any other tenant in relation to the taking of a lease of any other premises whether or not those premises are comparable with the Premises.
- (b) The term "inducement" means any inducement or incentive provided by the Landlord in respect of the Tenant's entry into this Lease and includes any payment, transfer of property, fit-out of premises or

pulladas

W RMackae

- 11 -

provision of services, assumption of obligation, rent moratorium or reduction, loan or gift.

#### Payment of Rent Pending Review 4.5

Until the determination of Rent has been made:

- the Tenant will continue to pay Rent at the same rate as immediately before (1) the review date;
- any variation in Rent as the result of any review under clauses 4.3 and 4.4 (2) will take effect on the review date and will apply until the next review date or the Termination Date whichever first occurs; and
- within 14 days of the determination, the Landlord will refund any overpaid (3) Rent or the Tenant will pay any shortfall.

#### Minimum Rent 4.6

Regardless of any other provision of this Lease:

- where the review date is the same under clause 4.3 and clause 4.4 of this (1) Lease then the Rent payable from each such review date will be the higher of the Rent determined under clause 4.3 or clause 4.4; and
- the Rent payable from any review date will never be less than the Rent (2) payable prior to the review date.

### 5. Operating Expenses

#### **Operating Expenses Year** 5.1

The term "operating expenses year" means each period of 12 months ending on the date stated in Item 9 of the Reference Schedule in each year regardless of any part of that 12 month period falling outside the Term.

#### **Definition of Operating Expenses** 5.2

The term "operating expenses" means the total of all amounts payable by the Landlord (or for which the Landlord may be or become liable) in any operating expenses year in connection with the Building or the Land whether or not by direct assessment including:

#### rates and charges (1)

rates, taxes, charges, assessments, duties, impositions and fees (excluding income and capital gains taxes payable by the Landlord) payable to any government, local government, semi-government or other competent authority in connection with the Land or the Building; state land tax will be included only if it is lawful to do so and then on the basis that the Land is the only land owned by the Landlord;

Kinfordos

V P.Machae

## (2) water, waste etc.

Costs for sewerage, water usage, supply of water and removal of waste and other garbage from the Building, the Premises and the Land;

### (3) insurances

amounts payable for insurances effected by the Landlord relating to the Building, its use and occupancy including, without limitation, insurance against fire and other usual risks on a replacement basis, public risk and plate glass insurance, workers' compensation insurance and loss of rents insurance;

### (4) repairs and maintenance

Costs of repairs or maintenance of the Building (excluding maintenance of a structural nature and major repairs and renovations) including Cost of operating, supplying, maintaining and repairing the Services and the plant and equipment required for those Services;

### (5) light, fuel and power

Costs incurred in providing lighting, fuel and power to the Building, the Common Areas and the Services not payable by specific tenants;

(6) pest control

Costs for the control of pest, vermin, insect or other similar infestation;

### (7) loading docks

Costs of running the loading docks, storage areas, public toilets and any similar facilities provided by the Landlord;

## (8) plants

Costs of purchasing, hiring, maintaining and servicing any indoor plants, gardens, lawns, potted shrubs, planted areas, fountains, artificial watercourses and associated facilities in or about the Common Areas or adjacent areas;

## (9) management

a management fee to cover the Landlord's Cost of having the Building managed by the Managing Agent, as agreed between the Landlord and the Managing Agent, and any other Cost relating to the management and promotion of the Buildingpointed that much fee shall not enced these for metallican granted terminated by the Refer Ref from time to time materials

Costs of materials used by the Landlord in maintaining and servicing the Landlord's Fixtures and the Services;

## (11) cleaning

Costs of cleaning the Premises and all other parts of the Building including general cleaning, window cleaning, annual carpet shampooing and the provision of toilet supplies but excluding any cost which are payable

p P.Mackae Killado

(10)

- 13 -

independently by any tenant or occupier in relation to any particular tenancy or area occupied;

# (12) security and caretaking

Costs of providing and maintaining security and caretaking services;

### (13) air conditioning

Costs of repairing, maintaining, servicing and running the Air Conditioning Equipment excluding cost payable separately by any tenant or occupier in relation to particular areas occupied; and

### (14) general expenditure

Costs properly incurred by the Landlord in the management, operation, control and maintenance of the Building, the Common Areas or the Land excluding expenditure related to items of a capital nature and commission or fees payable to any letting agent for the leasing of any part of the Building and any cost payable by Tenants separately under clauses 17.2 and 17.3 of this Lease.

# 5.3 Tenant's Proportion of Operating Expenses

- (1) The term "Tenant's proportion of the operating expenses" means that proportion which the net rentable area of the Premises bears to the net rentable area of the Building at the date of furnishing the statement under clause 5.6.
- (2) At the Commencement Date, the Tenant's proportion of the operating expenses is the percentage stated in Item 10 of the Reference Schedule.

# 5.4 Allocation of Operating Expenses

- (1) The Landlord may allocate to particular tenants, including the Tenant, operating expenses identifiable as incurred in respect of those tenants and exclude those operating expenses from the operating expenses for which all tenants are liable.
- (2) Where any operating expenses are not co-extensive with an operating expenses year, they will be deemed to accrue and be apportioned from day to day.

# 5.5 Variations in Tenant's Proportion

- (1) If the Landlord alters the net rentable area of the Building, the Landlord will be entitled to recalculate the Tenant's proportion in good faith on the same basis as the Tenant's proportion was calculated prior to the Commencement Date but taking into account any increased or reduced net rentable area of the Building.
- (2) Each recalculation will be binding from the date the Landlord notifies the Tenant of the recalculation.

Kulenkas

N P.Mackae

- 14 -

#### Statement of Operating Expenses 5.6

- The Landlord may determine the operating expenses payable by the Tenant in (1)respect of each operating expenses year or any broken period.
- As soon as practicable after the expiration of each operating expenses year. (2) the Landlord will furnish an itemised statement to the Tenant containing particulars of the operating expenses payable by the Tenant.
- Unless either party notifies the other of a manifest error in the notice within (3) 30 days of service, the statement will be conclusive.

#### Payment by the Tenant of Operating Expenses 5.7

The Tenant must pay to the Landlord the Tenant's proportion of the operating expenses within 30 days after service of the statement referred to in clause 5.6.

#### Payment by Tenant on Account of Operating Expenses 5.8

- Regardless of the provisions of clause 5.7, the Landlord may notify the (1) Tenant of the Landlord's estimate of the operating expenses payable by the Tenant for any period not exceeding 1 year in advance of the estimate.
- During that period the Tenant will pay to the Landlord the amount of the (2) estimate by equal monthly instalments in advance on the same days and in the same manner as the Tenant is required to pay Rent.
- Following a review of operating expenses, any necessary adjustment between (3) the estimated and actual operating expenses payable by the Tenant will be made.
- Any refund or further payment will be made by the relevant party. (4)

# 6. Damage and Destruction

- Abatement 6.1
  - If the Building or the Premises are damaged or destroyed so as to render any (1) part of the Premises wholly or substantially:
    - unfit for occupation and use by the Tenant; or (a)
    - inaccessible having regard to the nature and location of the Premises **(b)** and the normal means of access to them;

then:

- the Rent; (c)
- any other money payable by the Tenant; and (d)
- the covenant to repair and maintain; (e)

will abate according to the nature and extent of the damage or destruction sustained.

Mentes A. R. Mackae

- (2) The remedies for:
  - (a) recovery of the Rent and any other money or a proportionate part failing due after the damage or destruction; or
  - (b) enforcement of the covenant to repair and maintain;

will be suspended (or partially suspended as the circumstances require) until the Building and/or Premises are:

- (c) made fit for the Tenant's occupation and use;
- (d) made accessible; or
- (e) this Lease is terminated.

### 6.2 Termination

- (1) If either party considers the damage to the Premises renders it impractical or undesirable to restore the Premises or make them fit for the Tenant's occupation and use or renders them inaccessible to the Tenant, that party may terminate this Lease by giving not less than 1 month's notice to the other.
- (2) At the expiration of that notice this Lease will terminate.

# 6.3 Antecedent Rights

No liability will attach to either party because of termination of this Lease under clause 6.2 but that termination will be without prejudice to the rights of either party for any antecedent breach or non-observance of any provision of this Lease.

## 6.4 Dispute Resolution

- (1) Any dispute arising under this clause 6 will be determined by a member of the Australian Institute appointed by the president at the request of either the Landlord or the Tenant.
- (2) In making his determination, the appointed member will act as an expert and not as an arbitrator and his determination will be final and binding on both parties.
- (3) The Cost of the determination will be paid by both parties equally unless otherwise decided by the member of the Australian Institute.

## 6.5 Refusal of Indemnity

The provisions of clause 6.1 will not operate and clause 6.2 will not afford the Tenant a right to terminate this Lease if an insurer under any policy effected by the Landlord under clause 5.2(3) refuses indemnity because of any act, default, neglect, misconduct or omission of the Tenant.

# 6.6 Proceeds of Insurance

If the Premises are damaged or destroyed and the Lease is terminated under clause 6.2, the Tenant will have no interest in the insurance proceeds.

N P. Machae Willandos

### 7. Use of the Premises

# 7.1 Permitted Use

The Tenant must not use, occupy or permit the Premises to be used or occupied for any purpose other than as stated in Item 11 of the Reference Schedule.

# 7.2 Restrictions on Use

The Tenant will:

- (1) conduct the Tenant's business in the Premises as permitted under this Lease;
- (2) not use the Premises as a residence;
- (3) not keep any animals or birds in the Premises;
- (4) at its expense, keep the Premises free and clean of pests, insects and vermin and in default the Landlord will be entitled to employ pest exterminators to carry out any pest extermination at the Tenant's expense;
- (5) not carry on or permit any noxious or offensive act, trade, business, occupation or calling to be carried on in the Premises;
- (6) not cause annoyance, nuisance, grievance, damage or disturbance to other tenants or occupiers of the Building or adjacent premises;
- (7) not hold or permit any auction, bankrupt or fire sale to be held on the Premises;
- (8) not make any disturbing or irritating noises or install or use any appliance, engine or machine which causes or may be likely to cause noise or vibration in the Building; and
- (9) not prepare or cook food except in areas provided and approved by the Landlord for that purpose.
- 7.3 No Warranty as to Use
  - (1) The Landlord gives no warranty as to the suitability of the Premises for any purpose or the use to which the Premises may be put.
  - (2) The Tenant will:
    - (a) be deemed to have accepted this Lease with full knowledge of and subject to any prohibitions or restrictions on the use of the Premises under any Law or Requirement;
    - (b) at its expense, comply with all Laws and obtain and comply with the consents or approvals of any authority which may be necessary or appropriate for the Tenant's business; and
    - (c) not by any act or omission cause or permit any consent or approval referred to in clause 7.3(2)(b) to lapse or be revoked.

Kenfandas V

P. Mackae

- 17 -

# 7.4 Compliance with Laws and Requirements

- (1) At its expense, the Tenant must comply with and observe any Law or Requirement concerning:
  - (a) the Premises or any of the Tenant's Fittings; and
  - (b) the Tenant's particular use or occupation of the Premises including any which arise as a result of the sex or number of persons in the Premises;

whether or not the Law or Requirement is addressed to, or required to be complied with by either the Landlord or the Tenant or both or by any other person.

- (2) If any Law or Requirement is notified to or served upon the Tenant, it must promptly provide a complete copy to the Landlord.
- (3) If it is lawful to do so, before complying with any Law or Requirement under clause 7.4(1) the Tenant must obtain the Landlord's consent.

# 7.5 Landlord's Rights if Tenant Falls to Comply

The Landlord may:

- (1) without prejudice to any of its other rights in respect of non-compliance, elect to comply (wholly or partially) with any Law or Requirement under clause 7.4 at the Tenant's expense; and
- (2) if it exercises any rights under clause 7.5(1), elect to have the balance of any Law or Requirement complied with by the Tenant.

# 7.6 Structural Alterations

- (1) The Tenant will not be required under clauses 7.3, 7.4 and 7.5 to effect or pay for structural alterations or additions except those caused by, contributed to or arising from:
  - (a) the nature of its business;
  - (2) the number or sex of the persons comprising the Tenant or the Tenant's Employees; or
  - (3) any deliberate or negligent act or omission by the Tenant or the Tenant's Employees.
- (4) Subject to clause 7.6(1), upon demand the Tenant must pay to the Landlord all reasonable cost incurred by the Landlord in complying with any Law or Requirement under clauses 7.3, 7.4 and 7.5.

## 7.7 Exterior Signs

(1) The Tenant must not erect, paint, write or attach any sign upon any part of the Premises, the Building or the Land without the Landlord's prior consent.

Kujana V

P.Machae

- (2) Any sign erected, painted, written or attached upon any part of the Premises, the Building or the Land without the Landlord's prior consent may be taken away by the Landlord without notice at the Tenant's expense.
- (3) If any sign is erected, painted, written or attached upon any part of the Premises, the Building or the Land, on termination it must be removed and taken away by the Tenant and the Premises, the Building or the Land returned to their original condition at the Tenant's expense.

# 7.8 Overloading

- (1) The Landlord may prescribe the weight and position of any heavy article in the Premises.
- (2) Damage to any part of the Building or the Land caused by moving any heavy article will be repaired by the Tenant or, if the Landlord elects, by the Landlord at the Tenant's expense.
- (3) Within 7 days of the Landlord notifying the Tenant of the amount expended by the Landlord, the Tenant will pay that sum to the Landlord.
- (4) The Tenant must not install any equipment in the Premises that may overload any Services.
- (5) If the Landlord upgrades the Services to accommodate any equipment which the Tenant wishes to install, the Tenant will pay to the Landlord the Cost of any alterations.
- (6) The Landlord may require the Tenant to pay the estimated Cost of the alterations to the Landlord before the alterations are commenced.

# 7.9 Drains and Waste Pipes

- (1) The Tenant must not use any Landlord's Fixture other than for its intended purpose nor place in it any substance which it was not designed to receive.
- (2) The Tenant will pay to the Landlord the Cost of repairing any damage to the Landlord's Fixtures arising from misuse by the Tenant or the Tenant's Employees.

### 7.10 Antennae, Radios Etc.

The Tenant will not:

- (1) affix any television or radio mast, antennae, satellite dish or similar device to any part of the Building; or
- (2) use or operate any sound or picture producing equipment in the Premises which may cause irritation to other persons.

### 7.11 Rubbish Accumulation

The Tenant must not:

 throw anything out of the Premises, down any lift shafts or into any passages or Common Areas;

pupadas V

& Machae

- (2) deposit waste paper, unwanted or useless articles or rubbish anywhere except in proper receptacles; or
- (3) place anything on any external windowsill or ledge.
- 7.12 For Sale Signs Etc.

The Landlord may:

- (1) place advertisements and signs on any part of the Premises it reasonably considers appropriate if the Premises are for sale or lease; and
- (2) show any interested persons through the Premises after giving the Tenant reasonable prior notice.
- 7.13 Use of Common Areas
  - (1) In common with other persons authorised by the Landlord, the Tenant and the Tenant's Employees may use the Common Areas for the purposes for which they were designed or intended.
  - (2) The Tenant will comply with all reasonable directions given to it by the Landlord relating to conduct in the Common Areas.
  - (3) The Tenant will not interfere with or impede other persons using the Common Areas nor will it permit any of the Tenant's Employees to do so.

# 7.14 Emergency Evacuation Procedure and Drills

The Tenant must:

- (1) comply with all of the Landlord's requirements in relation to participation in emergency evacuation procedures and drills and ensure that the Tenant's Employees also participate in the procedures and drills; and
- (2) not make or permit any of the Tenant's Employees to make any Claim against the Landlord in connection with anything covered by this clause 7.14 and the Tenant indemnifies the Landlord against any Claim arising from the emergency evacuation procedures or drills.

## 7.15 Thermal or Fire Detectors

If any thermal or fire detectors are required to be installed in the Premises by any Law or Requirement because of the Tenant's use of the Premises, they will be installed at the Tenant's expense.

### 8. Assignment and Subletting

### 8.1 No Assignment Etc.

The Tenant will not assign, transfer, mortgage, charge or otherwise deal with its interest in the Premises or demise, sublet, part with possession of or grant any licence affecting the Premises.

Krilfandes W

& Machae

- 20 -

# 8.2 Assignment and Subletting Permitted

- (1) The Landlord will not withhold its consent to the assignment or subletting of the Premises if the Tenant first makes an application to the Landlord for consent and the following provisions have been observed by the Tenant:
  - (a) the Tenant is not in default under this Lease other than a default which has been waived by the Landlord;
  - (b) the Tenant proves to the Landlord's reasonable satisfaction that the incoming tenant is a respectable, responsible and solvent person capable of adequately carrying on the business permitted under this Lease;
  - (c) if the incoming tenant is a company other than a company whose shares are listed on an Australian stock exchange, the incoming tenant's obligations are guaranteed in a form acceptable to the Landlord;
  - (d) the Tenant obtains, at its expense, from the incoming tenant and any incoming guarantor an executed deed, in a form reasonably required by the Landlord, requiring the incoming tenant and incoming guarantor to perform and observe the Tenant's obligations under this Lease;
  - (e) in the case of a sublease, the Tenant proves to the Landlord's satisfaction that the incoming subtenant is obliged to pay a rent at least equal to the Tenant's Rent;
  - (f) the Tenant pays the Landlord's reasonable cost of giving its consent, whether or not the proposed assignment or sublease proceeds to completion; and
  - (g) the incoming tenant pays or gives to the Landlord a security deposit or bank guarantee of an amount determined by the Landlord (not in excess of 3 months' Rent) as security for the observance and performance of the Tenant's obligations.

# 8.3 Change in Ownership of Shares in Company

- (1) If the Tenant is a company, any change in the persons who beneficially own or control a majority of the company's voting shares at the date of this Lease except for the purposes of reconstruction or reorganisation will constitute an assignment of this Lease.
- (2) The Tenant will be in breach of clause 8.1 unless the Tenant obtains the Landlord's prior consent to the change in shareholding.
- (3) The consent will not be withheld in the following circumstances:
  - (a) the Tenant is not in default under this Lease other than a default which has been waived by the Landlord;
  - (b) the Tenant proves to the Landlord's reasonable satisfaction that any shareholder or director of the Tenant who was not a shareholder or director prior to the change in shareholding is a respectable,

Kenfalas N

RMachae

- 21 -

responsible and solvent person capable of adequately carrying on the business permitted under this Lease:

- (c) the Tenant obtains, at its expense, a guarantee from any new shareholder or director, in a form acceptable to the Landlord, guaranteeing the Tenant's obligations; and
- (d) the Tenant pays the Landlord's reasonable cost of giving its consent whether or not the proposed change in shareholding proceeds to completion.
- (4) This clause will not apply if the Tenant is a corporation whose voting shares are listed on an Australian stock exchange or if at least 80% of its voting shares are owned by another company whose voting shares are so listed.

#### 8.4 **Charges Over Tenant's Fittings**

- (1) Without the Landlord's prior consent, the Tenant will not mortgage, charge, lease or deal with any Tenant's Fittings if that requires or may require the Landlord to sign a waiver or similar document.
- (2) The consent will not be unreasonably withheld if:
  - (a) the Tenant wishes to enter into a mortgage, charge or lease in good faith as a means of financing the Tenant's Fittings; and
  - (b) the waiver is in a form acceptable to the Landlord and the Tenant pays the Landlord's reasonable cost in relation to it.

# 9. Utilities

#### 9.1 Source of Light and Power

The Tenant will only use light, power or heat generated by electrical current or gas supplied through meters except in the case of failure of supply when the Tenant may use other sources of energy except a naked flame.

#### 9,2 **No Alterations to Electrical Installations**

- (1) The Tenant will not make any alterations or additions to the electrical installations or wiring in the Premises without the Landlord's prior consent,
- (2) The Tenant will not install any electrical equipment which overloads the cables, switchboards or sub-boards through which electricity is conveyed to the Premises.

#### 9.3 **Charges for Utilities**

By the due date the Tenant will pay:

- (1) charges for electricity, gas and water consumed in the Premises;
- (2) charges for any telephone service connected to the Premises; and

Kuladas V

P. Machae

- 22 -

other charges and impositions imposed by any public authority for the supply (3) of any other Service to the Premises.

#### **Heating or Cooling Devices** 9.4

The Tenant will not use or install any heating or cooling device or machine which in the Landlord's opinion may interfere with the efficient running of the Air Conditioning Equipment or increase its running costs.

#### 9.5 Access to Equipment

The Tenant will not interfere with or obstruct access to the Air Conditioning Equipment or firs alarm or prevention system installed in the Premises.

### 10. Maintenance and Repair

#### **Repair of Premises** 10.1

- The Tenant must keep the Premises, the Tenant's Fittings and the Landlord's (1) Fixtures in good repair and condition except for:
  - fair wear and tear; and (a)
  - damage covered by insurances taken out by the Landlord in respect of **(b)** the Building.
- The exception in clause 10.1(b) will not apply if insurance money is (2) irrecoverable through the act, default, neglect, omission or misconduct of the Tenant or the Tenant's Employees.
- Nothing in clause 10.1(1) imposes any obligation on the Tenant in respect of (3) any major structural maintenance, replacement, renovation or repair unless required because of:
  - the act, omission, neglect, default or misconduct of the Tenant or the (a) Tenant's Employees;
  - the Tenant's use of the Premises; or (b)
  - other provisions of this Lease. (c)

#### Cleaning, Maintenance and Breakages 10.2

The Tenant will:

- subject to clause 10.3 keep the Premises thoroughly clean and tidy; (1)
- keep the Premises free from dirt and rubbish, store all trade waste, shop (2) refuse and garbage in proper receptacles and arrange for its regular removal from the Premises;
- keep the Tenant's Fittings clean and maintained in good order and condition; (3) and
- immediately repair or replace: (4)

Kulandoo H. P. Mackee

- (a) broken glass with glass of the same or similar quality;
- (b) damaged or inoperative electric light bulbs, globes, tubes and other means of illumination and light switches and power points which may become damaged or fail to operate; and
- (c) Landlord's Fixtures which are broken or damaged by the Tenant or the Tenant's Employees.

# 10.3 Cleaning Services

- (1) The Landlord will use its best endeavours to provide a satisfactory cleaning and waste removal service for the Premises, the Common Areas and other parts of the Building.
- (2) The Landlord will provide hand drying facilities and other toilet requisites in the toilets and washrooms.
- (3) The Tenant will only use the cleaning service provided by the Landlord.
- (4) Cleaning service employees will have access to the Premises during hours notified by the Landlord to the Tenant, including normal office hours, for the purpose of window cleaning and outside normal office hours for the purpose of other cleaning.
- (5) The Landlord will not be responsible to the Tenant for damage to the Premises or loss or damage to property in the Premises caused by the Landlord, its employees or contractors arising out of the Landlord's obligations under clause 10.3.

### 10.4 Landlord's Right to Inspect and Repair

- (1) Except in the case of emergency (when no notice will be required) after giving the Tenant prior reasonable notice, the Landlord and its agents may enter the Premises and view their condition.
- (2) The Landlord may serve a notice on the Tenant requiring it to repair any defect, the repair of which is the Tenant's obligation, within a reasonable time.
- (3) If the Tenant does not make the repairs to the Landlord's satisfaction, the Landlord and its agents may enter any part of the Premises and make the repairs at the Tenant's expense.
- (4) In exercising its powers under clause 10.4, the Landlord will endeavour to cause as little inconvenience to the Tenant as is practicable in the circumstances.

### 10.5 Landlord May Enter

- (1) The Landlord may enter the Premises with workmen and other authorised persons and necessary materials and appliances to:
  - (a) comply with any Law or Requirement involving the destruction of noxious animals, rodents or other pests;

Kenfadas I

P. Machae
- (b) carry out any repairs, alterations, renovations, extensions or works;
- (c) exercise the Landlord's rights; or
- (d) provide any Services to the Tenant or other tenants of the Building.
- (2) In carrying out work under clause 10.5 the Landlord will endeavour to cause as little inconvenience to the Tenant as is practicable in the circumstances.

# 10.6 Notice of Damage or Defect in Services

The Tenant will promptly give the Landlord notice of:

- any damage to, defect or disrepair in the Services or the Landlord's Fixtures; and
- (2) any circumstances likely to cause any danger risk or hazard to the Premises, the Building or any person.

# **11. Redecoration and Restoration**

# 11.1 Definition of Redecorate

The term "redecorate" includes:

- (1) washing down the interior of the Premises and the Tenant's Fittings; and
- (2) treatment as previously treated of all internal surfaces of the Premises to a specification previously approved by the Landlord;

but does not include replacing carpet or curtains unless they have been damaged by the act, neglect, default or omission of the Tenant or any of the Tenant's Employees.

#### 11.2 Tenant to Redecorate

The Tenant will redecorate the Premises to reasonable standards determined by the Landlord prior to the Tenant vacating the Premises or prior to the expiration of each period of  $3^{\prime}$  years calculated from the Commencement Date, whichever event occurs first.

# Landlord Approval Required

- (1) The Tenant must obtain the Landlord's approval prior to carrying out any redecoration.
- (2) The Landlord may specify the type and colour of paint, material and treatment to be used in connection with the redecoration.

# 11.4 Failure to Redecorate

- (1) If the Tenant fails to redecorate the Premises prior to vacating them, the Landlord may redecorate the Premises at the Tenant's expense.
- (2) The Landlord will be entitled to recover from the Tenant:

Killado V

R. Machae

86 Req:C145586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:27 of 54 Rel:IpT:Sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 25 -

- (a) the Cost of redecoration; and
- (b) if the Landlord redecorates the Premises as soon as practicable after the Tenant vacates, a sum equal to the Rent and any other money being paid by the Tenant at the time it vacated the Premises calculated from the date it vacated the Premises to the date the Landlord completes the redecoration.
- (3) The Tenant will pay interest at the Default Rate on any money payable by the Tenant under clause 11.4.

# 11.5 Disputes

- (1) Any disagreement between the Landlord and the Tenant concerning the standards to which the Premises are to be redecorated will be referred to an independent architect nominated by the president of the Royal Australian Institute of Architects at the request of either party.
- (2) The nominated architect will determine the disagreement as an expert and not as an arbitrator and his decision will be final and binding on the Landlord and the Tenant.
- (3) The cost of the nominated architect will be paid equally by the Landlord and the Tenant.

# 11.6 Materials

- (1) If the Tenant is required to redecorate the Premises, remove any partitions, fittings or fixtures, replace or repair any carpet or do any other work, the Tenant will at its expense restore the Premises to a new condition regardless of their condition at the commencement of the Tenant's occupation.
- (2) For the purposes of clause 11.6(1), "new condition" means replacing with new parts any parts of the Premises which have been lost, are missing or, in the Landlord's reasonable opinion, have been damaged and need to be replaced.

#### 12. Alterations

# 12.1 Alterations and Additions

- (1) The Tenant will not alter or add to the Premises, install or alter any partitions or install any heavy article likely to disturb the efficient operation of the Services without the Landlord's prior consent.
- (2) The Tenant must submit plans and specifications of the proposed alteration, addition or installation for the Landlord's consent.
- (3) As a condition of its approval the Landlord may require:
  - (a) the Tenant to engage the services of consultants approved by the Landlord to ensure that alterations and additions are in sympathy with the Building if the Tenant wishes to make structural alterations or additions to the Premises;

Kuladas V

Rallacka.

86 Req:C145586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Frt:27-Mar-2015 10:17 /Pgs:ALL /Seq:28 of 54 Ref:101:sixore /Src:w /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 26 -

- (b) the Tenant to use only the drawings, plans and specifications prepared by those consultants after first having them approved by the Landlord's consultants;
- (c) the Tenant to pay the Landlord's cost of the Landlord's consultants approving the plans and specifications referred to in clause 12.1(3)(a) and, if the Landlord pays those Costs, it may recover them from the Tenant as a debt;
- (d) any proposed work to be supervised by a person nominated by the Landlord and the cost of supervision will be paid by the Tenant;
- (e) any proposed work to be carried out by contractors or tradesmen approved by the Landlord;
- (f) the Tenant to obtain, at its expense, from any competent authority all approvals or permits necessary to enable the proposed work to be lawfully executed and, if requested by the Landlord, the Tenant must produce a copy of any approval and permit; and
- (g) the Tenant to reimburse the Landlord for cost incurred by it as a result of the installation, operation or removal of any equipment, fixture, fitting or machinery.
- (4) The Tenant must promptly provide the Landlord with a complete set of "as-built" drawings and commissioning data for the work carried out under this clause 12.1.
- 12.2 Installation of Equipment Etc.

Without the Landlord's prior consent, the Tenant must not:

- (1) install any water, gas or electrical fixtures, equipment or appliances or any apparatus for air conditioning, heating, cooling, ventilating or illuminating the Premises; or
- (2) mark, paint, drill, deface or damage any part of the Premises.
- 12.3 Partitions
  - (1) Prior to the Tenant installing or using any partitions in the Premises, the Tenant must submit to the Landlord for approval:
    - (a) a copy of the plans and specifications for the proposed work which must comply with Laws and Requirements and be previously approved by the relevant authorities; and
    - (b) details of the type, quality, colour and size of the partitions.
  - (2) The partitions must be installed:
    - (a) by a contractor approved by the Landlord under the supervision of the Landlord or its agent; and
    - (b) at the Tenant's expense including the cost of alterations or additions to the Services, the Premises or the Building.

Holanders V

P.Machas.

- 27 -

(3) The partitions remain the Tenant's property and the Tenant will maintain and insure them.

# 12.4 Reinstatement

- (1) If the Tenant breaches clauses 12.1, 12.2 or 12.3, the Landlord may give the Tenant a notice requiring it to reinstate the Premises to their former condition.
- (2) If the Tenant fails to comply with the notice, the Landlord may undertake the necessary work and the Landlord's cost of doing the work will be recoverable from the Tenant.

# 13. Insurances and Indemnities

13.1 Public Risk and Plate Glass Insurance

The Tenant will keep current:

- (1) an adequate and reasonable public risk insurance policy for the sum stated in Item 12 of the Reference Schedule or a greater sum as may be reasonably nominated by the Landlord from time to time; and
- (2) an adequate insurance policy covering any windows, doors, plate glass and display showcases forming part of or in the Premises for the full insurable reinstatement Cost.
- 13,2 Workers' Compensation Insurance

The Tenant will maintain workers' compensation insurance (including cover for unlimited common law liability) for:

- (1) the Tenant's Employees and persons who could assert a Claim for death or bodily injury against the Landlord; and
- (2) the sum approved by the Landlord whose approval shall not be unreasonably withheld.

# 13.3 Landlord May Insure

- (1) If the Tenant fails to maintain the insurances required by this clause 13, the Landlord may effect and maintain those insurances.
- (2) The Landlord's Cost in doing so will be paid by the Tenant.

# 13.4 Tenant's Insurances

The Tenant will ensure that all policies of insurance effected or required to be effected by the Tenant under this clause 13:

- (1) are taken out with an insurer approved by the Landlord;
- (2) are for an amount, cover risks and contain conditions which are acceptable to the Landlord and its insurer;

Kala

N

RMachae

- 28 -

- (3) have no exclusions, endorsements or alterations unless first approved by the Landlord;
- (4) are taken out in the joint names of the Landlord and the Tenant for their respective rights and interests; and
- (5) the approval of the Landlord under clause 13.4 will not be unreasonably withheld.
- 13.5 Payment and Production of Insurance Policies
  - (1) The Tenant will promptly pay all premiums and other money payable in respect of its insurances.
  - (2) If requested by the Landlord, the Tenant must produce policies of insurance which the Tenant is required to effect under this clause 13 and the receipt for the latest premium payable.

# 13.6 Effect on the Landlord's Insurance

- (1) Without the Landlord's prior consent, the Tenant will not do or omit to do anything to or upon the Premises or the Building which may:
  - (a) increase the rate of any insurance on the Premises, the Building or any property in them;
  - (b) vitiate or render void or voidable any insurance on the Premises or any property in the Premises; or
  - (c) conflict with any Law or Requirement, the requirements of the Landlord's insurer relating to fire, fire safety or fire prevention or any insurance policy in respect of the Premises or any property in them.
- (2) The Tenant must pay to the Landlord all extra insurance cost on the Premises, the Building or the Landlord's Fixtures on account of the extra risk caused by the Tenant's use or occupation of the Premises.
- (3) The Landlord's acceptance of any extra insurance cost will not constitute a waiver of any breach of clause 13.6 by the Tenant.
- 13.7 Inflammable Substances
  - (1) The Tenant will not store chemicals, inflammable liquids, acetylene, gas, alcohol, volatile or explosive oils, compounds or substances on the Premises except if it is necessary and proper for the Tenant's business permitted under this Lease.
  - (2) Only reasonably appropriate quantities may be kept.
- 13.8 Exclusions of Landlord's Liability
  - (1) Tenant's Risk
    - (a) All property in the Premises will be at the Tenant's sole risk.

Killadas

V P. Mackae

- (b) The Landlord will not be liable for any Claim by the Tenant, the Tenant's Employees or any person which may arise from:
  - (i) any fault in the construction or state of repair of the Building or the Landlord's Fixtures;
  - (ii) any defect in the Services or the Landlord's Fixtures; or
  - (iii) the flow, overflow, leakage, condensation or breakdown of any water, air conditioning, gas, oil or other source of energy or fuel from or in any part of the Building;

or from any other cause except as a result of the negligence of the Landlord, its employees, contractors, invitees or agents.

# (2) Refease

- (a) The Tenant will occupy and use the Premises at the Tenant's risk,
- (b) To the full extent permitted by Law, the Tenant releases the Landlord from liability for any Claim in respect of or arising from:
  - (i) any property in the Premises;
  - (ii) damage or injury to any person or property in the Premises or the Building or on the Land; or
  - (iii) any of the circumstances set out in clause 13.9;

unless the Claim results from the negligence of the Landlord, its employees, contractors, invitees or agents.

## 13.9 Indemnities

The Tenant indomnifies the Landlord against all Claims arising during or after the Term from:

- (1) any cause relating to the Premises, any property or any person inside or outside the Premises occasioned or contributed to by any act, neglect or default of the Tenant or the Tenant's Employees;
- (2) the negligent use or neglect of the Services and facilities in the Premises or the Landlord's Fixtures by the Tenant, the Tenant's Employees, any other person claiming by, through or under the Tenant or any trespasser while in the Premises;
- (3) the overflow or leakage of water from any source including the Services or the Landlord's Fixtures;
- (4) the Tenant's failure to give notice to the Landlord of any defect in the Services;
- (5) any person's accidental death or damage to property caused or contributed to by the use of the Premises by the Tenant or the Tenant's Employees; and

Miladao

RMachae

(6) plate and other glass caused or contributed to by any act or omission by the Tenant or the Tenant's Employees;

unless the Claim arises from the negligence of the Landlord, its employees, contractors, invitees or agents.

# 14. Landlord's Covenants

# 14.1 Quiet Enjoyment

If the Tenant performs and observes all its obligations under this Lease, it may use the Premises without interruption or disturbance from the Landlord or any person claiming by, through or under the Landlord.

## 14.2 Person Other Than Landlord Becoming Entitled to Rents Etc.

- (1) If any person other than the Landlord becomes entitled to receive the Rent, that person will have the benefit of all covenants by the Tenant under this Lease.
- (2) If required by and at the expense of the Landlord, the Tenant will enter into reasonable covenants with that other person.

# 14.3 Services to be Provided by Landlord

The Landlord will use reasonable endeavours to ensure that the Services are operational and functional during the times contained in the Rules.

# 14.4 Failure of Services

The Tenant will have no Claim against the Landlord or be entitled to terminate this Lease solely because:

- (1) the Services fail to operate;
- (2) the Landlord shuts down or removes any Services to repair, maintain or replace them or because of the provisions of any Law or Requirement; or
- (3) the Landlord closes the Building under the provisions of this Lease.

# 14.5 Structural Matters

The Landlord will endeavour to keep the Premises in a sound structural condition and carry out all structural repairs which are not the Tenant's obligation.

# **15. Default and Termination**

# 15.1 Default

The Tenant will be in default if:

the Rent or any money payable by the Tenant is unpaid for 14 days;

fillandas

N

P. Machae

- (2) repairs required by any notice are not carried out by the Tenant within the time specified in the notice;
- (3) the Tenant fails to perform or observe any of its covenants or obligations under this Lease;
- (4) the Tenant assigns its property for the benefit of creditors; or
- (5) the Tenant enters into any form of liquidation, is wound up or dissolved, enters into a scheme of arrangement for creditors, is placed under official management or a receiver and/or manager of any of its assets is appointed.

# 15.2 Forfeiture of Lease

Subject to giving any prior demand or notice required by any Law and without prejudice to any other Claim which the Landlord has or may have against the Tenant or any other person in respect of default, if the Tenant defaults as specified in clause 15.1 the Landlord may:

- (1) re-enter and take possession of the Premises (by force if necessary) and eject the Tenant and all other persons and this Lease will terminate;
- (2) by notice to the Tenant, terminate this Lease from the date of giving the notice; and/or
- (3) by notice to the Tenant, convert the unexpired portion of the Term into a tenancy from month to month and, after the notice and until the tenancy is terminated, the Tenant will occupy the Premises as tenant from month to month.

# 15.3 Landlord May Rectify

Without notice to the Tenant, any cost incurred by the Landlord in remedying a default may be treated by the Landlord as a liquidated debt payable by the Tenant.

# 15.4 Waiver

(1) Waiver Must Be in Writing

No waiver by the Landlord will be effective unless it is in writing.

- (2) No Waiver
  - (a) The Landlord's failure to take advantage of any default by the Tenant will not be construed as waiving the default.
  - (b) No custom or practice which evolves between the parties will constitute a waiver or lessen the Landlord's right to insist upon the Tenant's strict performance or observance of any provision of this Lease or to exercise any of the Landlord's other rights.

# (3) Acceptance or Demand for Rent Not Waiver

Regardless of the Landlord's knowledge at the time, a demand by it for Rent or other money payable under this Lease or the subsequent acceptance of

Windo

I R. P. Mackae

- 32 -

Rent or other money will not constitute a waiver of any earlier default by the Tenant.

# 15.5 Tender After Termination

In the absence of any election by the Landlord, any money tendered by the Tenant after termination and accepted by the Landlord will be applied:

- (1) firstly, on account of any unpaid Rent and other money due under this Lease at the date of termination; and
- (2) secondly, on account of the Landlord's Cost of re-entry.
- 15.6 Interest on Overdue Money
  - (1) Interest

The Tenant will pay interest to the Landlord at the Default Rate on any Rent, Cost or other money due to the Landlord and unpaid for 14 days.

(2) Conditions

Interest will:

- (a) accrue from day to day;
- (b) be capitalised on the last day of each month;
- (c) be payable on the first day of each month where an amount arose in the preceding month or months; and
- (d) be computed from the date for payment of the Rent, Cost or other money until payment.
- 15.7 Damages for Breach
  - (1) Repudiation
    - (a) If the Tenant's conduct constitutes breach of an essential provision of this Lease and the Landlord elects to treat that breach as repudiation or the conduct otherwise constitutes repudiation, the Tenant will compensate the Landlord for all loss or damage suffered by reason of or arising from the repudiation.
    - (b) The following covenants are deemed to be essential:
      - (i) to pay Rent [clause 4.1];
      - (ii) to pay operating expenses [clause 5.7];
      - (iii) to use the Premises for the use stated in Item 11 of the Reference Schedule [clause 7.1];
      - (iv) not to assign, sublet or charge without consent [clause 8.1];
  - (2) Damages for Repudiation

Kullanhee

P. Machae

- 33 -

Any loss or damage for the unexpired residue of the Term suffered by the Landlord as a result of the Tenant's repudiation may be recovered as damages at any time.

# (3) Landlord's Entitlement to Damages

The Landlord's entitlement to recover damages from the Tenant or any other person will not be limited or affected by any of the following:

- (a) if the Tenant abandons or vacates the Premises;
- (b) if the Landlord elects to re-enter the Premises or terminate this Lease;
- (c) if the Landlord accepts the Tenant's repudiation; or
- (d) if the parties' conduct (or that of any of their servants or agents) constitutes or may constitute a surrender by operation of law.

#### 15.8 Landlord to Mitigate Damages

- (1) If the Tenant vacates the Premises or if the Landlord accepts the Tenant's repudiation and terminates this Lease, the Landlord must take reasonable steps to mitigate its loss and endeavour to re-lease the Premises on reasonable terms.
- (2) The Landlord's entitiement to damages will be assessed on the basis that the Landlord has observed the obligation to mitigate damages.
- (3) The Landlord's conduct in mitigating its damages will not of itself constitute acceptance of the breach or repudiation or a surrender by operation of Law.

# 15.9 Calculation of Damages

Following repudiation by the Tenant if the Landlord terminates this Lease then, without prejudice to any other right or remedy, the Landlord may recover the difference between the aggregate of Rent and other money payable by the Tenant for the unexpired residue of the Term less any amount the Landlord obtains, or could in the Landlord's opinion reasonably be expected to obtain, by observing clause 15.8.

## 16. Termination of Term

#### 16.1 Tenant to Yield Up

On termination the Tenant will deliver the Premises to the Landlord in good repair, order and condition subject to clause 10.1.

# 16.2 Removal of Tenant's Fittings

At or before termination, the Tenant will remove all the Tenant's Fittings from the Premises.

V Kinfactas P. Machae

- 34 -

# 16.3 Tenant Not to Cause Damage

- (1) The Tenant will not damage the Premises in the removal of the Tenant's Fittings.
- (2) If the Tenant does so, it will repair any damage and leave the Premises clean.
- (3) If the Tenant fails to do so, the Landlord may repair and clean the Premises and recover the cost of doing so from the Tenant.

# 16.4 Failure by Tenant to Remove Tenant's Fittings

If the Tenant fails to remove the Tenant's Fittings as required by clause 16.2, the Landlord may:

- (1) remove and store the Tenant's Fittings at the Tenant's risk and expense: or
- (2) treat the Tenant's Fittings as if the Tenant had abandoned its interest in them and they had become the Landlord's property and deal with them as it thinks fit without being liable to account to the Tenant.

# 16.5 Tenant to Indemnify and Pay Landlord's Costs

The Tenant:

- (1) indemnifies the Landlord against the removal and storage of the Tenant's Fittings and against all Claims by any person claiming an interest in the Tenant's Fittings by reason of the Landlord's actions under clause 16.4; and
- (2) will pay the Landlord, as a liquidated debt payable on demand, any cost incurred by the Landlord in exercising its rights under clause 16.4 less any money received on disposal of the Tenant's Fittings.

# 16.6 Earlier Breaches

Termination of this Lease will not prejudice or affect any of the Landlord's rights or remedies against the Tenant for an earlier default by the Tenant.

## 17. Miscellaneous

#### 17.1 Notices

# (1) Execution of Landlord's Notice

Any notice by the Landlord under this Lease will be valid if signed by an officer or solicitor of the Landlord, the Managing Agent or any other person nominated by the Landlord.

# (2) Notice of Tenant's Address

The Tenant will promptly notify the Landlord of the address and facsimile number of the Tenant and any Guarantor and update the notice if any changes occur.

Juladas

. P.Mackae

# (3) Service of Notice on Tenant

Any notice which the Landlord elects to serve on the Tenant will be sufficiently served if:

- (a) served personally or addressed to the Tenant and left at the Promises;
- (b) sent to the Tenant's facsimile number; or
- (c) forwarded by prepaid security post to the Tenant's last known registered office, place of business or residence.
- (4) Service of Notice on Landlord

Any notice to the Landlord will be sufficiently served if:

- (a) served personally;
- (b) sent to the Landlord's facsimile number; or
- (c) forwarded by prepaid security post addressed to the Landlord.

All notices must be addressed to the Landlord at the address stated in this Lease or, if another address is later nominated by the Landlord, then at that address.

(5) Time of Service

Any notice sent by:

- (a) post will be deemed to be served on the second business day after the day it was posted; or
- (b) facsimile will be deemed to have been served at the time that the notice was transmitted unless the sender's facsimile machine indicates a malfunction in the transmission.

#### 17.2 Costs

The Tenant will pay the Landlord on demand:

- (1) stamp duty on this Lease including any fines and penalties except those due to the Landlord's default; and
- (2) the Landlord's reasonable legal fees and disbursements in connection with:
  - (a) the preparation, completion, stamping and registration of this Lease;
  - (b) any consent required under this Lease;
  - (c) obtaining the consent of any mortgagee of the Land and production of the title deed;
  - (d) any assignment or subletting;
  - (e) any surrender or termination of this Lease except through its expiry;

pula

. P.Mackae

- (f) any default by the Tenant or the Tenant's Employees in observing or performing the provisions of this Lease; and
- (g) any negotiations, alterations or changes to this Lease assessed on the basis of the Landlord's solicitors' usual hourly charges for that work.

# 17.3 Costs of Litigation

Except where a court determines otherwise and regardless of whether the Tenant is a party to the litigation, the Tenant will pay the Landlord's cost of litigation arising from the Tenant's occupancy of the Premises.

# 17.4 Right of Landlord to Make Rules

- (1) The Landlord may make Rules and amend them by notice to the Tenant so long as no Rule prejudices the Tenant's rights under this Lease.
- (2) If there is inconsistency between the provisions of this Lease and the Rules, the provisions of this Lease will prevail.

# 17.5 Compliance with Rules

The Tenant will observe and perform and cause the Tenant's Employees to observe and perform the Tenant's obligations under this Lease and the Rules.

# 17.6 Reservations

The Landlord reserves the following rights:

- (1) after reasonable notice (except in the case of an emergency when no notice will be necessary) the right to install, maintain, use, repair, alter, service and replace any Services;
- (2) the exclusive right to use any part of the Building or the Land which is not leased and the Landlord may erect or display advertising signs, grant leases or licences or authorise any person to use those areas for any purpose; and
- (3) by reasonable prior notice to the Tenant, the right to name or create a logo or change the name or logo of the Building and the Tenant will not make any Claim in relation to any name or logo.

# 17.7 Easements

- (1) For the purpose of:
  - (a) provision of access to the Premises;
  - (b) the support of structures erected on the Land or adjoining land;
  - (c) any Service;

the Landlord may dedicate, grant, transfer or otherwise enter into any arrangement in relation to the Land provided it does not prejudice the Tenant's rights under this Lease.

Kulance

Probackae

- 37 -

- This Lease will be deemed to be subject to any right granted by the Landlord (2) under clause 17.7.
- **Power of Attorney** 17.8
  - From the Commencement Date the Tenant irrevocably nominates and (1) appoints individually the Landlord, its officers and any nominee appointed in writing for the purposes of this clause 17.8 to be the attorney of the Tenant.
  - At any time after the power to re-enter contained in this Lease has arisen (2)(proof of which will be a statutory declaration of the Landlord or an officer of the Landlord) the attorney may execute and register a transfer or a surrender of this Lease or withdrawal of caveat.
  - In doing so, the attorney may use the Tenant's name and do anything relating (3) to the Premises which the Tenant could do.
  - The Tenant will ratify and confirm any lawful act of the attorney. (4)
  - The Director of Land Titles is authorised to act upon the statutory (5) declaration and to accept it as sufficient evidence of the termination of this Lease.

#### 18. Guarantees

#### 18.1 **Bank Guarantee**

Clause 18.2 applies if Item 14 of the Reference Schedule has been completed by inserting the amount of a bank guarantee.

#### Issue and Review of Bank Guarantee 18.2

The Tenant will:

- on or before executing this Lease, arrange for the issue of an unconditional (1) bank guarantee by a financial organisation approved by and in favour of the Landlord for the amount stated in Item 14 of the Reference Schedule;
- at each review date under clause 4 or at the later determination of each Rent (2) review, provide a replacement or additional bank guarantee so that the total amount guaranteed bears to the Rent payable from each review date the same proportion as the amount stated in Item 14 of the Reference Schedule bears to the Rent as at the Commencement Date;
- ensure that any bank guarantee is kept current and enforceable and will not (3) expire until at least 2 months after the Termination Date or after the date of termination of any further term arising from the exercise of an option to renew in this Lease or any further lease; and
- if the Landlord makes demand on any bank guarantee, provide a replacement (4) bank guarantee for the amount demanded by the Landlord.

Kulada W

R. Machae

- 38 -

# 18.3 Personal Guarantee

Clauses 18.4 to 18.11 apply if Item 15 of the Reference Schedule has been completed by inserting details of Guarantor.

## 18.4 Guarantee and Indemnity

In consideration of the Landlord granting this Lease to the Tenant at the Guarantor's request, the Guarantor guarantees to the Landlord:

- (1) the payment by the Tenant of the Rent and other money agreed to be paid; and
- (2) prompt performance and observance of all of the Tenant's covenants and obligations contained or implied in this Lease.

## 18.5 Indemnity

The Guarantor indemnifies the Landlord and agrees to keep the Landlord indemnified against all Claims which the Landlord may suffer or incur in connection with any breach or default by the Tenant under this Lease or any extension or renewal of the Term.

# 18.6 Liability of Guarantor

The Landlord's rights and the Guarantor's liability under clauses 18.4 and 18.5 will not be prejudiced or affected by:

- (1) the granting of any time, credit, forbearance, indulgence or concession by the Landlord to the Tenant or any Guarantor;
- (2) any absolute or partial release of the Tenant or any Guarantor or any compromise with the Tenant or any Guarantor;
- (3) any variation of this Lease, extension or renewal of the Term, holding over or continued occupation of the Premises by the Tenant;
- (4) any composition, compromise, release, discharge, arrangement, abandonment, waiver, variation, relinquishment or renewal of any security or right by the Landlord;
- (5) any assignment of this Lease or sublease of any part of the Premises;
- (6) the termination of this Lease;
- (7) the fact that the Rent or any other money may not be recoverable, may cease to be recoverable or may never have been recoverable or that any transaction affecting the Rent or the obligations contained in this Lease is or was wholly or partially void, voidable or unenforceable;
- (8) any failure to sue or agreement not to sue or any dealing, act or omission (whether constituting a waiver, election, estoppel or otherwise) by the Landlord with respect to the Rent, other money payable or the obligations under this Lease;

Kuladas

V & P. Machae

- any fact, circumstance, legal disability or incapacity which would otherwise (9) release the Tenant or any Guarantor from its obligations;
- non-execution of this Lease by one or more of the persons named as (10)Guarantor or the unenforceability of this guarantee and indemnity against one or more of the Guarantors; or
- the exercise or purported exercise by the Landlord of its right of re-entry. (11)
- Irrevocable 18.7

This guarantee and indemnity is irrevocable and will remain in force until the Tenant has performed and observed all its obligations under this Lease.

- 18.8 Guarantor Liable Regardless of Any Law
  - The Guarantor's liability will not be discharged by any payment to the (1)Landlord which is later avoided by Law.
  - (2) If that happens the Landlord, the Tenant and the Guarantor will be restored to their respective rights as if the payment had not been made.
- 18.9 Indemnity on Disclaimer

If a liquidator disclaims this Lease, the Guarantor indemnifies the Landlord against any resulting Claim by the Landlord for the residue of the Term.

- 18.10 Guarantor Not Prove in Liquidation
  - (1) The Guarantor will not prove or claim in any liquidation, composition, arrangement or assignment for the benefit of creditors until the Landlord has received 100¢ in the dollar of all money payable to it by the Tenant.
  - (2) The Guarantor will hold any proof, claim or dividend received by it on trust for the Landlord.
- 18.11 Guarantee to Continue
  - If this Lease is transferred or assigned, the benefit of this guarantee and (1) indemnity will extend to the transferee or assignee and continue concurrently for the benefit of the Landlord regardless of the transfer or assignment unless the Landlord releases the Guarantor in writing.
  - (2) This guarantee and indemnity covers the period while the Tenant occupies or is entitled to occupy the Premises as tenant or holds an equitable interest over the Premises under an agreement for lease or as a periodical tenant.
  - Without limiting clause 18.11(2) if registration of this Lease is required to (3) legally create a leasehold estate, then until this Lease is registered it will operate from the Commencement Date as an agreement for lease.
  - This guarantee and indemnity will apply whether this Lease is construed as (4) an agreement for lease, a periodical tenancy or otherwise.
  - (5) This guarantee and indemnity will extend to the lease, agreement for lease or periodical tenancy created by the Tenant's exercise of any option for renewal.

Kuladas W J. P. Mackae

86 Req;Cl45586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:42 of 54 Bf:1p1:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 40 -

#### 19. Option of Renewal

19.1 Option

If the Tenant:

- (1) wishes to lease the Premises for the further term stated in Item 16 of the Reference Schedule;
- (2) gives notice to that effect to the Landlord not less than 6 months before the Termination Date; and
- (3) at the date of giving notice and at the Termination Date, is not in default under this Lease;

the Landlord will grant a lease of the Premises ("further lease") to the Tenant upon the same terms and conditions as this Lease but subject to variations which are necessary in the Landlord's reasonable and bona fide opinion to render the further lease consistent with the terms and conditions of the standard lease for the Building then in use by the Landlord or to accommodate variations in:

- (4) the management of the Building;
- (5) the method of charging or collecting operating expenses;
- (6) the design or structure of the Building or Services; and
- (7) any Law or Requirement.

# 19.2 Further Lease

The further lease will contain:

- (1) a term as stated in Item 16 of the Reference Schedule;
- (2) a commencement date being the day after the Termination Date;
- (3) a termination date being the last day of the further term stated in Item 16 of the Reference Schedule; and
- (4) the amount to be inserted in Item 7 of the Reference Schedule shall be an amount to be agreed between the Landlord and the Tenant within 1 month of service on the Landlord of a notice under clause 19.1(2) and failing agreement within that 1 month period the amount will be determined by following the procedure set out in clause 4.4 as if the Termination Date was a review date but in any event the Rent will not be less than the Rent payable in the last year of the Term.

# 19.3 Omission of this Clause 19

This clause 19 will be omitted from the further lease but clause 20 of this Lease shall become clause 19 in the further lease.

Kulana W

. R.MacRae

# - 41 -

#### **19.4** Parties to Sign Further Lease

The Landlord, the Tenant and the Guarantor must sign the further lease within a reasonable time after service on the Landlord of a notice under clause 19.1(2).

#### 20. Option of Renewal

# 20.1 Option

If the Tenant:

- (1) wishes to lease the Premises for the further term stated in Item 17 of the Reference Schedule;
- (2) gives notice to that effect to the Landlord not less than 6 months before the Termination Date; and
- (3) at the date of giving notice and at the Termination Date, is not in default under this Lease;

the Landlord will grant a lease of the Premises ("further lease") to the Tenant upon the same terms and conditions as this Lease but subject to variations which are necessary in the Landlord's reasonable and bona fide opinion to render the further lease consistent with the terms and conditions of the standard lease for the Building then in use by the Landlord or to accommodate variations in:

- (4) the management of the Building;
- (5) the method of charging or collecting operating expenses;
- (6) the design or structure of the Building or Services; and
- (7) any Law or Requirement.

#### 20.2 Further Lease

The further lease will contain:

- (1) a term as stated in Item 17 of the Reference Schedule;
- (2) a commencement date being the day after the Termination Date;
- (3) a termination date being the last day of the further term stated in Item 16 of the Reference Schedule; and
- (4) the amount to be inserted in Item 7 of the Reference Schedule shall be an amount to be agreed between the Landlord and the Tenant within 1 month of service on the Landlord of a notice under clause 20.1(2) and failing agreement within that 1 month period the amount will be determined by following the procedure set out in clause 4.4 as if the Termination Date was a review date but in any event the Rent will not be less than the Rent payable in the last year of the Term.

pilladas

f. Machae

- 42 -

# 20.3 Omission of this Clause

This clause 20 will become clause 19 in the further lease, but it shall be omitted upon exercise of the second right to renew the lease.

# 20.4 Parties to Sign Further Lease

Mulanda W

P.Machae

The Landlord, the Tenant and the Guarantor must sign the further lease within a reasonable time after service on the Landlord of a notice under clause 20.1(2).

# - 43 -

Req:C1,45586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:45 of 54 f:1p1:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

86

#### **RULES AND REGULATIONS**

- 1. All areas outside the Premises including any and also the entrance halls passages lifts and stairways and garage areas of the Building shall be under the absolute control of the Landlord and shall not be obstructed by the Tenant or the Tenant's Employees or used by him or them for any other purpose except ingress and egress to and from the Premises.
- 2. With the exception of public holidays, the Building shall be open for business from 8.30 am until 11.00 pm every day of the year.
- 3. The Tenant shall be entitled to enter upon the Building and use the Premises at all times provided that if the Landlord shall become liable to pay any extra charges by reason of the Tenant's use of the Premises: on Sundays and public holidays or after 5.00 pm or any day then the Tenant shall pay to the Landlord all such charges.
- 4. The Tenant shall advise the Landlord and the Managing Agent of the Building for the time being of the private address of the Tenant or if the Tenant shall be a company of the Manager thereof or if there shall be more than one Tenant of each of them. The Landlord and the Managing Agent shall be promptly informed of any change in any such address.
- 5. If the Tenant fails to return to the Landlord at the termination of its occupancy all keys given by the Landlord to the Tenant or authorised by the Landlord then the Tenant will pay to the Landlord the cost of replacing the locks in the Premises.
- 6. The Tenant will not smoke and will cause each of the Tenant's Employees not to smoke in the Common Areas at all times during the Term.

Walanta

Prolachae

86 Reg:CL45586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seg:46 of 54 f:lpl:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 44 -

.

# **REFERENCE SCHEDULE**

ITEM 1.	LANDLORD (clause 1.1(11)	
	Isikit Pty Limited ACN 002 455 218 of 4/76 Campbell Parade, Bondi Beach	
ITEM 2.	TENANT (clause 1.1(22)	
	K MacRae Holdings Pty Ltd ACN 001 772 514 of Shop 4, 108-120 Station Street, Wentworthville	
ITEM 3.	PREMISES DESCRIPTION (clause 1.1(3))	
	Shop 4, 108-120 Station Street, Wentworthville	
ITEM 4.	COMMENCEMENT DATE (clause 1.1(5))	
	1 July 1993	
ITEM 5.	TERMINATION DATE (clause 1.1(26))	
	30 June 1998	
ITEM 6.	TERM (clause 1.1(25))	
	5 years	
ITEM 7.	<b>RENT</b> (clause 1.1(18) and 4.1)	
	\$56,167.80 per annum	
ITEM 8.	RENT REVIEW DATES (clause 4.3)	
	(a) Index review dates: 1 July 1994, 1 July 1995, 1 July 1996, 1 July 1997.	
	(b) Market review dates: Upon exercise of the option to renew the lease	
ITEM 9.	OPERATING EXPENSES YEAR (clause 5.1)	
	30 June	
ITEM 10.	TENANT'S PROPORTION OF OPERATING EXPENSES (clause 5.3)	
	30%	
ITEM 11.	USE OF PREMISES (clause 7.1)	
	Pharmacy and toy shop	
	Kulandero J. P. Machae	

Julantes

ITEM 12. AMOUNT OF PUBLIC RISK INSURANCE (clause 13.1(1))

\$10m

ITEM 13. DEFAULT RATE (clause 1.1(8)

Nil

ITEM 14. BANK GUARANTEE AMOUNT (clause 18.1(1))

Nil

ITEM 15. GUARANTOR (clause 1.1(9) and 18.3)

Kenneth Christopher MacRae, 39 Greenhaven Drive, Pennant Hills Patricia May MacRae, 39 Greenhaven Drive, Pennant Hills

ITEM 16, FIRST OPTION (clause 19)

5 years

ITEM 16. SECOND OPTION (clause 20)

5 years

Kenfanda W

P. Mackae

86 Reg:CL45586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:48 of 54 Ref:TFL:StAre /Src:w /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 46 -

EXECUTED as a deed on the date set out at the commencement of this Deed. 201 THE COMMON SEAL of ISIKIT PTY LIMITED Common was affixed in accordance with its Seal articles of association in the presence of: Director Director/Secretary KURANGIL VEDIT KUMANGIL . NWIT Name of Director/Secretary (BLOCK LETTERS) JOLDIM (Launman THE COMMON SEAL of KMACRAE HOLDINGS PTY LTD ≸ra was affixed in accordance with its Ă. C. N. articles of association in the 001 772 51 presence of: . A.M. 1.00 Director/Secretary Director PATRICIA MARY MACRAE KENNERY CHAISSORIER MINABLE Name of Director (BLOCK LETTERS) Name of Director/Secretary (BLOCK LETTERS) SIGNED SEALED AND DELIVERED Killado by KENNETH CHRISTOPHER MACRAE in the presence of: 6 Bestoces Signature of Witness

CHRISTINE BERTACCO. Name of Witness (BLOCK LETTERS)

CI-HUNT & HUNT Address of Witness EASTWOOP.

Name of Director (BLOCK LETTERS)

fridante

P.Mackae A

M

86 Req:Cl45586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seq:49 of 54 f:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

please

- 47 -

SIGNED SEALED AND DELIVERED by PATRICIA MAY MACRAE in the presence of: MARY

X

Signature of Witness

U

Name of Witness (BLOCK LETTERS)

Address of Witness EASTWOOD.

P. Mackae ) ) )

perfander & Prolackae M

# TABLE OF CONTENTS

86 Reg: C145586 /Doc:DL U760036 /Rev:16-Mar-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:17 /Pgs:ALL /Seg:50 of 54 of:lpl:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

j

1.	Definitions and Interpretation
	1.1 Definitions       1         1.2 Construction       3         1.3 Plurals and Genders       3         1.4 Parties Bound Jointly and Individually       4         1.5 Statutes and Regulations       4
	1.6 Covenants       4         1.7 Severability       4         1.8 Whole Agreement       4         1.9 Headings       4         1.10 Organisations       4         1.11 Areas and Measurement       5         1.12 Landlord's Consent       5         1.13 Written Notices       5
2.	Exclusion of Statutory Provisions
	2.1 Statutory Provisions Excluded
3.	Term and Holding Over
	3.1 Term         5           3.2 Holding Over         6
4.	Rent and Rent Reviews
	4.1 Rent       6         4.2 Definitions       6         4.3 Annual Rent Review: Consumer Price Index       7         4.4 Market Rent Review       8         4.5 Payment of Rent Pending Review       11         4.6 Minimum Rent       11
5.	Operating Expenses
	5.1 Operating Expenses Year115.2 Definition of Operating Expenses115.3 Tenant's Proportion of Operating Expenses135.4 Allocation of Operating Expenses135.5 Variations in Tenant's Proportion135.6 Statement of Operating Expenses145.7 Payment by the Tenant of Operating Expenses145.8 Payment by Tenant on Account of Operating Expenses14
б.	Damage and Destruction
	6.1 Abatement       14         6.2 Termination       15         6.3 Antecedent Rights       15         6.4 Dispute Resolution       15         6.5 Refusal of Indemnity       15         6.6 Proceeds of Insurance       15

Kulpho J. P. Machae

_			
	(ii)		
	7. Use of the Premises		16
	7.1 Permitted Use		16
	7.2 Restrictions on Use		16
	7.3 No Warranty as to Use		17
	7.5 Landlord's Rights if Tenant Fails to Comply		17
	7.6 Structural Alterations		17
	7.7 Exterior Signs		17
	7.8 Overloading		18
	7.9 Drains and Waste Pipes		18
	7.10 Antennae, Radios Étc	********	18 18
	7.12 For Sale Signs Etc.		19
	7.13 Use of Common Areas		19
	7.14 Emergency Evacuation Procedure and Drills		19
	7.15 Thermal or Fire Detectors		
			10
	3. Assignment and Subletting		17
	8.1 No Assignment Etc		19
	8.2 Assignment and Subletting Permitted		20
	8.3 Change in Ownership of Shares in Company		20
	8.4 Charges Over Tenant's Fittings		21
	). Utilities	<i>,</i> ,,,,,,,,,,,,	21
	9.1 Source of Light and Power		21
	9.2 No Alterations to Electrical Installations		21
	9.3 Charges for Utilities		21
	9.4 Heating or Cooling Devices	********	22
	9.5 Access to Equipment	* • • • • • • • • • •	<i>LL</i>
	0. Maintenance and Repair		22
	10.1 Repair of Premises		
	10.2 Cleaning, Maintenance and Breakages		
	10.3 Cleaning Services		
	10.4 Landlord's Right to Inspect and Repair		
	10.6 Notice of Damage or Defect in Services		24
	1. Redecoration and Restoration		24
	11.1 Definition of Redecorate		24
	11.2 Tenant to Redecorate		
	11.3 Landlord Approval Required		24
	11.4 Failure to Redecorate		24
	11.5 Disputes	* * • • • * * • • •	25 05
	11.6 Materials		25
	2. Alterations		25
	12.1 Alterations and Additions		25
	12.2 Installation of Equipment Etc.		

Jula J. P. Mackae

M

•		(iii)
•		
		12.3 Partitions
		12.4 Reinstatement
	13.	Insurances and Indemnities
	13,	
		13.1 Public Risk and Plate Glass Insurance 2
		13.2 Workers' Compensation Insurance 2
		13.3 Landlord May Insure
		13.4 Tenant's Insurances
		13.6 Effect on the Landlord's Insurance
		13.7 Inflammable Substances
		13.8 Exclusions of Landlord's Liability 2
		13.9 Indemnities
<u>.</u>	14.	Landiord's Covenants
	14,	
		14.1 Quiet Enjoyment
		14.2 Person Other Than Landlord Becoming Entitled to Rents Etc 3
		14.3 Services to be Provided by Landlord
		14.4 Failure of Services
	-	14.5 Structural Matters
	15.	Default and Termination
		15.1 Default
		15.2 Forfeiture of Lease
		15.3 Landlord May Rectify 33
		15.4 Waiver
		15.5 Tender After Termination         32           15.6 Interest on Overdue Money         32
		15.7 Damages for Breach
		15.8 Landlord to Mitigate Damages 33
	-	15.9 Calculation of Damages
	16.	Termination of Term
	1	16.1 Tenant to Yield Up
	1	16.2 Removal of Tenant's Fittings 33
		16.3 Tenant Not to Cause Damage
		16.4 Failure by Tenant to Remove Tenant's Fittings
		16.5 Tenant to Indemnify and Pay Landlord's Costs
		10.0 Lainei Dicacues
	17.	Miscellaneous
		17.1 Notices
		17.2 Costs
		17.3 Costs of Litigation
		17.4 Right of Landord to Make Rinos
		17.6 Reservations
	ť	17.7 Easements
	1	17.8 Power of Attorney
		Kelfandas A. R. Machae W

(iv)
------

18.	Guarantees	37
	18.1 Bank Guarantee	
	18.2 Issue and Review of Bank Guarantee	
	18.3 Personal Guarantee	
	18.4 Guarantee and Indemnity	38
	18.5 Indemnity	38 ·
	18.6 Liability of Guarantor	
	18.7 Irrevocable	
	18.8 Guarantor Liable Regardless of Any Law	
	18.9 Indemnity on Disclaimer	
	18.10 Guarantor Not Prove in Liquidation	
	18,11 Guarantee to Continue	37
19.	Option of Renewal	40
	19.1 Option	40
20.	Option of Renewal	40
	20.1 Option	40
	20.2 Further Lease	
	20.3 Omission of this Clause 19	
	20.4 Parties to Sign Further Lease	
	20.4 Farmes to Sign Further Lease	4T

٦

Kulada J. P. wlackae W

<del>DEE</del>E MEMORANDUM OF LEASE dated the **ANNEXURE TO** dav of One thousand nine hundred and from ISIKIT PTY LIMITED ACN: 002 455 218 to K C MACRAE HOLDINGS PTY LIMITED (hereinafter referred to as "the Lease")

goge is an ralia and New und Banking m Limited form itini dan

(2) Where the (2) where inc mortgage was given to one of the Banka named, delete reference to the other Bank and initial the alteration.

ACN: 001 772 514 AUSTRALIA AND NEW ZEALAND BANKING GROUP LIMITED A.C.N. 005 357 522 (hereinafter (1) Delete the words A referred to as "the Bank") being registered as the Mortgagee (1)(2) (formorly Australia and the set of the se New Zealand Bank Limited/The English Scottish and Australian Bank-Limited) under A Instrument of Mortgage No. I 191287 (1) (which has been vested in the Bank by virtue of the Australia and New Zealand Banking Group Act, 1970) of the premises the subject of the Lease (hereinafter referred to as "the Premises") HEREBY CONSENTS to the granting of the Lease subject to the following conditions and provisions namely that in the event of the Bank giving notice to the Lessee under the Lease by leaving the same at the Premises requiring the Lessee to pay the rent due under the Lease direct to the Bank the Lessee shall immediately thereupon enter into covenants with the Bank similar in all respects to the covenants with the Lessor expressed or implied in the Lease so that the Bank shall have the same rights and remedies as the Lessor by virtue of the Lease but subject to the Lessee's rights of quiet enjoyment of the Premises shall not be under any liability to observe or perform the covenants and agreements entered into by the Lessor and in the event of the Lessee failing to confer upon the Bank or its assigns the said rights and remedies the consent of the Bank to the granting of the Lease shall be deemed null and void and the rights of the Bank as Mortgagee shall be capable of being enforced without reference to the Lease. The Lease must not be varied rescinded or terminated without the written consent of the Bank.

> DATED at Sydney this 8TH

> > A

day of OCTOBER 1993

SIGNED in my presence by

GARRY CYRIL TETLEY

as the Attorney of AUSTRALIA AND **NEW ZEALAND BANKING GROUP** LIMITED he being personally known to me

Signature of Witness

LEANNE MARIE VELLA

Name of Witness (Block letters)

2964-12/92 FORM \$6/171

# AUSTRALIA AND NEW ZEALAND BANKING GROUP LIMITED

# By its Attorney

And I, the said Attorney, state that I have not received any Notices of the Revocation of the Power of Attorney Registered Book 4001 No. 878 under the authority of which I have just executed this document.

Detlen

Manager Branch Securities Department/District Manager

for the time being of Australia and New Zealand Banking

Group Limited for New South Wales.

Form: 07L Licence: 01-08-084 Licensee: Midware S Champion Legal	Systems Real Property Act 1900	
	PRIVACY NOTE: this information is legally required and will b	AB803964E
STAMP DUTY .	Office of State Revenue use only	NEW SOUTH WALES DUTY 16-08-2005 0002886494-001 LEASE - GENERAL DUTIABLE ANOUNT \$ \$\$\$\$\$\$\$\$337,318.00 DUTY
(A) TORRENS TITLE	Property leased: if appropriate, specify the part or premises	
	Part Folio Identifier B/410947 being Premises known as Sho Wentworthville	ops 10-12/108-120 Station Street
(B) LODGED BY	Deli 256 Name, Address or DX and Telephone Box CHAMPION LEGAL, Line 123167 DX 8220 PARRAMATTA Tel. (02) 9635.8266 Reference (optional): MCD:JN 50050 (1971)	
(C) <b>Lessor</b>		
	MICHAEL ELIAS & SAMIRA ELIAS	
(D)	The lessor leases to the lessee the property referred to above. Encumbrances (if applicable):	
	KATHIRVEL VEL SATHA SHAM	
(G) 1. TERM:	FIVE (5)YEARS	
	GDATE: 1 July 2005 GDATE: 30 June 2010	
<ol> <li>With an OPTI set out in clau</li> <li>With an OPTI</li> <li>Together with</li> <li>Incorporates in</li> </ol>	ON TO RENEW for a period of Five years	
No. N.A. 9. The <b>RENT</b> is a		PQI

83 Req:Ç145583 /Doc:DL AB803964 /Rev:30-Sep-2005 /Sts:OK.OK /Prt:27-Mar-2015 10:20 /Pgs:ALL /Seq:2 of 34 Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

1,7,05 DATE

(H) I certify that the person(s) signing opposite, with whom I am personally acquainted or as to whose identity I am otherwise satisfied, signed this instrument in my presence.

Signature of witness: Helean Name of witness: Annette ETics

Address of witness:

35 Coloola Rd Wentwoithille NSW 2145

Certified correct for the purposes of the Real Property Act 1900 by the lessor.

Signature of lessor:

& Elias

Note: where applicable, the lessor must complete the statutory declaration below

Certified correct for the purposes of the Real Property Act 1900 by the lessee,

Signature of FAJ PALANY 3/245 Castlerajh St Sydrey, NSW 2000

in expired lease No.

I certify that the person(s) signing opposite, with whom I am personally acquainted or as to whose identity I am otherwise satisfied, signed this instrument in my presence.

Signature of witness:

Name of witness:

Address of witness:

STATUTORY DECLARATION **(I)** 

# I,

solemnly and sincerely declare that -

1. The time for the exercise of option to

2. The lessee under that lease has not exercised the option.

I make this solemn declaration conscientiously believing the same to be true and by virtue of the provisions of the Oaths Act 1900.

Made and subscribed at	in the State of New South Wales
on	in the presence of:
Signature of witness:	Signature of lessor:
Name of witness:	

Address of witness:

Qualification of witness:

has ended:

**ANNEXURE "A" AND SCHEDULE** 

LANDLORD: MICHAEL ELIAS & SAMIRA ELIAS

TENANT: KATHIRVEL VEL SATHASEVAM

PREMISES: SHOPS 10 -12/108-120 STATION STREET WENTWORTHVILLE

#### 1. COVENANTS NEGATIVED

The covenants, powers and provisions implied in Leases by Sections 84 and 85 of the Conveyancing Act, 1919 are hereby expressly negatived except to the extent to which they are or may be included in the covenants hereinafter contained.

#### 2. TERM

# **Term of Lease**

- Subject to the provisions of this Lease, the Lease commences on the Commencement 2.1Date as set out in Item 1 (a) of the Schedule and ends on the Expiry Date as set out in Item 1 (b) of the Schedule.
- 3. RENT

# Rent

- 3.1 The Tenant covenants with the Landlord to pay free from all deductions the annual sum set out in Item 2 (a) of the Schedule for rent for the first year of the term by monthly payments set out in Item 2 (b) of the Schedule each in advance.
- 3.2 The Tenant must pay the first instalment on or before the Commencement Date.
- 3.3 The Tenant shall not withhold or be entitled to withhold the whole or part of any such payment by way of deduction, set off or counter claim in respect of any claim for damages or for compensation which the Tenant shall make or has made against the Landlord.

# **Review of Rent**

3.4. As and from each anniversary of the commencement date hereof (such anniversary occurring during the term or during any holding over period) the yearly rental for the ensuing year shall be the rent paid for the preceding year increased as set out in Item 3 of the Schedule.

DElias 3.F34

JN: M:\Docs\50050\193944.doc

# Monthly tenancy

,

.

- 3.5 Should the Tenant continue to occupy the Premises beyond the Expiry Date with the Consent of the Landlord otherwise than pursuant to a further lease granted by the Landlord, the Tenant shall do so as a monthly tenant only, on the terms and conditions of this Lease with necessary changes applicable to a monthly tenancy except that:
- 3.5.1 the monthly rent will be equal to one month's proportion of the rent hereby reserved increased annually by the percentage set out in Item 4 of the Schedule.
- 3.5.2 the first payment of rent is to made on the next day after the Expiry Date and thereafter payable monthly in advance
- 3.5.3 the tenancy may be determined at the will of either the Landlord or Tenant by one month's Notice in writing expiring on any day of the month and the tenancy will end one month after service of that Notice.

# 4 OUTGOINGS

# 4.1 Part contribution to outgoings

The Tenant will in addition to the rental reserved hereunder pay to the Landlord within fourteen days of the service upon the Tenant of a Notice of Contribution the percentage set out in Item 5 of the Schedule of the outgoings which have been assessed charged payable or imposed on or in respect of the Building or Land of which the Premises forms part AND for the purpose of this Lease the "outgoings" shall include but not be limited to:

- 4.1.1 council rates and charges.
- 4.1.2 water sewerage and drainage rates and charges payable to Sydney Water.
- 4.2 Estimate of outgoings

The Tenant's contribution to the outgoings of the estate shall be payable in the following manner:

- 4.2.1 The Landlord must give the Tenant a written Estimate of the outgoings to which the Tenant contributes under the Lease, itemising those outgoings under the item descriptions used in the List of Outgoings in the form of Disclosure Statement tendered to the Tenant before the commencement of this Lease.
- 4.2.2 The Estimate of outgoings must be given to the Tenant in respect of each accounting period of the Landlord during the term of the Lease at least one month before the commencement of the accounting period concerned.

JN: M:\Docs\50050\193944.doc

40834

3

4.2.3	The Landlord must make a written Expenditure Statement availab for examination by the Tenant detailing all expenditure by the Landlord on account of outgoings to which the Tenant contributes itemising those outgoings under the item descriptions used in the of Outgoings in the form of Disclosure Statement tendered to the Tenant before the commencement of this Lease.	
4.2.4	The Expenditure Statement must be made available annually during the Landlord's accounting periods during the term of the Lease (once	

the Landlord's accounting periods during the term of the Lease (once in relation to expenditure during the first six months of each such accounting period and once in relation to expenditure during the second six months of each such accounting period), and in each case must be made available within one month after the end of the six month period to which it relates.

# 4.3 Adjustment of contributions to outgoings

- 4.3.1 Within one month after the end of each period in respect of which the Tenant contributes to outgoings under the Lease, there is to be an adjustment between the Landlord and the Tenant to take account of any under-payment or over-payment by the Tenant in respect of those outgoings.
- 4.3.2 The adjustment is to be calculated on the basis of the difference between the total amount of outgoings in respect of which the Tenant contributed (that is, the estimated total expenditure by the Landlord on outgoings during the period concerned) and the total amount actually expended by the Landlord in respect of those outgoings during that period, but taking into account only expenditure properly and reasonably incurred by the Landlord in payment of those outgoings.

# Telephones, electricity, etc

4.4 The Tenant will pay to the proper authorities all charges for both rental and use of the telephone services for gas and electricity consumed on the Premises and for all water and sewerage usage charges, excess water grease trap fees and charges and all meter rents and if the Tenant makes default in the payment thereof it shall be optional for the Landlord to pay the same and in addition to the Landlord's remedies the Landlord may recover any amounts to be paid as if the same were overdue rent hereunder.

# **Registration of Strata Plan**

4.5 Should during the term of this Lease or any renewal thereof or during any holding over period a strata plan be registered in respect of the Land of which the Premises forms part so that the Premises becomes a lot in the strata plan then in addition to the rent payable hereunder the Tenant shall forthwith and in any event no longer than fourteen days after demand shall have been made therefore pay to the Landlord the following outgoings payable with respect to the Premises:

JN: M:\Docs\50050\193944.doc

1

5 of 34

4

4.5.1 Council rates and charges.

1

i.

- 4.5.2 Water sewerage and drainage rates and charges payable to Sydney Water.
- 4.5.3 Land tax PROVIDED that this shall be calculated on the basis that the Premises is the only land in respect of which the Landlord is required to lodge a Return for land tax purposes and the Land concerned was not subject to a special trust (within the meaning of the Land Tax Management Act 1956) and the Landlord was not a company classified under Section 29 of that Act as a non-concessional company.
- 4.5.4 Insurance premiums payable in respect of each risk or insurable interest which the Landlord may in its discretion effect.
- 4.5.5 all levies or contributions payable to the Owners Corporation (created upon registration of the strata plan in respect of the Property of which the Demised Premises forms part) pursuant to Part 3 Division 2 of the Strata Schemes Management Act 1996 or which may otherwise be properly raised by the Owners Corporation against the Landlord.

# Abatement of rent

- 4.6 The Lessee covenants with the Lessor to pay the initial rent herein referred to and such additional rent as may be payable from time to time pursuant to this Lease at the times and in the manner herein provided without any deduction or abatement whatsoever except as may be hereinafter provided that:
- 4.6.1 If after the commencement of the said term the Demised Premises or any part thereof be destroyed or damaged so as to be unfit for occupation or use or is inaccessible due to that damage the rent hereby reserved and any amount payable to the Lessor in respect of outgoings or a fair proportion thereof according to the nature and extent of the damage sustained shall be suspended until the Demised Premises shall have again been rendered fit for occupation or use.
- 4.6.2 (a) If the Lessor notifies the Lessee in writing that the Lessor considers that the damage is such as to make its repair impracticable or undesirable, the Lessor or the Lessee may terminate the Lease by giving not less than seven days' Notice in writing to the other and no compensation is payable in respect of that termination.
  - (b) If the Lessor fails to repair the damage within a reasonable time after the Lessee requests the Lessor in writing to do so, the Lessee may terminate the Lease by giving not less than seven days' Notice in writing of termination to the Lessor.
  - (c) the aforesaid provisions do not affect any right of the Lessor to recover damages from the Lessee in respect of any damage or destruction to which those provisions apply.

JN: M:\Docs\50050\193944.doc

6 of 34

- 4.6.3 Nothing in this section is to limit any liability of a party to the Lease to pay compensation to another party to the Lease in respect of damage to the Premises or the Building of which the Premises forms part.
- 4.6.4 Nothing in this section prevents the parties to a Lease from terminating the Lease by agreement if the Premises or the Building of which it forms part is damaged or destroyed.
- 4.6.5 If after the commencement of the renewed term the Demised Premises or any part thereof be adversely impacted due to redevelopment of the Building so as to be unfit or restricted for occupation or use or is inaccessible due to building works the rent hereby reserved and any amount payable to the Lessor in respect of outgoings or a fair proportion thereof according to the nature and extent of the redevelopment sustained shall be suspended until the Demised Premises shall have again been rendered fit for occupation or use.

# **BREACH OF ESSENTIAL TERMS**

- 5.1 In the event of a breach of an essential term of this Lease and if such breach shall continue for a period of seven (7) days after Notice in writing to the Tenant by the Landlord of such breach then thenceforth it shall be lawful for the Landlord or any person or persons duly authorised by the Landlord in that behalf to enter into or upon the Premises (forcibly if necessary) or any part thereof in the name of the whole and to repossess and enjoy the same as if these presents had not been made without prejudice to any right of action or remedy of the Landlord in respect of any antecedent breach by the Tenant of any essential term hereof.
- 5.2 Notwithstanding anything herein contained upon the Landlord becoming entitled to re-enter or determine this Lease the residue of the term hereof for the time being unexpired shall at the option of the Landlord immediately upon Notice of the exercise of such option being given by the Landlord to the Tenant become reduced to one week and thereafter the tenancy hereby created shall be and remain a tenancy from week to week at a weekly rate equal to one week's proportion of the rent hereby reserved (as adjusted hereunder from time to time) commencing from the date of service of such Notice and shall be determinable by either the Landlord or the Tenant giving to the other one week's Notice in writing expiring on any day of the week.
- 5.3 Upon the expiration or sooner determination of the term of this Lease the Tenant shall surrender to the Landlord all keys giving access to all parts of the Premises held by the Tenant the Tenant's servants or invitees irrespective of whether or not the same have been supplied by the Landlord, and remove at the Tenant's expense all lettering and distinctive marks or signs put up by the Tenant or by the Landlord for the Tenant on any part or parts of the Premises or the Building.

JN: M:\Docs\50050\193944.doc
- 5.4 Each covenant by the Tenant which is specified in this paragraph is an essential term of this Lease:
- 5.4.1 The covenant to pay rent and outgoings throughout the Lease term and any holding over period not later than the due date for the payment of each instalment of rent (Clause 2.1);
- 5.4.2 Clause 6.1 & 6.3; (To Repair);
- 5.4.3 Clause 7.1; (Alterations to Premises);
- 5.4.4 Clause 10; (Use of Premises);
- 5.4.5 Clause 13; (Assignment of Lease);
- 5.5 In respect of the Tenant's obligation to pay rent, the acceptance by the Landlord of arrears or of any late payment of rent shall not constitute a waiver of the essentiality of the Tenant's obligation to pay rent in respect of those arrears or of the late payments or in respect of the Tenant's continuing obligation to pay rent during the Lease term.
- 5.6 The Tenant covenants to compensate the Landlord in respect of any breach of an essential term of this Lease and the Landlord is entitled to recover damages from the Tenant in respect of such breaches. The Landlord's entitlement under this Clause is in addition to any other remedy or entitlement of the Landlord (including to terminate this Lease).
- 5.7 In the event that the Tenant's conduct (whether acts or omissions) constitutes a repudiation of the Lease (or of the Tenant's obligations under the Lease) or constitutes a breach of any Lease covenants, the Tenant covenants to compensate the Landlord for the loss or damage suffered by reason of the repudiation or breach.
- 5.8 The Landlord shall be entitled to recover damages against the Tenant in respect of repudiation or breach of covenant for the damage suffered by the Landlord during the entire term of this Lease.
- 5.9 The Landlord's entitlement to recover damages shall not be affected or limited by any of the following:
- 5.9.1 If the Tenant shall abandon or vacate the leased Premises;
- 5.9.2 If the Landlord shall elect to re-enter or to terminate the Lease;
- 5.9.3 If the Landlord shall accept the Tenant's repudiation;
- 5.9.4 If the parties conduct shall constitute a surrender by operation of law.
- 5.10 The Landlord shall be entitled to institute legal proceedings claiming damages against the Tenant in respect of the entire Lease term, including the periods before

and after the Tenant has vacated the lease Premises and before and after abandonment, termination, repudiation, acceptance of repudiation or surrender by operation of law referred to in clause 5.9, whether the proceedings are instituted either before or after such conduct.

5.11 In the event of the Tenant vacating the leased Premises, whether with or without the Landlord's Consent, the Landlord shall be obliged to take reasonable steps to mitigate his damages and to endeavour to lease the Premises at a reasonable rent and on reasonable terms. The Landlord's entitlement to damages shall be assessed on the basis that the Landlord should have observed the obligation to mitigate damages contained in this paragraph. The Landlord's conduct taken in pursuance of the duty to mitigate damages shall not by itself constitute acceptance of the Tenant's breach or repudiation or a surrender by operation of law.

## **TENANT'S OBLIGATIONS**

ı.

,

### 6. **Positive use obligations**

The Tenant must:

## 6.1 Maintenance and Repair

- 6.1.1 Repair and to keep the Premises, the exterior facade and all additions thereto and the Landlord's fixtures thereon and all drains soil and other pipes sewers sanitary and water apparatus glass pavings wall fences and railing and appurtenances in good and substantial repair and maintained paved and cleansed in every respect (excepting fair wear and tear damage by fire explosion earthquake aircraft flood tempest Act of God riot and civil commotion save where any insurance moneys are irrecoverable through the neglect and default or misconduct of the Tenant) AND ALSO will and so often as it shall be necessary to renew any fixtures belonging to the Premises or substitute other fixtures of a similar description and value to the satisfaction of the Landlord except where such substitution is due to damage by fire explosion earthquake aircraft flood tempest Act of God riot or civil commotion save where any insurance moneys are irrecoverable through the neglect and default or misconduct of the Tenant.
- 6.1.2 Paint the whole of the Premises, being the interior and exterior (if applicable), as reasonably required and prior to the expiration of the Lease.
- 6.1.3 Where applicable, clean and pump out every six months the inlet and drainage lines leading the grease arrester and the outlet pipes from the arrester to the Sydney Water Sewerage pipelines.

### 6.2 Replace fixtures

Replace from time to time during the said term such of the Landlord's fixtures and fittings as may have become worn out or unfit for use (excepting fair wear and tear) or lost by substituting others of a like quality and in new condition.

9 of 34

# 6.3 Enter renew and repair

- 6.3.1 The Landlord may by itself or through an independent property consultant enter upon the Premises every six months upon giving 48 hours notice (except in case of emergency as to the existence of which the unfettered opinion of the Landlord shall be conclusive, when no Notice shall be required) and view the state of repair thereof and the Landlord may require the Tenant by Notice in writing served upon the Tenant to rectify any defect (excluding latent defects and defects not arising from the act or omission of the Tenant its servants or agents) or carry out any repairs within 30 days in accordance with any covenant expressed or implied in the Lease.
- 6.3.2 In default of the Tenant repairing any defects according to the Notice the Landlord may from time to time enter the Premises and execute and carry out the required repairs as if it were the Tenant and for that purpose the Landlord its architects contractors and workmen and agents may enter upon the whole or any part of the Premises and there remain for the purpose of doing executing and effecting and carrying out any such repair AND the expenses and costs of carrying out such work shall forthwith be payable by the Tenant to the Landlord.
- 6.3.3 The Tenant will permit the Landlord upon 48 hours Notice (except in case of emergency as to the existence of which the unfettered opinion of the Landlord shall be conclusive when no Notice shall be required) to enter execute and carry out repairs renovations maintenance or alterations to the Premises or to any part thereof or to the Common Areas or any part thereof in compliance with the Landlord's obligations under the provisions of this Lease or otherwise deemed necessary or desirable by the Landlord and for the purpose the Landlord its architects, contractors and workmen and agents may enter upon the whole or any part of the Premises and there remain for the purpose of doing executing effecting and carrying out such repairs renovations maintenance or alteration PROVIDED ALWAYS THAT in the exercise of such power no undue inconvenience is caused to the Tenant.
- 6.3.4 If at any time any authority having jurisdiction or authority over or in respect of the Premises or the user thereof requests or requires any alterations, or additions, conversions, improvements or other works to be made to the Premises which the Landlord elects to do and for which the Tenant is not liable under its covenants herein contained the Landlord its architects contractors workmen servants and others and the servants and workmen of any of its contractors may enter into and on the Premises and there remain at all times for the purpose of making any such alterations additions conversions improvements or other works or any of them as aforesaid PROVIDED ALWAYS THAT in the exercise of any such power no undue inconvenience is caused to the Tenant.

10 of 34

# 6.4 **Plant and equipment**

Keep any air conditioning and ventilating equipment and all other plant and equipment installed in the Premises during the continuance of this Lease in proper working condition.

## 6.5 Pests

Take all proper precautions to keep the Premises free of white ants borers vermin and other like pests in particular to ensure that the Premises shall not be or create a hazard to health or otherwise infringe the provisions of Public Health Act or other legislation the provisions of which may be applicable thereto from time to time.

# 6.6 **Compliance with the Statutes**

Comply with all Statues, Ordinances, Proclamations, Orders, Regulations or Notices relating to the Premises and do and execute or cause to be done or executed all such works acts deeds matters and things as under or by virtue of any of the said Statutes, Ordinances, Proclamations, Orders, Regulations and Notices for the time being in force or shall be properly directed or necessary to be done or executed upon or in respect of the Premises or any part thereof whether by the owner Landlord Tenant or occupier in consequence of the occupation and use of the Premises by the Tenant.

# 6.7 Inspection by prospective Tenants

At all reasonable times permit the Landlord to exhibit to prospective tenants or purchasers the Premises and will at all times within the three months immediately preceding the termination of the this Lease allow the Landlord to affix and exhibit where the Landlord shall think fit at any time the usual "For Sale" and the usual "To Be Let" Notice and in each case with the name and address of the Landlord and/or its agent thereon and the Tenant will not remove any such Notice without the written Consent of the Landlord.

# 7. Negative Use Obligations

The Tenant must not:

# 7.1 Alterations

Without the previous written Consent of the Landlord (which shall not be unreasonably withheld):

11 of 34

- 7.1.1 erect permit or suffer to be erected or carried out any structural alteration or works on the Premises PROVIDED that if any structural alterations or works shall be carried out by the Tenant with the Consent of the Landlord as aforesaid the Tenant shall at the expiration of the term hereof (if so required by the Landlord) restore the Premises to the state and condition in which they were prior to the carrying out of such works having regard to the condition of the Building through fair wear and tear during the term hereof.
- 7.1.2 change the exterior colour of the Premises or of the Building in which the same are situate or any part thereof from the colour nor the size location or composition of any sign symbol or advertisement that may have been approved by the Landlord.
- 7.1.3 install any exterior lighting or plumbing facilities shades canopies awnings or window boxes or amplifiers or similar devices or use any advertising medium which may be heard or experienced outside the Premises such as loud speakers, tapes, CDs, television, video or radio broadcasts.
- 7.1.4 unless herein provided, mark paint drill write upon or in any way deface any wall ceiling floor wood stone or ironwork of the Premises.
- 7.1.5 in any manner to interfere with or attempt to control any part of the fire alarm or sprinkler systems or any other machinery of the Landlord of any kind now installed or hereafter to be installed in the Premises PROVIDED HOWEVER and it is hereby specifically agreed by and between the parties hereto that the Landlord shall not be responsible for and shall incur no liability for failure of any of the aforesaid equipment or machinery or for the ineffectual operation thereof or for any damage or loss occasioned or arising in connection with the same to the Tenant or to any person claiming by through or under the Tenant and the Tenant hereby indemnifies and agrees to keep indemnified the Landlord against all or any such claim.

## 7.2 Additions

,

Should the Landlord at the request of the Tenant construct or erect or permit the Tenant to construct or erect any walls, partitions or ceilings within the Premises whether at the cost of the Tenant or not the Tenant shall not be entitled to remove such walls partitions or ceiling on vacating the Premises (but the same shall remain on the Premises for the benefit of but at no cost to the Landlord).

12 of 34

### 7.3 Advertisement

.

Not affix or exhibit or suffer to be affixed or exhibited to or on any part of the exterior of the Premises any advertisements signs symbols posters placards neon lights or other light or other object or thing other than such as may relate solely to the Premises or some business lawfully carried on there and so that the same shall be displayed only in such position on and height about the Premises and be of such materials colour dimensions and general character as shall be first approved in writing by the Landlord nor hang place deposit or expose or permit or suffer any hanging placing depositing or exposing outside any part of the Buildings comprised in the Premises any goods articles or thing for sale.

### 7.4 Water closets

Not use the water closets or other water apparatus for any other purpose than those for which they are constructed and no sweepings, rubbish, rags, trade effluent or deleterious matter of any description shall be thrown into water closets sewers drains or watercourses serving the Premises, and any damage thereby resulting to such water closets or sewers drains or watercourse and the cost of repairing the same shall be borne by the Tenant.

### 7.5 Nuisance

Generally not do or permit or suffer to be done or permitted upon or in connection with the Premises anything which shall or tend to be a nuisance annoyance or cause of damage to the Landlord or to any adjoining or neighbouring property or to the owner or occupier thereof and to pay the Landlord all costs charges and expenses which may be incurred by the Landlord in abating any such nuisance and for executing all works as may be necessary to comply with any Notice served by a local or public authority or neighbouring premises for abating such nuisance created as a consequence of the occupation and use of the Premises by the Tenant.

### 7.6 Heavy objects

Not do or bring or permit or suffer to be done or to be brought in or upon the Premises anything which may be put on the Premises or any Buildings of which the same form part any weight or impose any strain in excess of that which such Premises are calculated to bear with due margin for safety and in particular not to suspend any weights from the roof trusses.

#### 7.7 **Obstruction of windows**

Not obstruct or permit or suffer to be obstructed any of the windows, doorways, lights or ventilators belonging to the Premises.

13 of 34

### 7.8 Rubbish

- 7.8.1 Not permit any rubbish or garbage to accumulate on the Premises or any part thereof unless confined in suitable containers so located as not to be visible to members of the public visiting the Premises.
- 7.8.2 Dispose of all rubbish, garbage and trade waste by way of minimum weekly domestic, trade or commercial disposal services in accordance with all laws, ordinances and local council policies and as approved or reasonably instructed by the Landlord.

## 7.9 Development Application

Not do or omit or permit or suffer to be done or omitted anything on or in connection with the Premises the doing or omission of which shall be a contravention of any Act of Parliament or Ordinance or the conditions of any development application approval or of any Notices Orders Licences Consents Permissions and Conditions (if any) served made granted or imposed thereunder and to indemnify the Landlord against all actions proceedings damages penalties costs charges claims and demands in respect of such acts and omissions or any of them except as may be caused by or result from the acts or omissions of the Landlord, its agents or employees, and against the cost of any application for planning permission and the works and things done in pursuance thereof.

### 7.10 Auction sale

Not conduct or permit an auction sale in the Premises.

### 7.11 Driveways and car parking

Not park or allow motor vehicles to be parked in the driveway nor to obstruct the driveway and not to park nor allow any employees or invitees of the Tenant to park motor vehicles on the Premises other than in allocated motor vehicles parking spaces.

## 8 Indemnities

8.1 The Tenant agrees to:

- 8.1.1 Occupy, use and keep the Premises at the risk of the Tenant and hereby releases to the full extent permitted by law the Landlord and its contractors and employees in the absence of any negligence on their part from all claims and demands of every kind and from all liabilities which may arise in respect of any accident or damage to property or injury to any person in or near the Premises or the Building and the Tenant expressly agrees that in the absence of any such negligence as aforesaid the Landlord shall have no responsibility or liability for any loss or damage to fixtures or personal property of the Tenant.
- 8.1.2 Hereby indemnify the Landlord and its contractors and employees in the absence of any negligence on their part from and against all actions claims

17 of 34

JN: M:\Docs\50050\193944.doc

demands losses damages costs and expenses incurred by the Landlord or for which the Landlord may become liable in respect of damage to property or injury to any person which may be suffered or sustained in upon or near any part of the Premises or the Building whether in the occupation of the Landlord or of the Tenant or of any other person.

- 8.1.3 Without limiting the generality of clauses 8.1.1 and 8.1.2 the Tenant will and does hereby indemnify the Landlord from and against all claims, actions, demands, losses, damages, costs and expenses incurred by the Landlord or with respect to which the Landlord may become liable in respect of or arising from:
- 8.1.3.1 the negligent or careless use, misuse waste or abuse by the Tenant or any contractor, sub-contractor, licensee, invitee, client customer or visitor of the Tenant or any other person claiming through or under the Tenant of the water, gas, electricity, lighting and other installations in and facilities of the Premises or the Building or arising from any faulty fittings or fixtures of the Tenant;
- \*8.1.3.2 overflow or leakage of water (including rain water) in or from the Premises or caused or contributed to by any act or omission on the part of the Tenant or any other person referred to in sub-clause 8.1.3.1 of this clause;
- 8.1.3.3 loss, damage or injury from any cause whatsoever to the Premises or the Buildings or to any property or person within or without the Premises or the Building occasioned or attributed to by any act, omission, neglect, breach or default by the Tenant or any other person as aforesaid.
- 9. Costs

.

.

The Tenant shall pay 50% of Landlord's legal costs, and shall pay 100% of any additional Landlord legal costs in respect to any amendments, all stamp duty, charges and expenses of and incidental to the preparation, completion stamping, registration of, and the obtaining of any Mortgagee's approval to, this Lease; and of any assignment or sub-letting and of any surrender or other termination of this Lease otherwise than by effluxion of time; and in case of default by the Tenant in performing or observing any covenants herein contained or implied the Tenant shall pay to the Landlord all legal costs and other costs, charges and expenses for which the Landlord shall become liable in consequence of or in connection with such default within seven days of the Landlord requesting him so to do.

### 10. Use of Premises

During the term hereby the Tenant will not at any time use the Premises other than as set out in Item 6 of the Schedule.

15 of 34

### DAMAGE TO PREMISES

.

### Damages

- 11.1 Should any damage occur to the Premises the Tenant shall forthwith give Notice thereof in writing to the Landlord and in particular upon the happening thereof to give to the Landlord Notice of any accident to or defects in the water pipes, gas pipes, lights or fittings used in connection with the water, gas or electrical services and the Tenant shall forthwith make good and repair any such damage or defect to the extent that the Tenant may be required to do so pursuant to the provisions of this Lease.
- 11.2 If the Premises or any part thereof be destroyed or damaged by any of the insured risks and the insurance money under any of the Insurance Policies against the same effected by the Landlord being wholly or partly irrecoverable by reasons solely or in part by any act or default of the Tenant then and in every such case the Tenant will forthwith (in addition to the said rent and any other moneys payable hereunder) pay to the Landlord the whole or (as the case may require) a fair proportion of the cost (including professional and other fees) of completely rebuilding and reinstating the same.
- 11.3 If after the commencement of the said term the Premises or any part thereof be destroyed or damaged so as to be unfit for occupation or use or is inaccessible due to that damage the rent hereby reserved and any amount payable to the Landlord in respect of outgoings or a fair proportion thereof according to the nature and extent of the damage sustained shall be suspended until the Premises shall have again been rendered fit for occupation or use.
- 11.3.1 If the Landlord notifies the Tenant in writing that the Landlord considers that the damage is such as to make its repair impracticable or undesirable, the Landlord or the Tenant may terminate the Lease by giving not less than seven days' Notice in writing to the other and no compensation is payable in respect of that termination.
- 11.3.2 If the Landlord fails to repair the damage within a reasonable time after the Tenant requests the Landlord in writing to do so, the Tenant may terminate the Lease by giving not less than seven days' Notice in writing of termination to the Landlord.
- 11.3.3 the aforesaid provisions do not affect any right of the Landlord to recover damages from the Tenant in respect of any damage or destruction to which those provisions apply.
- 11.4 Nothing in this section is to limit any liability of a party to the Lease to pay compensation to another party to the Lease in respect of damage to the Premises or the Building of which the Premises forms part.
- 11.5 Nothing in this section prevents the parties to a Lease from terminating the Lease by agreement if the Premises or the Building of which it forms part is damaged or destroyed.

16 oF34

## 12. Insurance

.

- 12.1 The Landlord shall keep insured or cause to be kept insured at all times the Premises.
- 12.2 The Tenant shall effect and keep current during the term hereof and any holding-over hereunder a Public Risk Policy for an amount in respect of any single accident or event of not less than that set out in Item 7 of the Schedule. Such Policy shall indemnify the Landlord against any actions, suits, claims and demands of any kind arising from the use by the Tenant, its employees and visitors of the Buildings.
- 12.3 The Tenant shall not at any time during the said term, do, permit or suffer to be done any act, matter or thing whereby any insurance in respect of or in relation to the Premises or the Building or the Land may be vitiated or rendered void or voidable or (except with the approval of the Landlord in writing) whereby the rate of premium on any insurance shall be liable to be increased.
- 12.3.1 In the event the Tenant's use of the premises causes the rate of premium to increase, the Tenant will pay 100% of the increase of the premiums.
- 12.4 The Tenant shall not store chemicals, inflammable liquids or solids acetylene gas or alcohol, volatile or explosive oils, compounds or substances upon or about the Premises or elsewhere in the Building and will not use any of such substances of fluids in or about the Premises or elsewhere in the Building for any purpose and will not in any way create any actual or potential fire hazard in the Premises or elsewhere in the Building. The Tenant will permit the Landlord at all times to enter upon the Premises and to abate any actual or potential fire hazard in or about the Premises and if such hazard in the reasonable opinion of the Landlord, is caused by or results from the use or occupancy of the Premises by the Tenant, the Tenant shall pay to the Landlord on demand the costs and expenses incurred by the Landlord in carrying out such work.
- 12.5 The Tenant will insure in the joint names of the Landlord and the Tenant in the full replacement value thereof all plate glass windows, doors, display showcases and shelves against the risk of breakage, hail, windstorm, flood, riot and civil commotion, explosion, earthquake, cost of temporary shattering, accidental damage to window frames, showcase frames and fittings.
- 12.6 The Tenant shall from time to time as and when required by Notice in writing from the Landlord forthwith pay all extra premiums payable in respect of insurance on the Building and its contents if any be required on account of extra risk caused by the use to which the Premises are put by the Tenant or by the bringing or keeping on the Premises of any materials or substances.
- 12.7 All Policies of Insurance liable or required to be effected by the Tenant hereunder, whether in respect of the property or risk of the Landlord or the Tenant, shall be taken out with an insurance office or company approved of by the Landlord. The Landlord hereby declares that such approval will not be

170f34

withheld where the Tenant intends to effect such insurance with a reputable and responsible insurance company.

12.8 The Tenant will in respect of any Policy of Insurance to be effected by the Tenant hereunder, if so required, produce to the Landlord the Policy of Insurance and the receipt for the premium last paid in respect thereof.

#### ASSIGNMENT, SUB-LETTING, ETC

ł

- 13.1 The Tenant covenants not to assign this Lease except in accordance with the provisions contained in this Clause.
- 13.2 If the Tenant desires to assign this Lease:
- 13.2.1 the Tenant shall, before requesting the Landlord's consent, furnish the proposed assignee with a copy of any Disclosure Statement given to the Tenant in respect of the Lease, together with details of any changes that have occurred in respect of the information contained in that Disclosure Statement since it was given to the Tenant (being changes of which the Tenant is aware or could reasonable be expected to be aware). If the Tenant cannot provide the proposed assignee with a copy of the Disclosure Statement, the Tenant may request the Landlord to provide a copy of that Disclosure Statement and if the Landlord is unable or unwilling to provide that copy within fourteen days after the request, the Tenant need not comply with this paragraph;
- 13.2.2 the Tenant shall request the Landlord's Consent to the assignment of this Lease in writing and shall furnish with that request:
- 13.2.2.1 information regarding the financial resources and financial standing and the business experience and retailing skills of the proposed assignce;
- 13.2.2.2 particulars of the use of the Leased Premises intended by the proposed assignee;
- 13.2.2.3 confirmation that the Tenant has complied with Clause 13.2.1
- 13.3 The Landlord:
- 13.3.1 agrees to deal expeditiously with the Tenant's request for Consent to assign this Lease;
- 13.3.2 is entitled to require the Tenant to furnish to the Landlord:
- 13.3.2.1 details of the Tenant's financial resources and retailing skills at the time of the request for Consent;

18 of 34

· ,

13.3.2.2	such further information as the Landlord may reasonably require concerning the financial standing and business experience of the proposed assignee;
13.3.3	is entitled to withhold Consent to the assignment of this Lease in any of the following circumstances:
13.3.3.1	if the proposed assignce proposes to change the use to which the Premises are put (unless the Landlord consents to the change of use in accordance with the provisions of this Lease);
13.3.3.2	if the proposed assignee has financial resources and retailing skills that are inferior to those of the Tenant;
13.3.3.3	if the Tenant has failed to comply with the provisions contained in this Clause for requesting and obtaining Consent to the assignment.
13.4	For the purposes of the Tenant making a request and the Landlord considering that request under this Clause:
13.4.1	if the assignce is a corporation, the retailing skills of the assignce include the retailing skills of those of its directors who have actively participated in the assignce's business and intend to actively participate in the business in the Leased Premises;
13.4.2	if the Landlord Consents to a change of use by the assignce, the assignce's retailing skills for that use should be compared with the Tenant's retailing skills for the Tenant's use of the Leased Premises;
13.5.	The Landlord may require the assignce to execute an Instrument in which the assignce will assume liability under the Lease for the residue of the lease term:
13.5.1	may require, if the assignee is a corporation whose shares are not listed on an Australian Stock Exchange, the Landlord may require, as a condition of the Landlord's Consent to the assignment, that at least two of its directors or shareholders (chosen by the Landlord) execute personal Guarantees in respect of the payment of rent and the observance and performance of the Lease covenants by the assignee during the lease term. It is agreed that in any change in the principal shareholding altering the effective control of the Tenant (if a company) shall be deemed to be an assignment of this Lease.
13.5.2	will require the Tenant will pay the Landlord's reasonable legal and other expenses of investigating the Tenant's Application for Consent and of preparing, executing, stamping and registering all the required Instruments or Dealings and the costs of obtaining the Mortgagee's Consent.
13.6	If the Tenant has complied with the requirements of Clause 13.5, the Landlord shall indicate, within forty two days after the Tenant has made its request under Clause 13.2 and furnished any further details required under Clause 13.3, whether the Landlord consents or withholds consent. If the Landlord does not provide that
JN: M:\Docs\\$(	1050\193944.doc

19 of 34

indication within forty two days, it shall be taken to have consented to the assignment, subject to the conditions in Clause 13.5.

- 13.7 The Tenant will not during the continuance of this Lease transfer demise sub-let or part with possession of the Premises (or any part thereof) or by any act or deed procure the Premises (or any part thereof) to be transferred sub-let unto or put into possession of any person or persons without the previous Consent in writing of the Landlord. Such consent shall not be unreasonably withheld where:
- 13.7.1 the Tenant not being in default under the covenants and agreements on the Tenant's part herein contained proposes sub-let to a sub-Tenant who proves to the satisfaction of the Landlord that such proposed sub-Tenant is a respectable responsible, solvent, fit and proper person capable of adequately carrying on the business in the Premises whilst performing and observing the terms and conditions of this Lease and the Tenant must provide the Landlord with such information as the Landlord may reasonably require concerning the financial standing and business experience of the proposed assignee, and
- 13.7.2 the Tenant pays to the Landlord all proper costs, charges and expenses incurred by the Landlord of and incidental to any enquiries which may be made by or on behalf of the Landlord as to respectability, responsibility, solvency, fitness and suitability of any proposed sub-Tenant, and
- 13.7.3 the Tenant pays the Landlord's solicitors costs in respect of the sub-lease; and
- 13.7.4 the sub-Tenant being a company the directors and also the principal shareholders of such company unconditionally guarantee to the Landlord the performance by the Company of the terms and conditions contained in this Lease or in the proposed sub-lease (as the case may be). It is agreed that in any change in the principal shareholding altering the effective control of the Tenant (if a company) shall be deemed to be an assignment of this Lease.

### LANDLORD'S OBLIGATIONS

7

14.1 The Landlord HEREBY COVENANTS with the Tenant that the Tenant paying the rent hereby reserved and observing and performing the covenants conditions and agreements herein contained on the Tenant's part to be observed and performed shall and may quictly enjoy the Premises during the said term without any interruption by the Landlord.

### 14.2 Tenant to be given Notice of alterations and refurbishment

The Landlord must not commence to carry out any alteration or refurbishment of the Building of which the Premises forms part which is likely to adversely affect the business of the Tenant unless:

14.2.1 the Landlord has notified the Tenant in writing of the proposed alteration or refurbishment at least three months before it is commenced; or

20 of 34

.

14.2.2 the alteration or refurbishment is necessitated by an emergency and the Landlord has given the Tenant the maximum period of notice that is reasonably practicable in the circumstances.

#### 14.3 Disturbance

- 14.3.1 Subject to 14.3.3, the Tenant is to be compensated for disturbance if the Landlord:
- 14.3.1.1 inhibits access of the Tenant to the Premises in any substantial manner; or
- 14.3.1.2 takes any action that would inhibit or alter, to a substantial extent, the flow of customers to the Premises; or
- 14.3.1.3 unreasonably takes any action that causes significant disruption of, or has a significant adverse effect on, trading of the Tenant in the Premises; or
  14.3.1.4 fails to take all reasonable stens to prevent or put a stop to anything that
- 14.3.1.4 fails to take all reasonable steps to prevent or put a stop to anything that causes significant disruption of, or which has a significant adverse effect on, trading of the Tenant in the Premises and that is attributable to causes within the Landlord's control; or
- 14.3.1.5 fails to rectify any breakdown of plant or equipment under the Landlord's care or maintenance, and the Landlord does not rectify the matter as soon as reasonably practicable after being requested in writing by the Tenant to do so, the Landlord is liable to pay the Tenant reasonable compensation for any loss or damage (other than nominal damage) suffered by the Tenant as a consequence.
- 14.3.2 In determining whether a Landlord has acted unreasonably for the purposes of sub-clause 14.3.1.3, due consideration is to be given to whether the Landlord has acted in accordance with recognised shopping centre management practices.
- 14.3.3 In respect of all occurrences the likelihood of which have been specifically drawn to the Tenant's attention in the Disclosure Statement given to the Tenant before entering into this Lease, the Tenant is precluded from making any claim for compensation for any loss or damage suffered by the Tenant as provided in Section 34 of the Retail Leases Act 1994 and the Landlord is exempted from any liability for compensation. In particular the Tenant acknowledges that during the term, option to renew period or holding over period the Landlord may carry out a redevelopment of the site. This redevelopment may cause inconvenience, loss of Tenant's business and/or interruption to the Tenant's business as disclosed. The Tenant acknowledges that he is prevented from making a claim for compensation during the term or any renewed term or holding over period under Section 34 of The Retail Leases Act as the plan to carry out extensive redevelopment during the term or any renewed term or holding over period has been disclosed. The Landlord and the Tenant agree that the rent payable by the Lessee, shall be reduced to a figure that is agreed upon between the parties, during any period of redevelopment, provided that if the parties cannot reach agreement on the reduced rental then the dispute shall be determined in accordance with the provisions of clause 27 of the Lease.

AA: C:\DOCUME~I\amanda\LOCALS~I\Temp\Elias and Svaan- Amended lease- 18.7.05.doc

21 of 34

14.4.4	The provisions implied by this section do not apply to any action taken by the Landlord:				
14.4.4.1	as a reasonable response to an emergency situation; or				

14.4.2 in compliance with any duty imposed by or under an Act or resulting from a requirement imposed by a public or local authority acting under the authority of an Act.

### **Bond or Bank Guarantee**

,

.

- 15.1 On signing of the Lease the tenant must deposit with the Landlord the amount set out in Item 8 of the Schedule as security for the due and punctual observance and performance of all the covenants obligations and provisions of the Tenant's part contained herein.
- 15.2 If at any time the Tenant fails to so duly and punctually observe and perform the Tenant's covenants in the Lease, then the Landlord may in the Landlord's sole discretion at any time appropriate and apply so much of or the whole of the deposit as may be necessary in the opinion of the Landlord to compensate the Landlord for loss or damage sustained or suffered by the Landlord by reason of such breach by the Tenant.
- 15.3 Any such appropriation by the Landlord shall not prejudice any other right of the Landlord arising from such breach.
- 15.4 Should the deposit or any portion thereof be appropriated by the Landlord as aforesaid then the Tenant shall upon demand by the Landlord and within seven days from the date thereof pay to the Landlord the amount of the sum so appropriated.
- 15.5 Should the Tenant comply with all the said covenants obligations and provisions and duly and punctually pay all of the rental hereby reserved and all other sum payable by the Tenant to the Landlord hereunder the said deposit less any sums appropriated by the Landlord in accordance with this clause and not paid by or received from the Tenant shall be refunded to the Tenant on the expiration of the term of this Lease or of any extension or renewal thereof.
- 15.6 The Landlord shall be at liberty to pay the said deposit less any sums appropriated by the Landlord in accordance with this clause and not paid or received from the Tenant to any assignee or transferee of the Landlord's interest in the Premises in the event that such interest of the Landlord is assigned or transferred and thereupon the Landlord shall be discharged from all liability to the Tenant or any other person with respect to such deposit.
- 15.7 The deposit paid by the Tenant to the Landlord as security for the performance of the Tenant's obligations under the Lease must be held by the Landlord on behalf of the Tenant in an account bearing interest. The Landlord must account to the Tenant for interest earned on the deposit but the Landlord is entitled to retain any

22 of 34

such interest and deal with it as money paid by the Tenant to the Landlord to form part of the deposit concerned.

### Termination

,

.

- 16.1 At the expiration or sooner determination of the said term the Tenant must:
- 16.1.1 quietly yield up unto the Landlord the Premises together with all additions and improvements made thereto and all fixtures in or upon the Premises or which during the said term may have been affixed or fastened to or upon the same and in such state and condition as shall in all respects be consistent with full and due performance by the Tenant of the covenants on the part of the Tenant herein contained.
- 16.1.2 If so required by the Landlord, remove any wall paper, carpet, floor tiles or any other covering which may have been affixed to the Premises whether by the Tenant or not and whether with the Consent of the Landlord or not and to resurface the walls ceiling and floors, in a manner approved by the Landlord consistent with the nature and character of the Building.
- 16.1.3 Repair any damage caused to the Premises by the removal of any of the Tenants fixtures and in particular make good any damage to the walls and ceiling caused by the removal of the air conditioning unit in the restaurant area and the cool room.
- 16.1.4 Remove at the Tenant's own expense all signs, advertisements or Notices inscribed painted or affixed by the Tenant (or by the Landlord at the Tenant's request) on or in any part of the Building and shall make good to the satisfaction of the Landlord any damage or disfigurement existing after such removal. In the event that the Tenant shall fail to make good any such damage or disfigurement this may be made good by the Landlord and the Tenant shall pay to the Landlord the cost thereof within seven days of the Landlord notifying the Tenant in writing of the amount thereof.
- 16.1.5 In the event that any part of the Premises which has been previously painted has not been painted during the period of two years immediately preceding the expiration or sooner determination of the said term then the Tenant shall to the satisfaction of the Landlord paint every such previously painted surface with good quality paint to a standard and in colours approved by the Landlord consistent with the nature and character of the Building.

### NOTICES

- 17.1 Any Notice required to be given under this Lease shall be in writing.
- 17.2 Any Notice to the Tenant (if a Corporation) shall be sufficiently served if left addressed to the Tenant on the Premises or sent by registered post or left at its registered office of the Tenant and (if the Tenant shall be a person) shall be sufficiently served if left addressed to the Tenant on the Premises or sent by registered post or left addressed to the Tenant at the Tenant's last known address in Australia.

23 of 34

- 17.3 Any Notice to the Landlord (if a Corporation) shall be sufficiently served if left addressed to the Landlord at or sent by registered post to the registered office of the Landlord and if the Landlord shall be a person shall be sufficiently served if sent by registered post to the address of the Landlord stated herein or at such other address as the Landlord may have in writing stipulated to the Tenant.
- 17.4 Any Notice sent by registered post shall be deemed to have been duly served at the expiration of twenty-four hours after the time of posting and in proving service it shall be sufficient to prove that the envelope containing the Notice was duly addressed to the Tenant or the Landlord (as the case may be) in accordance with this Clause and left at or posted to the address to which it is so addressed.

### **OPTION TO RENEW**

.

### 18. Tenant's option

- 18.1 The Landlord must grant a new Lease to the Tenant, at the cost of the Tenant on the Expiry Date to commence on the next day only if:
- 18.1.1 the Tenant gives the Landlord notice in writing no less than 6 months and no more than 9 months before the Expiry Date stating that it wants a new Lease for a further term set out in Item 9 of the Schedule
- 18.1.2 the Tenant in the meantime continues to duly and punctually pay the rent reserved by this Lease and duly observe and perform the covenants, obligations and agreements in accordance with the terms of the Lease
- 18.1.3 when the Tenant gives the Notice and on the Expiry date, the Tenant is not in breach of this Lease

#### New Lease

- 18.2. The new Lease is to be identical with this Lease except that:
- 18.2.1. this clause 18 is to be deleted, and
- 18.2.2. the term, commencement date, expiry date and review dates are to be as set out in item 10 of the schedule
- 18.2.3 the rent from the commencement date of the new Lease will be determined in accordance with clause 18.3.

#### **Market Review**

18.3 The Landlord shall within thirty five days of the receipt of Notice to be given by the Tenant in clause 18.1 inform the Tenant in writing of the Landlord's assessment of the rent for the first year of the new term.

18.3.1 The Tenant may accept the Landlord's assessment and in that event the Tenant fails or refuses to accept the Landlord's assessment within twenty eight days JN: M:\Docs\50050\193944.doc

after service of the assessment or if the Landlord and Tenant are unable to agree on some other amount as the current market rent for the first year of the new term the current market rent shall be determined in accordance with this clause 18.3.

- 18.3.2 The Parties must apply within 14 days to the President or the Principal Officer for the time being of the Australian Property Institute Inc (New South Wales Division) ("the Nominator") to nominate a person ("the Nominee") who is a licensed valuer and
- 18.3.2.1 has been a full Member of not less than ten years standing of the Australian Property Institute Inc (New South Wales Division);
- 18.3.2.2 must be the holder of a licence to practice as a Valuer of the kind of premises by this Lease;
- 18.3.2.3 must have at least ten years experience in valuing such kind of premises and be active in the market at the time of the appointment.
- 18.3.3 The Nominee shall act as an expert and not as an arbitrator. Each party must submit to the Nominee written Valuations and Submissions within 14 days after the Nominee has accepted the nomination to act, but may not make oral submissions or adduce any evidence. At the time of making any written Submissions or forwarding to the Nominee any written Valuation, that party shall forward to the other party a copy of all written materials submitted to the Nominee. Within 14 days after the Nominee has accepted the nomination to act each party may forward to the Nominee written comments on the other party's written Valuations and Submissions. The Nominee shall take into consideration any written Submissions received within those periods but is not to be fettered by them and shall determine the current market rent in accordance with his own Judgment in the opinion he forms. The Nominee shall conclude his Determination in writing and shall inform the parties of it forty five days after having accepted the nomination to act in writing giving detailed reasons for his Determination and must specify the matters to which he had regard for the purposes of making his Determination). The Nominee's Determination is final and is binding on the parties.
- 18.3.4 If the Nominee:

.

- 18.3.4.1 fails to accept the nomination to act;
- 18.3.4.2 fails to determine the current market rent within sixty days after accepting the nomination to act;
- 18.3.4.3 becomes incapacitated or dies;
- 18.3.4.4 resigns as the Nominee;

then either party may request the nominator to appoint another Nominee in accordance with sub-clause 18.3.2.

25 of 34

18.3.5 The parties shall bear equally the total cost of the rent determination for the rent for the first year of the renewed term. Each party shall bear its own costs of any experts for making valuations for the purpose of written Submissions.

If the Tenant fails to contribute its share of the rent determination within 7 days of the request to do so either by the Landlord or the Nominee, such failure will be a fundamental breach of the Lease thereby terminating the Tenant's rights to take up the option for a new Lease.

- 18.3.6 If the current market rent for the first year of the renewed term shall not be determined by the date of commencement of the renewed term:
- 18.3.6.1 the Tenant shall pay interim rent calculated by:
- 18.3.6.2 increasing the current instalments of rent due by 80% of the Landlord's assessment of the increase in rent, if such increase is less than \$10,000.00, or
- 18.3.6.3 increasing the current instalments of rent due by 60% of the Landlord's assessment of the increase in rent, if such increase is more than \$10,000.00

until the new rent has been determined;

- 18.3.7 Immediately upon the rent has been determined the Tenant shall pay the additional amount (if any) due for rent from the date of commencement of the renewed term to the date when the next monthly instalment of rent is payable on the rent day next succeeding the expiration of thirty days after the date when the Tenant has been informed of the rent determination.
- 18.3.8 If the new rent is less than the interim rent, the Landlord will reimburse the Tenant the over payments within 14 days.

### INTEREST ON OVERDUE MONEY

19. Without prejudice to the rights, powers and remedies of the Landlord otherwise under this Lease, the Tenant will pay to the Landlord interest as set out in Item 11 of the Schedule on any moneys due to the Landlord from the Tenant but unpaid for fourteen days after becoming due on any account whatsoever pursuant to this Lease. Such interest shall be computed from the due date for the payment of the money in respect of which the interest is chargeable until payment of such moneys in full and to be recoverable in a like manner as rent in arrears.

#### HEADINGS

s.

20. The headings to each Clause or paragraph of this Lease are inserted only for the purpose of ease of reference and are not to be used for the interpretation of this Lease.

26 of 34

## LEASE DOCUMENTATION

- 21.1 If the Lease is not to be registered, the Landlord must provide the Tenant with an executed copy of the stamped Lease within one month after the Lease is returned to the Landlord or the Landlord's lawyer or agent following payment of stamp duty on the Lease.
- 21.2 If the Lease is to be registered, the Landlord must lodge the Lease for registration within one month after the Lease is returned to the Landlord or the Landlord's lawyer or agent following payment of stamp duty on the Lease and the Landlord must provide the Tenant with an executed copy of the stamped and registered Lease within one month after the Lease is returned to the Landlord or the Landlord's lawyer or agent following registration of the Lease.
- 21.3 The periods specified in clauses 21.1. and 21.2 are to be extended for delays attributable to the need to obtain any Consent from a head Landlord or mortgagee.
- 21.4 In clauses 21.1 and 21.2 "registered" means registered under the Real Property Act 1900.
- 21.5 This clause does not affect the operation of the Stamp Duties Act 1920 or the Real Property Act 1900.

## GOODS AND SERVICES TAX

- 22.1 Unless the Rent and/or Other Moneys are expressly stated in this Lease to be inclusive of GST, the Rent and/or Other Moneys, price, fees, rates, charges, tax and expenses (:the Amounts") payable by the Lessee to the Lessor under this Lease do not include any goods and services tax ("GST").
- 22.2 Where the Rent and/or Other Moneys are expressly stated in this Lease to be inclusive of GST, the GST is 1/11<sup>th</sup> of the GST inclusive Rent. If the rate of GST is varied at any time during the term, the GST inclusive Rent shall be adjusted in accordance with any changes in the rate of GST.
- 22.3 If any GST is incurred or payable by, or collectable from the Lessor in respect of any supply under this Lease the Lessee must pay the GST to the lessor in addition to the Amounts provided that the payment or reimbursement will not in the aggregate exceed 10% of the Amounts or such other percentage as may be the statutory GST rate from time to time.
- 22.4 The GST must be paid by the Lessee to the Lessor at the earliest of:
  - 7 days prior to the date upon which the Lessor is required to remit the GST to the Australian Taxation Office; or
  - (ii) at the same time as the Amount is required to be paid by the Lessee to the Lessor.

27 of 34

- 22.5 The Lessor shall promptly provide the Lessee with a tax invoice in the form specified by the relevant GST regulations from time to time.
- 22.6 Any GST charged on Outgoings shall be reduced by the amount of input tax credits received by the Lessor in respect of GST on the cost of supply of those Outgoings to the Lessor.

### DISCLOSURE STATEMENT

23. The Tenant acknowledges the receipt of the Disclosure Statement required to be given to the Tenant under the Retail Leases Act 1994.

#### **GUARANTEE**

24.1 Intentionally Deleted

#### **DEFINITION AND INTERPRETATION**

25. In this Lease unless the context otherwise requires:

"the Building" means the building or buildings and other structures erected or to be erected on the Land within which the Premises are situated together with any modifications, additions, extensions or alterations thereto with the fixtures, fittings, furnishings, plant, machinery and equipment of the Landlord therein from time to time.

"the Landlord" shall mean and include where the Landlord is a company its successors and assigns and where the Landlord is a person the Landlord's executors, administrators and assigns.

"the Tenant" shall mean and include where the Tenant is a company its permitted successors and assigns and where the Tenant is a person the Tenant's executors administrators and permitted assigns.

In the event of there being two or more Landlords or two or more Tenants under this Lease (including whilst the Lease or the reversion is held by legal personal representatives, successors or assigns), each of them is jointly and severally liable to perform covenants and obligations under this Lease.

"the Premises" or "the Premises" means the premises hereby leased including the fixtures, fittings and appurtenances belonging thereto.

"the Guarantor" or "the Guarantors" means the person or persons (if any) named herein as guarantor and includes the guarantor's executors administrators successors and assigns and every two or more of them jointly and severally:

Words importing:

- (i) The singular number include the plural numbers and vice versa.
- (ii) Any gender include all other genders.

28 of 34

(iii) Persons include corporations.

3

(iv) References to Statutes include all Statutes amending consolidating or replacing the Statute referred to.

The expression "current market rental" means:

- 1. The best annual rent that can be reasonably obtained for the leased Premises;
- 2. On the basis that the Premises are available for leasing for the use to which the Premises may be put in accordance with the Lease with vacant possession by a willing Landlord to a willing Tenant for the residue of the current lease term (and Option for Renewal) at the review date;
- 3. On the terms and conditions contained in this Lease and such other matters as are relevant to the assessment of current market rent;
- 4. On the basis that the Tenant's Lease covenants and obligations shall have been fully performed at the review date;
- 5. Without taking into account any goodwill created by the Tenant's occupation of the Premises; improvements or fixture erected or installed at the Tenant's expense which the Tenant is permitted or required to remove at the termination of the Lease, except for permanent structural improvements to the leased Premises installed at the Tenant's expense which the Tenant is not permitted to remove at the termination of the Lease, which shall be taken into account;
- 6. Having regard to the rental values of comparable premises.
- 7. Rent concessions and other benefits that are frequently or generally offered to prospective Tenants of unoccupied premises are declared to be relevant matters to be taken into account in the assessment of current market rent.

The reference to "Consumer Price Index" "(CPI)" means the Consumer Price Index All Groups Sydney or the index officially substituted for it.

CPI adjustment of rent in accordance with the CPI:

- (i) take the yearly rent as of the last review date or if none, the rent at the commencement date (\$X),
- (ii) divide that rent by the Consumer Price Index Number for Sydney (All Groups) for the quarter ended just before that date (CPI 1),
- (iii) multiply the result by the Consumer Price Index Number for Sydney (All Groups) for the quarter ended just before the review date (CPI 2).

The product is the new rent for the year beginning on the review date (Y), written as a formula –

$$\frac{\$X}{CPI} \times CP2 = \$Y$$

The Landlord must calculate the new rent after each review date and give the tenant written notice of the new rent.

If the Australian Bureau of Statistics makes a change in the reference base of the index and there is a published co-relation between the old and new base then the published co-relation is to be applied to convert the CP1 1 figure to the new reference base. If there is none then the landlord and the tenant agree to accept the calculations of the landlord's solicitor who must be retained to determine a fair co-relation between the old and the new series of numbers.

### DEMOLITION

26.1

> In the event that the Landlord intends to demolish the Building of which the Premises forms part the Lessor may terminate the Lease by written Notice to the Tenant, but:-

- (a) The Lease cannot be terminated on that ground unless and until the Landlord has provided the Tenant with details of the proposed demolition sufficient to indicate a genuine proposal to demolish the Building within a reasonably practicable time after the Lease is to be terminated.
- (b) The Lease cannot be terminated by the Landlord on that ground without at least six months written Notice of termination.
- (c) If Notice of termination on that ground is given to the Tenant, the Tenant may terminate the Lease by giving the Landlord not less than seven days written Notice of termination at any time within six months before the termination date notified by the Landlord.
- 26.2 If the Lease is for a term of twelve months or less, the period of six months in sub-section (1)(b) and (c) is shortened in each case to three months.
- 26.3 If a Lease is terminated on such a ground and demolition of the Building is not carried out within a reasonably practicable time after the termination date notified by the Landlord, the Landlord is liable to pay the Tenant reasonable compensation for damage suffered by the Tenant as a consequence of the early termination of the Lease unless the Landlord establishes that at the time Notice of termination was given by the Landlord there was a genuine proposal to demolish the Premises within that time.
- 26.4 "demolition" of the Building of which the Premises forms part includes any substantial repair, renovation or reconstruction of the Building that cannot be carried out practicably without vacant possession of the Premises.

### **DISPUTE RESOLUTION**

27. If a dispute arises out of or relates to this agreement or the breach, termination, validity or subject matter thereof, or as to any claim in tort, in equity or pursuant

248 of 34

• •	29
	to any domestic or international statute or law, the parties to the agreement and to the dispute expressly agree to first endeavour in good faith to settle the dispute by mediation administered in accordance with the Retail Tenancy Unit before having recourse to arbitration or litigation
27.1	A party claiming that a dispute has arisen, must give written notice to the other parties to the dispute specifying the nature of the dispute.
27.2	On receipt of the notice specified 18.1 the parties to the dispute must within seven (7) days of receipt of said notice seek to resolve the dispute.
27.3	The mediation shall be conducted and administered in accordance with Retail Tenancy Mediation Rules which set out the procedures to be adopted, the process of selection of the mediator and the costs involved and which terms are hereby deemed incorporated.
27.5	This clause shall not merge upon completion.

.

.

.

219 of 34 31,

•

83 Req:C145583 /Doc:DL AB803964 /Rev:30-Sep-2005 /Sts:OK.OK /Prt:27-Mar-2015 10:20 /Pgs:ALL /Seq:32 of 34 Rof:Pri:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

1 1 15 to

è,

30

### **SCHEDULE**

### **ITEM:**

1. 1(a) 1 (b)	<b>Term (Clause 2.1) FIVE</b> Commencement Date Expiry Date	YEARS 1 July 2005 30 June 2010
2(a)		s GST of \$5,200.00 = \$57,200.00 per annum including
2 <b>(</b> b)	GST \$4,333.33 per month plus	GST of \$433.33 = \$4,766.66 per month including GST
3.	Rent Review (Clause 3.4) 4% annual increase on pre	
4.	Percentage on Holding O 5% annually	over Rent (Clause 3.5)
5.	Outgoings (Clause 4.1.1) 40%	
6.	Use of Premises (Clause Indian Restaurant	10)
7.	Public liability insurance \$10,000,000.00.	e (Clause 12.2)
8.	Bond (Clause 15.1) \$14,300:00 (\$ months rent	including GST) rent including GST)
9.	<b>Option (Clause 18.1.1)</b> Five years	
10.	<b>Option Review Dates (Cl</b> Lease Commencement Da Lease Expiry Date – 30 Ju Rent Review Dates –1 July	te – 1 July 2010 ne 2015
11.	Interest on Overdue Mor	ney (Clause 19)

- Twelve (12%) percent per annum
- 12. Guarantor(s) (Clause 24.1) Not Applicable

A Elias

JN: M:\Docs\50050\193944.doc

32 of 34

.

83 Reg:C145583 /Doc:DL AB803964 /Rev:30-Sep-2005 /Sts:OK.OK /Prt:27-Mar-2015 10:20 /Pgs:ALL /Seq:33 of 34 Ref:Tp1:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

31

IN WITNESS WHEREOF of the parties hereto have hereunto set their hands and affixed their seals the day and year first hereinbefore written.

SIGNED by the said Landlord MICHAEL ELIAS & SAMIRA ELIAS in the presence of:

SIGNED by the said Tenant KATHIRVEL VEL SATHASEVAM in the presence of:

. . . . . . . . . . . . . .

JN: M:\Docs\50050\193944.doc

33 of 34

.

.

# MORTGAGEE CONSENT

CHAMPION MORTGAGE SERVICES PTY LIMITED ACN 003 689 905 being the Mortgagee under Memorandum of Mortgage No 8987304 of the premises demised by the within Lease and the undersigned Guarantors (if any) HEREBY CONSENT to such Lease as from the registration thereof and not otherwise and subject to the following conditions and provisions, namely:-

- 1. That this consent shall be without prejudice to the rights, powers and remedies of the Mortgagee under the said Mortgage which shall remain in full force and effect as if this consent had not been given except that so long as the covenants, conditions and provisions of the said Lease are duly observed and performed the Mortgagee will in the event of the exercise of the power of sale or other power or remedy of the Mortgage on default under the said Mortgage exercise the same subject to the then subsisting rights of the Lessee under the said Lease.
- 2. That so long as the Mortgagee is registered as Mortgagee of the said premises the Lessee shall obtain the consent or approval of the Mortgagee in addition to the consent or approval of the Lessor in all cases where under the said Lease the consent or approval of the Lessor is required.
- 3. That upon the Mortgagee giving Notice to the Lessee of demanding to enter into receipt of the rents and profits of the said premises the covenants on the part of the Lessee expressed or implied in the said Lease shall be deemed to have been entered into by the Lessee with the Mortgagee and all the rights, powers and remedies of the Lessor under the said Lease shall vest in and be exercisable by the Mortgagee until such notice be withdrawn or the said Mortgage be discharged.
- 4. The Mortgagee shall in no way be bound to perform and shall not incur any liability in respect of the covenants and agreements expressed or implied in the said Lease and on the part of the Lessor to be performed and observed.
- 5. Whenever used herein the word "Lessee" shall mean and include the Lessee and the Lessee's assigns; the word "Mortgagee" shall mean and include the Mortgagee and the Mortgagee's assigns.

34 .4 34

### Dated 23 September 2005.

THE COMMON SEAL of CHAMPION MORTGAGE SERVICES PTY LIMITED A C N 003 689 905 was hereunto affixed by authority of the Board in the presence of:

Secretary



	STAMP DUTY	py Supplied by LPI	NSW 1QL	onveya	SPFYCe/USE	DWER DOBCD	Only.		
	OFFIC: 11111 (95)201 (9.5.9. 1990/01		4				1111 <u>5051</u>		
	M - REACTOR	LEASE REAL PROPERTY ACT (To be lodged in dupl (To be lodged in dupl See Instructions for Completion Jacu	icate)	·	L	}; [ 01 ] \$			
SCRIPTION		LAND of which LESS							
LAND ce (a)	Tarrens Tille Reference	Il Part or P PART	remises, See Note	) (8) (ii)	<u> </u>		Location		
	Vol 8442 Fol 95 now being FI B/410947	Shop No, 3, 108 Wentworthville		ion S	treet	Wen	tworthville		
350Fl 9 (b)	RALIAN PTY LIMITED ( (ACN 000 156 276)		ent, Manl	·у					
38EE	The abovenamed LESSOR) hereby leases to	the LESSEE					<b></b>		
101 0	SHARMAN MAY OF 7 WI	illow Street, Went	worthvill	.e		•	OFFICE USE ONLY		
• (c)	IS joint lenants/tenants in common			****	• • •,••••••••••••••••••••••••••••••••		OVER		
o (h) OR	ine land and above described, subject to the to premises		_	16	" · · · · · · · · · · · · · · · · · · ·				
XIMBRANCES (d)		2 ¥184206 (or a TERM of Three(3) commencing on 1 /11 /90 and TERMINATING on 31/10 93 . OFFICE USE ONLY							
M (8)	for a TERM of THIES (5) commencing	jon 1/11/90 and TERM	AINATING on J	· · ·	Annexu	ire "A"			
s (C) n (g) s (M)	(with an GHI CON-LO-PUNCHASE and/of an OPTION OF HENEVAL as set forth in clausador And together with and reserving the rights and liberties set forth in SCHEDULE-ONE hereto), and SUBJECT TO the covenants and provisions:								
	(ii) set forth in the Memorandum filed i	in the Land Tilles Office as Numb	âī		; a	ne	NQ 3Years		
	(iii) set forth in/SCHEDREE We hard Date 11 Activity (99) We hereby cently this leave to be correct for	r the purposes of the Real Property	1	LIMITE	ACA	rein.			
EXECUTION Noto (I)	Signed-in-my-presence by the lessor who is		A NY	( Umn Fe	भाषा ह	A D			
	The Common Seal of RAL Signature of Winners	FFILMAN ()		$\lambda$	$C \mathcal{N}$				
	Eignature of Winness LIMITED was hereunto a Name of Winness (DLOGK LCT2EAS)	- AMBIT		AN CO	C . St	Kla	Y		
	LIMITED was hereunto a	Di	rector/se	ereta	CANE I	irector	ve		
D (I)	Eightre of Winner LIMITED was hereunto a Many of Winner (DLOGK LEFTCH) in the presence of:	Di	rector/se	oreta	C ANY	K Ja	ve		
ə (I)	Signed in my presence by the lessee who is	personally known to me ALAWRICHMOND BRAWN SOLICITOR		A PARTICIPACITY OF THE PARTICI		irector Mo	v		
> (f) > (f)	Eignature of Winness LIMITED was hereunto a Name of Winness (DLOCK LETARS) in the presence of: Additional of the lasse who is Signed in my presence by the lasse who is Signature of Winness Name of Winness (BLDCK LETYERS) Address and occupation of Winness	personally known ja me	$\sim$		J	Bigrature of Le	-,		
> (i) > (i) > (i) > (i) > (i) > (ii) > (ii) > (ii) > (ii) > (ii) > (ii) > (ii)	Eignature of Winness LIMITED was hereunto a Name of Winness (BLOCK LETACRE) in the presence of: Additional Company of Winness Signed in my presence by the lesse with is Signature of Winness Hame of Winness (BLOCK LETTERS) Address and occupation of Winness LODGED BY.	Di.	~ 		J	neeroon Mb/	-,		
» (i) » (i) BE COMPLETED COCING PARTY es (j) and (k)	Eignature of Witness LIMITED was hereunto a Name of Witness (BLOCK LETTERS) in the presence of: Additional Statements of Witness Signed in my presence by the lesse who is Signature of Witness Name of Witness (BLOCK LETTERS) Address and occupation of Witness LODGED BY SDLICITORS 255 EL ZABETH STREE	Di. Di. Di. Di. Di. Di. Di. Di. Di. Di.	~ 		J	Bigrature of Le	-,		
» (i) » (i) BE COMPLETED COCING PARTY es (j) and (k)	LIMITED was hereunto a Name of Winess (BLOCK LETGERS) in the presence of: Minute of Winess (BLOCK LETGERS) Signed in my presence by the lessae wire is Signature of Winess Name of Winess (BLOCK LETTERS) Address und occupation of Winess LODGED BY CODGED BY SDLICITORS 255 EL IZABETH STREE 255 EL IZABETH STREE	Di. Di. Di. Di. Di. Di. Di. Di. Di. Di.	~ 			Signature of Le	-,		
» (i) » (i) » COMPLETEO COGING PARTY	LIMITED was hereunto a Name of Winness (BLOCK LETTERS) in the presence of: Additional Company of Winness Signed in my presence by the lesse who is Signature of Winness Hume of Winness (BLOCK LETTERS) Address und occupation of Winness LODGED BY CODGED BY SDLICITORS 255 EL 'ZABETH STREE SYONEY, 2000 TEL : 286 3000 BEF SCH/	Di. Di. Di. Di. Di. Di. Di. Di. Di. Di.	~ 				-,		
o (I) o (I) BE COMPLETED JODING PARTY es (j) and (k)	LIMITED was hereunto a Name of Witness (BLOCK LETTERS) in the presence of: Additional Statements of Witness Signed in my presence by the lesse who is Signed in my presence by the lesse who is Signed in my presence by the lesse who is Name of Witness (BLOCK LETTERS) Address and occupation of Witness LODGED, BY SDLICITORS 255 EL IZABETH STREE SYDNEY, 2000 Ref: TEL: 286 3000 REF SCH/ Delivery for Witness Checked Passed REGIS	Di. Di. Di. Di. Di. Di. Di. Di. Di. Di.	~ 		Herewith.		-,		
e (I) BE COMPLETED ODDING PARTY es (J) and (k)	LIMITED was hereunto a Name of Winness (BLOCK LETTERS) in the presence of: Additional Company of Winness Signed in my presence by the lesse who is Signed in my presence by the lesse Signed in my presence by the lesse who is Signed in my presence by the lesse who is Signed in my presence by the lesse who is Signed in my presence by the lesse Signed in my presence by the lesse Signed in my presence by the lesse Address and occupation of Witness LODGED BY Signed in Company in the lesse Signed in my presence by the lesse Signe	Di. Di. Di. Di. Di. Di. Di. Di. Di. Di.	CT OT	нся Нея	Herewith.		-,		

						· · ·	
····							
		·		X08	4)		
			······	602+	0ng		<u>.</u>
		3 years	445L92 ×		QPF	· · · · · · · · · · · · · · · · · · ·	i
י מר וניתנשה	ando 26-01-18 sandaro	Wentwo-thwille	1113E26 X		000		
ts noitnets	44 2400 No 3 108- 150				NO		
	8,11AT3KI			BAAL NGLEN		ABIRITNEOI OLIO	
		(+)  TE & OTHER DIRECTION	(a) DEVINC	(3)	្រៅ		
							•
	······································						·
· · · · · · · · · · · · · · · · · · ·	3MAM	·····	-	(0)	NOILOBHIO (8)	ASI DENTRIER	, · ·
			· · · · · · · · · · · · · · · · · · ·	(0)		X IO IDENTICIDO	<u> </u>
	······································	EDALE DIRECTIONS					
(6) aloşş	(0)0	TOPLE USE ONLY SOULE TWO, see annoxure(s) he	PCC contruction of SCH				
		)	ı				
			- 1				
							• •
							•
			7				•
					,		
							•.
(n) bus (m) soloM	01	оэняэлэя эролэвиэнэ)	SCHEDREE LMO H				
			•				
							۰.
m) bee (I) satoN			:saihedit	០០៩ ទាប់ពីរ	i Suimogol eus tiebhil	Oluu seviesei 10889.	aux
					is \$1		~ <b>4</b> F
					• •		
				· .			•
						• •	•
							. •
						۰.	. •
3							
no ore (i) soloN			seathedit brit	នៅរដ្ឋភ្លាំង ស្ព	riwollot ant to Manad	ori ovar lindz gesso.	₽YL
*		илэтэн элочзеиюлэны					

FILM WITH Z'bobISY

### STATUTORY DECLARATION

I, RALPH WILLIAM PAYNE of 12/55 The Crescent, Fairlight in the State of New South Wales do solemnly and sincerely swear as follows:-

1. I am a Director of Ralian Pty Limited ("the Company").

- 2. The Company is the registered proprietor of the whole of the land contained in Certificate of Title Volume 8442 Folio 95 now being Folio Identifier B/410947 and is the Lessor of Shop 3, 108-120 Station Street, Wentworthville ("the Premises").
- 3. A Lease of the Premises was granted to Frank Pellegrino and Joan Pellegrino being registered Lease X267544.
- 4. The Lessees advised the Company that they did not wish to take up the option for renewal available under the Lease.
- 5. The Company has entered into a new Lease with Sharmane May.

AND I MAKE this solemn declaration, conscientiously believing the same to be true, and by virtue of the provisions of the Oaths Act, 1900.

### Made and Subscribed

at Planty this 11th day of Jebruary

1991

before me:

Justice of the Peace/Solicitor

PAG00188/d1/23.01.91



SOLICITORS

LEVEL 16, 25 BLIGH STREET, BYONEY, 2000 DX 10140 SYDNEY STOCK EXCHANGE FACSIMILE (D2) 281 7424 TELEPHONE (D2) 233 1933

> Your ref; Our ref;

MO/AMI

25th March 1991

The Land Titles Office Sydney

Dear Sir

Farrow Mortgage Services Pty Ltd and Ralian Pty Ltd

On behalf of our abovenamed client the mortgagee under mortgages 1842-7 And X586116 we hereby consent to Lease in favour of Sharmane May in respect of Shop 3 at 108-120 Station Street, Wentworthville dated 11th February 1991 and authorised the use of folio identifier B/410947 for the registration of the said Lease. The said Title is at present under production with dealing 2525411.

87 Reg:C145587 /Doc:DL Z606154 /Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seg:4 of 28 Ren:1p1:Sixdre /Src:W /WARNING: A4 Copy Supplied by LP1 NSW for Conveyancing Purposes Only.

> Yours faithfully ANDREWS & BACKHOUSE

PARTNERS W. John Beckhouse Kenneth T. Andrews Robert J. Jeremish Neil S. Mathiaon Richard P. Anderson Michael L. Cietes

Geelong Office: P.O. Box 42, Lovel 3, 115 Myare Screet, Geeolong, 3220, Telephone: (D52) 21 7168 Fecsimile: (D52) 21 9262 DX 22019 (Geelong) Wernibas Office: Unit 10, 2-14 Station Street, Wernibae, 3030 Telephone: (03) 741 0077 Facsimile: (03) 742 8483 DX 30282 (Wernibae)



THIS AND THE FOLLOWING

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:5 of 28 A4 Copy supplied by LPE NSW for Conveyancing Purposes Only.

23 PAGES IS ANNEXURE "A" TO THE LEASE DATED THE

87 Reg:C145587 /Doc:DL Z606154

ŝ

1990 MADE BETWEEN RALIAN PTY LIMITED (AS LESSOR) DAY OF AND SHARMAIN MAY (AS LESSEE) PART 1 - INTERPRETATION AND DEFINITIONS In this Lease unless the context otherwise requires a reference to: 1.1 this Lease shall include the Reference Schedule to this Lease the contents of which Reference Schedule shall be 1.1.1 read and construed as if they were set out in the body of this Lease: 1.1.2 an Item number means the respective Items set out in the Reference Schedule to this Lease; 1.1.3 a clause or part number means the respective clauses and parts of this Lease, 1.1.4 one gender includes a reference to the other genders and each of them: 1.1.5 a person includes a reference to a corporation or firm and vice versa; words importing the singular number shall be 1.1.6 deemed tο include the plural number and vice versa; 1.1.7 any statutory provision shall be construed as a reference to that provision as respectively amended or re-enacted (either before or after the date of this Lease) from time to time; 1.1.8 any person or party to this Lease means and includes the legal personal representatives or permitted assigns of such person or party as the circumstances may require;

> the use of any word or phrase which includes any one or more parties persons documents facts securities or events shall be given effect to as extending to meaning and binding all such parties persons facts events documents or securities jointly and each and every one of them severally, as the circumstances may require;

decorate or decoration in the context of the Premises means to clean and prepare in a good and workmanlike manner and then to paint with at least two coats of paint all parts of the Premises previously painted and to paper varnish and otherwise treat all part of the Premises previously so treated and to clean repaint make good and colour all brickwork stonework and plaster; and

1.1.11 determination by a Valuer or other independent person of any dispute or matter arising pursuant to this Lease means determination by such independent person acting an expertand not as an arbitrator.

1.2 In this Lease unless the context otherwise requires the words and expressions set out below shall have the respective meanings attributed to them as follows:

SIN .

PAG00163/d1/13.11.90

1.1.9

1.1.10



NSW

Supp led

1.2.2

1.2.3

1.2.5

1.2.6

1.2.7

• hv

87 Reg: C145587 /Doc: DL Z606154

1.2.1 "Building" means the land and improvements erected thereon described in Item 2 as modified extended altered or added to at any time.

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:6 of 28

for Conveyancing Purposes Only

"Common Parts" means any passageways, walkways, stairceaes, parades, elevators, lifts, entrances, lobbles, foyers, foyer decorations, gardens, pavements; Lawns and other parts of the Building not demised or intended to be demised by the Lessor which are for the "common" use enjoyment or benefit of the Lessees and other occupants of premises in the Building notwithstanding that they may not be of equal: use enjoyment or benefit to all such lessees and occupants and notwithstanding that they may also be used enjoyed by or be of benefit to the public and Common Parts includes any car parking areas notwithstanding that any particular Lessee or occupant may have the right to use any such area or any part thereof.

"Further Term" means the period specified in Item 14.

1.2.4 "Guarantor" means the person or persons (if any) specified in Item 1.

> "fiead Lease" means the head lease (if any) referred τo 'in' Item 13.

"Index Number" means the Consumer Price Index Number for Sydney published from time to time by the Australian Bureau of Statistics.

"Initial Year" means the first year of this lease but does not include any year during the Further Term.

1.2.8 "Insurance Cost" means such yearly sum as shall represent;

> the premium or premiums including stamp duty 1.2.8.1 payable thereon (or in the event of the Building being insured with other premises the proportion of such premium or premiums as is determined by the Lessor as being attributable to the Building whose determination shall except in case of manifest error be final and binding on the Lessee) paid or payable by the Lessor for insuring the Building in such sum as the Lessor considers at any time to be the full reinstatement value thereof (and including the cost of professional fees and other incidental expenses, the costs of demolition and shoring up and the removal of debris) against the Insured Risker

1.2.8.2

1.2.8.3

the premium or premiums paid or payable by the Lessor for insuring against loss of reat from time to time payable pursuant to this lease and other leases of premises within the Building for such period as the Lessor shall require including the anticipated loss of rent following a rent review; and

the premium or premiums paid or payable by the Lessor for insuring all third party property owners and employers risks in relation

to the

100

PAG00183/BB/-2-/021190

Building and persons employed by the Lessor in or about the Building. ...

/Sea:7

"Insured Risks" means fire lightning storm tempest and such other risks as the Lessor may from time to time reasonably consider it expedient to insure the Building against.

"Lessee" means the person referred to as the Lessee on the front page of this Lease and includes his successors in title and permitted assigns.

"Lessee Party" means and includes the Lessee servants agents employees contractors or visitors and its or not by invitation) and any person claiming through or under the Lessee or any person under the control or direction of the Lessee.

"Leance's Percentage" means the percentage specified in Item 9.

"Lessor" means the person referred to as the Lessor on the front page of this Lease and includes the persons for the time being entitled to the reversion immediately expectant upon the term hereby granted.

"Lessor's Services" means the services specified in Item 4 and any other service now or hereafter provided by the Lessor (at its discretion) for the benefit of the lessees and other occupants of premises in the Building notwithstanding that any such service may not be of equal benefit to all those lessees and may also be of benefit to members of the public.

"Outgoings" means and includes:

1.2.15.1 all rates taxes (including State Land Tax but on the assumption that the Building is the only property owned by the Lessor) charges impositions and fees of any kind at any time payable by the Lessor to any Federal or State government or local or other authority in respect of the Building or any part thereof charged to or payable by the Lessor provided that if the Building is not separately assessed or charged in respect of any such rates taxes charges and impositions there shall be included in ... the Outgoings the Lessor's estimate (which shall be final and binding on the Lessee except in the case of manifest error) of the rates taxes charges impositions or fees (as the case may be) attributable to the Building;

1.2.15.2 all charges incurred by the Lessor in relation to the supply of electricity, gas, water sewerage and drainage to and the removal of waste and other garbage from the Building provided that if the Building is not separately charged in respect of any such service there shall be included in the Outgoings the Lessor's estimate (which shall be final and binding on the Lessee except in the case of manifest error) OÉ the charges attributable to the Building;

le de la completa de

PAG00183/BB/~3-/021190

1.2.14

7.2.15

1.2.12

1.2.13

1.2.10

1.2.9

87 Req:C145587 /Doc:DL Z606154

1.2.11

Req:C145587 /Doc:DL Z606154 /Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Fgs:ALL f:lpi:slxdre /Src:W /WARNING: A4 Copy Supplied by DPL NSW for Conveyancing Purposes Only



5.3 all charges for lighting, power, heating, air conditioning and ventilation incurred by the Lessor in relation to the Building;

/Pas:ALL

/Sec:8 of

28

1.2.15.4 the total cost from time to time to the Lessor of providing the Lessor's Services;

1.2.15.5 the Insurance Cost,

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18

/WARNING: A4 Copy Supplied by LPL NSW

1.2.15.6 in the event the Building comprises any one or more lots in a Strata Plan registered pursuant to the Strata Titles Act 1973, all levies and other charges made by the relevant Body Corporate in respect of the Building, and

1.2.15.7 if the Lessor shall at any time so determine such sum but not exceeding ten per centum (10%) of the aggregate of the Outgoings (but excluding this item) in any year as the Lessor may reasonably determine to set aside as a fund to cover repairs renovations replacements and maintenance of the Building of a substantial but infrequent nature.

"Premises" means that part of the Building referred to on the front page of this lease which for the purposes of obligation as well as grant shall exclude any Common Parts but include the surface of all internal walls floors and ceilings and all floor and wall coverings, all doors and door frames, all windows and window frames, all internal partitioning (erceted by the Lessor or the Lessee or any other person at any time) and all the Lessor's fixtures and fittings (excepting any heating air conditioning and ventilation plant not exclusively serving the Premises).

1.2.17 "Rent" means the initial yearly amount specified in Item 5 together with and including any increases thereto in accordance with the provisions of Clause 4.2.

1.2.18 "Rent Commencement Date" means the date specified in Item 6.

1.2.19 "Review Dates" means those dates specified in Item 8.

1.2.20 "Specified Rate" means the rate which is three per centum above the annual interest rate from time to time and at any time charged by the Lessors principal bankers on overdrafts of more than \$100,000.00.

1.2.21 "Term" means the term granted by this Lease.

1.2.22 "Valuer" means a valuer (with experience in assessing for properties of the same nature as the Fremises) agreed upon by the Lessor and the Lessee or failing such agreement a valuer appointed by the Lessor, but if otherwise required by the Lessee, and advised prior to the Lessor nominating a Valuer, by a Valuer appointed by the President or Secretary for the time being of the Australian Institute of Valuers.

1.3 When two or more persons comprise the Lessee or Guarantor all the covenants conditions terms and restrictions bind such persons jointly and each of them severally and also bind the respective personal representatives assigns and successors in title of each of them

PAG00183/BB/-4-/021190

1 2 16

87 Req:C145587 /Doc:DL Z606154 ef:lpi:sfxdre /Src:W /WARNING



er: pi:sixdre /src:w /wARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

87 Req:C145587 /Doc:DL Z606154

1.4

the Conveyancing Act 1919 (as amended) are hereby expressly negatived. except in so far as they or some part of parts thereof are included in the provisions expressed in this Lease. In the following cases any reference in this Lease to the Lessor is 1.5 deemed to include a reference to any superior lessor or any mortgagee from the Lessor or any superior lessor of the Premises (whether with or without other premises) or any part thereof: where there are rights easements 1.5.1 and reservations exercisable by or benefiting the Lessor; 1.5.2 where there is an obligation to obtain consent from the Lessor; and 1.5.3 where there are any indemnities in favour of the Lessor. Any covenant or agreement by the Lessee not to do or omit any act or 1.6 thing is deemed to extend to an obligation not to permit any third party to do or omit the same. Any approval consent permission or notice given pursuant to 1.7 this

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:9 of

The covenants powers and provisions implied in leases by virtue of

- 1.8 Marginal notes and headings where used in this Lease are for the purpose of identification and location only and not to be considered in the interpretation of the provisions of this Lease.
- 1.9 This Lease shall be read and construed and take effect in accordance with the laws of the State or Territory in which the Premises are situate.
- 1.10 The Lessor shall be entitled to exercise any right on its behalf expressed in or implied by this Lesse by itself, its employees agents servants or contractors.
- 1.11 To the extent permitted by law the application to this Lease and the provisions hereof or any moratorium or other Commonwealth or State statute ordinance rule or regulation which reduces or postpones the payment of rent or extends the Term or otherwise affects the operation of any of the provisions of this Lease to the detriment of the Lessor is hereby expressly excluded and negatived.
- 1.12 The Lessor and the Lessee hereby agree and declare that any provision of this Lease which is or shall be or become in breach of the Trade Practices Act, 1974 or any other Commonwealth or State statute rule or regulation and in consequence of such breach is void voidable unenforceable or invalid shall in any such case and for so long as it is in breach as aforesaid be severable from this Lease and this Lease shall be read and construed as if such provision was not expressed herein.

PART 2 - ANCILLARY RIGHTS, EXCEPTIONS AND RESERVATIONS

Lease is not valid unless in writing.

2.1 This demise shall include the rights (if any) specified in Item 3 and in the event that the rights so specified include the right to park any motor vehicles in any part of the Building the following provisions shall apply:

Sm

Sec. Sec.

2.1.1 The Lessee shall park the motor vehicles only in the positions designated from pime to time by the Lessor.

PAG00183/BB/~5-/021190


ef: 1pi /sixdre /Src:W /WARNING: A4 Copy Supplied by LPT NSW for Conveyancing Purposes

2.1.2 The Lessee shall have the right to identify its car parking spaces by painting appropriate identifying numbers letters or names in such spaces or by erecting at its own expense and with the consent in each oase of the Lessor appropriate identifying name plates adjacent to each space and the Lessee covenants to remove any such signs or painting at its own expense on the termination of this Lease and to make good any damage occasioned thereby.

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seg:10 of

- The Lessee shall not permit or allow any motor vehicle to be cleaned greased oiled washed or repaired in any part of the Building.
- 2.1.4 The Lessee shall not store or permit or suffer to be stored or kept in any part of such car parking area of the Building any petrol or other inflammable fuel except as is contained in the petrol or other fuel feed tanks forming a permanent part of a vehicle.
  - The Lessor shall not be held responsible for the loss of or damage to any car entering leaving or parked in the car parking area or for the loss of or damage to any article or thing in or upon any car or for any injury to any person howsoever such loss damage or injury may arise or be caused.

2.2 There is excepted and reserved from this demise as follows:

- the right to all times upon giving reasonable prior notice to the Lessee (except in the case of emergency) for the Lessor to enter upon the Premises for any purpose in connection with its rights duties and obligations expressed in or implied by this Lesse or otherwise arising from its possession of the Building;
- 2.2.2

2.1.3

2.1.5

2.2.1

87 Reg:C145587 /Doc:DL Z606154

- the free and uninterrupted passage and running at all times of water soil gas electricity and other services from and to all other parts of the Building and any adjoining or neighbouring premises whether belonging to the Lessor or not through and along the conduits which are now or may hereafter during the Term be in the Premises and the right at any time during the Term for the Lessor to Lay move remove replace and maintain any such conduits;
- 2.2.3 all rights of light or air now subsisting or which might (but for this exception) be acquired over any neighbouring land or premises; and
- 2.2.4 as may be specified in the Certificate(s) of Title or title deeds in respect of the land upon which the Building is erected.

PART 3 - HOLDING OVER AND OPTION FOR FURTHER TERM

3.1 If the Lessee with the consent of the Lessor continues to occupy the Premises beyond the expiration of the Term otherwise than pursuant to a further lease granted by the Lessor to the Lessee, he shall do so as a monthly tenant only, at a rental payable monthly in advance equal to one-twelfth of the Rent listly payable under this Lease and such tenancy shall be terminable at any time by either party giving to the other one (1) month's written notice but otherwise shall be subject to such of the provisions hereof as are not inconsistent with a monthly tenancy.

فالمواجبة بيادية والمراجع والمتحاط والمتحصين والمتحد والمحاجة

PAG00183/BB/-6-/021190

If the Lessee gives the Lessor not more than six (6) and not less than three (3) months prior written notice (from the date of expiry of the Term) that it wishes to renew this Lesse for the Further Term and provided the Lessee has duly performed all of its obligations pursuant to this Lease throughout the Term then the Lessor shall upon the expiry of this Lease grant to the Lesses a lease for the Further Term upon the same provisions (including; any guarantee) as are contained in this Lease but amended as follows:

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL

ef:1pi;sixdre /Src:W /WARMING: A4 Copy Supplied by LPI MSW-Mary Conveyancing

/Sec:11

Purpo

3.2.1. The initial yearly amount to be specified in Item 5 of the Reference Schedule of such further lease shall be the greater of:

3.2.1.1 the amount agreed upon by the Lassor and the Lessee as representing the full yearly open market value of the Premises (upon the assumption) that the Premises have been duly repaired decorated maintained and cleaned in accordance repaired with the Lessee's obligations pursuant to this Lease) including any car parking rights pursuant to Clause 2.1 at the date of commencement of such further term as between a willing lessor and a willing lessee with vacant possession for a term. equal to the said further term and upon the terms and conditions of the further lease provided that : if no agreement can be reached such amount shall be determined by a Valuer whose costs shall be borne equally by the Lessor and the Lessee unless otherwise awarded; and

3.2.1.2 the amount of the Rent lastly payable during the Term increased by the percentage specified in Item 17.

The rent commencement date to be specified in Item 6 of the Reference Schedule of such further lease shall be the commencement date of the Further Term.

The dates for review of rent to be specified in Item 8 of the Reference Schedule of such further lease shall be the dates specified in Item 15.

There shall be no option in such further lease for any further term and accordingly the further term to be specified in Item 14 of the Reference Schedule of such further lease shall be nil and the words "not applicable" shall be inserted in Item 15 of the Reference Schedule of such further lease.

PART 4 - RENT AND OUTGOINGS

4.1 The Lessee hereby covenants with the Lessor that it shall:

4-1-1

3.2.2

3.2.3

3.2.4

87 Reg:C145587 /Doc:DL Z606154

without demand or deduction pay to the Lessor from and including the Rent Commencement Date and thereafter throughout the Term (subject only to Clause 10.1.3) without any deduction and without demand the Rent at the times and in the manner specified in Item 7;

4.1.2

if the Lessor shall require pay the Rent by bank authority;

PAG00183/88/-7-/021190



87 Req: C145587 /Doc: DL Z606154 /Rev: 30-Jun-2010 /Sts: OK. OK /Prt: 27-Mar-2015 10:18 /Pgs: ALL /Seg: 12 of

/WARNING: A4 Copy Supplied by LPI NSW

4-1-3 .

if any Rent or other moneys (whether or not in the nature of rent) payable by the Lessee to the Lessor remain unpaid for saven (7) days after the same shall have become due (whether demanded or not) pay to the Lessor interest calculated daily on amounts remaining unpaid at the Specified Rate from the earlier of the date those moneys first became due by the Leesee or are outlaid or incurred by the Lessor (as the case may be) to the date upon which such moneys are paid or reimbursed to or recovered by the Lessor and the Lessor shall be entitled to recover any such interest as rent in arrears and the Lessee acknowledges that the payment to or the demand receipt or recovery by the Lessor of any such interest shall not prejudice or otherwise affect the Lessor's other rights upon the default by the Lessee in duly paying the Rent and any such other woneys. 1

for Conveyancing Purposes

The Lessee hereby acknowledges and agrees that the Rent shall be reviewed on each of the Review Dates in the following manner:

4-2-1 On each Review Date the Rent (being the Rent payable immediately prior to the relevant Review Date) shall automatically increase by the greater of the following:-

> 4.2.1.1 the same proportion as the Index Number has increased since the last Review Date (or in the case of the first Review Date since the commencement of the Term). For the purposes of ascertaining the increase aforesaid the relevant publication of the Index Number shall be that published immediately prior to each Review Date or the date of commencement of the Term as the case may be; or

4.2.1.2 the amount equal to the percentage specified in Item 17 of the Rent payable for the period from the last preceding Review Date (or the date of commencement of the Term in the circumstances of the first Review Date) so that such increases are compounded on an annual basis.

4.2.2

4.2.3

In any case the Rent following any review shall not be less than the amount which is payable immediately prior to such Review Date.

If at any Review Date the Index Number has ceased to be published or in the reasonable opinion of the Lessor has. substantially changed its character then upon any such Review Date the Rent shall (at the option of the Lessor) automatically increase to the amount agreed upon by the Lessor and the Lessee as representing the full yearly open market value of the Premises (upon the assumption that the Premises have been duly repaired, decorated, maintained and cleaned in accordance with the Lessee's obligations pursuant to this Lease) including any car parking rights. pursuant to clause 2.1 upon the relevant Review Date as between a willing lessor and a willing lesses with vacant possession for a term equal to the then unexpired residue of the Term together with the term of any further Leage which the Lessee has the option hereunder to take and upon the terms and conditions of this Lease provided that if no agreement can be reached such amount shall be determined at the request of either party by a Valuer.

PAG00183/88/-8-/021190

4.2

ef:lpi:sixdre /Src:W

4.2.4 For the purposes of agreeing or determining the Rent pursuant to Clause 3.2 or 4.2 the Lesson and the Lessee agree that the amount of the Rent at the commencement of the Term which is attributable to the right to park any cars in the Building (pursuant to Clause 2.1) is as specified in Item 16.

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL

ef:lpl:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing

4.2.5

4 2 5 2

/Sec:13 of

28

Where there has been any delay (for whatever reason) in agreeing or determining any increase in the Rent at a Review Date pursuant to Clause 4.2.1 or the new rent upon the grant of any further term pursuant to Clause 3.2 the Lessee shall

4.2.5.1 continue to pay Rent at the rate payable immediately prior to such Review Date or grant of a further term (as the case may be) up to the time any increase in the Rent or the new rent (as the case may be) is agreed or determined;

> within seven (7) days of any increase in the Rent or the new rent (as the case may be) being agreed or determined pay an amount equal to such increase for the period from the relevant Review Date or the commencement of the further term (as the case may be) to the date of such agreement or determination;

4-2.5.3 pay to the Lesson interest on the amount payable. pursuant to Clause 4.2. calculated at the Specified Rate from the relevant Review

Specified Rate from the relevant Review Date or commencement of the further term (as the case may be) to the date of its payment; and

4.2.5.4 pay the Rent as increased or the new rent (as the case may be) from the date any such increase is agreed or determined.

4.3 The Lessee covenants with the Lessor to pay the Lessee's Percentage of Outgoings to the Lessor as follows:

4.3.1 for the purposes of this Clause 4.3 a reference to a year shall mean a period of twelve (12) months commencing on the same day as the Term or such other date as the Lessor may at any time elect.

4.3.2 At the beginning of each year which is wholly or partly within the Term (or as soon as is practicable thereafter) the Lessor or his agent may prepare a reasonable estimate of the Outgoings in respect of that year ("Estimate").

4.3.3 The Lessor shall provide the Lessee with a copy of an Estimate prepared pursuant to Clause 4.3.2.

4.3.4

87 Reg:C145587 /Doc:DL Z606154

In respect of each year (or part of a year) during the Term the Lessee shall (if the Lessor has submitted an Estimate for that year) pay to the Lessor on account of the Lessee's Percentage of Outgoings for that year the Lessee's Percentage of the Estimate (or in the event that the year to which the Estimate relates is partly outside the Term such amount of the Estimate as is attributable to that part of the year within the Term) by equal monthly instalments in advance throughout the year to which the Estimate relates or that

PAG00183/BB/-9-/021190

B

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL

part of that year which is within the Term (as the case may be) which instalment shall be paid on the same day of the month that the instalments of Rent are payable pursuant to this Lease or such other day as the Lessor may from time to time specify to the Lessee.

conveyancing

/Sec.14 of

2.8

Immediately after the end of each year during the Term and in the event that the Term expires or is determined during the course of a year immediately after the end of that year (or as soon as is practicable thereafter) the Lessor or his agent shall prepare and provide the Lessee with a statement showing the total of the actual Outgoings for that year (or in the event that the year to which such statement relates is partly outside the Term such amount of the actual Outgoings as is attributable to that part of the year within the Term) and also showing the amount due to the Lessor or the Lessee (as the case may require after taking account of any payments by the Lessee pursuant to an Estimate) which amount shall be paid forthwith.

4.3.6 This Clause 4.3 shall continue in full force and effect notwithstanding the expiry or termination of this lease for the purpose of making any balancing adjustment as aforesaid.

4.4

ef: 1pi; sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for

4.3.5

87 Reg:C145587 /Doc:DL Z606154

The Lessee shall pay for all services supplied or provided to the Premises (including but without limiting the generality thereof gas electricity telephones and telex) provided that if the Premises are not separately metered or charged in respect of any such service (and the cost of such service is not included in the Outgoings) the Lessor shall pay for the provision of such service and the Lessee shall forthwith pay to the Lessor the Lessor's estimate (which shall be final and binding on the Lessee except in the case of manifest error) of the cost of the provision of any such service attributable to the Premises.

4.5 Notwithstanding anything contained in Part 4.3 at the option of the Lessor Outgoings are to be paid by the Lessee within fourteen (14) days of notification to the Lessee by the Lessor or the Lessor's Agent of the Lessee's Percentage of Outgoings.

PART 5 - DEFAULT AND DETERMINATION

5.1 Notwithstanding any provision to the contrary expressed in or implied by this Lease upon the bappening of any of the following events the Lessor shall be entitled to exercise any of the rights specified in Clause 5.2:

5.1.1 if the Rent or any part thereof or any other moneys (whether or not in the nature of Rent) shall be in arrears or unpaid for a period of fourteen (14) days (whether or not formally demanded);

5.1.2 If the the Lessee shall default in the due observance and performance of any covenant condition restriction agreement or regulation (not relating to the payment of Rent or other moneys (whether or not in the nature of Rent)) expressed in or implied by this Lease on its part to be observed and performed provided that in so far as such default is reasonably capable of remedy the Lessor shall first give the Lessee not less than twenty eight (28) days notice requiring the default to be remedied;

PAG00183/BB/-10-/021190



87 Req: C145587 /Doc: DL Z606154 /Rev: 30-Jun-2010 /Sts: OK. OK /Prt: 27-Mar-2015 10:18 /Pgs: ALL /Seq: 15 of

## if in the event of the Lessee being a corporation:

an order is made or a resolution is passed for its winding up or proceedings are initiated or a meeting called to obtain any such order or to pass such resolution;

Only

5.1.3.2

5.1.3.1

ef:lpi:sixdre /Src:W. /WARNING: A4 Copy Supplied by LPT NSW for Conveyancing Purposes

a receiver manager or receiver and manager of its undertaking or any part thereof is appointed or an official manager or provisional liquidator is appointed;

5.1.3.3

If the Lessee is not a public listed company and without the prior written consent of the Lessor any sale transfer or other disposition whatsoever of the mares in its issued capital or any issue or allotment of any new shares in its capital or any other act matter or thing whatsoever occurs is done or performed the effect of which is to transfer whether directly or indirectly to any person persons company or companies the effective management and control of the Lessee.

## 5.1.4 If the Lessee being an individual:

# 5.1.4.1 dies or becomes incapable of managing his own affairs;

5.1.4.2 is declared bankrupt or makes any arrangement with his creditors;

5.1.5 if any execution or other process of any Court or other authority issues out against or is levied upon any of the property of the Lessee;

5.1.6 if the Lessee stops or threatens to stop payment of its debts or without the prior written consent of the Lessor ceases or threatens to cease to carry on its business;

5.1.7 If any warranty or representation expressed in or implied by this Lease or otherwise made by or on behalf of the Lessee to the Lessor prior to entering into this Lease is found to be materially incorrect;

5.1.8 if a final judgment is entered in any Court, against the Lessee and is not estisfied within twenty eight (28) days thereafter;

- 5.1.9 If the Lessee without the prior written consent of the Lessor oreates or purports to create any charge or mortgage over its interest in this Lesse;
- 5.2 Upon the happening of any of the events specified in Clause 5.1 the. Lessor may at its absolute discretion:-
  - 5.2.1

immediately or at any time thereafter (unless prohibited by any statute) and without any notice (unless required by virtue of any statute) or provious demand re-enter (forceably if necessary) into and upon the Pramises or part or parts thereof in the name of the whole and repossess the same as of its former estate and expel and remove the Lessee and all other occupiers without liability for the tort of trespass or

PAG00183/BB/-11-/021190



87 Reg:C145587 /Doc:DL Z606154 /Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:16 of Ref:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Furposes Only.

to the Lessee for any liability it may incur and without prejudice to any remedies which might otherwise be available to the Lessor to recover arrears of Rent or in respect to any antecedent breach of the covenants conditions restrictions agreements and regulations on the part of the Lessee expressed in or implied by this Lease and such re-entry shall (unless the Lessor otherwise expressly elects) automatically cause this Lease to determine as if it had thereupon expired by effluxion of time but the Lessee shall remain liable for all rents due to the date of such re-entry and for all other moneys (if any) due hereunder; or

- 5.2.2
  - by giving written notice to the Lessee reduce the Term so that it expires on a date being not prior to the twenty eight days after the date of service of such notice.
- 5.3 In the event of any rementry pursuant to Clause 5.2.1 the Lessor may retain any furniture fittings fixtures or other items belonging to. the Lessee in the Premises and the Lessor shall have the right to sell such furniture fittings and fixtures or other items by public auction and apply the proceeds of such sale towards the payment of any moneys outstanding and payable to the Lessor pursuant to this Lease. . . . .
- 5.4
- In the event that the Lessee defaults in its obligations pursuant to this Lease in respect to the repair and decoration of the Premises then without prejudice to any other of the Lessor's rights and remedies hereunder it shall be lawful for the Lessor without further notice to enter upon the Pramises and repair and remedy the same and all expenses incurred by the Lessor in so doing shall be re-imbursed for thwith without the need for demand by the Lessee and if not so reimbursed shall be recoverable as rent in arrears without notice.
- If any of the events specified in Clause 5,1.3 occurs with respect to 5-5 any Guarantor being a corporation or if any of the events specified in Clause 5.1.4 occurs with respect to any Guarantor being an individual then the Lessee covenants with the Lessor that it shall within fourteen (14) days of such event procure an additional; guarantee of its obligations under this Lease (in the form contained in Clause 13) by a respectable responsible and solvent person acceptable to the Lessor.

PART 6 - REPAIR AND MAINTENANCE OF PREMISES

The Lesses hereby covenants with the Lessor that it shall;

- 6.1 At all times during the Term and during any period of holding over repair and keep the Premises in good and substantial repair. decoration and condition to the satisfaction of the Lessor (damage by any of the Insured Risks excepted provided that no policy of insurance effected by the Lessor has been rendered void or voidable by virtue of any act matter or thing done upon the Premises or the Building by the Lessee its servants agents invitees or any person claiming through or under the Lessee).
- If required by the Lessor enter into and maintain (at the cost of the 6.2 Lessee) throughout the Term and during any period of holding over fully comprehensive maintenance contracts for the maintenance of any heating ventilation fire prevention air conditioning and other equipment or plant which exclusively serves the Premises with reputable companies first approved by the Lessor (whose approval shall not be unreasonably withheld) and produce to the Lessor at any

PAG00183/BB/-12-/021190

time upon demand such contracts and evidence that any sums due thereunder have been fully paid.

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:17 of 28

Supplied by LPI NSW for Conveyancing Purposes Only.

From time to time immediately replace broken glass in the Premises. 6.3 with glass of the same or similar quality and all broken or damaged plumbing, lighting, heating and electrical equipment appliances and other fixtures and fittings of the Lessor in the Premises (excluding any heating and air conditioning plant not exclusively serving the Premises which is maintained by the Lessor as part of the Lessor's Services).

- At the expiration or sconer determination of the Term yield up to the 6.4 Lessor the Premises (but excluding any fittings and fixtures which are tenants trade fixtures or fittings or which are otherwise required to be then removed pursuant to this Lease) duly repaired and maintained in accordance with the covenants herein contained.
- 6.5 Permit the Lessor upon giving not less than two days at any time during the Term notice to the Lessee except in the case of emergency to enter upon the Premises;
  - to take a plan of or to examine the state of repair 6.5.1 and conditions of the Premises and to take inventories; and
  - 6.5.2 to execute repairs decorations alterations or other work to the Building and any adjoining or neighbouring land and for the purpose of building upon any adjoining of neighbouring land provided that the person or persons exercising such right shall make good in a reasonable manner all damage e 14 thereby occasioned to the Premises and cause as little inconvenience to the Lessee as is practical.
- At all times keep the Premises including external surfaces of windows 6.6 and doors clean and tidy and not place leave or permit to be placed or left any debris or rubbish in any part of the Premises.
- 6.7 At all times keep waste, trash and garbage in proper receptacles.
- At all times to keep the fire extinguishers hoses and other fire 6.8 fighting or prevention equipment exclusively serving the Premises in good working order and condition and inspected at least once in every year of the Term by the relevant fire authority.
- In the event that the Lessor does not as part of the Lessor's. 6.9 services arrange for the cleaning of the Premises the Lessee shall enter into and maintain a comprehensive cleaning contract for the daily cleaning of the Premises with a reputable company first approved by the Lessor and produce to the Lessor at any time upon demand a copy of such contract.

PART 7 - USE OF AND CONDUCT ON PREMISES

The Lessee hereby covenants with the Lessor that it shall:

- Not use the Premises or any part thereof: 7.1
  - 7.1.1 for any purpose other than as specified in Item 10 or such other purpose as the Lessor may consent to;
  - 7-1-2 for any illegal purpose or for any purpose in contravention. of any approval consent or zoning by the relevant authority;

PAG00103/BB/-13-/021190

WARNING A4

CODV

87 Reg; C145587 /Doc; DL Z606154



Ef: 1pi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI-NSW for Conveyancing Purposes

87 Reg:C145587 /Doc:DL Z606154

7.1.3 as a dwelling or sleeping place or to keep any animals birds or other livestock;

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seg:18 of 28

OnF

7.1.4 for the carrying on of any auction sale;

7.1.5 except during the times specified in Item 11.

- 7.2 Not without the prior written consent of the Lessor (who may require such plans and details as it considers necessary to be submitted to it by the Lessee) make or permit or suffer any alteration or addition whatsoever to the Premises (including the erection, demolition or alteration of any partitioning within the Premises) or any part thereof or mark drill cut main injure or deface or in any way damage any parts of the Premises, and at the end of the Term (or any subsequent term granted pursuant to any option herein contained) if the Lessor so requires remove any such alterations and additions.
- 7.3 Subject to Clause 12.7 at all times at the cost of the Lessee observe and comply with:
  - 7.3.1 the laws and regulations for the time being relating to fires (and the prevention thereof) and every requirement of the Fire and Accident Underwriters Association the Board of Fire Commissioners of the State in which the Premises are situated and any other competent authority; and
  - 7.3.2 the provisions of every Commonwealth and State Statute and every rule regulation ordinance and by-law made at any time under or in pursuance of any such Statute and the requirements at any time of every government or other competent authority.
- 7.4 Not cause any damage to or obstruction of any of the Common Parts or to any road or laneway serving the Building and not to place or store any goods outside the Premises.
- 7.5 Not erect or display any sign or advertisement on the exterior of the Premises (or within the Premises so that such sign or advertisement can be seen from the exterior of the Building) without the written consent of the Lessor (such consent not to be unreasonably withheld) and the consents of all relevant and competent authorities and uponthe termination of this Lesse to remove any such sign.
- 7.6 Not use the lavatories conveniences and water apparatus in the Building for any purpose other than those for which they were constructed and on demand pay to the Lessor an amount sufficient to compensate the Lessor for any damage resulting from any such misuse by the Lessee.
- 7.7 Not without the written consent of the Lessor install use place or permit or suffer to be brought into the Premises any plant machinery or other articles which may cause undue noise or vibrations or which are of a weight or size which may cause damage directly or indirectly to the Building.
- 7.8 Not bring on to the Premises any dangerous inflammable explosive noxious or offensive substances except in the ordinary course of the Lessee's business (permitted by this Lease to be carried on at the Premises) and provided that the Lessee shall ensure that all proper and prudent measures are taken in the storage and use of any such substances and that the Lessor is previously notified of the nature and extent of any such substances brought on to the Premises.

PAG00183/BB/-14-/021190

Not do any act or thing which in the opinion of the Lessor (as to 7.9 which it need not give reasons) may be offensive or cause a muisance damage or annoyance or inconvenience to the Lessor or to the owners. Lessees or occupiers of any adjoining or neighbouring premises. For the purposes of this Clause the use permitted by Item 11 will not contribute an offensive act. 7.10 Not overload the electric wires and cables serving the Premises. 7.11 Give to the Lessor full particulars of any permission notice order requirement recommendation or proposal given or issued in respect of or in connection with the Premises by any competent authority within seven days of its receipt by the Lessee. 7.12 Comply with such reasonable regulations as the Lessor may now or in the future make for the more efficient management of the Building (including its security and that of its lessees and occupants). 7.13 At all times observe and perform the restrictions stipulations and covenants (if any) referred to in the certificate(s) of title in

7 Req:C145587 /Doc:DL Z606154 /Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:19 of 28 ef:lpi':sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Fungers and Antion Conveyancing Fungers

87 Req:C145587 /Doc:DL Z606154

- 7.14 Not obstruct any of the windows or ventilators belonging to the Premises nor to permit any new window or ventilator or other encroachment or easement of which the Lessee is aware to be made against or over the Premises.
- 7.15 In carrying out any construction fitting act reinstatement or removal works or alterations to the Premises:
  - 7.15.1 ensure that such works and alterations are carried out in a good and workmanlike manner by licensed and reputable. tradespersons; and
  - 7.15.2 forthwith repair and reinstate any part of the Building damaged or defaced in carrying out such works or alterations. to the satisfaction of the Lessor.
- 7.16 Observe the covenants on the part of the Lesses contained in the Head Lease (if any) under which the Lessor holds the Premises (except' those relating to the payment of rent, insurances and those which are inconsistent with the terms hereof) as though the same were set out in full herein.
- 7.17 In the event that the Building comprises any one or more lots in a Strata Plan registered pursuant to the Strata Titles Act 1973, at all times to observe and comply with the provisions of the Strata Titles Act 1973 and its regulations and schedules and any by-laws of the relevant strata scheme.

PART 8 - INSURANCES AND INDEMNITIES

respect of the Building.

The Lessee hereby covenants with the Lessor that it shall:

At its own cost effect and at all times keep in full force and effect. 8.1 a policy of public risk insurance with a reputable and solvent insurer with respect to the Premises and the business carried on in the Premises in which limits of public risk shall be not less than the amount specified in Item 12 (or such other amount as the Lessor may from time to time reasonably require) as the amount payable in respect of liability arising out of any one single accident or event. and shall deliver to the Lessor on demand a copy of the policy and a ourrent certificate of insurance.

PAG00183/98/-15-/021190



87 Req:C145587 /Doc:DL Z606154 /Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:20 of

ef:lpf:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only

8.2 At its own cost effect and at all times keep in full force and effect plate glass insurance (for the full replacement value thereof) in respect of all plate glass attached to or forming part of the Premises and shall deliver to the Lessor on demand a copy of the policy and a current certificate of insurance.

8.3 Include as the insured parties in respect of each of the policies effected pursuant to Clauses 8.1 and 8.2 the Lessor, any superior lessor, any person or persons nominated by the Lessor as being mortgagees of the Building and the Lessee.

8.4 And hereby does indemnify and hold indemnified the Lessor from and against all actions claims demands losses damages costs and expenses which the Lessor may sustain or incur or for which the Lessor may become liable whether during or after the Term in respect of or arising from:

- 8.4.1 The neglect or default of any Lessee Party to observe or perform any of the terms covenants and conditions expressed in or implied in this Lease.
- 8.4.2 The negligent use or misuse waste or abuse by any Lessee Party of any water gas electricity or other services to the Building.
- 8.4.3 The overflow leakage or escape of water fire gas electricity or any other harmful agent whatsoever in or from the Premises.
- 8.4.4 The failure of the Lessee upon becoming aware of any defect in any of the air conditioning fire prevention equipment or other facilities presently available in relation to the Premises to notify the Lessor of such defect.
- 8.4.5 The use of the Premises and the Common Parts by any Lessee Party.
- 8.4.6 The carrying out of any additions or alterations or other works to the Premises by any Lessee Party.
- 8.4.7 The use by any Lessee Party of any car parking facilities in the Building permitted by this Lease.

8.4.8 The happening of any accident or event in or about the Premises.

and it is hereby agreed that the Lessor shall not be liable or in any way responsible to the Lessee or any Lessee Party for any injury loss or damage which may be suffered or sustained to any property or by any person on the Premises unless caused by the wilful act or omission of the Lessor its servants or agents.

8.5

Comply with all requirements and recommendations of the insurers of the Building and not do or omit to do anything on the Premises which may increase the premium above the ordinary rate or render any additional premium payable for the insurance of the Building or of any neighbouring premises or which may make void or voidable any policy of such insurance and reimburse the Lessor forthwith on demand any additional premium which may have been paid or become payable by reason of any such act or omission together with all expenses incurred by the Lessor in relation to the renewal of any policy of insurance (which shall not form part of the Insurance Cost).

PAG00183/BB/-16-/021190



/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18

Inform the Lessor of any reason arising from the Lessee's use of the Premises why the Lessor's insurable interest in the Premises or in any adjoining premises may be affected.

Conveyancing Purp

/Pas:ALL

/Sec:21

In the event of the Premises or any part thereof being damaged or destroyed by any of the Insured Risks at any time during the Term and the insurance money under the insurance policy effected thereon being wholly or partially irrecoverable by reason (solely or in part) of any act or default of a Lessee Party forthwith pay to the Lessor the whole or as the case may require a fair proportion of the cost of rebuilding and reinstating the same.

#### PART 9 - ALIENATION

/WARNING: A4 Copy Supplied by LPI NSW

9.7 The Lessee hereby covenants with the Lessor hot to assign, transfer sublet part with or share possession of the whole or any part of the Premises or its interest therein provided that the Lessee may assign of sub lesse the benefit of this Lesse as a whole having first obtained the written consent of the Lessor whose consent shall not be unreasonably withheld in the case of a respectable and responsible assignee or sub lessee and provided that any such assignment or subletting shall comply with the relevant provisions of Clauses 9.2, 9.3 and 9.4. For the purposes of this Clause 9.1 a change in the constitute a transfer of this Lesse.

9.2

8.6

8.7

87 Req: C145587 /Doc: DL Z606154

ef:lpf:sixdre /Src:W

If the intended assignes of sub leases shall be a company then the directors or principal shareholders of the same shall if the Lessor so requires act as guarantors for such company and shall (inter alia) jointly and severally covenant with and guarantee to the Lessor in the manner set out in Clause 13.

- 9.3 Upon any assignment or transfer of the Lease the assignee shall if the Lessor so requires enter into a direct covenant with the Lessor to observe and perform the novenants and conditions herein contained and on the part of the Lessee to be observed and performed.
- 9.4 Any subletting shall be upon the same terms of this Lease and at a rent which is no less than the Rent and shall be capable of being determined by the Lessor in the event of determination of this Lease.

PART 10 - DESTRUCTION OF PREMISES

10.1 If during the term of this Lease the Building shall be destroyed or damaged by any of the Insured Risks so as to render the Premises or any substantial part thereof substantially unfit for the use and occupation by the Lessee or so as to deprive the Lessee of substantial use of or access to the Premises:

10.1.1 This Lease may be terminated without compensation by either the Lessor or the Leasee by written notice to the other provided that the Leasee shall not be entitled to terminate the Lease inless the Premises shall not have been rendered fit for the use and occupation of the Lessee or the use of and access to the Premises shall not have been substantially restored within four (4) weeks of such destruction or damage.

10.1.2 Any such termination shall be without prejudice to the rights of either party in respect of any antecedent breach matter or thing.

PAG00183/BB/-17-/021190



10.1.3 On the happening of any such damage or destruction as aforesaid (provided that any insurance moneys that would have been payable to the Lessor are not wholly or partially irrecoverable by reason of any act or default of a Lessee Party) the Rent or a proportionate part thereof (according to the nature and extent of the damage sustained) shall abate and all or any remedies for the recovery of such Rent or such proportionate part thereof shall be suspended until the Premises shall have been rebuilt or reinstated or made fit for the occupation and use of the Lessee or until access thereto shall have been provided or until the Lesse shall be terminated pursuant to the provisions hereof as the case may be. Any dispute arising out of, this clause shall be determined by a Valuer whose fees shall be borne by the Lessor and Lessee in equal shares unless otherwise awarded.

10:18

/Pas:ALL

/Sec:22

10.2 Nothing expressed in or implied by this Lease shall be deemed to impose any obligation upon the Lessor to rebuild or reinstate or make fit for occupation the Premises or the Building in the event of damage thereto or destruction thereof.

PART 11 - LESSOR'S COVENANTS

87 Reg:C145587 /Doc:DL Z606154 /Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015

/WARNING: A4 Copy Supplied by LPI

ef:1p1:sixdre /Src:W

The Lessor hereby covenants with the Lessee as follows:

- 11.1 That the Lessee paying the Rent hereby reserved and performing and observing the covenants and stipulations herein contained shall peaceably hold and enjoy the Premises during the Term without any interference from the Lessor or any person rightfully claiming under or in trust for the Lessor.
- 11.2 That the Lessor will at all times during the Term (unless such insurance shall be avoided by any act or omission of the Lessee as aforesaid) insure and keep insured the Building against the Insured Risks and will whenever required (but not more than twice in every calendar year) produce a certificate to the Lessee in respect of such insurance.
- 17.3 That the Lessor will use its best endeavours to provide the Lessor's Services during the Term provided that in so far as the Lessor's Services relate to the provision of any heating lighting air conditioning or ventilation to the Building or the operation of any lifts in the Building the Lessor shall only be obliged to provide such Lessor's Services during normal working hours during the week (excluding public holidays) and provided further that the Lessor shall not be liable for any failure at any time to provide any of the Lessor's Services arising out of matters beyond its reasonable control.

#### PART 12 - GENERAL

- 12.1 The Lessee hereby covenants with the Lessor to pay to the Lessor of demand:
  - 12.1.1 all legal and other costs and disbursements (including stamp duty) incurred by the Lessor in relation to any application by the Lessee for consent to an assignment of this Lesse or any other dealing with the Lessee's interest herein or any other matter requiring the consent of the Lessor pursuant to provisions hereof or in connection with any breach or threatened breach of any of the terms hereof by the Lessee and in connection with any proceedings for enforcement of payment of rent or any other terms of this lesse.

PAG00183/BB/-18-/021190

12-1-2 Upon the signing of this lease the lessor's reasonable legal costs and disbursements in connection with this lease and the stamping and registration thereof.

87 Req:C145587 /Doc:DL Z606154 /Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:23 of 28 lef:lp1:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

12.2 Service of any notice required or authorised by this Lease may be effected in the manner permitted by Section 170 of the Conveyancing Act, 1919.

12.3 The Lessee shall permit the Lessor at any time during the last six months of the Term to enter the Premises and affix upon any suitable parts of the exterior thereof notice boards or other signs advertising the reletting or selling the same and the Lessee shall not remove or obscure such notices.

12.4 Nothing in this Lease shall imply or warrant that the Premises are fit or entitled to be used for any particular purpose and the Lessee hereby acknowledges that in entering into this Lease he has not relied upon any representation made by the Lessor, its solicitors or agents as to the suitability of the Premises for any particular purpose.

12.5 The Lessor will (at the cost of the Lessee) supply the Lessee (and, such of its employees as the Lessee may specify to the Lessor) with keys enabling the Lessee to gain access to the Premises at all times during the Term. The Lessee shall immediately inform the Lessee in the event of any of the keys supplied being lost and if so required by the Lessor pay for the changing of any locks and the supply of new keys deemed necessary by the Lessor as a result of such loss. The Lessee shall return all keys supplied to the Lessor at the end of the Term.

12.6 The Lessee shall be responsible for protecting and keeping eafe the Premises from theft and robbery and shall keep all windows and doors properly locked at all times that the Premises are unoccupied.

12.7 It is hereby acknowledged and agreed that the Lessee shall not be obliged by anything expressed in or implied by this Lesse to carry out any structural repairs or works to the Building unless the same arise (whether directly or indirectly) as a result of any of the following:

- 12.7.1 the neglect or default by any Lessee Party to observe or perform any of the terms covenants and conditions expressed in or implied by this Lesse;
- 12.7.2 the use or occupation of the Premises by the Lessee or any sub-tenant:
- 12.7.3 the employment of any person in the Premises by the Lessee or any sub-tenant;
- 12.7.4 the use of any fixtures fittings plant machinery or goods in the Premises by the Lessee or any sub-tenant;

12.7.5 the carrying out of any alterations or additions to the Premises by the Lessee or any sub-tenant or the reinstatement of the Premises following any alterations or additions thereto; and

12.7.6 the bringing on to the bremises by the Lessee or any subtemant of any plant machinery or other items (whether consented to by the Lessor or not)

PAG00183/BB/-19-/021190



/WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

87 Req:C145587 /Doc:DL Z606154 /Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:24 of 28

in which event the Lessee shall at the option of the Lessor either at its own cost forthwith repair and reinstate the structure of the Building so requiring repair or carry out such structural works as may be necessary to comply with the requirements of any competent authority to the reasonable satisfaction of the Lessor and all relevant authorities or if the Lessor has itself elected to carry out such repairs and reinstatement or works the Lesses shall forthwith upon demand by the Lessor pay to the Lessor all its costs and expenses (including all professional fees) incurred in connection therewith.

12.8

ef:lpf:sixdre /Src:W

Notwithstanding anything to the contrary expressed in or implied by this Lease:

- 12.8.1 The Lessee hereby assures the Lessor that it will at all times strictly and substantially comply with each and every covenant condition restriction rule and agreement expressed in or implied by this Lesse on the Lessee's part to be performed or observed and the Lessee acknowledges that the Lessor has entered into this Lesse on the basis of such assurance.
- 12.8.2 The Lessor and the Lessee hereby expressly agree and acknowledge that each covenant condition restriction rule and agreement expressed in or implied by this Lesse on the Lessee's part to be performed and observed is a fundamental and essential term of this Lesse.
- 12.8.3 The Lessor and the Lessee hereby expressly agree and acknowledge that if:
  - 12.8.3.1 any Rent is in arrears or unpaid for fourteen
    (14) days after it has become due and payable
    (whether or not any formal demand has been made
    for payment);
  - 12.8.3.2 the Lessee defaults in the due performance and observance of any other (than those relating to payment of Rent) covenant, condition, restriction, rule or agreement herein contained or implied on the Lessee's part to be performed or observed and in the case of a default which is capable of remedy has continued the default for more than twenty eight (29) days after notice from the Lessor requiring the default to be remedied;
  - 12.8.3.3 the Lessee (being a natural person) commits any act of bankruptcy or has his estate sequestrated in bankruptcy or assigns his estate for the benefit of creditors or enters into a deed of arrangement with his creditors or enters into an arrangement for the liquidation of his debts by composition or otherwise; or
  - 12.8.3.4 the Lessee (being a corporation) becomes subject to an order for liquidation whether compulsory or voluntary or commits any act of bankruptcy or has a receiver appointed over the whole or part of its assets or undertaking;

any such act matter or thing shall if the Lessor so elects (and the Lessor shall deemed to have so elected if it

PAG00183/BB/--20-/021190

exercises any right of re-entry pursuant to Clause 5.2) constitute and be deemed to be a repudiation by the Lessee of this lease so that the Lessor shall without prejudice to any other rights it may have be entitled to recover damages from the Lessee for the full extent of its loss arising out of such repudiation.

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:25 of

WARNING: A4 Copy Supplied by LPI NSW: for Conveyancing Purposes Only.

28

#### REDEVELOPMENT OF FREMISES

13.1

PAG00183/BB/-21-/021190

87 Reg:C145587 /Doc:DL Z606154

The Lessor may give the Lessee six (6) months written notice of his intention to redevelop the premises.

13.2 At any time after the date of expiration of the six (6) month period stated in Part 13.1 the Lessor may by giving fourteen (14) days written notice to the Lesson determine the residue of the term of this Lease effective on and from the date of expiration of the Notice pursuant to this Part 13.2 but without prejudice to the rights and remedies of the Lessor against the Lessee in respect of any antecedent claim or breach of covenant by the Lessee.

13.3 Nothing herein contained or implied shall modify release vary or affect the liability of the Lessee to pay the Rent up to and including the date of determination of the lease pursuant to this Part nor entitle the Lessee to make any claim or demand on the Lessor as a result of the determination of the residue of the term of this Lease.

## THE REFERENCE SCHEDULE

#### GUARANTOR:

A: 101: sixdre /Src:W /WARNINGY A4 Copy Supplied by AFT NSW for

**ITEM 1:** 

ITEM 2:

ITEM 3

COEM.

#### N11 .

1.

2.

з.

5.

6.

7.

8.

#### BUILDING:

The whole of the Land comprised in Certificate of Title Yolume 8442 Folio 95, now being Folio Identifier B/410947, together with the building erected thereon known as 108-120 Station Street, Wentworthville.

#### ANCILLARY RIGHTS:

87 Req:C145587 /Doc:DL Z606154 /Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:26 of 28

The right to use the Common Parts in so far as is necessary for the purposes of access to and egress from the Premises;

### LESSOR'S SERVICES:

The lighting, cleaning, maintenance, repairing (but not any repairing of a structural nature) and decorating of the Common Parts and the exterior and other parts of the Building not comprised or intended to be comprised in any lease of premises within the Building.

The provision of air conditioning heating and air ventilation to the Building during such times of the year as the Lessor may reasonably determine provided that if the necessary plant and equipment is not present and operational in the Building at the commencement of the Term the Lessor shall not be obliged to install such plant and equipment or provide air conditioning heating and air ventilation (as the case may be).

The provision of such caretaking and security services to the Building as the Lessor may from time to time reasonably deem necessary.

The operation and maintenance of any lifts and elevators in the Building.

The maintenance of the Lessor's fixtures in the Building (including any air conditioning heating and ventilation plant and equipment and lifts and elevators but excluding those included in the Fremises).

The provision and maintenance of all necessary fire fighting and fire prevention equipment to the Common Parts.

The maintenance of any landscaped areas of the Building including the cost of mowing irrigation and the replacement of plants and trees.

All other services from time to time and at any time provided by the Lessor for the more efficient running of the Building and the benefit of lesses of premises within the Building.

PAG00183/d1/13.11.90

AND where the Lessors Services include the maintenance of any item the Lessor shall be entitled to enter into maintenance contracts for the maintenance of such items and the cost to the Lessor of entering into, such contracts shall be part of the cost of providing the Lessors Services for the purposes of Clause 1.2.14.4. of Annexure "A".

conveyancine

ITEM 5:

f; lpT; sixdre /Src:W /WARNING; 74 Copy Supplied by LPI NSW for

#### INITIAL YEARLY AMOUNT:

87 Req:C145587 /Doc:DL Z606154 /Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seq:27 of 28

Seventeen thousand and thirty five dollars and twenty cents (\$17,035.20)

RENT COMMENCEMENT DATE:

1 November 1990

ITEM 7:

ITEM 6:

MANNER OF FAYMENT OF RENT:

By equal monthly instalments of one thousand four hundred and nineteen dollars and sixty cents (\$1,419.60) in advance on the first day of each month. Notwithstanding the aforesaid the Lessee shall pay a reduced monthly instalment of one thousand two hundred and ninety dollars and fifty four cents (\$1,290.54) for the initial six months of the Term. The reduced monthly instalment shall be payable by the Lessee provided that each instalment is paid to the Lessor on or before the due date. In the event the Lessee neglects to pay the reduced monthly instalment on or before the due date the Lessee shall pay rent in accordance with this Lease as if no reduced monthly instalment was available to the Lessee. The Lessee shall pay a proportionate part of such instalment or reduced monthly instalment on the signing of this Lease for the period from the Rent Commencement Date to the first day of the following month. In the event the term expires on a day other than the last day of a month the last instalment shall be reduced proportionately.

ITEM B:

REVIEW DATES:

1 November 1991 1 November 1992

ITEM 9:

LESSEE'S PERCENTAGE

7% provided that in the event that the net lettable area of the Building is increased or decreased the Lessee's percentage shall be reduced or increased accordingly.

ITEM 10:

PERMITTED USE: Hairdressing Salon.

ITEM 11:

ACCESS TIMES:

ITEM 12:

At all times.

PUBLIC LIABILITY COVER:

\$5,000,000.00

PAG00183/BB/-23-/021190

**ITEM 13:** HEAD LEASE: Not applicable. ITEM 14: FURTHER TERMI Three (3) years. ٠. • REVIEW DATES IN FURTHER LEASE: ITEM 15: 1 November 1994. 1 November 1995 ITEM 16: INITIAL YEARLY AMOUNT ATTRIBUTABLE TO CAR PARKING: Not applicable. . • . PERCENTAGE RATE FOR RENT INCREASES: ITEM 17:

/Rev:30-Jun-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:18 /Pgs:ALL /Seg:28 of 28 A4. Copy Supplied by LPI NSW for Conveyancing Purposes Only.

10% -

87 Reg:C145587 /Doc:DL Z606154

7Src:

weintoff:sixare

WARNING A4: CODY

PAG00183/BB/-24-/021190

С <u>к</u>			
89 Req:C14 )5	99 /Doc:DL Z193326 /Rev:20-Jul re /Src:W /WARNING: A4 Copy Suj	-2010 /Sts:OK.OK /Prt:27-Mar-	2015 10:43 /Pgs:ALL /Seg:1 of 2
RP 1 1985	STAMP DUTY		<b>7</b>
See an	C		193526
	μ (10 μη Α. 2009) Δ (10 μη Α. 2009) Δ (10 μη Α. 2009)		I SUCCE AND A SUCCESSION AND A SUCCESSIO
		LEASE	CAL " X DI
	AND THE REAL PROPERTY	REAL PROPERTY ACT. (900 L (To be lodged in duplicate)	
	- Contraction of the second seco	roctions for Completion issued as Form RP 18)	\$106
OCCUPTION	Torrens Title Reference	LAND of which LESSOR is registered proprietor If part or premises, see note (a) (ii)	Location
DESCRIPTION OF LAND Note (a)	FI B/410947	- <del>WHOLS</del> - PART	Wentworthville
		The premises described in the Reference Schedule	
	·	hereto being part building erected on the said land.	
		See anneyure A	
LESSOR Note (b)			
\~/	RALIAN PTY LIMITED of Un	it 53, Kilburn Towers, 1 Ad	dison Road, Manly.
	(the shares and   ECC/33) have be been as the L BOOPP	whether and and the charge	
LESSEE	(the abovenamed LESSOR) hereby leases to the LESSEE		OFFICE USE ONLY
WHICH OF STATE REVEN		ED of I Ruhe St Easth	buur
ALIERATION NOTE			OVER.
Note (h)	she land and above described, subject to the following Pi	NOR ENCUMBRANCES I. X586116	
PRIOR ENCUMBRANCES Note (d)	2Y184206	3	· · · · · · · · · · · · · · · · · · ·
TERM	for a TERM of Five (5) years co	mmencing on 26 41 / 89 and TERMINATING	0425 11 / 94 OFFICE USE ONLY
Note (8) Note (1)	(with an OPTION TO PURCHASE and/or an OPTION OF	Part: RENEWAL as set forth in <del>classe(s)-</del> 3 of :	ONE SCHEDULE TWO hereto, S. 11. 1994
Note (g) Note (h)	together with and reserving the rights and liberties set for SUBJECT TO the covenants and provisions:	rth in SCHEDULE ONE hereto), and	Purchasa Renewat
		ig Act, 1919 as are not expressly negatived or modified	
	(iii) set forth in SCHEDULE TWO hereto, which cov	enants and provisions shall be deemed to be Incorporat	
EXECUTION	DATE 24 HAN 1990 We hereby certify this lesse to be correct for the purpose	es of the Real Property Act, 1900.	PTY.
Nate (I)	. Signed in my presence by the lesser who is personally kno		ETAD B
	The Common Seal of RALIAN PT	l .	Je Seal
	LIMITED was hereunto affixed	~ Dela	a spent of
	in the presence of:	Director/Secretary Directo	Signature State
Note (i)	FOR EXECUTION BY GUARANTOR SH Signed is my presence by the leaves who is personally know	E OVER	INGS
	The Common Seal of J K N HOLD		mmon 2
	PTY LIMITED was hereunto affi	[2]	Seal ) - V
00	in the presence of:	Sedar .	J.S Hund a
	Address and occupation of Wilness . 4.	irector/Secretary Directo	-Signature of Larres-
TO BE COMPLETED BY LODGING PARTY	LODGED BY PHILLIPS FOX	CT OTHER	DCATION OF DOCUMENTS
Notes (j) and (k)	17th LEVEL,		Harewith.
	20 BOND STREET, SYDNEY. 2000		In R.G.O. with
	Delivery Bot Mignin 89-8907 REF. SCH/RE Checked D.Kessi C7 REGISTERED 602		Produced by
OFFICE USE ONLY		LI -19 Secondary Directions	
C	EB KA RF18 Signed Extra Fen	SE2 1990	
720		Dalivery Directions	
	l	<b>L</b>	
-		2001	05762

об Without Interest MARIE WAY AGER. MARIE WAY AGER. MARIE WAY AGER. MARIE WAY AGER. MARIE WAY AGER. MARY CHEDULE TWO, see anexurely hered) MARY SCHEDULE TWO, see anexurely hered MARY SCHEDULE TWO, see anexurely hered) MARY SCHEDULE TWO, see anexurely hered MARY SCH	Signed in my presence who is perconally know signature of Witness Signature of Witness (Hiod) Mame of Witness (Hidd) Mame of Witness (Hidd) Mame of Witness (Hidd) Mame of Witness (North (for continuation of the MAN of (for continuation of (for continuation of (	Signed in my presence by who is personally known signed in my presence by Name of Witness (HJOCK , ich MAR(E 'HOD Mane of Witness (HLOCK , ich Spire , ich MAR(E 'HOD Mane of Witness (HLOCK , ich Spire , ich MAR(E 'HOD MAR (for continuation of ) (for continuation of ) (for continuation of ) (for continuation of ) (for continuation of ) (for continuation of ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) ) )	Note: (n) and     Note: (n) and       Signed in my presence by the Guarantor, merenses to an in the frequence attended to may presence by the Guarantor, may be the frequence of main	<ul> <li>Каранана и и ургаваного у Сове со у Сове с</li></ul>	A more standard in my presence by the Guarantor, standard in my presence by the Guarantor, where of Mitness (BLOCK LETTRR)) (MAL Standards & Occupation of Mitness Malabase & Occupation of Mitness Mitness Malabase & Occupation of Mitness Malabase & Occupation of Mitness Malabase Malabase Malabase & Occupation of Mitness Malabase Malab	All and the second of the second seco			T	<u> </u>	╾╤╤╌╴	
об Without Interest MARIE WAY AGER. MARIE WAY AGER. MARIE WAY AGER. MARIE WAY AGER. MARIE WAY AGER. MARY CHEDULE TWO, see anexurely hered) MARY SCHEDULE TWO, see anexurely hered MARY SCHEDULE TWO, see anexurely hered) MARY SCHEDULE TWO, see anexurely hered MARY SCH	Signed in my presence who is perconally know signature of Witness Signature of Witness (HLOCI Mame of Witness (HLOCI (for centinution of fice MAN of (for centinution of fice MAN of (for centinution of fice MAN of (for centinution of fice MAN of (for centinution of fice MAN of	Signed in my presence by Signed in my presence by Name of withess (G) nume (for continuation of Nddress & occupation of Nddress & occupation of Nddress & occupation of Nddress & occupation of Nddress & man and Nddress & man and Ndd	Non (m) 2000 мани (л. 2000 на поли на развидатован станиватова на поли на выската с поли на	A       Who is personate by the Guarantor, signed in my presence by the Guarantor, who is personative by the Guarantor, i	All the state of the Guardinator of the Guardinator, who is presented by the Guardinator, who is the presence by the Guardinator, who is the control of white is the forter is the presence by the Guardinator is the forter is th	<ul> <li></li></ul>			ļ			
M. Marker Monther M.	Signed in my presence who is percondity know signature of Witness Mame of Witness (Hood Mame of Witness (Hood	Signed in my presence by who is personally known signature of Witness Mame of Witness (Hock Name of Mater Name of Witness (Hock Name of Mater Name of Mater Na	Иссы (п) ала и пуравелов ву the Guaranter инстидатор, иссы (п) ала и пуравелов у the Guaranter инстидатор, исс. (п) ала и пуравелов у the Guaranter инстидатор, исс. (пос. и и пуравелов у the Guaranter и и пос. (пос. и и пуравелов и и пос. (пос. и и пуравелов и и и и и и и и и и и и и и и и и и и	الله المستقدم المستقدين المستقدين المستقدين المستقدم المستقدين المستقد المستقدين المستقديين المستقدين المستقدين المستقدين المستقدين المستقدين المستقدي	<ul> <li></li></ul>	Standards for and handle of the Charachical Contract of the Charachical Contractio			ናትር ካን	i i		
OE WITNESS       ALD MAR MARK MARK MARK MARK MARK MARK MARK	Signed in my presence who is perconally know signature of Witness (Ho Name of Witness (Hoo Address & Occupation of size MANAA Of size MANAA Of size MANAA	FOUCIDENTIFIER (E) DIRECTION (F) NOTAL VANCE DI FOUCIDENTIFIER (E) DIRECTION (F) NOTAL (G) DENTIFIER FOUCIDENTIFIER (B) DIRECTION (C) NAME FOUCIDENTIFIER (B) DIRECTION (B) DIRECTION (C) NAME FOUCIDENTIFIER (B) DIRECTION (B) DIR	мое (а) под накачиватова натанато на под накачиватова натанато то под накачито и под накачите и	الله المن المن المن المن المن المن المن المن	الله المستقدمة المستقد	Manual Line Strand for a Manual Stand for a Manual Line Strand for a Manual Line Stand for a Manual Line Stand Line Line Stand Line Manual Line Stand Line Stand Line Line Stand Line Line Stand		· ·	1			
он интерек маларии Средски и продактивно и продактор об Истовая маларии об Сентер 108 - 120 540000 Моц. Мара Сентер 108 - 120 5400000 Моц. Мара Сентер 108 - 120 5400000 Моц. Мара Сентер 108 - 120 5400000000000000000000000000000000000	Signed in my presence who is perconally know signature of Witness (Ho Name of Witness (Hoo Address & Occupation of size MANAA Of size MANAA Of size MANAA	FOLO IDENTIFIER (E) DIRECTION (F) NOTAL (G) DENTIFIER (G) DIRECTION (C) NAME (G) DIRECTION (C) NAME (G) DIRECTION OF (G) DIRECTION OF (G) DIRECTION OF (G) DIRECTION OF (G) NAME (G) DIRECTION (G) DIRECTION (G) NAME (G) DIRECTION (G) NAME (G) DIRECTION (G) DIRECTION (G) NAME (G) DIRECTION (G)	Иссе (а) али и полька и польк	A Who is personally known to met Signiful to be personally known to met Who is personally known to met When of Witness Mane of Witness (BLOCK LETTERS) Mane of W	All the second states of the second of the second states and the second states and second states and the second states are second at the second states of th	All the stand definition of Without All and the ad Market All the stand definition of Without and Weller All the stand definition of Without and the advantant for meri Signature of Without LETTER (MARKET) All the stand for the Stand for the Stand for the Stand Market Market (Market Stand Letters) Market				Î		
Image: State of Mathematical State of Street No. 11/2     11/2/14/14/20 of Street No. 11/2       OF WITHORS     MANAACE A.       MARK CONTINUETION OF WITHORS     MANAACE A.       OCCUPATION OF WITHORS     MANAACE A.       MANAACE A.     MANAACE A.       MANAACE A.     MANAACE A.       MANAACE A.     MANAACE A.	Signed in my presence who is perconally know signature of Witness (Ho Name of Witness (Hoo Address & Occupation of size MANAA Of size MANAA Of size MANAA	FOUCIDENTIFIER (E) DIRECTION (F) WITH (G) DENTIFIER (E) DIRECTION (C) WARE (G) DIRECTION (G) DIRECTI	Staned in my presence by the Guarantor, who is personally known to merivativer, staned in my presence by the Guarantor, who is personally known to me: Name of Mitness (Hock LETTERS)     Mit Continued (for continued of SchEDULE TWO, so anoxuret) hareo)       Name of Mitness (Hock LETTERS)     Mits schemut find of Mits schemut fi	Alioned in my presence by the Guarantor,	Алек наявия наявляе и по	Кара санала или разволае при сър свое при сър с полека и про сър сър с состатърка) и про сър сър с состатърка) и про сър сър	for maily or 4PPI-11-25 saving	Wentworthyille. Ex		i		, <b>→</b> ⊾
OF WITHORS MARY CREDULE TWO, SEO ANDEXINE(S) REFOLD MARY CREDULE TWO, SEO ANDEXINE(S) REFOLD MARY CREDULE TWO, SEO ANDEXINE(S) REFOLD OCCUPATION OF WITHORS MARY AGER, MARY AGER, M	Signed in my presence who is perconally know signature of Witness (Ho Name of Witness (Hoo Address & Occupation of size MANAA Of size MANAA Of size MANAA	FOLO IDENTIFIER (E) DIRECTION (F) NOTAL (G) DENTIFIER (G) DIRECTION (C) NAME (G) DIRECTION (C) NAME (G) DIRECTION OF (G) DIRECTION OF (G) DIRECTION OF (G) DIRECTION OF (G) NAME (G) DIRECTION (G) DIRECTION (G) NAME (G) DIRECTION (G) NAME (G) DIRECTION (G) DIRECTION (G) NAME (G) DIRECTION (G)	Колов (м)         Колов (M)         <	<ul> <li>A who is personally known to mei</li> <li>A who</li></ul>			tearts noitets osi-sc	Ralian Centre, 10	1	ι Ι	No	
ог илгени милтерии постратиона милтерии постратиона милически милическ	Signed in my presence who is perconally know signature of Witness is anature of Witness Mame of Witness (HLOCI Name of Witness (HLOCI Adress & Occupation office MAN of Office MAN of MAN (c) NAME	Signed in my presence by who is personally known signature of Witness Mame of Witness (BLOCK , JEAN MARIE (F), HOD Name of Witness (BLOCK , JEAN MARIE (F), NAM AGE OFFICE MAM AGE OFFICE MAM AGE	Norse (m) and Signed in my presence by the Guarantor, Signature of Witness Market C. Honger, Market C. H	<ul> <li>المستقدم المستقدم المستقدمة المستقدة المستقدة</li></ul>	الالمالي المالالمالي المالي المالي المالي المالي المالي المعالمالي المعالمالي المعالمالي المعالمالي المعالمالي المعالم المعالمالي المعالم المعالم المعالمالي المعالم المعالمالي المعالم المعالم المعالم المعالم المعالم المعالم المعالم المعالم المعالم المعالم المعالم المعالم المعالم المعالم المعالم معالم المعالم	الالمحافية بالأسفاء بالغانا بلماء بالمابا بلالغاما بلالغاما بلالغاما بلالغاما با	How were for all the			·	<u> </u>	
of Witness MARIE	Another is personally know who is personally know who is personally know the second of Window Wame of Window Wame of Window Watess & Occupation of the manual Office MAM of Wates & Occupation Office MAM of	Algored in my presence by who is personally known signature of Witness bidnates & Occupation of Address & Occupation of Office MANAGE	Nores (m) and Stand in my presence by the Guarantor, who is personally known to me: Signed in my presence by the Guarantor, Signature of Witness Mane of Witness (BLOCK LETTERS) Mane of Witnes				DETAILS	(H)	DEVING	(e) NOICN (e)		
of Witness MARIE	Another is personally know who is personally know who is personally know the second of Window Wame of Window Wame of Window Watess & Occupation of the manual Office MAM of Wates & Occupation Office MAM of	Algored in my presence by who is personally known signature of Witness bidnates & Occupation of Address & Occupation of Office MANAGE	Nores (m) and Stand in my presence by the Guarantor, who is personally known to me: Signed in my presence by the Guarantor, Signature of Witness Mane of Witness (BLOCK LETTERS) Mane of Witnes							•		
of Witness MARIE	Another is personally know who is personally know who is personally know the second of Window Wame of Window Wame of Window Watess & Occupation of the manual Office MAM of Wates & Occupation Office MAM of	Algored in my presence by who is personally known signature of Witness bidnates & Occupation of Address & Occupation of Office MANAGE	Nores (m) and Stand in my presence by the Guarantor, who is personally known to me: Signed in my presence by the Guarantor, Signature of Witness Mane of Witness (BLOCK LETTERS) Mane of Witnes									
ог Witness Of Witness MARIE (Horcentinuation of SCHEDULE TWO, see annexure(s) hereto) (For centinuation of SCHEDULE TWO, see annexure(s) hereto) steer MANA-GER, (For centinuation of SCHEDULE TWO, see annexure(s) hereto)	Signed in my presence who is perconally know bignature of Witness is nature of Witness Home of Witness (BLOC) Name of Witness (BLOC)	Signed in my presence by who is personally known signature of Witness Name of Witness (HLOCK , 55, EAST MARGE, RX, N Name of Witness (HLOCK , 55, EAST MARGE, RX, 0 Name of Witness (HLOCK , 55, EAST MARGE, RX, 0 N, 10 N,	Norse (m) and Stand in my presence by the Guarantor, who is personally known to me: Standture of Witness Mill	المعلم المسلم المس	Stand in my presence by the cuarantoric with a submission of the submission of the cuarantoric is who is personally known to me: Significate of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS)	هنال المرافق المرافق المرافق من المحملة المحملة المحملة المحملة المحملة المحملة المحملة المحملة             هنال المرافق المرق			TINDEHOS LSEIL	ICI NYWE	NOITO3410 (8)	Valan David Offica
об Witness of Witness MARIE, Ro., N. EPPING Steptig. Ro., N. EPPING Occupation of Witness User MANAGER, NANAGER,	Signed in my presence who is personally know signature of Witness JEAN, MARIE, Ho Name of Witness (BLOCI Name of Witness (BLOCI NAME of Witness (BLOCI NAME of Witness (BLOCI OFFICE, MAN A	Signed in my presence by who is personally known Signature of Witness Mame of Witness (BLOCK 	Area (m) and Signed in my presence by the Guarantor, who is personally known to me:	Address is personally known to mei Signed in my presence by the Guarantor, الاس Signed in my presence by the Guarantor, الاسمارية Signed in my presence by the Guarantor, Signed in my presence by the Guarantor, Mame of Witness Signed in my presence by the Guarantor, Mame of Witness (BLOCK LETTERS) Mame of Witness Signed in my presence by the Guarantor, Mame of Witness Mame	المعند مستقد المستقدان المعادية ال معادية معادية المعادية المعادي المعادية المعادي معادية معادية المعادية الم	المعند مستمسل الله المالمانة بالالمعاد المعادين. Signed in my presence by the Guarantor, ال المعادين المحمد المحمد المحمد المحمد ا						
об Witness of Witness itness (Block Letters) stepter Rp. N. EPPING. Occupation of Witness	who is personally know who is personally know the second of Witness second Witness Mame of Witness (HLOCI Name of Witness (HLOCI Name of Witness (HLOCI Name of Witness (NLOCI Name OF	Signed in my presence by who is personally known Signature of Witness Name of Witness (HLOCK Name of Witness (HLOCK Name of Witness (HOCK Name of Witness (HOCK)	Nores (m) and Signed in my presence by the Guarantor, who is personally known to me: 	<ul> <li>A who is personally known to me: Signed in my presence by the Guarantor, Signed in my presence by the Guarantor, Mame of Witness (BLOCK LETTERS)         Signed in my presence by the Guarantor, Who is personally known to me: Signed in my presence by the Guarantor, Much dec. Much dec.</li></ul>	<ul> <li></li></ul>	Alterest attered in my presence by the Guarantor. Signed in my presence by the Guarantor. Signed in my presence by the Guarantor. Mame of Witness (BLOCK LETTERS) Signature of Witness Signature of Witness Signature of Witness Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness Signature of Witness Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness Mame of Witness (BLOCK LETTERS) Mame of Witness Mame of Witness	(X) asoN	Т.У.О, seo аппохите(s) hereto)	attion of \$CREDULE	(Гог севеілі		
об Witness of Witness itness (Block Letters) stepter Rp. N. EPPING. Occupation of Witness	who is personally know who is personally know the second of Witness second Witness Mame of Witness (HLOCI Name of Witness (HLOCI Name of Witness (HLOCI Name of Witness (NLOCI Name OF	Signed in my presence by who is personally known Signature of Witness Name of Witness (HLOCK Name of Witness (HLOCK Name of Witness (HOCK Name of Witness (HOCK)	Nores (m) and Signed in my presence by the Guarantor, who is personally known to me: 	<ul> <li>A who is personally known to me: Signed in my presence by the Guarantor, Signed in my presence by the Guarantor, Mame of Witness (BLOCK LETTERS)         Signed in my presence by the Guarantor, Who is personally known to me: Signed in my presence by the Guarantor, Much dec. Much dec.</li></ul>	<ul> <li></li></ul>	Alterest attered in my presence by the Guarantor. Signed in my presence by the Guarantor. Signed in my presence by the Guarantor. Mame of Witness (BLOCK LETTERS) Signature of Witness Signature of Witness Signature of Witness Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness Signature of Witness Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness Mame of Witness (BLOCK LETTERS) Mame of Witness Mame of Witness			•			
об Witness of Witness itness (Block Letters) stepter Rp. N. EPPING. Occupation of Witness	who is personally know who is personally know the second of Witness second Witness Mame of Witness (HLOCI Name of Witness (HLOCI Name of Witness (HLOCI Name of Witness (NLOCI Name OF	Signed in my presence by who is personally known signature of Witness Name of Witness (HLOCK Name of Witness (HLOCK Name of Witness (HLOCK Name of Witness (HLOCK Name of Witness of Occupation of	Nores (m) and Signed in my presence by the Guarantor, who is personally known to me: 	<ul> <li>A who is personally known to me: Signed in my presence by the Guarantor, Signed in my presence by the Guarantor, Mame of Witness (BLOCK LETTERS)         Signed in my presence by the Guarantor, Who is personally known to me: Signed in my presence by the Guarantor, Much dec. Much dec.</li></ul>	<ul> <li></li></ul>	Alterest states in ny presence by the Guarantor. Signed in my presence by the Guarantor. Signed in my presence by the Guarantor. Name of Witness (BLOCK LETTERS) Alteress is Occupation of Witness Signature of Witness Signature of Witness Mame of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS) Name of Witness Signature of Witness Name of Witness Name of Witness (BLOCK LETTERS) Name of Witness Name of Witness (BLOCK LETTERS) Name of Witness Name of Witness						
об Witness of Witness itness (Block Letters) stepter Rp. N. EPPING. Occupation of Witness	who is personally know who is personally know the second of Witness second Witness Mame of Witness (HLOCI Name of Witness (HLOCI Name of Witness (HLOCI Name of Witness (NLOCI Name OF	Signed in my presence by who is personally known signature of Witness Name of Witness (HLOCK Name of Witness (HLOCK Name of Witness (HLOCK Name of Witness (HLOCK Name of Witness of Occupation of	Nores (m) and Signed in my presence by the Guarantor, who is personally known to me: 	<ul> <li>A who is personally known to me: Signed in my presence by the Guarantor, Signed in my presence by the Guarantor, Mame of Witness (BLOCK LETTERS)         Signed in my presence by the Guarantor, Who is personally known to me: Signed in my presence by the Guarantor, Much dec. Much dec.</li></ul>	<ul> <li></li></ul>	Alterest attered in my presence by the Guarantor. Signed in my presence by the Guarantor. Signed in my presence by the Guarantor. Mame of Witness (BLOCK LETTERS) Signature of Witness Signature of Witness Signature of Witness Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness Signature of Witness Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS) Mame of Witness Mame of Witness (BLOCK LETTERS) Mame of Witness Mame of Witness						
DE WILLERS) DE WILLERS MARIE, RD, N. EPRING. MARIE, RD, N. EPRING.	who is personally know who is personally know the second of Witness signature of Witness MAMC	Signed in my presence by Wame of Witness (HLOCK Signature of Witness JU JU Signature of Witness JU Signature of Witness JU JU Signature of Witness JU JU Signature of Witness JU JU Signature of Witness JU Signature of Signature of Witness JU Signature of Signature of Sign	Vores (m) and Standard in my presence by the Guarantor, who is personally known to me: 	الم	<ul> <li></li></ul>	Address in my presence by the Guarantor. Who is personally known to me: who is personally known to me:		Det	14904'	ເ¥ຟ ,∄: ການທັກລາ	nis≓0 ∨ s asitut	
DE WIERS (BLOCK LETTERS)	Signed in my presence who is personally know signature of Witness Here of Witness Hame of Witness Name of Witness	Signed in my presence by Wame of Witness Signature of Witness JU JU Signature of Witness Mame of Mitness Mame of Mitness	Nores (m) and Signed in my presence by the Guarantor, Signature of Witness Name of Witness (BLOCK LETTERS)	الألك المسلحة المس	Altered in my presence by the Guarantor, Signed in my presence by the Guarantor, Name of Witness (BLOCK LETTERS) Signed in my presence by the Guarantor, Signed in my presence by the Guarantor, Mame of Witness Signed in my presence by the Guarantor, Mame of Witness Mame of Witness Signed in my presence by the Guarantor, Mame of Witness Mame of Wi	Algore representation of which is personally known to me: Signed in my presence by the Guarantor, Name of Witness (BLOCK LETTERS) Signed in my presence by the Guarantor, Mame of Witness (BLOCK LETTERS) Signed in my presence by the Guarantor, Signed in my presence by the Guarantor, Mame of Witness Signed in my presence by the Guarantor, Mame of Witness Signed in my presence by the Guarantor, Mame of Witness Mame of W		וּאַכּי	1999. N K.	1 3100	1.55. E Asi	
MARIE . HODGE	who is personally who is benets who is personally know who is personally know who is when is when is when is when is when is	Signed in my presence by signed in my presence by Signature of Witness Signature of Witness Signature of Witness	Neres (m) 20105 TWO HEREINBEFORE REFERENCE TWO HEREINBEFORE REFERENCE TWO HEREINBEFORE REFERENCE TWO Who is personally known to me: Nin	الالله       الالله         المالية       المالية	<ul> <li>Kanan the following tights and libration.</li> <li>Signed in my presence by the Guarantor, who is personally known to me:</li> <li>Signed in my presence by the Guarantor, who is personally known to me:</li> <li>Signed in my presence by the Guarantor, who is personally known to me:</li> <li>Signed in my presence by the Guarantor, who is personally known to me:</li> </ul>	<ul> <li>Kanon cannot be called in my presence by the Guarantor, signed in my presence by the Guarantor, who is personally known to me;</li> <li>Signed in my presence by the Guarantor, signed in my presence by the Guarantor, officies for MANE (Popper, signed in my presence by the Guarantor, who is personally known to me;</li> </ul>						
	eoneserq ym ni bengig Word yn my presence Word yn	nwony presence by presence by presence by presence by	Vores (m) зала Signed in my presence by the Guarantor, who is personally known to me:	<ul> <li>المجمع في معرف المحتفيل المسلم المحتود المحتفيل المسلم المحتود المحتفيل المسلم المحتود المحتفيل المسلم المحتفيل المحت المحتفيل المحتفيل المحت محتفيل المحتفيل الحتا الححتا الححتفي الححتفي الححتفي الححتفي الح</li></ul>	<ul> <li>Stigned in my presence by the Guarantor.</li> <li>Stigned in my presence by the Guarantor.</li> <li>Stignet in my presence by the Guarantor.</li> <li>Mame of Witness (BLOCK LETTERS)</li> <li>Mame of Witness (BLOCK LETTERS)</li> <li>Address &amp; Occupation of Witness</li> <li>Stigned in my presence by the Guarantor.</li> </ul>	<ul> <li>Signed in my presence by the Guarantor.</li> <li>Signed in my presence by the Guarantor.</li> <li>Signed in my presence by the Guarantor.</li> <li>Mame of Witness (BLOCK LETTERS)</li> <li>Address &amp; Occupation of Witness</li> <li>Address &amp; Occupation of Witness</li> <li>Signed in my presence by the Guarantor.</li> </ul>			Hodde	. อิเงงเ	YEAN A	7,≢
	eoneserq ym ni bengig Word yn my presence Word yn	nwony presence by presence by presence by presence by	Vores (m) зала Signed in my presence by the Guarantor, who is personally known to me:	<ul> <li>المجمع في معرف المحتفيل المسلم المحتود المحتفيل المسلم المحتود المحتفيل المسلم المحتود المحتفيل المسلم المحتفيل المحت المحتفيل المحتفيل المحت محتفيل المحتفيل الحتا الححتا الححتفي الححتفي الححتفي الححتفي الح</li></ul>	المحمد مستحمد مستحمد المسلم المعالمة: المحمد مستحمد مستحما المعالمة: المحمد مستحمد مستحمد المعالمة: المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد محمد محمد محمد محمد محمد محمد محمد	<ul> <li>Signed in my presence by the Guarantor.</li> <li>Signed in my presence by the Guarantor.</li> <li>Signed in my presence by the Guarantor.</li> <li>Mame of Witness (BLOCK LETTERS)</li> <li>Address &amp; Occupation of Witness</li> <li>Address &amp; Occupation of Witness</li> <li>Signed in my presence by the Guarantor.</li> </ul>	$\rho$	•	<i>?</i> ?.	өцттм Э	o etnieusi	S
	Signed in my presence	Signed in my presence b	Stgned in my presence by the Guarantor, Netservise of and in my presence by the Guarantor,	<ul> <li>Stgned in my presence by the Guarantor, who is personally known to me:</li> <li>Stgned in my presence by the Guarantor, is and my presence by the Guarantor.</li> <li>Stgned in my presence by the Guarantor, Stgned in my presence by the Guarantor.</li> </ul>	Address in the fillowing represence by the Guarantor, signed in my presence by the Guarantor, by the is personally known to me:	Address & Occupation of Withous Address & Occupation & Occup	AID Hune Electer			50	<i>y</i>	
	CIERCHED .	"CMEDAT	pur (m) soton	A who is personally known to me: who is personally known to me: Signeture of Witness Wame of Witness (BLOCK LETTERS) Wame of Witness (BLOCK LETTERS) Water of Witness (BLOCK LETTERS) Water of Witness (BLOCK LETTERS) Water of Witness (BLOCK LETTERS)	المندمة معدمة المعدة المعدة عنومة في معرف المعدة المعدة المالية عن المعدة المحافة المعدة المعدة المعنة المعدة معدة المعدة المعلم المعلم الم	Leaver reterent and himself the following rights and liberide: Signed in my presence by the Guarantor. Signeture of Witness Name of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS)		enarantor, Guarantor,	киоми го ше исе рћ гие	vileac	m ni bengi areq ai or	4m FS
	CENED	Indemos.	pur (m) soton	A who is personally known to me: who is personally known to me: Signeture of Witness Wame of Witness (BLOCK LETTERS) Wame of Witness (BLOCK LETTERS) Water of Witness (BLOCK LETTERS) Water of Witness (BLOCK LETTERS) Water of Witness (BLOCK LETTERS)	Lourn reterves and pinnell the following rights and liberide: Signed in my presence by the Guarantor. (۱)) Signeture of Witness Name of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS)	Learce reares and planning regions refer to make the following regions in the following region of the Guarantor. Signed in my presence by the Guarantor. (۱)) Signeture of Witness (۱)) Mame of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS)						
			Address & Occupation of Witness Office Anam Ceres.	Signed in my presence by the Guarantor, who is personally known to me: ()) Signature of Witness (HLOCK LETTERS) Vame of Witness (HLOCK LETTERS) ()) ()) ()) ()) ()) ()) ()) ()) ()) (	Leaver reserves and planell the following rights and liberide: Signed in my presence by the Guarantor, Mame of Witness (BLOCK LETTERS) Signeture of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS) Signet of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS)	Leuce reures und planell the following right and liberide: Signed in my presence by the Guarantor, Mame of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS)		₩ <u>₩₩₩₩₩₩₩₩₩₩₩₩₩₩₩</u> ₩	CWT-BILLOW	72		
ISE MANAGER.	OFFICE MANAG	ANAM BUTTO	SSOUIM IO NOIIBUUDDO 3 SSOTABA	Signed in my presence by the Guarantor, who is personally known to me: ()) Signature of Witness (HLOCK LETTERS) Vame of Witness (HLOCK LETTERS) ()) ()) ()) ()) ()) ()) ()) ()) ()) (	Leaver reserves and planell the following rights and liberide: Signed in my presence by the Guarantor, Mame of Witness (BLOCK LETTERS) Mame of Witness (BLOCK LETTERS)	Leaver new way with the following right and liberide: Signed in my presence by the Guarantor, who is personally known to me: Signeture of Witness Name of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS)			AGER.	NAM 3	91330	
Cocupation of Withess Occupation of Withess	n roitsouso 3 searbha	Tricest Proposition & Section 4	'SRILLE' KO'' W EWINE'	Signed in my presence by the Guarantor, Signeture of Witness Wame of Witness (BLOCK LETTERS) Wame of Witness (BLOCK LETTERS)	Leaver reserves and planell the fellowing rights and the relation. Signed in my presence by the Guarantor, Signeture of Witness Mame of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS)	Leuce neuron unico himmell the following rights and liberide: Signet in my presence by the Guarantor, Signeture of Witness Name of Witness (BLOCK LETTERS) Name of Witness (BLOCK LETTERS)		550	Philip Price Price	ערב גיי ערב גיי	olecs south	₩¥ ₩
		7 N Y J J J J J J J J J J J J J J J J J J		Signed in my presence by the Guarantor. No. 10, 10, 10, 10, 10, 10, 10, 10, 10, 10,	Leaver reserves and planell the relieves read in my presence by the Guarantor. Signed in my presence by the Guarantor. I(.) II(.) Signeture of Witness II(.)	Learner and himsell the relieved the relieved to the Guarantor. Signed in my presence by the Guarantor. My My Michael Signeture of Medicaeles My My Michaeles My My Michaeles My My My Michaeles My My M						
			AND O MILLOY NOTAL SOULTM ID OUDN	Signed in my presence by the Guarantor, Who is personally known to me: 10, 11, 11, 12, 11, 12, 11, 12, 11, 12, 11, 12, 12	Leaver reserves and planell the fellowing right and liberide: Signed in my presence by the Guarantor, My for is personally known to me: 10,1	Leuror neuron unico himmell the following rights and liberide: Signed in my presence by the Guarantor, My for the Guarantor, If, If, If, If, If, If, If, If		••		IA) SSOU		
	Name of Witness (BLOCH	Name of Witness (BLOCK ]		Signed in my presence by the Guarantor.	Leuror returnes unto himmell the following right and internet. Signed in my presence by the Guarantor, II(,	Leuror resurves unto himmell the following right and the more the Guarantor.		••	LOCK LETTER			
	AFAN NAAL Name of Witness (BLOCP	NAME OF WITHOSS (HLOCK )	JEAN MARE HODDE	Signed in my presence by the Guarantor,	Concernant units have all the following right and fiber de Cuarantor, concernant units have a fiber de Cuarantor.	Concor reserves unto himmell the following right and fiberries:		••	LOCK LETTER	. FIAAM	NUST	
	Signeture of Witness Signeture of Witness Vame of Witness (BLOCH	II (, Signature of Witness Name of Witness (BLOCK)	Il. John Moder (101) Brenten Stender Brenden MARK NASL	,	soistedil-bac stilgit gaiwollai udt lleunid otau tevuses rosset	seistedil bac stägt gaiwolla) udt lleunid atnu soviusen rossed.	6 Work (SS)	( <u>5</u> )	LOCK LETTER	entine: ₩AAR	ле¥и одибтике)ог	rs N(`
soustly known to me:	who is personally know Signature of Witness Mame of Witness (BLOCH	When the personally known is personally known it. Wi, when the search of Witness (BLOCK) is the sea	Who is personally known to me: 10, 11, 11, 11, 11, 11, 11, 11,				in the find a	( 5)	LOCK LETTER	MAAR.	JEAN	Г5 "И(` цм ү
	A A C A C A C A C A C A C A C A C A C A	A NO TRASTERNA ON N.A		Signed in my presence by the Guarantor. () Signeture of Witness Signeture of Witness	Leaver same himsel the following signed in my presence by the Guarantor.	Leuror reserves unto himmel the following rights and liberide: Signed in my presence by the Guarantor, I(c) I(c		••	***		оти то ещи	
			Name of Witness (BLOCK LETTERS)	Signed in my presence by the Guarantor.	(10) Who is personally known to me: Signed in my presence by the Guarantor, MWD is personally known to me:	Lauor resurves unto himself the relation of the cuaranter for the cuaranter of the function of the cuaranter		••		(A) SSAN		
	Name of Witness (BLOCH	Name of Witness (BLOCK ]		Signed in my presence by the Guarantor.	Leuror reserves unto himmell the following right and the cuartent come: Signed in my presence by the Guarantor, If () If ()	Leucor result value de following right and fiberdet: Leucor result of the following right and fiberdet: Leucor result of the following right of the Guarantor, Leucor result of the following right of the following: Leucor result of the following right		••	LOCK LETTER			
	AFAN NAAL Name of Witness (BLOCP	Vame of Witness (BLOCK)	JEAN MAAR HODDE	Ano is personally known to me: Signed in my presence by the Guarantor,	Anto is personally known to me: Signed in my presence by the Guarantor.	Kenter marken unto his pressonal.Ly known to me: Signed in my presence by the Guarantor.		••	LOCK LETTER	. FIAAM	NUST	
	Vame of Witness (BLOCH	Signeture) of Witness (BLOCK ) Vame of Witness (BLOCK )	JEAN MAAR HODGE	Signed in my presence by the Guarantor,	Concorrent with price of the full of the f	Concorner and the following file and filestrict by the Guarantor, and the following th		••	LOCK LETTER	. FIAAM	NUST	TS.
	Signeture of Witness Signeture of Witness Vame of Witness (BLOCH	N(, Signeture) of Witness Name of Witness (BLOCK) Name of Witness (BLOCK)	H() Denne Denne Abordes Brandture De With Mail BAAM NABL	,	seistedil ber stägt griwollel udt lleunid eten sees resser	seistedil bac sidgit gaiwolle) ude lleunid ann souwsen rossed.	of my way	( <u>5</u> )	LOCK LETTER	entine: ₩AAR	ле¥и одибтике)ог	rs N(`
soustly known to me:	who is personally know Signature of Witness Mame of Witness (BLOCH	Who is personally known signature of Witness Mame of Witness (BLOCK)	Who is personally known to me: 1/1/1/2 Aborder 1/1/1/2 Aborder 1/2/1/2	186/1668/ DUC 3100/11 301/0010 / PUT 300/200/200/200/200/200/200/200/200/200/			icen freeday	( 5)	LOCK LETTER	MAAR.	JEAN	Г5 "И(` цм ү
ROUBLY KNOWN TO THE: WY DIESENCE DY THE GUSTANTOL'	Signed in my presence who is personally know figneture of Witness Signeture of Witness Mame of Witness (BLOCH	Signed in my presence by who is personally known (, )   Signeture of Witness (BLOCK ) Mame of Witness (BLOCK )	Lenor neuron anto himmell the following right and the market of the Guarantor. Signature of Witness N() Signature of Witness N() Signature of Witness N() N() N() N() N() N() N() N()					( 5)	COCK FELLES HOR CELLES COOM TO ME COOM TO ME TOG DA FUG	WWWE MITUS MITUS MITUS MITUS MITUS	u is perso o is perso of ture of of ture of t	te dw λ t t 2
ROUBLY KNOWN TO THE: WY DIESENCE DY THE GUSTANTOL'	Signed in my presence who is personally know figneture of Witness Signeture of Witness Mame of Witness (BLOCH	Signed in my presence by who is personally known (, )   Signeture of Witness (BLOCK ) Mame of Witness (BLOCK )	Lenor neuron anto himmell the following right and the market of the Guarantor. Signature of Witness N() Signature of Witness N() Signature of Witness N() N() N() N() N() N() N() N()					( 5)	COCK FELLES HOR CELLES COOM TO ME COOM TO ME TOG DA FUG	WWWE MITUS MITUS MITUS MITUS MITUS	o is perso of ture of of ture of ture of of ture of ture of of ture of ture of ture of of ture of ture of ture of of ture of ture of ture of ture of ture of ture of of ture of ture o	te dw λ t t 2
ROUBLY KNOWN TO THE: WY DIESENCE DY THE GUSTANTOL'	Signed in my presence who is personally know figneture of Witness Signeture of Witness Mame of Witness (BLOCH	Signed in my presence by who is personally known (, )) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,)))) (,))) (,)))) (,)))) (,)))) (,))))) (,))))))))	Lenor reverses and blandl the following right and the roll the following the rad the function. Signed in my presence by the Guarantor, I(,) I					( 5)	COCK FELLES HOR CELLES COOM TO ME COOM TO ME TOG DA FUG	WWWE MITUS MITUS MITUS MITUS MITUS	o is perso of ture of of ture of ture of of ture of ture of of ture of ture of ture of of ture of ture of ture of of ture of ture of ture of ture of ture of ture of of ture of ture o	te dw λ t t 2
ROUBLY KNOWN TO THE: WY DIESENCE DY THE GUSTANTOL'	Signed in my presence who is personally know figneture of Witness Signeture of Witness Mame of Witness (BLOCH	Signed in my presence by who is personally known (, )   Signeture of Witness (BLOCK ) Mame of Witness (BLOCK )	Lenor neuron anto himmell the following right and the market of the Guarantor. Signature of Witness N() Signature of Witness N() Signature of Witness N() N() N() N() N() N() N() N()					( 5)	COCK FELLES HOR CELLES COOM TO ME COOM TO ME TOG DA FUG	WWWE MITUS MITUS MITUS MITUS MITUS	o is perso of ture of of ture of ture of of ture of ture of of ture of ture of ture of of ture of ture of ture of of ture of ture of ture of ture of ture of ture of of ture of ture o	te dw λ t t 2
ROUBLY KNOWN TO THE: WY DIESENCE DY THE GUSTANTOL'	Signed in my presence who is personally know figneture of Witness Signeture of Witness Mame of Witness (BLOCH	Signed in my presence by who is personally known (, )   Signeture of Witness (BLOCK ) Mame of Witness (BLOCK )	Lenor neuron anto himmell the following right and the market of the Guarantor. Signature of Witness N() Signature of Witness N() Signature of Witness N() N() N() N() N() N() N() N()					( 5)	COCK FELLES HODOG COOMU FO MG COOMU FO MG TOG DA FUG	WWWE MITUS MITUS MITUS MITUS MITUS	o is perso of ture of of ture of ture of of ture of ture of of ture of ture of ture of of ture of ture of ture of of ture of ture of ture of ture of ture of ture of of ture of ture o	te dw λ t t 2
ROUBLY KNOWN TO THE: WY DIESENCE DY THE GUSTANTOL'	Signed in my presence who is personally know figneture of Witness Signeture of Witness Mame of Witness (BLOCH	Signed in my presence by who is personally known (, )   Signeture of Witness (BLOCK ) Mame of Witness (BLOCK )	Lenor neuron anto himmell the following right and the market of the Guarantor. Signature of Witness N() Signature of Witness N() Signature of Witness N() N() N() N() N() N() N() N()					( 5)	COCK FELLES HODOG COOMU FO MG COOMU FO MG TOG DA FUG	WWWE MITUS MITUS MITUS MITUS MITUS	o is perso of ture of of ture of ture of of ture of ture of of ture of ture of ture of of ture of ture of ture of of ture of ture of ture of ture of ture of ture of of ture of ture o	te dw λ t t 2
ROUBLY KNOWN TO THE: WY DIESENCE DY THE GUSTANTOL'	Signed in my presence who is personally know figneture of Witness Signeture of Witness Mame of Witness (BLOCH	Signed in my presence by who is personally known (, )) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,))) (,)))) (,))) (,))) (,)))) (,)))) (,)))) (,)))) (,)))) (,))))))))	Leure neuron unic blanning right and überden. Signed in my presence by the Guarantor. M(A) Signeture of Witness M(A) Signeture of Witness M(A) M(A) M(A) M(A) M(A) M(A) M(A) M(A)					( 5)	COCK FELLES HODOG COOMU FO MG COOMU FO MG TOG DA FUG	WWWE MITUS MITUS MITUS MITUS MITUS	o is perso of ture of of ture of ture of of ture of ture of of ture of ture of ture of of ture of ture of ture of of ture of ture of ture of ture of ture of ture of of ture of ture o	te dw λ t t 2
ROUBLY KNOWN TO THE: WY DIESENCE DY THE GUSTANTOL'	Signed in my presence who is personally know figneture of Witness Signeture of Witness Mame of Witness (BLOCH	Signed in my presence by who is personally known (, )   Signeture of Witness (BLOCK ) Mame of Witness (BLOCK )	Lenor neuron anto himmell the following right and the market of the Guarantor. Signature of Witness N() Signature of Witness N() Signature of Witness N() N() N() N() N() N() N() N()					( 5)	COCK FELLES HODOG COOMU FO MG COOMU FO MG TOG DA FUG	WWWE MITUS MITUS MITUS MITUS MITUS	o is perso of ture of of ture of ture of of ture of ture of of ture of ture of ture of of ture of ture of ture of of ture of ture of ture of ture of ture of ture of of ture of ture o	te dw λ t t 2
Property known to me: My presence by the Guarantor, Moust Hy-and (m)	o himeil the following (ght and therein: who is personally know who is personally know	Who is personally known (, ) II (, ) Signeture of Witness (BLOCK ) (Ame of Witness (BLOCK )	Noret for an any presence by the Guarantor. Who is personally known to me: (A) (A) (A) (A) (A) (A) (A) (A)					( 5)	LOCK LETTER	Sentin 3	Here	FS IN(' FS wild cann any page 2000



25

#### SCHEDULE ONE HEREINBEFORE REFERRED TO

f:lpi;sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

Req:C145589 /Doc:DL Z193326

#### PART 1 - INTERPRETATION AND DEFINITIONS

- 1.1 In this Lease unless the context otherwise requires a reference to:
  - 1.1.1 this Lease shall include the Reference Schedule to this Lease the contents of which Reference Schedule shall be read and construed as if they were set out in the body of this Lease;

/Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:3 of

- 1.1.2 an Item number means the respective Items set out in the Reference Schedule to this Lease;
- 1.1.3 a clause or part number means the respective clauses and parts of this Lease;
- 1.1.4 one gender includes a reference to the other genders and each of them;
- 1.1.5 a person includes a reference to a corporation or firm and vice versa;
- 1.1.6 words importing the singular number shall be deemed to include the plural number and vice versa;
- 1.1.7 any statutory provision shall be construed as a reference to that provision as respectively amended or re-enacted (either before or after the date of this Lease) from time to time;
- 1.1.8 any person or party to this Lease means and includes the legal personal representatives or permitted assigns of such person or party as the circumstances may require;
- 1.1.9 the use of any word or phrase which includes any one or more parties persons documents facts securities or events shall be given effect to as extending to meaning and binding all such parties persons facts events documents or securities jointly and each and every one of them severally as the circumstances may require;
- 1.1.10 decorate or decoration in the context of the Premises means to clean and prepare in a good and workmanlike manner and then to paint with at least two coats of paint all parts of the Premises previously painted and to paper varnish and otherwise treat all part of the Premises previously so treated and to clean repaint make good and colour all brickwork stonework and plaster at the times required by the Lessor but not more frequently than once during the Term; and
- 1.1.11 determination by a Valuer or other independent person of any dispute or matter arising pursuant to this Lease means determination by such independent person acting an expert and not as an arbitrator.
- 1.2 In this Lease unless the context otherwise requires the words and expressions set out below shall have the respective meanings attributed to them as follows:
  - 1.2.1 "Building" means the land and improvements erected thereon described in Item 2 as modified extended altered or added to at any time.
  - 1.2.2 "Common Parts" means any passageways, walkways, staircases, parades, elevators, lifts, entrances, lobbies, foyers, foyer decorations, gardens, pavements, lawns and other parts of the

AG90004/BB

*ዚ* ፓ ከ 4<sub>CPD1</sub>

TOLISIXORE /STC:W



Building not demised or intended to be demised by the Lessor which are for the common use enjoyment or benefit of the lessees and other occupants of premises in the Building notwithstanding that they may not be of equal use enjoyment or benefit to all such lessees and occupants and notwithstanding that they may also be used enjoyed by or be of benefit to the public and Common Parts includes any car parking areas notwithstanding that any particular Lessee or occupant may have the right to use any such area or any part thereof.

- 1.2.3 "Further Term" means the period specified in Item 14.
- "Guarantor" means the person or persons (if any) specified in 1.2.4 Item 16.
- 1.2.5 "Index Number" means the Consumer Price Index Number for Sydney all groups as published from time to time by the Australian Bureau of Statistics.
- 1.2.6 "Insurance Cost" means such yearly sum as shall represent:
  - 1.2.6.1 the premium or premiums (or in the event of the Building being insured with other premises the proportion of such premium or premiums as is determined by the Lessor as being attributable to the Building whose determination shall except in case of manifest error be final and binding on the Lessee) paid or payable by the Lessor for insuring the Building in such sum as the Lessor considers at any time to be the full reinstatement value thereof (and including the cost of professional fees and other incidental expenses, the costs of demolition and shoring up and the removal of debris) against the Insured Risks

the premium or premiums paid or payable by the 1.2.6.2 Lessor for insuring against loss of rent from time to time payable pursuant to this Lease and other leases of premises within the Building for such period as the Lessor shall require including the anticipated loss of rent following a rent. review; and

1.2.6.3 the premium or premiums paid or payable by the Lessor for insuring all third party property owners and employers risks in relation to the Building and persons employed by the Lessor in or about the Building.

- 1.2.7 "Insured Risks" means fire lightning storm tempest and such other risks as the Lessor may from time to time reasonably consider it expedient to insure the Building against.
- 1.2.8 "Lessee" means the person referred to as the Lessee on the front page of this Lease and includes his successors in title and permitted assigns.
- 1.2.9 "Lessee Party" means and includes the Lessee and its servants agents employees contractors or visitors (whether or not by invitation) and any person claiming through or under the Lessee or any person under the control or direction of the

June Log 00004/BB/ kem



1.2.10 "Lessee's Percentage" means the percentage specified in Item 9.

- 1.2.11 "Lessor" means the person referred to as the Lessor on the front page of this Lease and includes the persons for the time being entitled to the reversion immediately expectant upon the term hereby granted.
- 1.2.12 "Lessor's Services" means the services specified in Item 4 and any other service now or hereafter provided by the Lessor (at its discretion) for the benefit of the lessees and other occupants of premises in the Building notwithstanding that any such service may not be of equal benefit to all those lessees and may also be of benefit to members of the public.

1.2.13 "Outgoings" means and includes:

89 Reg:C145589 /Doc:DL Z193326

L: IDI: SIXARE /STC:W

1.2.13.1 all rates taxes (including State Land Tax but on the assumption that the Building is the only property owned by the Lessor) charges impositions and fees of any kind at any time payable by the Lessor to any Federal or State government or local or other authority in respect of the Building or any part thereof charged to or payable by the Lessor provided that if the Building is not separately assessed or charged in respect of any such rates taxes charges and impositions there shall be included in the Outgoings the Lessor's estimate (which shall be final and binding on the Lessee except in the case of manifest error) of the rates taxes charges impositions or fees (as the case may be) attributable to the Building;

1.2.13.2

all charges incurred by the Lessor in relation to the supply of electricity, gas, water sewerage and drainage to and the removal of waste and other garbage from the Building provided that if the Building is not separately charged in respect of any such service there shall be included in the Outgoings the Lessor's estimate (which shall be final and binding on the Lessee except in the case of manifest error) of the charges attributable to the Building;

1.2.13.3 all charges for lighting, power, heating, air conditioning and ventilation incurred by the Lessor in relation to the Building;

- 1.2.13.4 the total cost from time to time to the Lessor of providing the Lessor's Services;
- 1.2.13.5 the Insurance Cost;

1.2.13.6 all management fees.

1.2.14 "Premises" means that part of the Building referred to on the front page of this Lease which for the purposes of obligation as well as grant shall exclude any Common Parts but include the surface of all internal walls floors and ceilings and all floor and wall coverings, all doors and door frames, all windows and window frames, all internal partitioning (erected by the Lessor or the Lessee or any other person at any time) and all the Lessor's fixtures and fittings (excepting any

June Lod PAG90004/BB/-3-ASD kom



heating air conditioning and ventilation plant not exclusively serving the Premises).

- 1.2.15 "Rent" means the initial yearly amount specified in Item 5 together with and including any increases thereto in accordance with the provisions of Clause 4.2.
- 1.2.16 "Rent Commencement Date" means the date specified in Item 6.
- 1.2.17 "Review Dates" means those dates specified in Item 8.
- 1.2.18 "Specified Rate" means the rate which is three per centum above the annual interest rate from time to time and at any time charged by the Lessors principal bankers on overdrafts of more than \$100,000.00.
- 1.2.19 "Term" means the term granted by this Lease.
- 1.2.20 "Valuer" means a valuer (with experience in assessing for properties of the same nature as the Premises) agreed upon by the Lessor and the Lessee or failing such agreement a valuer appointed by the Lessor, but if otherwise required by the Lessee, and advised prior to the Lessor nominating a Valuer, by a Valuer appointed by the President or Secretary for the time being of the Australian Institute of Valuers.
- 1.3 When two or more persons comprise the Lessee or Guarantor all the covenants conditions terms and restrictions bind such persons jointly and each of them severally and also bind the respective personal representations assigns and successors in title of each of them jointly and severally.
- 1.4 The covenants powers and provisions implied in leases by virtue of the Conveyancing Act 1919 (as amended) are hereby expressly negatived except in so far as they or some part of parts thereof are included in the provisions expressed in this Lease.
- 1.5 In the following cases any reference in this Lease to the Lessor is deemed to include a reference to any superior lessor or any mortgagee from the Lessor or any superior lessor of the Premises (whether with or without other premises) or any part thereof:
  - 1.5.1 where there are rights easements and reservations exercisable by or benefiting the Lessor;
  - 1.5.2 where there is an obligation to obtain consent from the Lessor; and
  - 1.5.3 where there are any indemnities in favour of the Lessor.
- 1.6 Any covenant or agreement by the Lessee not to do or omit any act or thing is deemed to extend to an obligation not to permit any third party to do or omit the same.
- 1.7 Any approval consent permission or notice given pursuant to this Lease is not valid unless in writing.
- 1.8 Marginal notes and headings where used in this Lease are for the purpose of identification and location only and not to be considered in the interpretation of the provisions of this Lease.
- 1.9 This Lease shall be read and construed and take effect in accordance with the laws of the State or Territory in which the Premises are situate.

Kem

PAG90004/88/-4-/13-12-89

June Dodst Kullander



/Sec:7

of 25

1.10 The Lessor shall be entitled to exercise any right on its behalf expressed in or implied by this Lease by itself, its employees agents servants or contractors.

/Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL

f:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only

89 Req: C145589 /Doc: DL Z193326

- 1.11 To the extent permitted by law the application to this Lease and the provisions hereof or any moratorium or other Commonwealth or State statute ordinance rule or regulation which reduces or postpones the payment of rent or extends the Term or otherwise affects the operation of any of the provisions of this Lease to the detriment of the Lessor is hereby expressly excluded and negatived.
- 1.12 The Lessor and the Lessee hereby agree and declare that any provision of this Lease which is or shall be or become in breach of the Trade Practices Act, 1974 or any other Commonwealth or State statute rule or regulation and in consequence of such breach is void voidable unenforceable or invalid shall in any such case and for so long as it is in breach as aforesaid be severable from this Lease and this Lease shall be read and construed as if such provision was not expressed herein.

## PART 2 - ANCILLARY RIGHTS, EXCEPTIONS AND RESERVATIONS

- 2.1 This demise shall include the rights (if any) specified in Item 3 and in the event that the rights so specified include the right to park any motor vehicles in any part of the Building the following provisions shall apply:
  - 2.1.1 The Lessee shall park the motor vehicles only in the positions designated from time to time by the Lessor.
  - 2.1.2 The Lessee shall not permit or allow any motor vehicle to be cleaned greased oiled washed or repaired in any part of the Building.
  - 2.1.3 The Lessor shall not be held responsible for the loss of or damage to any car entering leaving or parked in the car parking area or for the loss of or damage to any article or thing in or upon any car or for any injury to any person howsoever such loss damage or injury may arise or be caused.
- 2.2 There is excepted and reserved from this demise as follows:
  - 2.2.1 the right to all times upon giving reasonable prior notice to the Lessee (except in the case of emergency) for the Lessor to enter upon the Premises for any purpose in connection with its rights duties and obligations expressed in or implied by this Lease or otherwise arising from its possession of the Building;
  - 2.2.2 the free and uninterrupted passage and running at all times of water soil gas electricity and other services from and to all other parts of the Building and any adjoining or neighbouring premises whether belonging to the Lessor or not through and along the conduits which are now or may hereafter during the Term be in the Premises and the right at any time during the Term for the Lessor to lay move remove replace and maintain any such conduits:
  - 2.2.3 all rights of light or air now subsisting or which might (but for this exception) be acquired over any neighbouring land or premises; and
  - 2.2.4 as may be specified in the Certificate(s) of Title or title deeds in respect of the land upon which the Building is erected.

AG90004/BB/~ 5-/13,12.89

KCM



/Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:8 of

/WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

PART 3 - HOLDING OVER AND OPTION FOR FURTHER TERM

89 Req:C145589 /Doc:DL Z193326

: Ip1:S1xdre /Src:W

- 3.1 If the Lessee with the consent of the Lessor continues to occupy the Premises beyond the expiration of the Term otherwise than pursuant to a further lease granted by the Lessor to the Lessee, he shall do so as a monthly tenant only, at a rental payable monthly in advance equal to one-twelfth of the Rent lastly payable under this Lease and such tenancy shall be terminable at any time by either party giving to the other one (1) month's written notice but otherwise shall be subject to such of the provisions hereof as are not inconsistent with a monthly tenancy.
- 3.2 If the Lessee gives the Lessor not more than six (6) and not less than three (3) months prior written notice (from the date of expiry of the Term) that it wishes to renew this Lease for the Further Term and provided the Lessee has duly performed all of its obligations pursuant to this Lease throughout the Term then the Lessor shall upon the expiry of this Lease grant to the Lesses a lease for the Further Term upon the same provisions (including any guarantee) as are contained in this Lease but amended as follows:
  - 3.2.1 The initial yearly amount to be specified in Item 5 of the Reference Schedule of such further lease shall be the greater of:
    - 3.2.1.1 the amount agreed upon by the Lessor and the Lessee as representing the full yearly open market value of the Premises (upon the assumption that the Premises have been duly repaired decorated maintained and cleaned in accordance with the Lessee's obligations pursuant to this Lease) including any car parking rights pursuant to Clause 2.1 at the date of commencement of such further term as between a willing lessor and a willing lessee with vacant possession for a term equal to the said further term and upon the terms and conditions of the further lease provided that if no agreement can be reached such amount shall be determined by a Valuer whose costs shall be borne equally by the Lessor and the Lessee unless Otherwise awarded; and
      - the amount of the Rent lastly payable during the Term increased by the percentage specified in Item 1.
  - 3.2.2 The rent commencement date to be specified in Item 6 of the Reference Schedule of such further lease shall be the commencement date of the Further Term.
  - 3.2.3 The dates for review of rent to be specified in Item 8 of the Reference Schedule of such further lease shall be the dates specified in Item 15.
  - 3.2.4 There shall be no option in such further lease for any further term and accordingly the further term to be specified in Item 14 of the Reference Schedule of such further lease shall be nil and the words "not applicable" shall be inserted in Item 15 of the Reference Schedule of such further lease.

PART 4 - RENT AND OUTGOINGS

3.2.1.2

4.1 The Lessee hereby covenants with the Lessor that it shall:

AJD June De PAG90004/BB/\\$-/13 Keni



/Sect9

4.1.1 without demand or deduction pay to the Lessor from and including the Rent Commencement Date and thereafter throughout the Term (subject only to Clause 10.1.3) without any deduction and without demand the Rent at the times and in the manner specified in Item 7;

/Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL

f:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

89 Req:C145589 /Doc:DL Z193326

- 4.1.2 if the Lessor shall require pay the Rent by bank authority; and
- 4.1.3 if any Rent or other moneys (whether or not in the nature of rent) payable by the Lessee to the Lessor remain unpaid for seven (7) days after the same shall have become due (whether demanded or not) pay to the Lessor interest calculated daily on amounts remaining unpaid at the Specified Rate from the earlier of the date those moneys first became due by the Lessee or are outlaid or incurred by the Lessor (as the case may be) to the date upon which such moneys are paid or reimbursed to or recovered by the Lessor and the Lessor shall be entitled to recover any such interest as rent in arrears and the Lessee acknowledges that the payment to or the demand receipt or recovery by the Lessor of any such interest shall not prejudice or otherwise affect the Lessor's other rights upon the default by the Lessee in duly paying the Rent and any such other moneys.
- 4.2 The Lessee hereby acknowledges and agrees that the Rent shall be reviewed on each of the Review Dates in the following manner:
  - 4.2.1 On each Review Date the Rent (being the Rent payable immediately prior to the relevant Review Date) shall automatically increase by the greater of the following:-
    - 4.2.1.1 the same proportion as the Index Number has increased since the last Review Date (or in the case of the first Review Date since the commencement of the Term). For the purposes of ascertaining the increase aforesaid the relevant publication of the Index Number shall be that published immediately prior to each Review Date or the date of commencement of the Term as the case may be.
    - 4.2.1.2 the amount equal to the percentage specified in Item 1 so that such increases are compounded on an annual basis.
  - 4.2.2 In any case the Rent following any review shall not be less than the amount which is payable immediately prior to such Review Date.
  - 4.2.3 If at any Review Date the Index Number has ceased to be published or in the reasonable opinion of the Lessor has substantially changed its character then upon any such Review Date the Rent shall (at the option of the Lessor) automatically increase to the amount agreed upon by the Lessor and the Lessee as representing the full yearly open market value of the Premises (upon the assumption that the Premises have been duly repaired, decorated, maintained and cleaned in accordance with the Lessee's obligations pursuant to this Lease) including any car parking rights pursuant to clause 2.1 upon the relevant Review Date as between a willing lesser and a willing lessee with vacant possession for a term equal to the then unexpired residue of the Term together with the term of any further Lease which the Lessee has the option hereunder

61.J N

PAG90004/88

89 Reg:C145589 /Doc:DL Z193326



to take and upon the terms and conditions of this Lease provided that if no agreement can be reached such amount shall be determined at the request of either party by a Valuer.

4.2.4 Where there has been any delay (for whatever reason) in agreeing or determining any increase in the Rent at a Review Date pursuant to Clause 4.2.1 or the new rent upon the grant of any further term pursuant to Clause 3.2 the Lessee shall :-

4.2.4.1 continue to pay Rent at the rate pavable immediately prior to such Review Date or grant of a further term (as the case may be) up to the time any increase in the Rent or the new rent (as the case may be) is agreed or determined;

within seven (7) days of any increase in the Rent 4.2.4.2 or the new rent (as the case may be) being agreed or determined pay an amount equal to such increase for the period from the relevant Review Date or the commancement of the further term (as the case may be) to the date of such agreement or determination;

#### 4.2.4.3 pay the Rent as increased or the new rent (as the case may be) from the date any such increase is agreed or determined.

- 4.3 The Lessee covenants with the Lessor to pay the Lessee's Percentage of Outgoings to the Lessor as follows:
  - for the purposes of this Clause 4.3 a reference to a year 4.3.1 shall mean a period of twelve (12) months commencing on the same day as the Term or such other date as the Lessor may at any time elect.
  - 4.3.2 At the beginning of each year which is wholly or partly within the Term (or as soon as is practicable thereafter) the Lessor or his agent may prepare a reasonable estimate of the Outgoings in respect of that year ("Estimate").
  - 4.3.3 The Lessor shall provide the Lessee with a copy of an Estimate prepared pursuant to Clause 4.3.2.
  - 4.3.4 In respect of each year (or part of a year) during the Term the Lessee shall (if the Lessor has submitted an Estimate for that year) pay to the Lessor on account of the Lessee's Percentage of Outgoings for that year the Lessee's Percentage of the Estimate (or in the event that the year to which the Estimate relates is partly outside the Term such amount of the Estimate as is attributable to that part of the year within the Term) such payment shall be made within fourteen (14) days of the Estimate being forwarded to the Lesses.
  - 4.3.5 Immediately after the end of each year during the Term and in the event that the Term expires or is determined during the course of a year immediately after the end of that year (or as soon as is practicable thereafter) the Lessor or his agent shall prepare and provide the Lessee with a statement showing the total of the actual Outgoings for that year (or in the event that the year to which such statement relates is partly outside the Term such amount of the actual Outgoings as is attributable to that part of the year within the Term) and also showing the amount due to the Lessor or the Lessee (as the case may require after taking account of any payments by

ብ እ PAG90004/BB/ /13.12.89

89 Req:C145589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:11 et:lpi:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

the Lessee pursuant to an Estimate) which amount shall be paid forthwith.

of

- 4.3.6 This Clause 4.3 shall continue in full force and effect notwithstanding the expiry or termination of this Lease for the purpose of making any balancing adjustment as aforesaid.
- 4.4 The Lessee shall pay for all services supplied or provided to the Premises provided that if the Premises are not separately metered or charged in respect of any such service (and the cost of such service is not included in the Outgoings) the Lessor shall pay for the provision of such service and the Lessee shall forthwith pay to the Lessor the Lessor's estimate (which shall be final and binding on the Lessee except in the case of manifest error) of the cost of the provision of any such service attributable to the Premises.
- 4.5 Notwithstanding anything contained in Part 4.3 at the option of the Lessor Outgoings are to be paid by the Lessee within fourteen (14) days of notification to the Lessee by the Lessor or the Lessor's Agent of the Lessee's Percentage of Outgoings.

#### PART 5 - DEFAULT AND DETERMINATION

- 5.1 Notwithstanding any provision to the contrary expressed in or implied by this Lease upon the happening of any of the following events the Lessor shall be entitled to exercise any of the rights specified in Clause 5:2:
  - 5.1.1 If the Rent or any part thereof or any other moneys (whether or not in the nature of Rent) shall be in arrears or unpaid for a period of fourteen (14) days (whether or not formally demanded);
  - 5.1.2 if the the Lessee shall default in the due observance and performance of any covenant condition restriction agreement or regulation (not relating to the payment of Rent or other moneys (whether or not in the nature of Rent)) expressed in or implied by this Lease on its part to be observed and performed provided that in so far as such default is reasonably capable of remedy the Lessor shall first give the Lessee not less than twenty eight (28) days notice requiring the default to be remedied:
  - 5.1.3 if in the event of the Lessee being a corporation:
    - 5.1.3.1 an order is made or a resolution is passed for its winding up or proceedings are initiated or a meeting called to obtain any such order or to pass such resolution;
    - 5.1.3.2 a receiver manager or receiver and manager of its undertaking or any part thereof is appointed or an official manager or provisional liquidator is appointed;
    - 5.1.3.3 if the Lessee is not a public listed company and without the prior written consent of the Lessor any sale transfer or other disposition whatsoever of the shares in its issued capital or any issue or allotment of any new shares in its capital or any other act matter or thing whatsoever occurs is done or performed the effect of which is to transfer whether directly or indirectly to any person persons company or companies the effective management and control of the Lessee.

AJD Y

#### 5.1.4 if the Lessee being an individual:

n	5.1.4.1	
Wan.	P3090004 (PP1-0-412 40 0	
	PAG90004/Bh/-9-/13.12.89	, ,
NUM		<b>}</b>

dies or becomes incapable of managing his own affairs,

Kon v findante



7WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

## 5.1.4.2 is declared bankrupt or makes any arrangement with his creditors;

/Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:12 of 25

- 5.1.5 if any execution or other process of any Court or other authority issues out against or is levied upon any of the property of the Lessee;
- 5.1.6 if any warranty or representation expressed in or implied by this Lease or otherwise made by or on behalf of the Lessee to the Lessor prior to entering into this Lease is found to be materially incorrect;
- 5.1.7 if a final judgment is entered in any Court against the Lessee and is not satisfied within twenty eight (28) days thereafter;
- 5.1.8 if the Lessee without the prior written consent of the Lessor creates or purports to create any charge or mortgage over its interest in this Lease;
- 5.2 Upon the happening of any of the events specified in Clause 5.1 the Lessor may at its absolute discretion:-
  - 5.2.1 immediately or at any time thereafter (unless prohibited by any statute) and without any notice. (unless required by virtue of any statute) or previous demand re-enter (forceably if necessary) into and upon the Premises or part or parts thereof in the name of the whole and repossess the same as of its former estate and expel and remove the Lessee and all other occupiers without liability for the tort of trespass or to the Lessee for any liability it may incur and without prejudice to any remedies which might otherwise be available to the Lessor to recover arrears of Rent or in respect to any antecedent breach of the covenants conditions restrictions agreements and regulations on the part of the Lessee expressed in or implied by this Lease and such re-entry shall (unless the Lessor otherwise expressly elects) automatically cause this Lease to determine as if it had thereupon expired by effluxion of time but the Lessee shall remain liable for all rents due to the date of such re-entry and for all other moneys (if any) due hereunder; or
  - 5.2.2 by giving written notice to the Lessee reduce the Term so that it expires on a date being not prior to the twenty eight days after the date of service of such notice.
- 5.3 In the event of any re-entry pursuant to Clause 5.2.1 the Lessor may retain any furniture fittings fixtures or other items belonging to the Lessee in the Premises and the Lessor shall have the right to sell such furniture fittings and fixtures or other items by public auction and apply the proceeds of such sale towards the payment of any moneys outstanding and payable to the Lessor pursuant to this Lease.
- 5.4

89 Req:C145589 /Doc:DL Z193326

7STC:W

In the event that the Lessee defaults in its obligations pursuant to this Lease in respect to the repair and decoration of the Premises then without prejudice to any other of the Lessor's rights and remedies hereunder it shall be lawful for the Lessor without further notice to enter upon the Premises and repair and remedy the same and all expenses incurred by the Lessor in so doing shall be re-imbursed forthwith without the need for demand by the Lessee and if not so reimbursed shall be recoverable as rent in arrears without notice.

No pour Dodal Kom fullactor G90004/88/-10-



#### PART 6 - REPAIR AND MAINTENANCE OF PREMISES

The Lessee hereby covenants with the Lessor that it shall:

89 Reg:C145589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:13 of 25 Ref: Private /Src:w /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 6.1 At all times during the Term and during any period of holding over repair and keep the Premises in good and substantial repair decoration and condition to the satisfaction of the Lessor (damage by any of the Insured Risks excepted provided that no policy of insurance effected by the Lessor has been rendered void or voidable by virtue of any act matter or thing done upon the Premises or the Building by the Lessee its servants agents invitees or any person claiming through or under the Lessee).
- 6.2 From time to time immediately replace broken glass in the Premises with glass of the same or similar quality and all broken or damaged plumbing, lighting, heating and electrical equipment appliances and other fixtures and fittings of the Lessor in the Premises (excluding any heating and air conditioning plant not exclusively serving the Premises which is maintained by the Lessor as part of the Lessor's Services).
- 6.3 At the expiration or sconer determination of the Term yield up to the Lessor the Premises (but excluding any fittings and fixtures which are tenants trade fixtures or fittings or which are otherwise required to be then removed pursuant to this Lease) duly repaired and maintained in accordance with the covenants herein contained.
- 6.4 Permit the Lessor upon giving not less than two days at any time during the Term notice to the Lessee except in the case of emergency to enter upon the Premises;
  - 5.4.1 to take a plan of or to examine the state of repair and conditions of the Premises and to take inventories; and
  - 6.4.2 to execute repairs decorations alterations or other work to the Building and any adjoining or neighbouring land and for the purpose of building upon any adjoining or neighbouring land provided that the person or persons exercising such right shall make good in a reasonable manner all damage thereby occasioned to the Premises and cause as little inconvenience to the Lessee as is practical.
- 6.5 At all times keep the Premises including external surfaces of windows and doors clean and tidy and not place leave or permit to be placed or left any debris or rubbish in any part of the Premises.
- 6.6 At all times keep waste, trash and garbage in proper receptacles.
- 6.7 At all times to keep the fire extinguishers hoses and other fire fighting or prevention equipment exclusively serving the Premises in good working order and condition and inspected at least once in every year of the Term by the relevant fire authority.

PART 7 - USE OF AND CONDUCT ON PREMISES

PAG90004/BB//

The Lessee hereby covenants with the Lessor that it shall:

Sal

- 7.1 Not use the Premises or any part thereof: .
  - 7.1.1 for any purpose other than as specified in Item 10 or such other purpose as the Lessor may consent to;
  - 7.1.2 for any illegal purpose or for any purpose in contravention of any approval consent or zoning by the relevant authority;

AND George Dadola Kom fallana



- 7.1.3 as a dwelling or sleeping place or to keep any animals birds or other livestock;
- 7.1.4 for the carrying on of any auction sale;

89 Req:C145589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:14

.: pl:s1xdre /Src:W

7.1.5 except during the times specified in Item 11.

/WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 7.2 Not without the prior written consent of the Lessor (who may require such plans and details as it considers necessary to be submitted to it by the Lessee) make or permit or suffer any alteration or addition whatsoever to the Premises (including the erection, demolition or alteration of any partitioning within the Premises) or any part thereof or mark drill cut main injure or deface or in any way damage any parts of the Premises, and at the end of the Term (or any subsequent term granted pursuant to any option herein contained) if the Lessor so requires remove any such alterations and additions.
- 7.3 At all times at the cost of the Lessee observe and comply with:
  - 7.3.1 the laws and regulations for the time being relating to fires (and the prevention thereof) and every regularement of the Fire and Accident Underwriters Association the Board of Fire Commissioners of the State in which the Premises are situated and any other competent authority; and
  - 7.3.2 the provisions of every Commonwealth and State Statute and every rule regulation ordinance and by-law made at any time under or in pursuance of any such Statute and the requirements at any time of every government or other competent authority.
- 7.4 Not cause any damage to or obstruction of any of the Common Parts or to any road or laneway serving the Building and not to place or store any goods outside the Premises.
- 7.5 Not erect or display any sign or advertisement on the exterior of the Premises (or within the Premises so that such sign or advertisement can be seen from the exterior of the Building) without the written consent of the Lessor (such consent not to be unreasonably withheld) and the consents of all relevant and competent authorities and upon the termination of this Lesse to remove any such sign.
- 7.6 Not use the lavatories conveniences and water apparatus in the Building for any purpose other than those for which they were constructed and on demand pay to the Lessor an amount sufficient to compensate the Lessor for any damage resulting from any such misuse by the Lessee.
- 7.7 Not without the written consent of the Lessor install use place or permit or suffer to be brought into the Premises any plant machinery or other articles which may cause undue noise or vibrations or which are of a weight or size which may cause damage directly or indirectly to the Building.
- 7.8 Not bring on to the Premises any dangerous inflammable explosive noxious or offensive substances except in the ordinary course of the Lessee's business (permitted by this Lease to be carried on at the Premises) and provided that the Lessee shall ensure that all proper and prudent measures are taken in the storage and use of any such substances and that the Lessor is previously notified of the nature and extent of any such substances brought on to the Premises.
- 7.9 Not do any act or thing which in the opinion of the Lessor (as to which it need not give reasons) may be offensive or cause a mulsance damage or annoyance or inconvenience to the Lessor or to the owners Lessees or occupiers of any adjoining or neighbouring premises. For

04/BB/-12-/13.1/2.89 Kor



the purposes of this Clause the use permitted by Item 10 will not constitute an offensive act.

7.10 Not overload the electric wires and cables serving the Pramises.

89 Req:C145589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:15 of 25 Ref: pr:sixare /Src:w /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 7.11 Give to the Lessor full particulars of any permission notice order requirement recommendation or proposal given or issued in respect of or in connection with the Premises by any competent authority within seven days of its receipt by the Lessee.
- 7.12 Comply with such reasonable regulations as the Lessor may now or in the future make for the more efficient management of the Building (including its security and that of its lessees and occupants).
- 7.13 At all times observe and perform the restrictions stipulations and covenants (if any) referred to in the certificate(s) of title in respect of the Building.
- 7.14 Not obstruct any of the windows or ventilators belonging to the Premises nor to permit any new window or ventilator or other encroachment or easement of which the Lessee is aware to be mada against or over the Premises.
- 7.15 In carrying out any construction fitting act reinstatement or removal works or alterations to the Premises:
  - 7.15.1 ensure that such works and alterations are carried out in a good and workmanlike manner by licensed and reputable tradespersons; and
  - 7.15.2 forthwith repair and reinstate any part of the Building damaged or defaced in carrying out such works or alterations to the satisfaction of the Lessor.

#### PART 8 - INSURANCES AND INDEMNITIES

The Lessee hereby covenants with the Lessor that it shall;

- 8.1 At its own cost effect and at all times keep in full force and effect a policy of public risk insurance with a reputable and solvent insurer with respect to the Premises and the business carried on in the Premises in which limits of public risk shall be not less than the amount specified in Item 12 (or such other amount as the Lessor may from time to time reasonably require) as the amount payable in respect of liability arising out of any one single accident or event and shall deliver to the Lessor on demand a copy of the policy and a current certificate of insurance.
- 8.2 At its own cost effect and at all times keep in full force and effect plate glass insurance (for the full replacement value thereof) in respect of all plate glass attached to or forming part of the Premises and shall deliver to the Lessor on demand a copy of the policy and a current certificate of insurance.
- 8.3 Include as the insured parties in respect of each of the policies effected pursuant to Clauses 8.1 and 8.2 the Leesor, any person or persons nominated by the Lessor as being mortgagees of the Building and the Lessee.
- 8.4 And hereby does indemnify and hold indemnified the Lessor from and against all actions claims demands losses damages costs and expenses which the Lessor may sustain or incur or for which the Lessor may become liable whether during or after the Term in respect of or arising from:

AID June Devlot Kom Kell -0



25

89 'Req:C145589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:16 of ef:101:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 8.4.1 The neglect or default of any Lessee Party to observe or perform any of the terms covenants and conditions expressed in or implied in this Lesse;
- 8.4.2 The negligent use or misuse waste or abuse by any Lessee Party of any water gas electricity or other services to the Building;
- 8.4.3 The overflow leakage or escape of water fire gas electricity or any other harmful agent whatsoever in or from the Frances,
- 9.4.4 The use of the Premises and the Common Parts by any Lessee Party;
- 8.4.5 The carrying out of any additions or alterations or other works to the Premises by any Lessee Party;
- 8.4.6 The use by any Lessee Party of any car parking facilities in the Building permitted by this Lease; and
- 8.4.7 The happening of any accident or event in or about the Pramises.

and it is hereby agreed that the Lessor shall not be liable or in any way responsible to the Lessee or any Lessee Party for any injury loss or damage which may be suffered or sustained to any property or by any person on the Premises unless caused by the wilful act or omission of the Lessor its servants or agents.

- 8.5 Comply with all requirements and recommendations of the insurers of the Building and not do or omit to do anything on the Premises which may increase the premium above the ordinary rate or render any additional premium payable for the insurance of the Building or of any neighbouring premises or which may make void or voidable any policy of such insurance and reimburse the Lessor forthwith on demand any additional premium which may have been paid or become payable by reason of any such act or comission together with all expenses incurred by the Lessor in relation to the renewal of any policy of insurance (which shall not form part of the Insurance Cost).
- 8.6 Inform the Lessor of any reason arising from the Lessee's use of the Premises why the Lessor's insurable interest in the Premises or in any adjoining premises may be affected.
- 8.7 In the event of the Premises or any part thereof being damaged or destroyed by any of the Insured Risks at any time during the Term and the insurance money under the insurance policy effected thereon being wholly or partially irrecoverable by reason (solely or in part) of any act or default of a Lessee Party forthwith pay to the Lessor the whole or as the case may require a fair proportion of the cost of rebuilding and reinstating the same.

#### PART 9 - ALIENATION

in Briddy

9.1 The Lessee hereby covenants with the Lessor not to assign, transfer sublet part with or share possession of the whole or any part of the Premises or its interest therein provided that the Lessee may assign or sub lease the benefit of this Lesse as a whole having first obtained the written consent of the Lessor whose consent shall not be unreasonably withheld in the case of a respectable and responsible assignee or sub lessee and provided that any such assignment or subletting shall comply with this Part. For the purposes of this Clause 9.1 a change in the controlling shareholding of the Lessee if a proprietary company shall constitute a transfer of this/Lesse.

ANDC Golol PAG90004/BB/-14-/13.12 Kom



25

89 Req:C145589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:17 of ef:101:sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 9.2 If the intended assignee or sub lesses shall be a company then the directors or principal shareholders of the same shall if the Lessor so requires act as guarantors and provide a guarantee in the form required by the Lessor.
- 9.3 Upon any assignment or transfer of the Lease the assignee shall if the Lessor so requires enter into a direct covenant with the Lessor to observe and perform the covenants and conditions herein contained and on the part of the Lessee to be observed and performed.
- 9.4 Any sublatting shall be upon the same terms of this Lease and at a rent which is no less than the Rent and shall be capable of being determined by the Lessor in the event of determination of this Leage.

### PART 10 - DESTRUCTION OF PREMISES

- 10.1 If during the term of this Lease the Building shall be destroyed or damaged by any of the Insured Risks so as to render the Premises or any substantial part thereof substantially unfit for the use and occupation by the Lessee or so as to deprive the Lessee of substantial use of or access to the Premises;
  - 10.1.1 This Lease may be terminated without compensation by either the Lessor or the Lessee by written notice to the other provided that the Lessee shall not be entitled to terminate the Lease unless the Premises shall not have been rendered fit for the use and occupation of the Lessee or the use of and access to the Premises shall not have been substantially restored within four (4) weeks of such destruction or damage;
  - 10.1.2 Any such termination shall be without prejudice to the rights of either party in respect of any antecedent breach matter or thing;
  - 10.1.3 On the happening of any such damage or destruction as aforesaid (provided that any insurance moneys that would have been payable to the Lessor are not wholly or partially irrecoverable by reason of any act or default of a Lessee Party) the Rent or a proportionate part thereof (according to the nature and extent of the damage sustained) shall abate and all or any remedies for the recovery of such Rent or such proportionate part thereof shall be suspended until the Premises shall have been rebuilt or reinstated or made fit for the occupation and use of the Lessee or until access thereto shall have been provided or until the Lesse shall be terminated pursuant to the provisions hereof as the case may be. Any dispute arising out of this clause shall be determined by a Valuer whose fees shall be borne by the Lessor and Lessee in equal shares unless otherwise awarded.
- 10.2 Nothing expressed in or implied by this Lease shall be deemed to impose any obligation upon the Lessor to rebuild or reinstate or make fit for occupation the Premises or the Building in the event of damage thereto or destruction thereof.

PART 11 - LESSOR'S COVENANTS

The Lessor hereby covenants with the Lessee as follows;

11.1 That the Lessee paying the Rent hereby reserved and performing and observing the covenants and stipulations herein contained shall peaceably hold and enjoy the Premises during the Term without any interference from the Lessor or any person rightfully claiming under or in trust for the Lessor.

PAG90004//#B/-15-/13. Ala ture

Ass Jame Doold. Kom Kulandas


11.2 That the Lessor will at all times during the Term (unless such insurance shall be avoided by any act or omission of the Lessee as aforesaid) insure and keep insured the Building against the Insured Risks and will whenever required (but not more than twice in every calendar year) produce a certificate to the Lessee in respect of such insurance.

89 Req:C145589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:18 of 25 Rel: pr:stxdre /Srd:w /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

#### PART 12 - GENERAL

- 12.1 The Lessee hereby covenants with the Lessor to pay to the Lessor on demand:
  - 12.1.1 all legal and other costs and disbursements (including stamp duty) incurred by the Lessor in relation to any application by the Lessee for consent to an assignment of this Lease or any other dealing with the Lessee's interest herein or any other matter requiring the consent of the Lessor pursuant to provisions hereof or in connection with any breach or threatened breach of any of the terms hereof by the Lessee and in connection with any proceedings for enforcement of payment of rent or any other terms of this Lease.
  - 12.1.2 Upon the signing of this Lease the Lessor's legal costs and disbursements in connection with this Lease and the stamping and registration thereof.
  - 12.1.3 Upon demand by the Lessor all Financial Institutions or similar duty incurred by the Lessor in relation to the payment to it of the Rent and any other moneys payable hereunder (excluding moneys payable pursuant to this Clause).
- 12.2 Service of any notice required or authorised by this Lease may be effected in the manner permitted by Section 170 of the Conveyancing Act, 1919.
- 12.3 The Lessee shall permit the Lessor at any time during the last six months of the Term to enter the Premises and affix upon any suitable parts of the exterior thereof notice boards or other signs advertising the releting or selling the same and the Lessee shall not remove or obscure such notices.
- 12.4 Nothing in this Lease shall imply or warrant that the Premises are fit or entitled to be used for any particular purpose and the Lessee hareby acknowledges that in entering into this Lease he has not relied upon any representation made by the Lessor, its solicitors or agents as to the suitability of the Premises for any particular purpose.
- 12.5 The Lessee shall be responsible for protecting and keeping safe the Premises from theft and robbery and shall keep all windows and doors properly locked at all times that the Premises are unoccupied.
- 12.6 It is hereby acknowledged and agreed that the Lessee shall not be obliged by anything expressed in or implied by this Lease to carry out any structural repairs or works to the Building unless the same arise (whether directly or indirectly) as a result of any of the following:
  - 12.6.1 the neglect or default by any Lessee Party to observe or perform any of the terms covenants and conditions expressed in or implied by this Lesse;
  - 12.6.2 the use or occupation of the Premises by the Lesses aub-tenant;

PAG90004//BB/-16-/13.12.89

AJ& June Dodol Korn fallaties



25

12.6.3 the employment of any person in the Premises by the Lessee or any sub-tenant;

89'Req:C:45589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Fgs:ALL /Seq:19 of

I: 1p1: SIXdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- 12.6.4 the use of any fixtures fittings plant machinery or goods in the Premises by the Lessee or any sub-tenant;
- 12.6.5 the carrying out of any alterations or additions to the Premises by the Lessee or any sub-tenant or the reinstatement of the Premises following any alterations or additions thereto; and
- 12.6.6 the bringing on to the Premises by the Lessee or any subtenant of any plant machinery or other items (whether consented to by the Lessor or not)

in which event the Lessee shall at the option of the Lessor either at its own cost forthwith repair and reinstate the structure of the Building so requiring repair or carry out such structural works as may be necessary to comply with the requirements of any competent authority to the reasonable satisfaction of the Lessor and all relevant authorities or if the Lessor has itself elected to carry out such repairs and reinstatement or works the Lessee shall forthwith upon demand by the Lessor pay to the Lessor all its costs and expenses (including all professional fees) incurred in connection therewith.

- 12.7 Notwithstanding anything to the contrary expressed in or implied by this Lease:
  - 12.7.1 The Lessee hereby assures the Lessor that it will at all times strictly and substantially comply with each and every covenant condition restriction rule and agreement expressed in or implied by this Lease on the Lessee's part to be performed or observed and the Lessee acknowledges that the Lessor has entered into this Lease on the basis of such assurance.
  - 12.7.2 The Lessor and the Lessee hereby expressly agree and acknowledge that the covenants conditions restrictions rules and agreements expressed in or implied by Parts 3, 4, 6, 7, 8 and 9 of this Lesse on the Lessee's part to be performed and observed are fundamental and essential terms of this Lesse.
  - 12.7.3 The Lessor and the Lessee hereby expressly agree and acknowledge that if:
    - 12-7-3-1 any Rent is in arrears or unpaid for fourteen (14) days after it has become due and payable (whether or not any formal demand has been made for payment);
    - 12.7.3.2 the Lessee defaults in the due performance and observance of any other (than those relating to payment of Rent) covenant, condition, restriction, rule or agreement herein contained or implied on the Lessee's part to be performed or observed and in the case of a default which is capable of remady has continued the default for more than twenty eight (28) days after notice from the Lessor requiring the default to be remedied;

12.7.3.3 the Lessee (being a natural person) commits any act of bankruptcy or has his estate sequestrated in bankruptcy or assigns his estage for the

the



Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:20 of 25 /WARNING: A4 Copy Supplied by LPT NSW for Conveyancing Purposes Only,

> benefit of creditors or enters into a deed of arrangement with his creditors or enters into an arrangement for the liquidation of his debts by composition or otherwise; or

12.7.3.4

the Lesses (being a corporation) becomes subject to an order for liquidation whether compulsory or voluntary or commits any act of bankruptcy or has a receiver appointed over the whole or part of its assets or undertaking;

any such act matter or thing shall if the Lessor so elects (and the Lessor shall deemed to have so elected if it exercises any right of re-entry pursuant to Clause 5.2) constitute and be deemed to be a repudiation by the Lessee of this Lesse so that the Lessor shall without prejudice to any other rights it may have be entitled to recover damages from the Lessee for the full extent of its loss arising out of such repudiation.

#### FART 13 - GUARANTEE

89 Req: C145589 /Doc: DL Z193326

- 13.1 In consideration of the Lessor at the request of the person or persons named in Item 16 of the Reference Schedule (which person or persons are hereinafter called "the Guarantor") entering into the Lease with the Lessee the Guarantor covenants and agrees with the Lessor that;-
  - 13.1.1 The Guarantor guarantees to the Lessor that it will be with the Lessee jointly and severally liable to the Lessor for the due payment of all Rents and other moneys to be paid by the Lessee under the Lesse and the due performance and observance by the Lessee of all the covenants, terms and conditions of the Lesse on the part of the Lessee to be performed and observed.
  - The Guarantor will indemnify the Lessor and agrees at all 13.1.2 times hereafter to keep the Lessor indemnified from and against all losses and expenses which the lessor may suffer or incur in consequence of any breach or non-observance of any of the covenants, terms and condition of the Lease on the part of the Lessee to be performed or observed and the Guarantor agrees that it shall remain liable to the Lessor under this indemnity notwithstanding as a consequence of such breach or non-observance the Lessor has exercised any of its rights under the Lease including its rights of reentry and notwithstanding that the Lessee (being a Company) may be wound up or dissolved or being a natural person may be declared bankrupt and notwithstanding that the guarantee given by any person or corporation named in Item 16 of the Reference Schedule may for any reason whatsoever unenforceable either in whole or in part.
  - 13.1.3 On any default or failure by the Lessee to observe and perform any of the covenants, terms and conditions of the Lease the Guarantor will forthwith on demand by the Lessor pay the Rent and make good to the Lessor all losses and expenses sustained or incurred by the Lessor by reason or in consequence of any such default or failure by the Lessee in the payment of rent or in performing or observing any of the covenants, terms and conditions of the Lease without the necessity of any prior demand having been made on the Lessee.

Ass June Doold 0004/BB/-/18-/13.12



89 Req:C145589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:21 of 25 Ref: Provide /Src:w /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

- The liability of the Guarantor under this guarantee and 13.1.4 indemnity shall not be affected by the granting of time or any other indulgence to or by the compounding, compromise, release, abandonment, waiver, variation or renewal of any of the rights of the Lessor against the Lessee or any person or corporation named in Item 16 of the Reference Schedule or by any neglect or omission to enforce such rights or by any other thing which under the law relating to sureties would or might but for this provision release the Guarantor in whole or in part from its obligations under this guarantee.
- 13.1.5 Notwithstanding that as between the Guarantor and the Lessee the Guarantor may be a surety only nevertheless as hetween the Guarantor and the Lessor the Guarantor shall be deemed to be a primary debtor and contractor jointly and severally with the Lessee.
- 13.1.6 To the fullest extent permitted by law the Guarantor hereby waives such of its rights as surety or indemnifier (legal, equitable, statutory or otherwise) which may at any time be inconsistent with any of the provisions of the guarantee and indemnity contained in this Part.
- 13.1.7 The covenants and agreements made or given by the Guarantor shall not be conditional or contingent in any way or dependent upon the validity or enforceability of the covenants and agreements of any other person and shall be and remain binding notwithstanding that any other person shall not have executed or duly executed the Lease or this guarantee and indemnity.
- If the Lease shall contain an option for a further lease 13.1.8 and the Lessee shall exercise such option the grant of such a further lease shall be subject to the Guarantor guaranteeing the obligations of the Lessee under such further Lease and indemnifying the Lessor in respect thereof in the terms of the guarantee and indemnities contained in this Part.
- 13.1.9 The obligations of the Guarantor under the guarantee and indemnity contained in this Part shall continue to remain in force until all rent or other moneys payable pursuant to the Lease shall have been paid and until all other obligations and indemnities shall have been performed, observed and satisfied and such obligations shall not be reduced or affected by any notice to guit given by either party to the Lease or the death, imprisonment, mental illness, insolvency, liquidation or dissolution of the Lessee or any person or corporation named in Item 16 of the Reference Schedule.

NJD frome Doold Kong Kelland



#### THE REFERENCE SCHEDULE

#### ITEM 1: PERCENTAGE RATE FOR RENT INCREASES:

89 Reg:C:,45589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seg:22 of 25

/WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

10%.

ITEM 2: BUILDING:

SIXURE /SPC:W

Shop No.4, Ralian Centre, 108-120 Station Street, Wentworthville as shown on the Plan annexed to the Lease between the parties and marked with the letter "A".

ITEM 3: ANCILLARY RIGHTS:

Subject as contained in the Lease, full right and liberty to use in common with the Lessor and other Lessees and tenants of the building and all other persons entitled thereto during usual business hours the toilets at the rear of Shop 5 shown on the Plan, the covered walkway fronting Station Street, passages and areas giving access to the Premises and the said toilets (but for the purpose of access and egress to and from the Premises and the said toilets only. And whilst on foot without animals or vehicles except for truck bay in which case vehicles shall only stand a reasonable time for the purpose of loading and unloading.

### ITEM 4: LESSOR'S SERVICES:

- The lighting, cleaning, maintenance, repairing (but not any repairing of a structural nature) and decorating of the Common Parts and the exterior and other parts of the Building not comprised or intended to be comprised in any lease of premises within the Building.
- The provision and maintenance of all necessary fire fighting and fire prevention equipment to the Common Parts.
- 3. The maintenance of any landscaped areas of the Building including the cost of mowing irrigation and the replacement of plants and trees.

4. All other services from time to time and at any time provided by the Lessor for the more efficient running of the Building and the benefit of lessess of premises within the Building.

> AND where the Lessor's Services include the maintenance of any item the Lessor shall be entitled to enter into maintenance contracts for the maintenance of such items and the cost to the Lessor of entering into such contracts shall be part of the cost of providing the Lessors Services for the purposes of calculating the Outgoings.

ITEM 5:

#### INITIAL YEARLY AMOUNT:

-20 - /13

Forty six thousand four hundred and nineteen dollars and seventy two cents (\$46,419.72).

Aso June Dodd



89 Reg:C.45589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:23 of 25 Reg:Tp::sixdre /Src:W /WARNING: A4 Copy Supplied by LPI NSW for Conveyancing Purposes Only.

ITEM 6:	RENT COMMENCEMENT DATE:
	26 November 1989.
ITEM 7:	MANNER OF PAYMENT OF RENT:
	By equal monthly instalments in advance on the first day of each month. The Lessee shall pay a proportionate part of such instalment on the signing of this Lease for the period from the Rent Commencement Date to the first day of the following month. In the event the Term expires on a day other than the last day of a month the last instalment shall be reduced proportionately.
ITEM 8:	REVIEW DATES:
	Rach anniversary of the date of commencement of the term of the Leage.
ITEM 9:	LESSEE'S PERCENTAGE
	30% provided that in the event that the net lettable area of the Building is increased or decreased the Lessee's percentage shall be reduced or increased accordingly.
ITEM 10:	PERMITTED USE:
	Pharmacy and Toy Shop.
ITEM 11:	ACCESS TIMES:
	At all times.
ITEM 12:	PUBLIC LIABILITY COVER:
	\$5,000,000.
ITEN 13;	HEAD LEASE:
	Not applicable.
ITEM 14:	FURTHER TERM:
	Five (5) years.
ITEM 15:	REVIEW DATES IN FURTHER LEASE:
	Each anniversary of the date of commencement of the term of the Further Lease.
ITEN 16:	guarantee
	(i) Kenneth Christopher Macrae, 39 Greenhaven Drive, Pennant Hills.
	(11) AUGREY JUNE (11) June Audrey Dodd, 23 Ernest Street, Hunters Hill.

Non Konfarla PAG90004/45/421-/13.12 89



•...

#### INDEX

89 Req:C145589 /Doc:DL Z193326 /Rev:20-Jul-2010 /Sts:OK.OK /Prt:27-Mar-2015 10:43 /Pgs:ALL /Seq:24 of 25 Her:Ipt:state /Sfc:W /WARMING: A4 Copy Supplied by LPE NSW for Conveyancing Purposes Only.

•

ç,

~ .

.

(PAGE 1)	PART 1:	INTERPRETATION AND DEPINITIONS
(PAGE 5)	PART 2;	ANCILLARY RIGHTS, ACCEPTIONS AND RESERVATIONS
(PAGE 6)	PART 3:	HOLDING OVER AND OPTION FOR FURTHER TERM .
(PAGE 7)	PART 4:	RENT AND OUTGOINGS
(PAGE 9)	PART 5:	DEFAULT AND DETERMINATION
(PAGE 11)	PART 6:	REPAIR AND MAINTENANCE OF PREMISES
(PAGE 11)	PART 7:	USE OF AND CONDUCT ON PREMISES
(PAGE 13)	PART 8:	INSURANCES AND INDEMNITIES
(PAGE 14)	PART 9:	ALIENATION
(PAGE 15)	PART 10:	DESTRUCTION OF PREMISES
(PAGE 15)	PART 11:	LESSOR'S COVENANTS
(PAGE 16)	PART 12:	GENERAL
(PAGE 18)	PART 13:	guarantee

PAG90004 48/13.12.89 ay N in the second



# **APPENDIX E**

# CURRENT AND HISTORICAL AERIAL PHOTOGRAPHS



## **HISTORICAL AERIAL PHOTOGRAPHS - 1928**



	ABN 75 050 212 710	Aargus Pty Limited	Environment – Reme	diation – Geotechnic	cal Engineerin
PROJECT DETAILS				DRAWING DETAIL	.S
Project Title	Preliminary Site Investigation			Figure No.	А
Project No.	ES6206			Scale	As above
Client	Beaini Projects Pty Ltd			Drawn by	СК
Site Address	108 Station Street, Wentworthville NSV	N	Aargus	Approved by	МК

## HISTORICAL AERIAL PHOTOGRAPHS – 1951



PROJECT DETAILS			DRAWING DETAI	LS		
Project Title	Preliminary Site Investigation		Figure No.	В	Rev No.	0
Project No.	ES6206		Scale	As above	Size	A3
Client	Beaini Projects Pty Ltd		Drawn by	СК	Date	27.03.2015
Site Address	108 Station Street, Wentworthville NSW	Aargus	Approved by	МК	Date	27.03.2015

Site Location

N

## **HISTORICAL AERIAL PHOTOGRAPHS - 1970**



	ABN 75 050 212 710	Aargus Pty Limited	Environment – Reme	ediation – Geotechni	cal Engineeri
PROJECT DETAILS				DRAWING DETAI	LS
Project Title	Preliminary Site Investigation			Figure No.	С
Project No.	ES6206			Scale	As above
Client	Beaini Projects Pty Ltd			Drawn by	СК
Site Address	108 Station Street, Wentworthville NSV	V	Aargus	Approved by	МК

# **HISTORICAL AERIAL PHOTOGRAPHS - 1991**



	ABN 75 050 212 710	Aargus Pty Limited	Environment – Remed	liation – Geotechni	cal Engineering		
PROJECT DETAILS				DRAWING DETAI	LS		
Project Title	Preliminary Site Investigation			Figure No.	D	Rev No.	0
Project No.	ES6206			Scale	As above	Size	A3
Client	Beaini Projects Pty Ltd			Drawn by	СК	Date	27.03.2015
Site Address	108 Station Street, Wentworthville NSV	N	Aargus	Approved by	МК	Date	27.03.2015

## **CURRENT AERIAL PHOTOGRAPHS - 2015**



**PROJECT DETAILS** Project Title Project No. Client Site Address

# **APPENDIX F**

# **NSW EPA RECORDS**





Home > Contaminated land > Record of notices

#### Search results

Your search for:LGA: Holroyd City Council

		to 2 sites	5
		S	earch Again
		F	Refine Search
Suburb	Address	Site Name	Notices related to this site
Merrylands	Corner Walpole and Peel Street	Merrylands Substation PCB Storage	ge 3 former
Yennora	6 Foray Street	Former Tetra Pak Site, Yennora	2 current and 4 former

Page 1 of 1

27 March 2015

Matched 9 notices relating

Environment, Healthy Community, Health

Connect	Feedback	Contact	Government	About	
Y	Web support Public consultation	Contact us Offices Report pollution	NSW Government jobs.nsw	Accessibility Disclaimer Privacy Copyright	

Healthy

Environment, Healthy Community, Health



<u>Home</u> > <u>Environment protection licences</u> > <u>POEO Public Register</u> > <u>Search for licences, applications and notices</u>

#### Search results

Your search for: Notices with the following criteria

Notice type - Penalty Notice LGA - HOLROYD

returned 10 results

Export to excel	1 of 1 Pages			Search Again
<u>Number</u> <u>Name</u>	Location	<u>Type</u>	<u>Status</u>	Issued date
<u>3085764835</u> ALCOA AUSTRALIA ROLLED PRODUCTS PTY LIMITED	KIORA CRESCENT, YENNORA, NSW 2161	Penalty Notice	Issued	25 Jul 2012
3085765210ALCOA AUSTRALIA ROLLED PRODUCTS PTY LIMITED	KIORA CRESCENT, YENNORA, NSW 2161	Penalty Notice	Issued	13 Aug 2012
3085769152BORAL RESOURCES (NSW) PTY LTD	RECONCILIATION ROAD, PROSPECT, NSW 2148	Penalty Notice	Issued	18 Mar 2013
3085773424BORAL RESOURCES (NSW) PTY LTD	MORT STREET, GRANVILLE, NSW 2142	Penalty Notice	Issued	05 Mar 2014
3085763881 PTY LTD	230 Toonganbbie Rd, GIRRAWEEN, NSW 2145	Penalty Notice	Issued	25 May 2012
3085766265 PTY LTD	230 Toonganbbie Rd, GIRRAWEEN, NSW 2145	Penalty Notice	Issued	22 Nov 2012
<u>3085773626</u> REDOX PTY LTD	144 GILBA ROAD, GIRRAWEEN, NSW 2145	Penalty Notice	Issued	31 Mar 2014
3085773195VISY PAPER PTY LTD	6 HERBERT PLACE, SMITHFIELD, NSW 2164	Penalty Notice	Issued	06 Feb 2014
<u>3085774570</u> VISY PAPER PTY LTD	6 HERBERT PLACE, SMITHFIELD, NSW 2164	Penalty Notice	Issued	29 Jul 2014
3085774560VISY PAPER PTY LTD	6 HERBERT PLACE, SMITHFIELD, NSW 2164	Penalty Notice	Issued	29 Jul 2014
				27 March 2015



27 March 2015



**GROUNDWATER BORE SEARCH** 



allwaterdata.water.nsw.gov.au/wgen/users/060333251//gw113429.wsr.htm

### NSW Office of Water Work Summary

#### GW113429

Licence:	10BL600157	Licence Status:	ACTIVE
		Authorised Purpose(s): Intended Purpose(s):	
Work Type:	Bore		
Work Status:	Equipped		
Construct.Method:			
Owner Type:	Private		
Commenced Date: Completion Date:	27/03/2006	Final Depth: Drilled Depth:	
Contractor Name:	Terratest		
Driller:	Craig Williams		
Assistant Driller:			
Property:	MOBIL OIL AUST LTD 344 - 348 GREAT WESTERN HWY WENTWORTHVILLE 2145 NSW	Standing Water Level:	
GWMA: GW Zone:		Salinity: Yield:	
Gw Zone.		neiu.	
Site Details			

#### Site Chosen By:

	County Form A: CUMBE Licensed:	<b>Parish</b> CUMBE.41	<b>Cadastre</b> 38/3/963
Region: 10 - Sydney South Coast	CMA Map:		
River Basin: - Unknown Area/District:	Grid Zone:	So	cale:
Elevation: 0.00 m (A.H.D.) Elevation Unknown Source:	Northing: 6256490.0 Easting: 312098.0		ude: 33°48'54.4"S ude: 150°58'11.4"E
GS Map: -	MGA Zone: 0		nate Unknown Irce:

#### Construction

Negative depths indicate Above Ground Level; C-Cemented; SL-Slot Length; A-Aperture; GS-Grain Size; Q-Quantity; PL-Placement of Gravel Pack; PC-Pressure Cemented; S-Sump; CE-Centralisers

Π	Hole	Pipe	Component	Туре	From	То	Outside	Inside	Interval	Details
					(m)	(m)	Diameter	Diameter		
							(mm)	(mm)		

#### Water Bearing Zones

From (m)	To (m)	Thickness (m)	WBZ Туре	S.W.L. (m)	D.D.L. (m)	(L/s)	Hole Depth	Duration (hr)	Salinity (mg/L)
							(m)		

#### Geologists Log Drillers Log

From 1	То	Thickness	Drillers Description	Geological Material	Comments
1 1			I		

(m) (m) (m)		
1		

#### Remarks

30/07/2014: Nat Carling, 30-July-2014; Added status, drill method & depth, updated work type.

\*\*\* End of GW113429 \*\*\*

Warning To Clients: This raw data has been supplied to the NSW Office of Water by drillers, licensees and other sources. The NOW does not verify the accuracy of this data. The data is presented for use by you at your own risk. You should consider verifying this data before relying on it. Professional hydrogeological advice should be sought in interpreting and using this data.

allwaterdata.water.nsw.gov.au/wgen/users/060333251//gw113430.wsr.htm

### NSW Office of Water Work Summary

#### GW113430

Licence:	10BL600157	Licence Status:	ACTIVE
		Authorised Purpose(s): Intended Purpose(s):	MONITORING BORE
Work Type:	Bore		
Work Status:	Equipped		
Construct.Method:			
Owner Type:	Private		
Commenced Date: Completion Date:		Final Depth: Drilled Depth:	
Contractor Name:	Terratest		
Driller:	Craig Williams		
Assistant Driller:			
Property:	MOBIL OIL AUST LTD 344 - 348 GREAT WESTERN HWY WENTWORTHVILLE 2145 NSW	Standing Water Level:	
GWMA: GW Zone:		Salinity: Yield:	
GW Zone.		neiu.	
Site Details			

#### Site Chosen By:

	County Form A: CUMBE Licensed:	<b>Parish</b> CUMBE.41	<b>Cadastre</b> 10/3/963
Region: 10 - Sydney South Coast	СМА Мар:		
River Basin: - Unknown Area/District:	Grid Zone:	So	cale:
Elevation: 0.00 m (A.H.D.) Elevation Unknown Source:	Northing: 6256479.0 Easting: 312097.0		ude: 33°48'54.8"S ude: 150°58'11.3"E
GS Map: -	MGA Zone: 0		nate Unknown Irce:

#### Construction

Negative depths indicate Above Ground Level; C-Cemented; SL-Slot Length; A-Aperture; GS-Grain Size; Q-Quantity; PL-Placement of Gravel Pack; PC-Pressure Cemented; S-Sump; CE-Centralisers

H	ole	Pipe	Component	Туре	From	То	Outside	Inside	Interval	Details
					(m)	(m)	Diameter	Diameter		
							(mm)	(mm)		

#### Water Bearing Zones

From (m)	To (m)	Thickness (m)	WBZ Туре	S.W.L. (m)	D.D.L. (m)	(L/s)	Hole Depth	Duration (hr)	Salinity (mg/L)
							(m)		

#### Geologists Log Drillers Log

From To	Thickness	Drillers Description	Geological Material	Comments
1 1		1	I I	1

(m) (m) (m)		
1		

#### Remarks

30/07/2014: Nat Carling, 30-July-2014; Added status, drill method & depth, updated work type.

\*\*\* End of GW113430 \*\*\*

Warning To Clients: This raw data has been supplied to the NSW Office of Water by drillers, licensees and other sources. The NOW does not verify the accuracy of this data. The data is presented for use by you at your own risk. You should consider verifying this data before relying on it. Professional hydrogeological advice should be sought in interpreting and using this data.

allwaterdata.water.nsw.gov.au/wgen/users/060333251//gw113431.wsr.htm

### NSW Office of Water Work Summary

#### GW113431

Licence:	10BL600157	Licence Status:	ACTIVE
		Authorised Purpose(s): Intended Purpose(s):	MONITORING BORE
Work Type:	Bore		
Work Status:	Equipped		
Construct.Method:			
Owner Type:	Private		
Commenced Date: Completion Date:	20/03/2006	Final Depth: Drilled Depth:	
Contractor Name:	Terratest		
Driller:	Craig Williams		
Assistant Driller:			
Property:	MOBIL OIL AUST LTD 344 - 348 GREAT WESTERN HWY WENTWORTHVILLE 2145 NSW	Standing Water Level:	
GWMA:		Salinity:	
GW Zone:		Yield:	
Site Details			

#### Site Chosen By:

	County Form A: CUMBE Licensed:	<b>Parish</b> CUMBE.41	<b>Cadastre</b> 8/3/963
Region: 10 - Sydney South Coast	CMA Map:		
River Basin: - Unknown Area/District:	Grid Zone:	S	cale:
Elevation: 0.00 m (A.H.D.) Elevation Unknown Source:	Northing: 6256454.0 Easting: 312114.0		ude: 33°48'55.6"S ude: 150°58'12.0"E
GS Map: -	MGA Zone: 0		nate Unknown Irce:

#### Construction

Negative depths indicate Above Ground Level; C-Cemented; SL-Slot Length; A-Aperture; GS-Grain Size; Q-Quantity; PL-Placement of Gravel Pack; PC-Pressure Cemented; S-Sump; CE-Centralisers

Π	Hole	Pipe	Component	Туре	From	То	Outside	Inside	Interval	Details
					(m)	(m)	Diameter	Diameter		
							(mm)	(mm)		

#### Water Bearing Zones

From (m)	To (m)	Thickness (m)	WBZ Туре	S.W.L. (m)	D.D.L. (m)	(L/s)	Hole Depth	Duration (hr)	Salinity (mg/L)
							(m)		

#### Geologists Log Drillers Log

F	rom	То	Thickness	Drillers Description	Geological Material	Comments

(m) (m) (m)		
1		

#### Remarks

30/07/2014: Nat Carling, 30-July-2014; Added status, drill method & depth, updated work type.

\*\*\* End of GW113431 \*\*\*

Warning To Clients: This raw data has been supplied to the NSW Office of Water by drillers, licensees and other sources. The NOW does not verify the accuracy of this data. The data is presented for use by you at your own risk. You should consider verifying this data before relying on it. Professional hydrogeological advice should be sought in interpreting and using this data.

allwaterdata.water.nsw.gov.au/wgen/users/060333251//gw113432.wsr.htm

### NSW Office of Water Work Summary

#### GW113432

Licence:	10BL600157	Licence Status:	ACTIVE
		Authorised Purpose(s): Intended Purpose(s):	MONITORING BORE
Work Type:	Bore		
Work Status:	Equipped		
Construct.Method:			
Owner Type:	Private		
Commenced Date: Completion Date:	27/03/2006	Final Depth: Drilled Depth:	
Contractor Name:	Terratest		
Driller:	Craig Williams		
Assistant Driller:			
Property:	MOBIL OIL AUST LTD 344 - 348 GREAT WESTERN HWY WENTWORTHVILLE 2145 NSW	Standing Water Level:	
GWMA:		Salinity:	
GW Zone:		Yield:	
Site Details			

Site Chosen By:

	County Form A: CUMBE Licensed:	<b>Parish</b> CUMBE.41	<b>Cadastre</b> 6/3/963
Region: 10 - Sydney South Coast	CMA Map:		
River Basin: - Unknown Area/District:	Grid Zone:	S	cale:
Elevation: 0.00 m (A.H.D.) Elevation Unknown Source:	Northing: 6256462.0 Easting: 312131.0		ude: 33°48'55.4"S ude: 150°58'12.7"E
GS Map: -	MGA Zone: 0		nate Unknown Irce:

#### Construction

Negative depths indicate Above Ground Level; C-Cemented; SL-Slot Length; A-Aperture; GS-Grain Size; Q-Quantity; PL-Placement of Gravel Pack; PC-Pressure Cemented; S-Sump; CE-Centralisers

Π	Hole	Pipe	Component	Туре	From	То	Outside	Inside	Interval	Details
					(m)	(m)	Diameter	Diameter		
							(mm)	(mm)		

#### Water Bearing Zones

From (m)	To (m)	Thickness (m)	WBZ Туре	S.W.L. (m)	D.D.L. (m)	(L/s)	Hole Depth	Duration (hr)	Salinity (mg/L)
							(m)		

#### Geologists Log Drillers Log

From To	Thickness	Drillers Description	Geological Material	Comments
1 1	1		1	1

(m) (m) (m)		
1		

#### Remarks

30/07/2014: Nat Carling, 30-July-2014; Added status, drill method & depth, updated work type.

\*\*\* End of GW113432 \*\*\*

Warning To Clients: This raw data has been supplied to the NSW Office of Water by drillers, licensees and other sources. The NOW does not verify the accuracy of this data. The data is presented for use by you at your own risk. You should consider verifying this data before relying on it. Professional hydrogeological advice should be sought in interpreting and using this data.

allwaterdata.water.nsw.gov.au/wgen/users/060333251//gw113433.wsr.htm

### NSW Office of Water Work Summary

#### GW113433

Licence:	10BL600157	Licence Status:	ACTIVE
		Authorised Purpose(s): Intended Purpose(s):	MONITORING BORE
Work Type:	Bore		
Work Status:	Equipped		
Construct.Method:			
Owner Type:	Private		
Commenced Date: Completion Date:		Final Depth: Drilled Depth:	
Contractor Name:	Terratest		
Driller:	Craig Williams		
Assistant Driller:			
Property:	MOBIL OIL AUST LTD 344 - 348 GREAT WESTERN HWY WENTWORTHVILLE 2145 NSW	Standing Water Level:	
GWMA: GW Zone:		Salinity: Yield:	
GW Zone.		neiu.	
Site Details			

#### Site Chosen By:

	County Form A: CUMBE Licensed:	ParishCadastreCUMBE.4110/3/963		
Region: 10 - Sydney South Coast	CMA Map:			
River Basin: - Unknown Area/District:	Grid Zone:	So	cale:	
Elevation: 0.00 m (A.H.D.) Elevation Unknown Source:	Northing: 6256461.0 Easting: 312096.0		ude: 33°48'55.4"S ude: 150°58'11.3"E	
GS Map: -	MGA Zone: 0		nate Unknown Irce:	

#### Construction

Negative depths indicate Above Ground Level; C-Cemented; SL-Slot Length; A-Aperture; GS-Grain Size; Q-Quantity; PL-Placement of Gravel Pack; PC-Pressure Cemented; S-Sump; CE-Centralisers

Π	Hole	Pipe	Component	Туре	From	То	Outside	Inside	Interval	Details
					(m)	(m)	Diameter	Diameter		
							(mm)	(mm)		

#### Water Bearing Zones

From (m)	To (m)	Thickness (m)	WBZ Туре	S.W.L. (m)	D.D.L. (m)	(L/s)	Hole Depth	Duration (hr)	Salinity (mg/L)
							(m)		

#### Geologists Log Drillers Log

F	rom	То	Thickness	Drillers Description	Geological Material	Comments
		1		I		

(m) (m) (m)		
1		

#### Remarks

30/07/2014: Nat Carling, 30-July-2014; Added status, drill method & depth, updated work type.

\*\*\* End of GW113433 \*\*\*

Warning To Clients: This raw data has been supplied to the NSW Office of Water by drillers, licensees and other sources. The NOW does not verify the accuracy of this data. The data is presented for use by you at your own risk. You should consider verifying this data before relying on it. Professional hydrogeological advice should be sought in interpreting and using this data.

allwaterdata.water.nsw.gov.au/wgen/users/060333251//gw113434.wsr.htm

### NSW Office of Water Work Summary

#### GW113434

Licence:	10BL600157	Licence Status: /	ACTIVE
		Authorised / Purpose(s): Intended Purpose(s): /	MONITORING BORE
Work Type:	Bore		
Work Status:	Equipped		
Construct.Method:			
Owner Type:	Private		
Commenced Date: Completion Date:	27/03/2006	Final Depth: Drilled Depth:	
Contractor Name:	Terratest		
Driller:	Craig Williams		
Assistant Driller:			
Property:	MOBIL OIL AUST LTD 344 - 348 GREAT WESTERN HWY WENTWORTHVILLE 2145 NSW	Standing Water Level:	
GWMA:		Salinity:	
GW Zone:		Yield:	
Site Details			

Site Chosen By:

	County Form A: CUMBE Licensed:	<b>Parish</b> CUMBE.41	<b>Cadastre</b> 6/3/963
Region: 10 - Sydney South Coast	CMA Map:		
River Basin: - Unknown Area/District:	Grid Zone:	So	cale:
Elevation: 0.00 m (A.H.D.) Elevation Unknown Source:	Northing: 6256468.0 Easting: 312139.0		ude: 33°48'55.2"S ude: 150°58'13.0"E
GS Map: -	MGA Zone: 0		nate Unknown Irce:

#### Construction

Negative depths indicate Above Ground Level; C-Cemented; SL-Slot Length; A-Aperture; GS-Grain Size; Q-Quantity; PL-Placement of Gravel Pack; PC-Pressure Cemented; S-Sump; CE-Centralisers

Π	Hole	Pipe	Component	Туре	From	То	Outside	Inside	Interval	Details
					(m)	(m)	Diameter	Diameter		
							(mm)	(mm)		

#### Water Bearing Zones

From (m)	To (m)	Thickness (m)	WBZ Туре	S.W.L. (m)	D.D.L. (m)	(L/s)	Hole Depth	Duration (hr)	Salinity (mg/L)
							(m)		

#### Geologists Log Drillers Log

F	rom	То	Thickness	Drillers Description	Geological Material	Comments

(m) (m) (m)		
1		

#### Remarks

30/07/2014: Nat Carling, 30-July-2014; Added status, drill method & depth, updated work type.

\*\*\* End of GW113434 \*\*\*

Warning To Clients: This raw data has been supplied to the NSW Office of Water by drillers, licensees and other sources. The NOW does not verify the accuracy of this data. The data is presented for use by you at your own risk. You should consider verifying this data before relying on it. Professional hydrogeological advice should be sought in interpreting and using this data.

allwaterdata.water.nsw.gov.au/wgen/users/060333251//gw113436.wsr.htm

### NSW Office of Water Work Summary

#### GW113436

Licence:	10BL600157	Licence Status:	ACTIVE
		Authorised Purpose(s): Intended Purpose(s):	MONITORING BORE
Work Type:	Bore		
Work Status:	Equipped		
Construct.Method:			
Owner Type:	Private		
Commenced Date: Completion Date:	20/03/2006	Final Depth: Drilled Depth:	
Contractor Name:	Terratest		
Driller:	Craig Williams		
Assistant Driller:			
Property:	MOBIL OIL AUST LTD 344 - 348 GREAT WESTERN HWY WENTWORTHVILLE 2145 NSW	Standing Water Level:	
GWMA:		Salinity:	
GW Zone:		Yield:	
Site Details			

#### Site Chosen By:

	County Form A: CUMBE Licensed:	<b>Parish</b> CUMBE.41	<b>Cadastre</b> 6/3/963
Region: 10 - Sydney South Coast	СМА Мар:		
River Basin: - Unknown Area/District:	Grid Zone:	S	cale:
Elevation: 0.00 m (A.H.D.) Elevation Unknown Source:	Northing: 6256449.0 Easting: 312131.0		ude: 33°48'55.8"S ude: 150°58'12.6"E
GS Map: -	MGA Zone: 0		nate Unknown Irce:

#### Construction

Negative depths indicate Above Ground Level; C-Cemented; SL-Slot Length; A-Aperture; GS-Grain Size; Q-Quantity; PL-Placement of Gravel Pack; PC-Pressure Cemented; S-Sump; CE-Centralisers

Π	Hole	Pipe	Component	Туре	From	То	Outside	Inside	Interval	Details
					(m)	(m)	Diameter	Diameter		
							(mm)	(mm)		

#### Water Bearing Zones

From (m)	To (m)	Thickness (m)	WBZ Туре	S.W.L. (m)	D.D.L. (m)	Yield (L/s)	Hole Depth (m)	Duration (hr)	Salinity (mg/L)
							(m)		

#### Geologists Log Drillers Log

Fr	rom	То	Thickness	Drillers Description	Geological Material	Comments
			I I			

(m) (m) (m)		
1		

#### Remarks

30/07/2014: Nat Carling, 30-July-2014; Added status, drill method & depth, updated work type.

\*\*\* End of GW113436 \*\*\*

Warning To Clients: This raw data has been supplied to the NSW Office of Water by drillers, licensees and other sources. The NOW does not verify the accuracy of this data. The data is presented for use by you at your own risk. You should consider verifying this data before relying on it. Professional hydrogeological advice should be sought in interpreting and using this data.

#### **NSW Groundwater Works**

Preformatted reports containing information about licensed water bores, wells and excavations.



# **APPENDIX H**

# LOCAL METEOROLOGY



#### Monthly Rainfall (millimetres)

#### GREYSTANES (BATHURST STREET)

Station Number: 067017 · State: NSW · Opened: 2001 · Status: Open · Latitude: 33.81°S · Longitude: 150.94°E · Elevation: 70 m

Year	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec	Annual
2001						1.4	69.6	28.6	22.4	35.8	82.2	39.8	
2002	86.0	275.0	102.6	25.2	47.4	10.6	11.2	6.6	10.8	8.0	19.4	70.0	672.8
2003	19.6	100.0	85.0	72.8	179.0	37.2	27.2	25.8	4.8	87.2	56.6	78.4	773.6
2004	47.6	102.6	46.6	36.6	9.2	11.8	18.2	38.4		209.2	56.0	71.6	
2005		114.6	59.2		33.0	68.2	11.8	3.4	41.8	50.4	127.2	17.8	
2006	101.2	83.6	14.6	6.2	10.4	71.0	47.4	46.0	138.2	8.2	29.2	57.6	613.6
2007	30.8	172.4	69.6	72.6	17.4	292.8	26.6			26.0	133.4	86.8	
2008	71.6	261.6	73.2	73.0	4.2	101.8	33.6	29.0	56.2	58.4	77.2	71.6	911.4
2009	34.2	133.2	65.4		85.0	42.0		8.6	17.2	93.4	11.0	63.4	
2010	34.8	200.4	105.8	23.4	72.0	71.4	44.8	22.6	29.0	93.8	115.6	91.2	904.8
2011	87.6	20.2	78.6	105.8	67.4	53.6	94.2	35.6	66.2	33.8	138.4	150.8	932.2
2012	131.0	219.0	199.4	154.0	17.2	130.6	20.8	6.2	21.4	27.4	36.4	42.0	1005.4
2013	123.8	163.6	78.2	69.6	37.2	150.6	7.4	8.4	16.8	7.8	169.2	31.6	864.2
2014	19.0	66.4	170.4	39.2	7.6	28.8	11.8	168.2	30.6	87.8	34.6	193.6	858.0
2015	170.4	38.8											

Quality control: 12.3 Done & acceptable, 12.3 Not completed or unknown



© Copyright Commonwealth of Australia 2015, Bureau of Meteorology. Prepared using Climate Data Online, Bureau of Meteorology http://www.bom.gov.au/climate/data. Contact us using details on http://www.bom.gov.au/climate/how/contacts.shtml. We have taken all due care but cannot provide any warranty nor accept any liability for this information. http://www.bom.gov.au/other/copyright.shtml

#### Monthly Rainfall (millimetres)

#### **GREYSTANES (BATHURST STREET)**

Station Number: 067017 · State: NSW · Opened: 2001 · Status: Open · Latitude: 33.81°S · Longitude: 150.94°E · Elevation: 70 m

	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec	Annual
Lowest	19.0	20.2	14.6	6.2	4.2	1.4	7.4	3.4	4.8	7.8	11.0	17.8	613.6
Highest	170.4	275.0	199.4	154.0	179.0	292.8	94.2	168.2	138.2	209.2	169.2	193.6	1005.4

#### Statistics for this station calculated over all years of data

1) Calculation of statistics

Summary statistics, other than the Highest and Lowest values, are only calculated if there are at least 20 years of data available.

2) Gaps and missing data

Gaps may be caused by a damaged instrument, a temporary change to the site operation, or due to the absence or illness of an observer.

3) Further information

http://www.bom.gov.au/climate/cdo/about/about-rain-data.shtml.



Product code: IDCJAC0001 reference: 18266734 Created on Fri 27 Mar 2015 16:29:54 PM EST

© Copyright Commonwealth of Australia 2015, Bureau of Meteorology. Prepared using Climate Data Online, Bureau of Meteorology http://www.bom.gov.au/climate/data. Contact us using details on http://www.bom.gov.au/climate/how/contacts.shtml. We have taken all due care but cannot provide any warranty nor accept any liability for this information. http://www.bom.gov.au/other/copyright.shtml

# **APPENDIX I**

# **IMPORTANT INFORMATION ABOUT YOUR REPORT**





#### IMPORTANT INFORMATION ABOUT YOUR ENVIRONMENTAL SITE ASSESSMENT

These notes have been prepared by Aargus (Australia) Pty Ltd and its associated companies using guidelines prepared by ASFE (The Association) of Engineering Firms Practising in the Geo-sciences. They are offered to help you in the interpretation of your Environmental Site Assessment (ESA) reports.

#### **REASONS FOR CONDUCTING AN ESA**

ESA's are typically, though not exclusively, carried out in the following circumstances:

- as pre-acquisition assessments, on behalf of either purchaser or vender, when a property is to be sold;
- as pre-development assessments, when a property or area of land is to be redeveloped or have its use changed for example, from a factory to a residential subdivision;
- as pre-development assessments of greenfield sites, to establish "baseline" conditions and assess environmental, geological and hydrological constraints to the development of, for example, a landfill; and
- as audits of the environmental effects of an ongoing operation.

Each of these circumstances requires a specific approach to the assessment of soil and groundwater contamination. In all cases however, the objective is to identify and if possible quantify the risks that unrecognised contamination poses to the proposed activity. Such risks may be both financial, for example, cleanup costs or limitations on site use, and physical, for example, health risks to site users or the public.

#### THE LIMITATIONS OF AN ESA

Although the information provided by an ESA could reduce exposure to such risks, no ESA, however, diligently carried out can eliminate them. Even a rigorous professional assessment may fail to detect all contamination on a site. Contaminants may be present in areas that were not surveyed or sampled, or may migrate to areas which showed no signs of contamination when sampled.

#### AN ESA REPORT IS BASED ON A UNIQUE SET OF PROJECT SPECIFIC FACTORS

Your environmental report should not be used:

- when the nature of the proposed development is changed, for example, if a residential development is proposed instead of a commercial one;
- when the size or configuration of the proposed development is altered;
- when the location or orientation of the proposed structure is modified;
- when there is a change of ownership
- or for application to an adjacent site.

To help avoid costly problems, refer to your consultant to determine how any factors, which have changed subsequent to the date of the report, may affect its recommendations.

#### ESA "FINDINGS" ARE PROFESSIONAL ESTIMATES

Site assessment identifies actual subsurface conditions only at those points where samples are taken, when they are taken. Data derived through sampling and subsequent laboratory testing are interpreted by geologists, engineers or scientists who then render an opinion about overall subsurface conditions, the nature and extent of contamination, its likely impact on the proposed development and appropriate remediation measures. Actual conditions may differ from those inferred to exist, because no professional, no matter how qualified, and no subsurface exploration program, no matter how comprehensive, can reveal what is hidden by earth, The actual interface between rock and time. materials may be far more gradual or abrupt than a report indicates. Actual conditions in areas not sampled may differ from predictions. Nothing can be done to help minimise its impact. For this reason owners should retain the services of their consultants

through the development stage, to identify variances, conduct additional tests which may be needed, and to recommend solutions to problems encountered on site.

#### SUBSURFACE CONDITIONS CAN CHANGE

Natural processes and the activity of man change subsurface conditions. As an ESA report is based on conditions, which existed at the time of subsurface exploration, decisions should not be based on an ESA report whose adequacy may have been affected by time. Speak with the consultant to learn if additional tests are advisable.

#### ESA SERVICES ARE PERFORMED FOR SPECIFIC PURPOSES AND PERSONS

Every study and ESA report is prepared in response to a specific brief to meet the specific needs of specific individuals. A report prepared for a consulting civil engineer may not be adequate for a construction contractor, or even some other consulting civil engineer. Other persons should not use a report for any purpose, or by the client for a different purpose. No individual other than the client should apply a report even apparently for its intended purpose without first conferring with the consultant. No person should apply a report for any purpose other than that originally contemplated without first conferring with the consultant.

#### AN ESA REPORT IS SUBJECT TO MISINTERPRETATION

Costly problems can occur when design professionals develop their plans based on misinterpretations of an ESA. To help avoid these problems, the environmental consultant should be work with appropriate retained to design professionals to explain relevant findings and to review the adequacy of their plans and specifications relative to contamination issues.

#### LOGS SHOULD NOT BE SEPARATED FROM THE ENGINEERING REPORT

Final borehole or test pit logs are developed by environmental scientists, engineers or geologists based upon their interpretation of field logs (assembled by site personnel) and laboratory evaluation of field samples. Only final logs customarily included in our reports. These logs should not under any circumstances be redrawn for inclusion in site remediation or other design drawings, because drafters may commit errors or omissions in the transfer process. Although photographic reproduction eliminates this problem, it does nothing to minimise the possibility of contractors misinterpreting the logs during bid preparation. When this occurs, delays, disputes and unanticipated costs are the all-too-frequent result.

To the likelihood of boring reduce log misinterpretation, the complete report must be available to persons or organisations involved in the project, such as contractors, for their use. Those who o not provide such access may proceed under the mistaken impression that simply disclaiming responsibility for the accuracy of subsurface information always insulates them from attendant liability. Providing all the available information to persons and organisations such as contractors helps prevent costly construction problems and the adversarial attitudes that may aggravate them to disproportionate scale.

#### READ RESPONSIBILITY CLAUSES CLOSELY

Because an ESA is based extensively on judgement and opinion, it is necessarily less exact than other disciplines. This situation has resulted in wholly unwarranted claims being lodged against consultants. To help prevent this problem, model clauses have been developed for use in transmittals. These are not exculpatory clauses designed to foist liabilities onto some other party. Rather, they are definitive clauses that identify where your consultant's responsibilities begin and end. Their use helps all parties involved recognise their individual responsibilities and take appropriate action. Some of these definitive clauses are likely to appear in your ESA report, and you are encouraged to read them closely. Your consultant will be pleased to give full and frank answers to your questions.